Outer Dowsing Offshore Wind

Compulsory Acquisition Information

4.1 Book of Reference

Date: July 2024

Document Reference: 4,1 Pursuant to APFP Regulatio Rev: 3.0 Clean

Company: Out		Outer Dowsing Offshore Wind		Asset:		Whole Asset		
Project: WI		Whole Wind Farm		Sub Project/Package: Whole		Whole /	Asset	
Document Title or Description: 4.1 Book		Book of Reference	e					
Internal Document Pf Number:		РР	P1-ODOW-DEV-CS-TOR-0001_03		3 rd Party Doc No (If applicable):			
Rev No. Date			Status / Reason for Issue	Author	Checked by	Re ^v by	viewed	Approved by
1.0	1.0 March 2024		DCO Application	Dalcour Maclaren	Shepherd & Wedderburn	Ou Do	ter wsing	Outer Dowsing
2.0 June 2024		24	Section 59 Certificate of Compliance	Dalcour Maclaren	Shepherd & Wedderburn		ter wsing	Outer Dowsing
3.0 July 2024		Response to Section 51 Advice	Dalcour Maclaren	Shepherd & Wedderburn		ter wsing	Outer Dowsing	

Glossary of Acronyms

R Book of Reference

Glossary of Terminology

the Order	Development Consent Order
the authorised project	Outer Dowsing Offshore Wind Farm
the Regulations	Regulation 5(2)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

1.1 This Book of Reference ("BoR") accompanies the proposed Development Consent Order ("the Order") for the Outer Dowsing Offshore Wind Farm ("the authorised project"), as required by Regulation 5(2)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 ("the Regulations"). It describes all the land, and identifies all the interests, affected by the Order.

1.2 The land described in this BoR, which is to be subject to powers of compulsory acquisition and/or temporary possession, is required to deliver the works comprising the authorised project which are described in detail in Part 1 of Schedule 1 of the Order and shown on the works plans.

1.3 Every parcel of land that is affected is identified on a plot by plot basis and a unique number has been ascribed to each plot. The plots are shown on the land plans which accompany the Order.

1.4 This BoR is comprised of five Parts, in accordance with Regulation 7(1) of the Regulations, as follows:

- Part 1 contains the names and addresses of those who own, lease, occupy or have another interest in the land that will be affected by the authorised project and the rights contained in the Order;
- Part 2 contains the names and addresses of those whose land, while not directly affected by the authorised project, may be entitled to claim compensation for loss resulting from the implementation of the Order and use of the authorised project;
- Part 3 contains the names and addresses of those entitled to enjoy easements or other private rights which it is proposed may be extinguished, suspended or interfered with in connection with the authorised project, pursuant to the Order;
- Part 4 identifies plots in which there is a Crown interest that will be affected by the authorised project and the rights contained in the Order (please note that the interests of the Crown in these plots will <u>not</u> be subject to powers of compulsory acquisition); and
- Part 5 identifies plots which constitute "special category land" for the purposes of Section 132 of the Planning Act 2008 that will be affected by the authorised project and the rights contained in the Order.

1.5 The Order seeks powers to compulsorily acquire land and new rights (both temporary and permanent) for the purposes of the construction and operation of the authorised project.

1.6 The majority of the plots identified in Part 1 of the BoR will be subject to the acquisition of Permanent Rights (including restrictive covenants) pursuant to Article 22 (Compulsory acquisition of rights) and Schedule 7 (Land in which only new rights etc. may be acquired) of the Order. These plots are shown coloured blue on the land plans.

1.7 A number of the plots identified in Part 1 of the BoR will be subject to freehold acquisition pursuant to Article 20 (Compulsory acquisition of land) of the Order. These plots are shown coloured pink on the land plans.

1.8 The land shown coloured blue and pink on the land plans will also be subject to powers of temporary possession for the purpose of carrying out the authorised project (by virtue of Article 28 (Temporary use of land for carrying out the authorised project)).

1.9 Plots that are subject to powers of temporary possession only, such as for the purpose of access to and/or use as a temporary construction compound, are also listed in Schedule 9 (Land of

which temporary possession may be taken) of the Order and are shown coloured yellow on the land plans.

1.10 By virtue of Article 29 (Temporary use of land for maintaining authorised project) of the Order, any land within the Order limits which is reasonably required for the purpose of maintaining the authorised project may be entered and/or temporarily possessed.

1.11 The Tracked Version of the Clean Book of Reference highlights the updates to land interest information since the previous submission of the Book of Reference (Doc 4.1) as follows:

- those entries struck through and highlighted in red have been removed;

- those entries highlighted in blue have been added; and
- those entries highlighted in orange have an updated name, address, or qualifier.

Categories of New Rights sought under the Outer Dowsing Offshore Wind Farm Order 202X

The descriptions of the New Rights sought by the Undertaker set out in the Table of New Rights below shall apply in the column headed "extent, description and situation of the land" in Part 1 of this Book of Reference. For this purpose the letter in column 1 of the Table of New Rights is used to indicate the New Right(s) sought over each relevant plot of land. The details are described in column 2 of the Table of New Rights. The New Rights are sought for the purposes of construction, installation, operation, maintenance and decommissioning of the authorised project.

In the Table of New Rights:

"adjoining land"	means such other parts of the land within the Order limits as may
"authorised development"	be required for the authorised project; has the meaning defined in the Outer Dowsing Offshore Wind Farm Order 202X
"authorised project"	has the meaning defined in the Outer Dowsing Offshore Wind Farm Order 202X
"Cables"	means collectively electricity cables together with telemetry and fibre optic lines, ducting, earthing measures, thermal protection, temperature sensing measures, cathodic protection and heat dissipation measures, and other apparatus, bedding measures, protection measures, safety measures and other equipment ancillary to the purposes of transmitting electricity along such electricity cables;
"Jointing Installations"	means collectively jointing bays, cross bonding leads, cross bonding pits, link boxes, inspection chambers, covers for pits, link boxes and chambers, and marker posts;
"Land"	means the relevant plot shown on the land plan and described in this Book of Reference;
"Order land"	means the land shown on the land plan which is within the limits of land to be acquired or used and described in this Book of Reference;
"terrestrial work activities"	 means laying down, constructing, installing, adjusting, altering, testing, maintaining, repairing, renewing, upgrading, surveying, cleansing, relaying, connecting into, diverting, protecting, making safe, making incapable of operation, replacing and removing any part or parts of the authorised development and such other works as may be necessary or expedient for the purposes of or in connection with the relevant part of the authorised development and which fall within the scope of the work assessed by the environmental statement, including— a) haul roads, ramps, bridges, means of access and footpaths, creation and enhancement of tracks and footpaths; b) bunds, embankments, swales, landscaping, ground preparation, vegetation clearance, ground-raising, signage, fencing and boundary treatments; c) habitat creation and enhancement; d) spoil storage and associated control measures;

joint bays, link boxes, earth pits, cable ducts, cable protection, joint protection, manholes, marker posts, underground cable marker, tiles and tape, and lighting and other works associated with duct installation, cable laying and pulling cables through cable ducts;
works for the provision of apparatus, including cabling, water and electricity supply works, foul drainage provision, surface water management systems, flumes and culverting;
works to alter the position of apparatus, including mains, sewers, drains and cables;
works to alter the course of, or otherwise interfere with, nonnavigable rivers, streams or watercourses;
landscaping and related works;
other works to mitigate any adverse effects of the
construction, maintenance or operation of the authorised project;
works for the benefit or protection of land affected by the authorised project;
working sites in connection with the construction of the authorised project, construction lay down areas and
compounds, storage compounds and their restoration;
works of restoration; and
fencing or other means of enclosure.
GT R4 Limited (company number 13281221);

Right	Right description
A	A. all rights and restrictions necessary for the
Work No. 11 (landfall connection works	Undertaker and/or those authorised by the Undertaker
consisting of up to four underground	1. to install the Cables by the use of directional drilling
cable circuits and up to six associated	or other trenchless techniques only;
cable ducts to Work No. 12)	2. to retain and use the Cables for the purposes of the
	transmission of electricity and telecommunications;
	3. to benefit from continuous vertical and lateral
	support for the Cables;
	B. a restrictive covenant over the Land for the benefit
	of the remainder of the Order land to prevent anything
	being done in or upon the Land or any part thereof
	which interferes with or might interfere with the
	exercise of the Rights or the use of the Cables or in any
	way render the Cables in breach of any statute or
	regulation for the time being in force and applicable
	thereto and without prejudice to the generality of the
	foregoing to prevent the construction of any buildings
	on, the surfacing of, the carrying out of any excavations
	or works to a depth greater than 0.7 metre on or in, or
	the planting of any trees or shrubs on, the Land.
B	A. all rights and restrictions necessary for the
Work No. 12 (up to four underground	Undertaker and/or those authorised by the Undertaker
cable circuits and up to six associated	1. to install the Cables by the use of directional drilling
cable ducts to Work No. 13)	or other trenchless techniques only;
	2. to retain and use the Cables for the purposes of the
	transmission of electricity and telecommunications;
	3. to benefit from continuous vertical and lateral
	support for the Cables;
	B. a restrictive covenant over the Land for the benefit
	of the remainder of the Order land to prevent anything being done in or upon the Land or any part thereof
	which interferes with or might interfere with the
	exercise of the Rights or the use of the Cables or in any
	way render the Cables in breach of any statute or
	regulation for the time being in force and applicable
	thereto and without prejudice to the generality of the
	foregoing to prevent the construction of any buildings
	on, the surfacing of, the carrying out of any excavations
	or works to a depth greater than 0.7 metre on or in, or
	the planting of any trees or shrubs on, the Land.
C	A. all rights and restrictions necessary for the
Work No. 13 (works consisting of:	Undertaker and/or those authorised by the Undertaker
a. up to six trenchless technique	to enter onto, pass and repass and remain on the Land
drilling launch pits;	with or without all necessary plant, vehicles, machinery,
5 - 5 - F ,	materials, drilling fluids, apparatus, temporary structures
	and equipment and
	and adaption and

b.	up to four underground cable	1	the right to install the Cables by way of (but not
	circuits and up to six associated		limited to) digging trenches, the use of directional
	cable ducts to Work No. 14;		drilling, auger boring, thrust boring, micro
с.	up to four underground cable		tunnelling or pipe ramming or other similar
	circuits and associated cable ducts	2	trenchless techniques;
d.	from Work No. 14 to Work No. 15; a landfall temporary works area;	2.	the right to retain and use the Cables and Jointing Installations for the purposes of the transmission of
и. е.	storage areas;		electricity and telecommunications;
f.	drainage works;	3.	the right to benefit from continuous vertical and
g.	construction of a haul road; and		lateral support for the Cables and Jointing
h.	vehicular access tracks,		Installations;
	bellmouths and footpaths)	4.	the right to benefit from continuous vertical and
			lateral support for the pipes, cables, conduits,
			service media and apparatus installed in, under, or
	rk No. 14 to six transition joint have including	5.	over the Land; the right to carry out terrestrial work activities;
	to six transition joint bays including und preparation)	5. 6.	the right to access adjoining land and highway;
gro		7.	the right to place and use plant, vehicles,
			machinery, materials, drilling fluids, apparatus,
			equipment and temporary structures on and within
			the Land;
		8.	the right to remove and discharge water from the
		0	Land;
		9.	the right to remove, store and stockpile materials (including excavated material) within the Land;
		10	the right to retain, maintain, straighten, widen,
		-0.	repair, alter, upgrade and use existing access routes
			for the purposes of accessing the Land, adjoining
			land and highway;
		11.	the right to erect temporary signage and provide
			measures for the benefit of public and personnel
		12	safety;
		12.	the right to fell, prune, cut, coppice, alter, lop, uproot and replant trees, shrubs and hedges, and
			remove roots of trees, shrubs and hedges;
		13.	the right to install, execute, implement, retain,
			repair, improve, renew, remove, relocate and plant
			trees, woodland, shrubs, hedgerows, seeding and
			other ecological measures together with the right to
			maintain, inspect and replant trees, shrubs and
		1.4	landscaping;
		14.	the right to install, execute, implement, retain, repair, improve, renew, relocate, maintain and
			carry out mitigation, maintenance, remediation
			and/or enhancement works;
		15.	the right to lay down, construct, install, retain,
			adjust, alter, test, use, maintain, repair, renew,

	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove pipes, cables,
	conduits, service media or apparatus that are used
	for the purposes of transmitting or distributing
	electricity, communications, gas, oil, water supply,
	sewerage and drainage (including the pipes, cables,
	conduits, service media or apparatus of statutory
	undertakers);
16	. the right to remove archaeological artefacts where
	they would prevent or cause it to be materially
	more difficult to undertake terrestrial work
	activities or where leaving such archaeological
	artefacts in situ would materially increase the cost
	of the terrestrial work activities;
17	. the right to remove fences, hedges, gates or other
1,	barriers during any period during which terrestrial
	work activities are being undertaken (subject to the
	prior erection of any temporary stock proof fencing
	as is reasonably required) and the replacement,
	replanting or re-instatement of fences, hedges or
	other barriers;
19	the right to carry out surveys including for site
10	investigation and environmental surveys (including
	without prejudice to the generality of the foregoing,
	topographic, drainage, archaeological,
	geotechnical, geoenvironmental, ecological and
	ground stability site investigations and surveys),
	making of boreholes, trial pits and archaeological
	trenches and field stripping, auger hole sampling
	and the taking of soil and other samples;
19	. the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove cable marker posts
	to identify the location of the Cables as required for
	routine integrity testing;
20	the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove temporary access
	roads, hard standing and other surface materials
	including (but not limited to) matting, aggregate,
	trackway, stone, tarmacadam, terram;

 the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove public and private drains, watercourses, sewers, ponds or culverts, and to drain into and manage waterflows in any public and private drains, watercourses, sewers, ponds or culverts including (but not limited to) by way of damming and overpumping; the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove supporting or protective structures (including the bridging over of or protection of the apparatus of the statutory undertakers); the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of or protective structures (including the bridging over of or protection of the apparatus of the statutory undertakers); the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of
operation, replace and remove temporary haul
 roads; 24. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove temporary welfare facilities (including but not limited to portakabins, portaloos and welfare equipment); 25. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew,
upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove fencing, gates, walls, barriers or other means of enclosure;
B. a restrictive covenant over the Land for the
benefit of the remainder of the Order land to prevent anything being done in or upon the Land or any part
thereof which interferes with or might interfere with
the exercise of the Rights or the use of the Cables or in any way render the Cables in breach of any statute or
regulation for the time being in force and applicable
thereto and without prejudice to the generality of the
foregoing to prevent the construction of any buildings on, the surfacing of, the carrying out of any excavations
on, the surfacing of, the carrying out of any excavations

	or works to a depth greater than 0.7 metre on or in, or the planting of any trees or shrubs on, the Land.
 D Work No. 15 (works consisting of: a. up to four underground cable circuits and associated cable ducts to Work No. 16; b. up to 680 link boxes and/or earth pits; c. up to 680 joint bays; d. temporary trenchless technique compounds; e. storage areas; f. drainage works; g. construction of a haul road; and h. vehicular access tracks, bellmouths and footpaths.) 	 A. all rights and restrictions necessary for the Undertaker and/or those authorised by the Undertaker to enter onto, pass and repass and remain on the Land with or without all necessary plant, vehicles, machinery, materials, drilling fluids, apparatus, temporary structures and equipment and 1. the right to install the Cables by way of (but not limited to) digging trenches, the use of directional drilling, auger boring, thrust boring, micro tunnelling or pipe ramming or other similar trenchless techniques; 2. the right to retain and use the Cables and Jointing Installations for the purposes of the transmission of electricity and telecommunications; 3. the right to benefit from continuous vertical and lateral support for the Cables and Jointing Installations; 4. the right to benefit from continuous vertical and lateral support for the pipes, cables, conduits, service media and apparatus installed in, under, or over the Land; 5. the right to carry out terrestrial work activities; 6. the right to place and use plant, vehicles, machinery, materials, drilling fluids, apparatus, equipment and temporary structures on and within the Land; 9. the right to remove, store and stockpile materials (including excavated material) within the Land; 10. the right to erect temporary signage and provide measures for the benefit of public and personnel safety; 12. the right to fell, prune, cut, coppice, alter, lop, uproot and replant trees, shrubs and hedges; 13. the right to install, execute, implement, retain, repair, improve, renew, remove, relocate and plant trees, woodland, shrubs, hedgerows, seeding and other ecological measures together with the right to

ГТ	
	maintain, inspect and replant trees, shrubs and
	landscaping;
14.	the right to install, execute, implement, retain,
	repair, improve, renew, relocate, maintain and
	carry out mitigation, maintenance, remediation
	and/or enhancement works;
15.	the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove pipes, cables,
	conduits, service media or apparatus that are used
	for the purposes of transmitting or distributing
	electricity, communications, gas, oil, water supply,
	sewerage and drainage (including the pipes, cables,
	conduits, service media or apparatus of statutory
	undertakers);
16	the right to remove archaeological artefacts where
10.	they would prevent or cause it to be materially
	more difficult to undertake terrestrial work
	activities or where leaving such archaeological
	artefacts in situ would materially increase the cost
	of the terrestrial work activities;
17.	the right to remove fences, hedges, gates or other
	barriers during any period during which terrestrial
	work activities are being undertaken (subject to the
	prior erection of any temporary stock proof fencing
	as is reasonably required) and the replacement,
	replanting or re-instatement of fences, hedges or
	other barriers;
18.	the right to carry out surveys including for site
	investigation and environmental surveys (including
	without prejudice to the generality of the foregoing,
	topographic, drainage, archaeological,
	geotechnical, geoenvironmental, ecological and
	ground stability site investigations and surveys),
	making of boreholes, trial pits and archaeological
	trenches and field stripping, auger hole sampling
	and the taking of soil and other samples;
19.	the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove cable marker posts
	to identify the location of the Cables as required for
	routine integrity testing;
	routine integrity testing,

Гт	
	 the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove temporary access roads, hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram; the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove public and private drains, watercourses, sewers, ponds or culverts, and to drain into and manage waterflows in any public and private drains, watercourses, sewers, ponds or culverts including (but not limited to) by way of damming and overpumping; the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove supporting or protective structures (including the bridging over of or protection of the apparatus of the statutory
	undertakers); 23. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove temporary haul roads;
	 24. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove temporary barriers for the protection of fauna;
	25. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove temporary noise alleviation measures;
	26. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect

	 into, divert, protect, make safe, make incapable of operation, replace and remove temporary paths and bridleways for public use; 27. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove temporary welfare facilities (including but not limited to portakabins, portaloos and welfare equipment); 28. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove fencing, gates, walls, barriers or other means of enclosure; B. a restrictive covenant over the Land for the benefit of the remainder of the Order land to prevent anything being done in or upon the Land or any part thereof which interferes with or might interfere with the exercise of the Rights or the use of the Cables or in any way render the Cables in breach of any statute or regulation for the time being in force and applicable thereto and without prejudice to the generality of the foregoing to prevent the construction of any buildings on, the surfacing of, the carrying out of any excavations or works to a depth greater than 0.7 metre on or in, or the planting of any trees or shrubs on, the Land.
Freehold Acquisition	Freehold Acquisition
Work No. 16 (works consisting of:	
a. an onshore HVAC substation;	
b. up to four underground cable	
circuits and associated cable ducts	
between Work No. 15 and the	
onshore HVAC substation;	
c. up to two underground cable	
circuits and associated cable ducts	
between the onshore HVAC	
substation and Work No. 17;	
d. construction of a haul road;	
e. vehicular access tracks,	
bellmouths and footpaths;	
f. temporary works areas to support	
the construction activities;	
g. storage areas and	
h. ground preparation and ground-	

Work No. 23 (works consisting of landscaping works including bunding and planting, drainage works, and formation of footpaths and access) and Work No. 25 (works consisting of drainage works, sustainable drainage system ponds, surface water management systems, formation of footpaths and access)	
 Fortputtis und uccess) E Work No. 17 (connection works consisting of: a. up to two underground cable circuits and associated cable ducts between the onshore HVAC substation forming part of Work No. 15 and a National Grid substation, including a connection above ground; (b) electrical engineering works within or around the National Grid substation buildings and compound; (c) up to 20 link boxes and/or earth pits; (d) up to 20 joint bays; (e) temporary trenchless technique compounds; (f) storage areas; (g) drainage works; (h) construction of a haul road; and 	 A. all rights and restrictions necessary for the Undertaker and/or those authorised by the Undertaker to enter onto, pass and repass and remain on the Land with or without all necessary plant, vehicles, machinery, materials, drilling fluids, apparatus, temporary structures and equipment and 1. the right to install the Cables by way of (but not limited to) digging trenches, the use of directional drilling, auger boring, thrust boring, micro tunnelling or pipe ramming or other similar trenchless techniques; 2. the right to retain and use the Cables and Jointing Installations for the purposes of the transmission of electricity and telecommunications; 3. the right to benefit from continuous vertical and lateral support for the Cables and Jointing Installations; 4. the right to benefit from continuous vertical and lateral support for the pipes, cables, conduits, service media and apparatus installed in, under, or over the Land; 5. the right to carry out terrestrial work activities;
(i) vehicular access tracks, bellmouths and footpaths.	 the right to carry out terrestrian work activities, the right to access adjoining land and highway; the right to place and use plant, vehicles, machinery, materials, drilling fluids, apparatus, equipment and temporary structures on and within the Land; the right to remove and discharge water from the Land; the right to remove, store and stockpile materials (including excavated material) within the Land; the right to erect temporary signage and provide measures for the benefit of public and personnel safety; the right to fell, prune, cut, coppice, alter, lop, uproot and replant trees, shrubs and hedges;

12	the right to lay down, construct, install, retain,
12.	
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove pipes, cables,
	conduits, service media or apparatus that are used
	for the purposes of transmitting or distributing
	electricity, communications, gas, oil, water supply,
	sewerage and drainage (including the pipes, cables,
	conduits, service media or apparatus of statutory
	undertakers);
13.	the right to remove archaeological artefacts where
	they would prevent or cause it to be materially
	more difficult to undertake terrestrial work
	activities or where leaving such archaeological
	artefacts in situ would materially increase the cost
	of the terrestrial work activities;
14	the right to remove fences, hedges, gates or other
±	barriers during any period during which terrestrial
	work activities are being undertaken (subject to the
	prior erection of any temporary stock proof fencing
	as is reasonably required) and the replacement,
	replanting or re-instatement of fences, hedges or
	other barriers;
15.	the right to carry out surveys including for site
	investigation and environmental surveys (including
	without prejudice to the generality of the foregoing,
	topographic, drainage, archaeological,
	geotechnical, geoenvironmental, ecological and
	ground stability site investigations and surveys),
	making of boreholes, trial pits and archaeological
	trenches and field stripping, auger hole sampling
	and the taking of soil and other samples;
16.	the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove cable marker posts
	to identify the location of the Cables as required for
	routine integrity testing;
17.	the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove temporary access
	roads, hard standing and other surface materials
	round, hard standing and other surface materials

	including (but not limited to) matting, aggregate,
	trackway, stone, tarmacadam, terram;
12	8. the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove public and private
	drains, watercourses, sewers, ponds or culverts,
	and to drain into and manage waterflows in any
	public and private drains, watercourses, sewers,
	ponds or culverts including (but not limited to) by
	way of damming and overpumping;
1	
T	9. the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove supporting or
	protective structures (including the bridging over of
	or protection of the apparatus of the statutory
	undertakers);
20	0. the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove temporary haul
	roads;
22	2. the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove temporary welfare
	facilities (including but not limited to portakabins,
	portaloos and welfare equipment);
2	3. the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove fencing, gates,
	walls, barriers or other means of enclosure;
В	a restrictive covenant over the land for the benefit
0	f the remainder of the Order land to prevent anything
b	eing done in or upon the Land or any part thereof
	hich interferes with or might interfere with the
	kercise of the Rights or the use of the Cables or in any
	ay render the Cables in breach of any statute or
	gulation for the time being in force and applicable
	nereto and without prejudice to the generality of the
	,

	foregoing to prevent the construction of any buildings on, the surfacing of, the carrying out of any excavations or works to a depth greater than 0.7 metre on or in, or the planting of any trees or shrubs on, the Land.
F Work No. 18 (temporary vehicular access tracks, bellmouths and associated footpaths)	Land subject to temporary possession for the purposes of access to other parts of the Order land and subject to all rights and restrictions necessary for the Undertaker and/or those authorised by the Undertaker to enter onto, pass and repass and remain on the Land with or without all necessary plant, vehicles, machinery, materials, drilling fluids, apparatus, temporary structures and equipment and 1. the right to carry out terrestrial work activities; 2. the right to access adjoining land and highway.
G Work No. 19 (temporary works consisting of: temporary works areas to support the construction activities; temporary construction ramps; storage areas to assist with the onshore transmission works; and drainage works; and vehicular access tracks, bellmouths and footpaths)	 Land subject to temporary possession for the purpose of temporary works areas and subject to all rights and restrictions necessary for the Undertaker and/or those authorised by the Undertaker to enter onto, pass and repass and remain on the Land with or without all necessary plant, vehicles, machinery, materials, drilling fluids, apparatus, temporary structures and equipment and the right to carry out terrestrial work activities; the right to place and use plant, vehicles, machinery, materials, drilling fluids, apparatus, temporary structures on and within the Land; the right to remove and discharge water from the Land; the right to remove, store and stockpile materials (including excavated material) within the Land; the right to fell, prune, cut, coppice, alter, lop, uproot and replant trees, shrubs and hedges; the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove pipes, cables, conduits, service media or apparatus that are used for the purposes of transmitting or distributing electricity, communications, gas, oil, water supply, sewerage and drainage (including the pipes, cables,

ГГ	· · · · · ·
	conduits, service media or apparatus of statutory undertakers);
	 the right to remove archaeological artefacts where they would prevent or cause it to be materially more difficult to undertake terrestrial work activities or where leaving such archaeological artefacts in situ would materially increase the cost of the terrestrial work activities;
	10. the right to remove fences, hedges, gates or other barriers during any period during which terrestrial work activities are being undertaken (subject to the prior erection of any temporary stock proof fencing as is reasonably required) and the replacement, replanting or re-instatement of fences, hedges or other barriers;
	11. the right to carry out surveys including for site investigation and environmental surveys (including without prejudice to the generality of the foregoing, topographic, drainage, archaeological, geotechnical, geoenvironmental, ecological and ground stability site investigations and surveys), making of boreholes, trial pits and archaeological trenches and field stripping, auger hole sampling and the taking of soil and other samples;
	12. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove temporary access roads, hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone, tarmacadam, terram;
	13. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove public and private drains, watercourses, sewers, ponds or culverts, and to drain into and manage waterflows in any public and private drains, watercourses, sewers, ponds or culverts including (but not limited to) by way of damming and overpumping;
	14. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove supporting or

	 protective structures (including the bridging over of or protection of the apparatus of the statutory undertakers); 15. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove temporary welfare facilities (including but not limited to portakabins, portaloos and welfare equipment); 16. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove fencing, gates, walls, barriers or other means of enclosure.
Freehold Acquisition	Freehold Acquisition
Work No. 20 (permanent vehicular	
access tracks to serve Work Nos. 13 and 16, bellmouths, drainage works	
and associated footpaths)	
H	A. Land subject to temporary possession and to be
Work No. 21 (highway alterations including widening of the highway, provision of passing bays, culverting and drainage works)	 Land subject to temporary possession and to be subject to all rights and restrictions necessary for the Undertaker and/or those authorised by the Undertaker to enter onto, pass and repass and remain on the Land with or without all necessary plant, vehicles, machinery, materials, drilling fluids, apparatus, temporary structures and equipment and the right to benefit from continuous vertical and lateral support for the pipes, cables, conduits, service media and apparatus installed in, under, or over the Land; the right to carry out terrestrial work activities; the right to place and use plant, vehicles, machinery, materials, apparatus, equipment and temporary structures on and within the Land; the right to remove and discharge water from the Land; the right to fell, prune, cut, coppice, alter, lop, uproot and replant trees, shrubs and hedges; the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew,

	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove pipes, cables,
	conduits, service media or apparatus that are used
	for the purposes of transmitting or distributing
	electricity, communications, gas, oil, water supply,
	sewerage and drainage (including the pipes, cables,
	conduits, service media or apparatus of statutory
	undertakers);
9.	the right to remove archaeological artefacts where
	they would prevent or cause it to be materially
	more difficult to undertake terrestrial work
	activities or where leaving such archaeological
	artefacts in situ would materially increase the cost
	of the terrestrial work activities;
10	the right to remove fences, hedges, gates or other
	barriers during any period during which terrestrial
	work activities are being undertaken (subject to the
	prior erection of any temporary stock proof fencing
	as is reasonably required) and the replacement,
	replanting or re-instatement of fences, hedges or
	other barriers;
11	the right to lay down, construct, install, retain,
11	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove hard standing and
	other surface materials including (but not limited
	to) matting, aggregate, trackway, stone,
	tarmacadam, terram;
12.	the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove public and private
	drains, watercourses, sewers, ponds or culverts,
	and to drain into and manage waterflows in any
	public and private drains, watercourses, sewers,
	ponds or culverts including (but not limited to) by
	way of damming and overpumping;
13	the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove supporting or
	protective structures (including the bridging over of

	 or protection of the apparatus of the statutory undertakers); 14. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove fencing, gates, walls, barriers or other means of enclosure; B. a restrictive covenant over the land for the benefit of the remainder of the Order land but for the duration only of the period for which temporary possession of the land is required to prevent anything being done in or upon the Land or any part thereof which interferes with or might interfere with the exercise of the Rights and without prejudice to the generality of the foregoing to prevent the construction of any buildings on, the surfacing of, the carrying out of any excavations or works to a depth greater than 0.7 metre on or in, or the planting of any trees or shrubs on, the Land.
	 walls, barriers or other means of enclosure; B. a restrictive covenant over the land for the benefit of the remainder of the Order land but for the duration only of the period for which temporary possession of the land is required to prevent anything being done in or upon the Land or any part thereof which interferes with or might interfere with the exercise of the Rights and without prejudice to the generality of the foregoing to prevent the construction of any buildings on, the surfacing of, the carrying out of any excavations or
1	All rights and restrictions necessary for the Undertaker
Work No. 22 (works comprising the reinforcement or replacement of the	and/or those authorised by the Undertaker to enter onto, pass and repass and remain on the Land with or without
bridge)	all necessary plant, vehicles, machinery, materials, drilling fluids, apparatus, temporary structures and equipment and
	1. the right to carry out terrestrial work activities;
	 the right to access adjoining land and highway including use of the existing or any replacement bridge;
	3. the right to place and use plant, vehicles,
	machinery, materials, drilling fluids, apparatus, equipment and temporary structures on and within the Land;
	 the right to construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, protect, make safe, replace and remove the bridge and supporting structures on or over or in the Land;
	5. the right to remove, store and stockpile materials
	(including excavated material) within the Land;6. the right to retain, maintain, straighten, widen,
	for the purposes of accessing the Land, adjoining land and highway;
	 the right to erect temporary signage and provide measures for the benefit of public and personnel safety;

8.	the right to fell, prune, cut, coppice, alter, lop,
	uproot and replant trees, shrubs and hedges, and
	remove roots of trees, shrubs and hedges;
9.	the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove pipes, cables,
	conduits, service media or apparatus that are used
	for the purposes of transmitting or distributing
	electricity, communications, gas, oil, water supply,
	sewerage and drainage (including the pipes, cables,
	conduits, service media or apparatus of statutory
	undertakers);
10	the right to remove archaeological artefacts where
	they would prevent or cause it to be materially
	more difficult to undertake terrestrial work
	activities or where leaving such archaeological
	artefacts in situ would materially increase the cost
	of the terrestrial work activities;
11	the right to remove fences, hedges, gates or other
	barriers during any period during which terrestrial
	work activities are being undertaken (subject to the
	prior erection of any temporary stock proof fencing
	as is reasonably required) and the replacement,
	replanting or re-instatement of fences, hedges or
	other barriers;
12	the right to carry out surveys including for site
	investigation and environmental surveys (including
	without prejudice to the generality of the foregoing,
	topographic, drainage, archaeological,
	geotechnical, geoenvironmental, ecological and
	ground stability site investigations and surveys),
	making of boreholes, trial pits and archaeological
	trenches and field stripping, auger hole sampling
	and the taking of soil and other samples;
13	the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect
	into, divert, protect, make safe, make incapable of
	operation, replace and remove hard standing and
	other surface materials including (but not limited
	to) matting, aggregate, trackway, stone,
	tarmacadam, terram;
14	the right to lay down, construct, install, retain,
	adjust, alter, test, use, maintain, repair, renew,
	upgrade, inspect, survey, cleanse, re-lay, connect

	 into, divert, protect, make safe, make incapable of operation, replace and remove supporting or protective structures (including the bridging over of or protection of the apparatus of the statutory undertakers); 15. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove fencing, gates, walls, barriers or other means of enclosure.
J Work No. 24 (drainage works, formation of footpaths and access)	 A. all rights and restrictions necessary for the Undertaker and/or those authorised by the Undertaker to enter onto, pass and repass and remain on the Land with or without all necessary plant, vehicles, machinery, materials, drilling fluids, apparatus, temporary structures and equipment and 1. the right to carry out terrestrial work activities; 2. the right to access adjoining land and highway; 3. the right to place and use plant, vehicles, machinery, materials, apparatus, equipment and temporary structures on and within the Land; 4. the right to remove and discharge water from the Land; 5. the right to remove, store and stockpile materials (including excavated material) within the Land; 6. the right to retain, maintain, straighten, widen, repair, alter, upgrade and use existing access routes for the purposes of accessing the Land, adjoining land and highway; 7. the right to fell, prune, cut, coppice, alter, lop, uproot and replant trees, shrubs and hedges; 9. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove pipes, cables, conduits, service media or apparatus that are used for the purposes of transmitting or distributing electricity, communications, gas, oil, water supply, sewerage and drainage (including the pipes, cables, conduits, service media or apparatus of statutory undertakers);

 the right to remove archaeological artefacts where they would prevent or cause it to be materially more difficult to undertake terrestrial work activities or where leaving such archaeological artefacts in situ would materially increase the cost of the terrestrial work activities; the right to remove fences, hedges, gates or other barriers during any period during which terrestrial work activities are being undertaken (subject to the prior erection of any temporary stock proof fencing as is reasonably required) and the replacement, replanting or re-instatement of fences, hedges or other barriers;
 the right to carry out surveys including for site investigation and environmental surveys (including without prejudice to the generality of the foregoing, topographic, drainage, archaeological, geotechnical, geoenvironmental, ecological and ground stability site investigations and surveys), making of boreholes, trial pits and archaeological trenches and field stripping, auger hole sampling and the taking of soil and other samples; the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove hard standing and other surface materials including (but not limited to) matting, aggregate, trackway, stone,
 tarmacadam, terram; 14. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove public and private drains, watercourses, sewers or culverts, and to drain into and manage waterflows in any public and private drains, watercourses, sewers or culverts; 15. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove supporting or
protective structures (including the bridging over of or protection of the apparatus of the statutory undertakers);

	 16. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove temporary paths and bridleways for public use; 17. the right to lay down, construct, install, retain, adjust, alter, test, use, maintain, repair, renew, upgrade, inspect, survey, cleanse, re-lay, connect into, divert, protect, make safe, make incapable of operation, replace and remove fencing, gates, walls, barriers or other means of enclosure; B. a restrictive covenant over the land for the benefit of the remainder of the Order land to prevent anything being done in or upon the Land or any part thereof which interferes with or might interfere with the exercise of the Rights and without prejudice to the generality of the foregoing to prevent the construction of any buildings on, the surfacing of, the carrying out of any excavations or works on or in, or the planting of any trees or shrubs on, the Land.
K Work No. N/A (temporary vehicle access)	Land subject to temporary possession with the right for the Undertaker and/or those authorised by the Undertaker to enter onto, pass and repass and remain on the Land with or without all necessary plant, vehicles, machinery, materials, drilling fluids, apparatus, temporary structures and equipment and 1. the right to carry out terrestrial work activities; 2. the right to access adjoining land and highway.
L Work No. N/A. (permanent vehicular access in connection with Works No. 23 and 24)	All rights and restrictions necessary for the Undertaker and/or those authorised by the Undertaker to enter onto, pass and repass and remain on the Land with or without all necessary plant, vehicles, machinery, materials, drilling fluids, apparatus, temporary structures and equipment and 1. the right to carry out terrestrial work activities; 2. the right to access adjoining land and highway;

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
01-001	A	Permanent Rights over 11097 square metres of foreshore seaward of the Mean High Water (East of Roman Bank) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	NONE		
01-002	A	Permanent Rights over 104734 square metres of foreshore seaward of the Mean High Water, tidal pond and groynes (east of Roman Bank) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	NONE		
01-004	В	Permanent Rights over 1636 square metres of foreshore landward of the Mean High Water, scrubland and path (east of Roman Bank)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	Lincolnshire Wildlife Trust Banovallum House Manor House Street HORNCASTLE Lincolnshire LN9 5HF	Lincolnshire Wildlife Trust Banovallum House Manor House Street HORNCASTLE Lincolnshire LN9 5HF	NONE		
01-005	В	Permanent Rights over 83744 square metres of foreshore landward of the Mean High Water, scrubland and path (east of Roman Bank)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	Lincolnshire Wildlife Trust Banovallum House Manor House Street HORNCASTLE Lincolnshire LN9 5HF	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Ande 18/1))	NONE		

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
01-005 cont'd					Lincolnshire Wildlife Trust Banovallum House Manor House Street HORNCASTLE Lincolnshire LN9 5HF	
01-006	В	Permanent Rights over 93177 square metres of agricultural land, drain, ponds and path (East of Roman Bank)	Lincolnshire Wildlife Trust Banovallum House Manor House Street HORNCASTLE Lincolnshire LN9 5HF	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	Alco Estates Limited Boathouse Sandhurst Road Sandilands MABLETHORPE Lincolnshire LN12 2RH (in respect of rights and restrictive covenants stated in Tranfer dated 30 September 2009) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted by Deed dated 14 January 1941)
01-007	В	Permanent Rights over 553 square metres of verge (Roman Bank)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Unknown (in respect of rights and restrictive covenants stated in Tranfer dated 30 September 2009)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
01-007 cont'd			Lincolnshire Wildlife Trust Banovallum House Manor House Street HORNCASTLE Lincolnshire LN9 5HF					
01-008	В	e e	Lincolnshire Wildlife Trust Banovallum House Manor House Street HORNCASTLE Lincolnshire LN9 5HF (as assumed owner) Unknown	NONE	Lincolnshire Wildlife Trust Banovallum House Manor House Street HORNCASTLE Lincolnshire LN9 5HF (as assumed owner) Unknown	Unknown		
01-009	В	square metres of public road and verges (Roman Bank)	Alco Estates Limited Boathouse Sandhurst Road Sandilands MABLETHORPE Lincolnshire LN12 2RH (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus and abandoned underground sewerage apparatus)		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
01-009 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Lincolnshire Wildlife Trust Banovallum House Manor House Street HORNCASTLE Lincolnshire LN9 5HF (assumed in respect of subsoil beneath public highway) R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (assumed in respect of subsoil beneath public highway) Unknown		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 7(1)(a) of the Infrastructure Planning (A Regulations 2009	pplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans	Extent of acquisition of use	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
01-010	C	Permanent Rights over 2154 square metres of verge (Roman Bank)	Sandhurst Road Sandilands MABLETHORPE Lincolnshire LN12 2RH (as assumed owner) R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) Unknown	NONE	Alco Estates Limited Boathouse Sandhurst Road Sandilands MABLETHORPE Lincolnshire LN12 2RH (as assumed owner) R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus and abandoned underground sewerage apparatus) Unknown			
01-011	Freehold Acquisition	Freehold over 177 square metres of public road, verge and drain (Roman Bank)	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner)	NONE	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus and abandoned underground sewerage apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
01-011 cont'd			Unknown		Unknown	Unknown		
01-012	Freehold Acquisition	Freehold over 569 square metres of agricultural land (west of Roman Bank)	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) Unknown	NONE	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) Unknown	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown		
01-013	С	Permanent Rights over 22897 square metres of agricultural land (west of Roman Bank)	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) Unknown	NONE	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) Unknown	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
01-014	C	Permanent Rights over 8249 square metres of agricultural land (west of Roman Bank)	Alco Estates Limited Boathouse Sandhurst Road Sandilands MABLETHORPE Lincolnshire LN12 2RH (as assumed owner) Unknown	R.H. Mowbray Limited Bank House Anderby SKEGNESS LincoInshire PE24 5XJ	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ Unknown	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown		
01-015	C	Permanent Rights over 20085 square metres of agricultural land (west of Roman Bank)	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) Unknown	NONE	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges)		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
01-015 cont'd						Unknown		
01-016	C	Permanent Rights over 34738 square metres of agricultural land (west of Roman Bank)	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) Unknown	NONE	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown		
01-017	D	Permanent Rights over 43296 square metres of agricultural land (west of Roman Bank)	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Ande 19/3))	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)		

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
01-017 cont'd 01-018	D	Permanent Rights over 6314 square metres of agricultural land, drain and access track (west of Roman Bank)	Unknown R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ	NONE	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) Unknown R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of rights reserved by Conveyance dated 16 March 1965)
01-019	К	Temporary Rights over 812 square metres of agricultural land and drain (west of Roman Bank)	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Ande 19/3))	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
01-019 cont'd			Unknown			Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown			
02-001	G	square metres of access track	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Ande 19/1)) R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)			

	Land v	vhich is proposed to be subject to	BOOK OF RI DOCK OF RI DOC (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-001 cont'd					Unknown	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown
02-002	G	square metres of agricultural land and access track (west of Roman Bank)	James Christopher Mowbray Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) John Michael Mowbray Bank House Farm Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner)	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground sewerage apparatus) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))		
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
02-002 cont'd			Robert Edward Mowbray 18 Tothby Lane ALFORD Lincolnshire LN13 0AG (as assumed owner) Unknown			Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown		
02-003	K	Temporary Rights over 719 square metres of agricultural land and drain (west of Roman Bank)	Alco Estates Limited Boathouse Sandhurst Road Sandilands MABLETHORPE Lincolnshire LN12 2RH (as assumed owner) Unknown	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
02-004	К	Temporary Rights over 2278 square metres of agricultural land and hedgerow (west of Roman Bank) and public footpath (LL Chap 19/5)	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ (as assumed owner) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Chap 19/5)) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown			
02-005	K	Temporary Rights over 491 square metres of agricultural land (north of Ember Lane) and public footpath (LL Chap 19/5)		R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Chap 19/5))	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)			

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-005 cont'd			Unknown		R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ Unknown	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown
02-006	D	square metres of agricultural land (north of Ember Lane)	Alco Estates Limited Boathouse Sandhurst Road Sandilands MABLETHORPE Lincolnshire LN12 2RH (as assumed owner) Unknown	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ	R.H. Mowbray Limited Bank House Anderby SKEGNESS Lincolnshire PE24 5XJ Unknown	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1))	a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-007	D	Permanent Rights over 1386 square metres of public road and verges (Ember Lane) and public footpath (LL Chap 21/4)	Alco Estates Limited Boathouse Sandhurst Road Sandilands MABLETHORPE Lincolnshire LN12 2RH (assumed in respect of subsoil beneath public highway) Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Unknown	

	Land	which is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights to ry of Lincolnshire	carry out protective works (Regulation 7(1)	ı(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-008	D	Permanent Rights over 751 square metres of agricultural land (south of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL (in respect of a rights stated in Transfer dated 28 May 2009 and Conveyance dated 22 October 1979) Unknown (in respect of rights granted by Conveyance dated 11 October 1993)	
02-009	D	Permanent Rights over 6152 square metres of agricultural land (south of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	NONE	
02-010	D	Permanent Rights over 1528 square metres of agricultural land (south of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 SRJ	NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	NONE	

	Land	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t y of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-011	D	Permanent Rights over 178 square metres of agricultural land (south of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner) Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Unknown	NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner) Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) LOUTH Lincolnshire LN11 8PG (as assumed owner) UNTH SPG (as assumed owner)	Unknown

	Land v	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-012	D	Permanent Rights over 15650 square metres of agricultural land (south of Ember Lane) and public footpath (LL Chap 27/3)	-	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Chap 27/3))	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground sewerage apparatus)	
02-013	D	Permanent Rights over 15141 square metres of agricultural land (east of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	NONE	

	Land v	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t y of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (App Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-014	K	public footpath (LL Chap 27/3)	Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL I Chap I 27/3))	NONE
02-015	K	square metres of hedgerow (south of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner)	NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)

	Land w	hich is proposed to be subject	BOOK OF REFE t to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ght to use the land, and/or (iii) rights Lincolnshire	to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
02-015 cont'd			Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL (as assumed owner) Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Unknown		Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL (as assumed owner) Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Unknown	Unknown

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-016	К			Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Chap 27/3))	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
02-017	K	Temporary Rights over 1 square metres of agricultural land (south of Ember Lane)	Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner)	NONE	Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
02-017 cont'd			Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Unknown		Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Unknown	Unknown			
02-018	K	(south of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner) Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner)	NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner) Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
02-018 cont'd			Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of mines and minerals) Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Unknown		Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 &PG (as assumed owner) Unknown				
02-019	K	Temporary Rights over 28 square metres of agricultural land (south of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner)		Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
02-019 cont'd			Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Unknown		Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Unknown	Unknown			
02-020	К	Temporary Rights over 2153 square metres of agricultural land (east of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Land	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
02-020 cont'd			Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of mines and minerals)			Unknown (in respect of reservation of mines and minerals and rights to work the same)
02-021	K	Temporary Rights over 16 square metres of agricultural land (south of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner) Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 & PG (as assumed owner) Richard John Hill Highfield Authorpe LOUTH Lincolnshire LOUTH Lincolnshire LOUTH Lincolnshire (as assumed owner) UNTH	NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner) Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (as assumed owner) UNTH Lincolnshire LN11 8PG (as assumed owner)	Unknown

	Land	which is proposed to be subject to	BOOK OF BOOK OF I	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t y of Lincolnshire	o carry out protective works (Regulation 7(1))(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-001	K	Temporary Rights over 47 square metres of agricultural land (east of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner) Unknown	NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
03-002	К	Temporary Rights over 485 square metres of agricultural land (east of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 SRJ	NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
03-003	K	Temporary Rights over 30 square metres of agricultural land (east of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner)	NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-003 cont'd			Unknown		Unknown	Unknown
03-004	К	Temporary Rights over 3 square metres of agricultural land and drain (east of Ember Lane)		NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of reservation of mines and minerals and rights to work the same)
03-005	K	Temporary Rights over 61 square metres of agricultural land and drain (east of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of mines and minerals)	NONE	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of reservation of mines and minerals and rights to work the same)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Extent of acquisition of use	n or use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
03-006	K	Temporary Rights over 248 square metres of agricultural land (east of Ember Lane)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of mines and minerals) Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL	NONE	Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of a rights as stated in Deed dated 28 February 1980 and and in Transfer dated 22 December 1997) Clydesdale Bank PLC 177 Bothwell Street Glasgow G2 7ER (as mortgagee for Pridgeon Farms Limited) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation						
Land Plans		n or use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
03-007	K	Temporary Rights over 106 square metres of agricultural land (east of Ember Lane)	Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL	NONE	Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (in respect of a right to enter and service media rights stated in Transfer dated 28 May 2008) Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 & PG (in respect of rights and restrictive covenants as stated in Transfer dated 28 May 2009) Clydesdale Bank PLC 177 Bothwell Street Glasgow G2 7ER (as mortgagee for Pridgeon Farms Limited)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
03-007 cont'd						Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (in respect of rights and restrictive covenants as stated in Transfer dated 28 May 2009)			
03-008	D	Permanent Rights over 1044 square metres of agricultural land (east of Ember Lane)	Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL	NONE	Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 5RJ (in respect of a right to enter and service media rights stated in Transfer dated 28 May 2008)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Future of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-008 cont'd						Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (in respect of rights and restrictive covenants as stated in Transfer dated 28 May 2009) Clydesdale Bank PLC 177 Bothwell Street Glasgow G2 7ER (as mortgagee for Pridgeon Farms Limited) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-008 cont'd						Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (in respect of rights and restrictive covenants as stated in Transfer dated 28 May 2009)			
03-009	D	square metres of watercourse (Wigg Drain)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of mines and minerals) Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL (as assumed owner) Unknown	NONE	Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown			

	Land	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-010	D	public footpath (LL Hogs 34/4, LL Hogs 28/1)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of mines and minerals) Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpaths (LL Hogs 28/1, LL Hogs 34/4)) Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Clydesdale Bank PLC 177 Bothwell Street Glasgow G2 7ER (as mortgagee for Pridgeon Farms Limited) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition of use	Description of faild	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-010 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)
03-011	К		Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of mines and minerals) Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL	NONE	Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of a rights as stated in Deed dated 28 February 1980 and and in Transfer dated 22 December 1997) Clydesdale Bank PLC 177 Bothwell Street Glasgow G2 7ER (as mortgagee for Pridgeon Farms Limited)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
03-011 cont'd						Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)				
03-012	K	metres of access track (east of Ember Lane)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of mines and minerals) Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL (as assumed owner) Unknown		Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
03-013	D	Permanent Rights over 7847 square metres of agricultural land (north of Langham Road)	Alan Ransom Chestnut Farm Langham Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QF	NONE	Alan Ransom Chestnut Farm Langham Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QF	Unknown (in respect of such right of way as may have existed as at 24 March 1977)			
03-014	К	Temporary Rights over 942 square metres of agricultural land (north of Langham Road)	Alan Ransom Chestnut Farm Langham Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QF	NONE	Alan Ransom Chestnut Farm Langham Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QF	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown (in respect of such right of way as may have existed as at 24 March 1977)			

	Land	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights t of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		Description on tand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-015	D	Permanent Rights over 890 square metres of public road and verge (Langham Road)	Alan Ransom Chestnut Farm Langham Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QF (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL (assumed in respect of subsoil beneath public highway) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-016	D	Permanent Rights over 45 square metres of verge and drain (Langham Road)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of mines and minerals) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL (in respect of subsoil beneath the public highway) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE			

	Land	which is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights y of Lincolnshire	s to carry out protective works (Regulation 7(1))(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Ap Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
03-017	D	Permanent Rights over 16933 square metres of agricultural land (south of Langham Road)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of mines and minerals) Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL	NONE	Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL	Clydesdale Bank PLC 177 Bothwell Street Glasgow G2 7ER (as mortgagee for Pridgeon Farms Limited)
03-018	D	Permanent Rights over 6428 square metres of agricultural land (north of Lowgate Road)	Kenneth Neil Grantham The Barn Langham Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QF	NONE	Kenneth Neil Grantham The Barn Langham Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QF	NONE
03-019	К	Temporary Rights over 1367 square metres of agricultural land (west of Langham Road)	Kenneth Neil Grantham The Barn Langham Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QF	NONE	Kenneth Neil Grantham The Barn Langham Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QF	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acculation or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans	Extent of acquisition or use	Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
03-020	К	Temporary Rights over 1725 square metres of agricultural land (west of Langham Road)	Dennis Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE Irene Annie Paul Wesley House Langham Road Mumby ALFORD Lincolnshire LN13 9SH Martin Paul Lowgate Farm Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR (trading as J Cragg & Son)	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR (trading as J Cragg & Son)	NONE				

	Land	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t y of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-021	D	Permanent Rights over 4522 square metres of agricultural land, hedgerow and drain (north of Lowgate Road)	Dennis Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE Irene Annie Paul Wesley House Langham Road Mumby ALFORD Lincolnshire LN13 9SH Martin Paul Lowgate Farm Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR (trading as J Cragg & Son)	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR (trading as J Cragg & Son)	NONE

	Land v	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-022	D	Permanent Rights over 6180 square metres of agricultural land, hedgerow and drain (north of Lowgate Road)	Dennis Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE Irene Annie Paul Wesley House Langham Road Mumby ALFORD Lincolnshire LN13 9SH Martin Paul Lowgate Farm Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE	Martin Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE	Martin Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunications apparatus)

	Land	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Deceription of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		or use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-023	К	Temporary Rights over 1205 square metres of agricultural land, hedgerow and drain (north of Lowgate Road)	Dennis Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE Irene Annie Paul Wesley House Langham Road Mumby ALFORD Lincolnshire LN13 9SH Martin Paul Lowgate Farm Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE	Martin Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE	Martin Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
03-024	K		Dennis Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE (as assumed owner) Irene Annie Paul Wesley House Langham Road Mumby ALFORD Lincolnshire LN13 9SH (as assumed owner) Martin Paul Lowgate Farm Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE (as assumed owner) Unknown	NONE	Dennis Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE (as assumed owner) Irene Annie Paul Wesley House Langham Road Mumby ALFORD Lincolnshire LN13 9SH (as assumed owner) Martin Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE (as assumed owner) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecoms apparatus) Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on		Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Extent of acquisition of use	f acquisition or use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
03-025	D	Permanent Rights over 122 square metres of agricultural land (south of Lowgate Road)	Dennis Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE Irene Annie Paul Wesley House Langham Road Mumby ALFORD Lincolnshire LN13 9SH Martin Paul Lowgate Farm Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE	Martin Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE	Martin Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE	NONE				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-026	D	Permanent Rights over 675 square metres of public road and verges (Lowgate Road)	Dennis Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE (assumed in respect of subsoil beneath public highway) Irene Annie Paul Wesley House Langham Road Mumby ALFORD Lincolnshire LN13 9SH (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown			

	Land	which is proposed to be subject to	BOOK OF F bo: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-027	К	Temporary Rights over 2424 square metres of agricultural land and drain (south of Lowgate Road)	Dennis Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR (trading as J Cragg & Son)	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR (trading as J Cragg & Son)	Unknown (in respect of rights of drainage reserved by a Transfer dated 5 September 2000)
03-028	D	Permanent Rights over 23668 square metres of agricultural land and drain (west of Lowgate Road)	Dennis Paul Lowgate Farm Lowgate Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QE	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR (trading as J Cragg & Son)	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR (trading as J Cragg & Son)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer apparatus) Unknown (in respect of rights of drainage reserved by a Transfer dated 5 September 2000)
03-029	D	Permanent Rights over 20779 square metres of agricultural land and access track (south of Lowgate Road) and public footpaths (LL Hogs 58/2, LL Hogs 57/1)	Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB	NONE	Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-029 cont'd					Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpaths (LL Hogs 58/2 and LL Hogs 57/1)) Kim Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son) Thomas Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 19 April 2019)
03-030	D	Permanent Rights over 4889 square metres of agricultural land (north of Bracken Lane)	Pell Holdings Ltd Listoft Farm Helsey Lane Hogsthorpe Skegness Lincolnshire PE24 SPE	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Pell Holdings Ltd)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
03-030 cont'd						Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Lease dated 12 December 2018)			
03-031	K	square metres of agricultural land and access track (north of Bracken Lane)	Jack Ward Hillcrest Cumberworth ALFORD LincoInshire LN13 9LB	NONE	Lincolnshire LN13 9LB (trading as L. H. Ward & Son)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 19 April 2019)			

	Land	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-031 cont'd					Thomas Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son)	
04-001	K	Temporary Rights over 242 square metres of agricultural land (west of Lowgate Road)	Pell Holdings Ltd Listoft Farm Helsey Lane Hogsthorpe Skegness Lincolnshire PE24 5PE	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Pell Holdings Limited) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights granted by the Deed of Grant dated 12 December 2018)
04-002	К	Temporary Rights over 621 square metres of agricultural land (north of Bracken Lane)	Pell Holdings Ltd Listoft Farm Helsey Lane Hogsthorpe Skegness Lincolnshire PE24 5PE	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR	Darren Cragg The Elms Orby Road Addlethorpe SKEGNESS Lincolnshire PE24 4TR	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Pell Holdings Ltd)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-002 cont'd						Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Lease dated 12 December 2018)			
04-003	K	square metres of agricultural land and access track (west of Lowgate Road)	Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB	NONE	Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 19 April 2019)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-003 cont'd					Thomas Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son)				
04-004	D	square metres of agricultural land (north of Bracken Lane)	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD	NONE	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 SPD	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in a Lease dated 26 October 2018)			
04-005	K	access track and drain (north of Bracken Lane)	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD (as assumed owner)	NONE	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD (as assumed owner)	Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of an assumed right of access)			

	Land w	hich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii)	ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of occurring or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Land Plans Extent of acquisition or use Des	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-005 cont'd			Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5PB (as assumed owner) Benjamin John Maurice Lowe St. Catherines House All Saints Green Norwich NR1 3GA (as assumed owner) Lindsay Jane Whitehead 2 Penrice Road Little Plumstead NORWICH NR13 5FP (as assumed owner) Unknown		Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5PB (as assumed owner) Benjamin John Maurice Lowe St. Catherines House All Saints Green Norwich NR1 3GA (as assumed owner) Lindsay Jane Whitehead 2 Penrice Road Little Plumstead NORWICH NR13 5FP (as assumed owner) Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-006	D	Permanent Rights over 22399 square metres of agricultural land and electricity cable and pylon (north of Bracken Lane)	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD	NONE	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 26 October 2018)			
04-007	К		Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD (as assumed owner)	NONE	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD (as assumed owner)	Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of an assumed right of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
04-007 cont'd			Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5PB (as assumed owner) Benjamin John Maurice Lowe St. Catherines House All Saints Green Norwich NR1 3GA (as assumed owner)		Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5PB (as assumed owner) Benjamin John Maurice Lowe St. Catherines House All Saints Green Norwich NR1 3GA (as assumed owner)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Caution dated 12 August 2021)			
			Lindsay Jane Whitehead 2 Penrice Road Little Plumstead NORWICH NR13 5FP (as assumed owner) Unknown		Lindsay Jane Whitehead 2 Penrice Road Little Plumstead NORWICH NR13 5FP (as assumed owner) Unknown	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 1 1)(a) of the Infrastructure Planning (Appl Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-008	K	Temporary Rights over 16 square metres of agricultural land and access track (north of Bracken lane)	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD (as assumed owner) Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (as assumed owner) Unknown	NONE	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD (as assumed owner) Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (as assumed owner) Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown			
04-009	K	Temporary Rights over 370 square metres of agricultural land (north of Bracken Lane)	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD	NONE	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
04-009 cont'd						Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 26 October 2018)				
04-010	K		Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD (as assumed owner) Unknown	NONE	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD (as assumed owner)	Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5PB (in respect of an assumed right of access) Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of an assumed right of access)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-010 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown			
04-011	D	square metres of agricultural	Thomas Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB	NONE	Thomas Ward Hillcrest Cumberworth ALFORD Lincolnshire (trading as L. H. Ward & Son) Kim Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son)	L H Ward & Son Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of rights granted by Deed dated 09 April 2019) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deeds dated 9 April 2019 and 26 October 2018)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans		Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
04-011 cont'd					Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son)					
04-012	D	Permanent Rights over 22 square metres of drain (north of Mumby Road, A52)	Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5PB (as assumed owner) Thomas Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (as assumed owner) Unknown	NONE	Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5PB (as assumed owner) Thomas Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (as assumed owner) Unknown	Unknown				

	Land	which is proposed to be subject to	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights t / of Lincolnshire	o carry out protective works (Regulation 7(1))(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-013	F	Temporary Rights over 714 square metres of agricultural land, access track and copse (north of Mumby Road, A52)	Thomas Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB	NONE	Thomas Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son) Kim Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son) Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son)	L H Ward & Son Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of rights granted by Deed dated 09 April 2019) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deeds dated 9 April 2019 and 26 October 2018)	

	Land	which is proposed to be subject to	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights t / of Lincolnshire	o carry out protective works (Regulation 7(1))(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-014	G	Temporary Rights over 7467 square metres of agricultural land, access track and copse (north of Mumby Road, A52)	Thomas Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB	NONE	Thomas Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son) Kim Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son) Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son)	L H Ward & Son Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of rights granted by Deed dated 09 April 2019) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deeds dated 9 April 2019 and 26 October 2018)	

	Land	which is proposed to be subject to	BOOK OF I c: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights y of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-015	K	Temporary Rights over 483 square metres of agricultural land, access track and copse (north of Mumby Road, A52)	Thomas Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB	NONE	Thomas Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son) Kim Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son) Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (trading as L. H. Ward & Son)	L H Ward & Son Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of rights granted by Deed dated 09 April 2019) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deeds dated 9 April 2019 and 26 October 2018)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
04-016	D	Permanent Rights over 14713 square metres of agricultural land (west of Bracken Lane)	Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5PB	NONE	Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5PB	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)				
04-017	D	Permanent Rights over 1499 square metres of public road, verges and hedgerow (Mumby Road, A52)	Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5PB (assumed in respect of subsoil beneath public highway) Brian Edward Taylor Slackholme House Farm Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		e Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers				
04-017 cont'd			David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (assumed in respect of subsoil beneath public highway) Joyce Margaret Blanchard Orchard Farm West End Hogsthorpe SKEGNESS Lincolnshire PE24 5PA (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)			Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-017 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (assumed in respect of subsoil beneath public highway) Unknown						
04-018	D	Permanent Rights over 29900 square metres of agricultural land and access track (south of Mumby Road, A52)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (trading as R G Taylor & Sons) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (trading as R G Taylor & Sons)	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (trading as R G Taylor & Sons) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (trading as R G Taylor & Sons)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Agreement dated 31 December 1973) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to • of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-018 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)
04-019	D	Permanent Rights over 219 square metres of drain (south of Bracken Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
04-019 cont'd			H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (as assumed owner) Jean Patricia Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (as assumed owner) Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)		H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (as assumed owner) Jean Patricia Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (as assumed owner) Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
04-019 cont'd			The Executor of the Estate of the Late Roger Glyn Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (as assumed owner) Unknown		The Executor of the Estate of the Late Roger Glyn Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (as assumed owner) Unknown					
04-020	D	square metres of agricultural	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	NONE	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	Andrew Michael Snell The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 5PG (in respect of rights granted in Conveyance dated 21 January 1993) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as the mortgagee of H. Bradley & Sons (Produce) Limited dated 13th May 2021)				

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-020 cont'd						Gillian Mary Sharpe The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 5PG (in respect of rights granted in Conveyance dated 21 January 1993) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB
04-021	D	square metres of agricultural land and drain (north of Listoft Lane)	Jean Patricia Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (in respect of rights granted by Deed of Grant dated 12 December 2018)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans	Extent of acquisition or use Description of land		Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
04-021 cont'd			The Executor of the Estate of the Late Roger Glyn Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunications apparatus)			
04-022	К	square metres of agricultural land and access splay (north of	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	NONE	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	Andrew Michael Snell The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 5PG (in respect of rights granted in Conveyance dated 21 January 1993) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as the mortgagee of H. Bradley & Sons (Produce) Limited dated 13th May 2021)			

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-022 cont'd						Gillian Mary Sharpe The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 5PG (in respect of rights granted in Conveyance dated 21 January 1993) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunications apparatus) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by Deed of Grant dated 12 December 2018)
04-023	D	Permanent Rights over 3964 square metres of agricultural land and drain (north of Listoft Lane)	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	NONE	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	Andrew Michael Snell The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 5PG (in respect of rights granted in Conveyance dated 21 January 1993)

	Land v	which is proposed to be subject to	BOOK OF RI bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
04-023 cont'd						Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as the mortgagee of H. Bradley & Sons (Produce) Limited dated 13th May 2021) Gillian Mary Sharpe The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 5PG (in respect of rights granted in Conveyance dated 21 January 1993) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunications apparatus)
05-001	К	Temporary Rights over 407 square metres of agricultural land (south of Listoft Lane)	Jean Patricia Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by Deed of Grant dated 12 December 2018)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
05-001 cont'd			The Executor of the Estate of the Late Roger Glyn Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ							
05-002	D	Permanent Rights over 1379 square metres of public road (Listoft Lane)	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (assumed in respect of subsoil beneath public highway) Jean Patricia Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-002 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) The Executor of the Estate of the Late Roger Glyn Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (assumed in respect of subsoil beneath public highway) Unknown			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown			
05-003	D	Permanent Rights over 16258 square metres of agricultural land and drain (south of Listoft Lane)	Jean Patricia Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)			

	Land v	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition of use	Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-003 cont'd			The Executor of the Estate of the Late Roger Glyn Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ			Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by Deed of Grant dated 12 December 2018)
05-004	D	Permanent Rights over 222 square metres of drain (Willoughby High Drain, south of Listoft Lane)	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD (as assumed owner) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner)	NONE	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD (as assumed owner) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-004 cont'd			Jean Patricia Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (as assumed owner) The Executor of the Estate of the Late Roger Glyn Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (as assumed owner) Unknown		Jean Patricia Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (as assumed owner) The Executor of the Estate of the Late Roger Glyn Bradley Southlands Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (as assumed owner) Unknown				
05-005	F	Temporary Rights over 1375 square metres of agricultural land and copse (west of Sloothby High Lane)	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD	NONE	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
05-005 cont'd						Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Deed dated 26 October 2018)			
05-006	D	square metres of agricultural land and copse (west of Sloothby High Lane)	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD		Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Deed dated 26 October 2018)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans		bir of use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
05-007	D	Permanent Rights over 1064 square metres of public road and verges (Sloothby High Lane)	Adrian Martin Walker Ivy House Farm Mumby Road Hogsthorpe SKEGNESS Lincolnshire PE24 5PD (assumed in respect of subsoil beneath public highway) Brenda Jean Wright Wyche Farm Wyche Hogsthorpe SKEGNESS Lincolnshire PE24 5NT (assumed in respect of subsoil beneath public highway) Karen Verena Wright The Elements Beldings Close Firsby SPILSBY Lincolnshire PE23 5QS (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
05-007 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown			
05-008	D	square metres of agricultural land (east of Sloothby High Lane) and drain (Wyche Drain)	Brenda Jean Wright Wyche Farm Wyche Hogsthorpe SKEGNESS Lincolnshire PE24 5NT Karen Verena Wright The Elements Beldings Close Firsby SPILSBY Lincolnshire PE23 5QS	NONE	David Pridgeon The Grange Stain Lane Withern ALFORD LN13 OPF	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Land	which is proposed to be subject to	BOOK OF BOOK OF i	Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) right y of Lincolnshire	s to carry out protective works (Regulation 7(1))(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-009	D	Permanent Rights over 106 square metres of drain (Wyche Drain, east of Sloothby Lane)	Brenda Jean Wright Wyche Farm Wyche Hogsthorpe SKEGNESS Lincolnshire PE24 5NT (as assumed owner) Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (as assumed owner) Karen Verena Wright The Elements Beldings Close Firsby SPILSBY Lincolnshire PE23 5QS (as assumed owner) Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	NONE	Brenda Jean Wright Wyche Farm Wyche Hogsthorpe SKEGNESS Lincolnshire PE24 5NT (as assumed owner) Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (as assumed owner) Karen Verena Wright The Elements Beldings Close Firsby SPILSBY Lincolnshire PE23 5QS (as assumed owner) Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to r of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use Description of land	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-009 cont'd			Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)		Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) David Pridgeon The Grange Stain Lane Withern ALFORD LN13 0PF Unknown	
05-010	D	Permanent Rights over 5883 square metres of agricultural land and copse (east of Sloothby High Lane)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989)

	Land	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to • of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-010 cont'd			Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS			Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)
05-011	D		Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)

	Land	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-011 cont'd			Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS			
05-012	К	Temporary Rights over 42 square metres of agricultural land (east of Sloothby High Lane)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans	Land Plans		Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
05-012 cont'd			Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS						
05-013	К	land (east of Sloothby High Lane)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (in respect of right to lay a copper pipe as stated in a Deed dated the 30 October 1961. Also in respect of right to pass as stated in a Transfer dated the 13 May 2004) British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989)			

	Land w	hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) r	nd Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights to o f Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		·	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-013 cont'd			Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS			David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (in respect of right to lay a copper pipe as stated in a Deed dated the 30 October 1961. Also in respect of right to pass as stated in a Transfer dated the 13 May 2004) Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (in respect of right to lay a copper pipe as stated in a Deed dated the 30 October 1961. Also in respect of a right to pass as stated in a Transfer dated the 13 May 2004)

	Land	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
05-014	D	Permanent Rights over 784 square metres of public road (south Ings Lane)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
05-014 cont'd			Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (assumed in respect of subsoil beneath public highway) Unknown						
05-015	D	Permanent Rights over 1258 square metres of agricultural land and copse (east of Sloothby High Lane)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Hogsthorpe Skegness Lincolnshire	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	NONE			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Plans	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
05-015 cont'd			Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS						
05-016	D	Ŭ	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1))	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-016 cont'd			Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS			
05-017	D	square metres of agricultural	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	NONE

	Land	which is proposed to be subject to	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to y of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-018	D	square metres of agricultural	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)

	Land v	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
05-019	K	square metres of agricultural land (east of South Ings Lane)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Appl Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
05-020	Κ	square metres of access track (east of South Ings Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of rights pursuant to transfer dated 5 January 2021)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on		Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
05-021	D	Permanent Rights over 19541 square metres of agricultural land, access track, drain (east of South Ings Lane) and public footpath (LL Hogs 48/1)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL I Hogs I 48/1))	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of rights pursuant to transfer dated 5 January 2021)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	F 1	Descriptions (local	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Land Plans Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
05-021 cont'd					Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS					
06-001		square metres of drain (east of South Ings Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
06-001 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown				
06-002	D	square metres of agricultural land and access track (east of South Ings Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Land Plans	Description of faild	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
06-002 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS					
06-003	K	square metres of access track (east of South Ings Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Land Plans	Description of faild	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
06-003 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS					
06-004	Κ	square metres of drain (east of South Ings Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
06-004 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown				
06-005	D	square metres of drain (east of South Ings Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
06-005 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown				
06-006	K		Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Future of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Land Plans Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
06-006 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS					
06-007		square metres of agricultural land, drain and copse (east of South Ings Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
Land Plans	Land Plans	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
06-007 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS					
06-008	D	square metres of agricultural	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of transfer dated 5 January 2021)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans	Land Plans Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
06-008 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS					
06-009	K	square metres of access track (east of South Ings Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans	Land Plans Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
06-009 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS					
06-010	D	square metres of agricultural	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Land Plans	Description of faild	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
06-010 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS					
06-011	K	square metres of agricultural	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Brian Edward Taylor and David Robert Norman Taylor and Maxine Hayley Taylor)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Id Plans	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
06-011 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS				
06-012	D	square metres of agricultural	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Brian Edward Taylor and David Robert Norman Taylor and Maxine Hayley Taylor)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Futorst of occursition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	and Plans Extent of acquisition or use Description of land	Description of faild	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
06-012 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS				
06-013	К	square metres of agricultural land (east of South Ings Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Brian Edward Taylor and David Robert Norman Taylor and Maxine Hayley Taylor)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans	d Plans	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
06-013 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS				
06-014	K		Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Unknown (in respect of subjections as stated in conveyance dated 29 September 1989)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	d Plans	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
06-014 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS				
06-015	K	square metres of agricultural	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Brian Edward Taylor and David Robert Norman Taylor and Maxine Hayley Taylor)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans	i Plans	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
06-015 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS				
06-016	К	square metres of agricultural land and drain (east of South Ings Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Unknown (in respect of subjections as stated in conveyance dated 29 September 1989)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	d Plans	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
06-016 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS				
06-017	D	square metres of agricultural	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Unknown (in respect of subjections as stated in conveyance dated 29 September 1989)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
06-017 cont'd			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS				
06-018	D	square metres of public road	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (assumed in respect of subsoil beneath public highway) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
06-018 cont'd			Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (assumed in respect of subsoil beneath public highway)						

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
06-018 cont'd			Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (assumed in respect of subsoil beneath public highway) Maxine Hayley Taylor Slackholme House Farm Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (assumed in respect of subsoil beneath public highway) Unknown					
06-019	D	Permanent Rights over 1583 square metres of agricultural land (west of South Ings Lane)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)		

	Land v	which is proposed to be subject to	BOOK OF RE b: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to c of Lincolnshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans		Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
06-019 cont'd			Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS			British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989)
06-020	K	square metres of agricultural	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Unknown (in respect of subjections as stated in conveyance dated 29 September 1989)

	Land	which is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-020 cont'd			David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	
06-021	D	Permanent Rights over 5950 square metres of agricultural land and drain (west of South Ings Lane)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	NONE

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
06-021 cont'd			Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS			
07-001	K	square metres of agricultural land (west of South Ings Lane)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	NONE

	Land v	which is proposed to be subject to	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to y of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-002	K	Temporary Rights over 1283 square metres of agricultural land and drain (west of South Ings Lane)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)

	Land v	which is proposed to be subject to	BOOK OF RI DOCK OF RI DOC (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-003	D	Ings Lane and north of Marsh Lane)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SNS 6PB (in respect of rights stated in Deed dated 28 January 2019)

	Land v	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t y of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans		Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-004	K	square metres of agricultural land (north of Marsh Lane)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Deed dated 28 January 2019)

	Land	which is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights y of Lincolnshire	s to carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-005	D	Permanent Rights over 199 square metres of drain (north of Marsh Lane)	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (as assumed owner) Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (as assumed owner) Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (as assumed owner) Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	NONE	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (as assumed owner) Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (as assumed owner) Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (as assumed owner) Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
07-005 cont'd			Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown		Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown				
07-006	D	square metres of agricultural land and copse (north of Marsh Lane)	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG	NONE	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for David Peter Hand and Graham Kenneth Hand) Unknown (in respect of the rights granted by the Deed dated 07 December 1966)			

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to [,] of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-007	D	Permanent Rights over 205 square metres of agricultural land and copse (north of Marsh Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ Unknown (in respect of the rights granted by the Deed dated 07 December 1966)
07-008	G	Temporary Rights over 8330 square metres of agricultural land (north of Marsh Lane)	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG	NONE	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for David Peter Hand and Graham Kenneth Hand)

	Land	which is proposed to be subject to	BOOK OF RI b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-008 cont'd						Unknown (in respect of the rights granted by the Deed dated 07 December 1966)	
07-009	G	Temporary Rights over 120 square metres of agricultural land and copse (north of Marsh Lane)	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG	NONE	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of personal covenants stated in Transfer dated 5 January 1994) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
07-010	F	Temporary Rights over 240 square metres of access splay (south of Marsh Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (in respect of an assumed right of access)	

	Land v	vhich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) r	nd Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights to f Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Futorst of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition or use	e Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
07-010 cont'd						Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (in respect of an assumed right of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS (in respect of an assumed right of access)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
07-010 cont'd						Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (in respect of an assumed right of access) Unknown (in respect of covenants contained in Conveyance dated 22 February 1991)			
07-011	F	Temporary Rights over 1026 square metres of agricultural land (south of Marsh Lane)	Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner)	NONE	Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL (as assumed owner) Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ (as assumed owner)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (in respect of an assumed right of way)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
07-011 cont'd			Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ (as assumed owner) Unknown		Unknown	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (in respect of an assumed right of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS (in respect of an assumed right of access)			

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
07-011 cont'd						Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (in respect of an assumed right of access) Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (in respect of an assumed right of access) Unknown
07-012	G	square metres of agricultural land (south of Marsh Lane)	Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL (as assumed owner)	NONE	Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL (as assumed owner)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
07-012 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ (as assumed owner) Unknown		Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ (as assumed owner) Unknown	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (in respect of a right of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown			
08-001	F	Temporary Rights over 17 square metres of access splay (north of Marsh Lane)	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (as assumed owner)	NONE	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Land v	vhich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-001 cont'd			Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (as assumed owner) Unknown		Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (as assumed owner) Unknown	Unknown
08-002	F			NONE	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (as assumed owner) Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of an option agreement dated 29 March 2019) Unknown

	Land	which is proposed to be subject to	BOOK OF F b: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t y of Lincolnshire	o carry out protective works (Regulation 7(1)	ı(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Appl Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-003	F	Temporary Rights over 626 square metres of agricultural land and access track (north of Marsh Lane)	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG	NONE	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for David Peter Hand and Graham Kenneth Hand) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants granted by Deed dated 29 March 2019) Unknown (in respect of the rights granted by the Deed dated 07 December 1966)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-004	D	Permanent Rights over 1 square metres of drain (north of Marsh Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (in respect of rights as stated in Conveyance dated 18 February 1991) Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (in respect of rights as stated in Conveyance dated 18 February 1991)			
08-005	D	Permanent Rights over 844 square metres of drain (north of Marsh Lane)	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)			

	Land w	hich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii) r	ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-005 cont'd			Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL (assumed in respect of subsoil beneath public highway) Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)			Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
08-005 cont'd			Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ (assumed in respect of subsoil beneath public highway) Unknown						
08-006	D	square metres of agricultural	Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner)	NONE	Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown			

	Land v	which is proposed to be subject to	BOOK OF RI DOCK OF RI DOC (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
08-006 cont'd			Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ (as assumed owner) Unknown		Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ (as assumed owner) Unknown	
08-007	D	square metres of agricultural land (south of Marsh Lane)	Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (trading as F Simpson & Son Ltd)	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (trading as F Simpson & Son Ltd)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
08-007 cont'd						RLS House Limited Alex House 260-268 Chapel Street SALFORD Greater Manchester M3 5JZ (in respect of informal rights agreement for car parking) Unknown (in respect of a right of way as contained in Conveyance dated 06 April 1962)			
08-008	F		Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ	SKEGNESS Lincolnshire	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (trading as F Simpson & Son Ltd)	RLS House Limited Alex House 260-268 Chapel Street SALFORD Greater Manchester M3 5JZ (in respect of informal rights agreement for car parking) Unknown (in respect of a right of way as contained in Conveyance dated 06 April 1962)			

	Land	which is proposed to be subject to	BOOK OF F bo: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-009	G	Temporary Rights over 7624 square metres of agricultural land (south of Marsh Lane)	Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (trading as F Simpson & Son Ltd)	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (trading as F Simpson & Son Ltd)	RLS House Limited Alex House 260-268 Chapel Street SALFORD Greater Manchester M3 5JZ (in respect of informal rights agreement for car parking) Unknown (in respect of a right of way as contained in Conveyance dated 06 April 1962)	
08-010	К	Temporary Rights over 48 square metres of agricultural land (south of Marsh Lane)	Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (trading as F Simpson & Son Ltd)	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (trading as F Simpson & Son Ltd)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) RLS House Limited Alex House 260-268 Chapel Street SALFORD Greater Manchester M3 5JZ (in respect of informal rights agreement for car parking)	

	Land v	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-010 cont'd						Unknown (in respect of a right of way as contained in Conveyance dated 06 April 1962)	
08-011	Κ	Temporary Rights over 7 square metres of access track and drain (east of Skegness Stadium, Marsh Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL (as assumed owner)	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-011 cont'd			Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ (as assumed owner) Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown		Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ (as assumed owner) Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown	
08-012	К	Temporary Rights over 77 square metres of access track (east of Skegness Stadium)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
08-012 cont'd			David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of a Wayleave Agreement relating to the construction and maintenance of an electric line dated 01 April 1992)			
08-013	D	Permanent Rights over 99 square metres of drain (east of Skegness Stadium)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Land w	/hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) r	ind Development Consent Order FERENCE - PART 1 ight to use the land, and/or (iii) rights f Lincolnshire	to carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-013 cont'd			David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL (as assumed owner) Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ (as assumed owner) Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)		David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Eileen Clarkson 148 High Street Hook GOOLE DN14 5PL (as assumed owner) Mary Mackinder Broadview Toynton All Saints SPILSBY Lincolnshire PE23 5AJ (as assumed owner) Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	Unknown	

	Land v	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of hand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-013 cont'd			Unknown		Unknown		
08-014	D	Permanent Rights over 29191 square metres of agricultural land (south of Marsh Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	NONE	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of a Wayleave Agreement relating to the construction and maintenance of an electric line dated 01 April 1992)	
			Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS		

	Land	which is proposed to be subject t	BOOK OF BOOK OF I	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) right y of Lincolnshire	s to carry out protective works (Regulation 7(1)	ı(a))		
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
08-015	D	Permanent Rights over 123 square metres of access track (south of Marsh Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (as assumed owner) Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	NONE	Brian Edward TaylorSlackholme House FarmSlackholme EndHogsthorpeSKEGNESSLincolnshirePE24 5NS(as assumed owner)David Robert Norman TaylorSlackholme House FarmSlackholme EndHogsthorpeSKEGNESSLincolnshirePE24 5NS(as assumed owner)Julia Clare PoundAyshford CourtWestleighTIVERTONEX16 7HL(as assumed owner)Mark Leopold CaudwellHowlett HouseSlackholme EndHogsthorpeSKEGNESSLincolnshirePE24 5NS(as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown		

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-015 cont'd			Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown		Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown	
08-016	D	Permanent Rights over 48130 square metres of agricultural land, access track and drain (north of Ingoldmells Road)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer apparatus)

	Land	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of Iana	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-016 cont'd			Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS			Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (In respect of rights as stated in deed dated 28 January 2019) Unknown (in respect of rights contained in Conveyance dated 08 September 1948)
08-017	К	Temporary Rights over 763 square metres of access track (north of Ingoldmells Road)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer apparatus)

	Land	which is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
08-017 cont'd			Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS			Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (In respect of rights as stated in deed dated 28 January 2019) Unknown (in respect of rights contained in Conveyance dated 08 September 1948)
09-001	K	Temporary Rights over 26 square metres of access splay (south of Ingoldmells Road)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (as assumed owner)	NONE	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-001 cont'd			Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown		Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Unknown	Unknown			
09-002	D	Permanent Rights over 976 square metres of public road and drain (Ingoldmells Road)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
09-002 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (assumed in respect of subsoil beneath public highway) Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (assumed in respect of subsoil beneath public highway) Unknown			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and			

	Land	which is proposed to be subject t	BOOK OF I o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t y of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-003	D	Permanent Rights over 152 square metres of drain (south of Ingoldmells Road)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (as assumed owner) Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) PE24 5NS (as assumed owner) Unknown	NONE	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (as assumed owner) Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) E24 5NS (as assumed owner) Unknown	Unknown

	Land v	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use	Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-004	D	Permanent Rights over 59860 square metres of agricultural land and drains (south of Ingoldmells Road and north of Younger's Lane)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of right of way for a drift road reserved by a conveyance dated 8 September 1948)

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to / of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-005	K	Temporary Rights over 9375 square metres of access track and drain (north of Younger's Lane and south of Ingoldmells Road)	Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness Lincolnshire PE24 5NS	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of right of way for a drift road reserved by a conveyance dated 8 September 1948)

	Land	which is proposed to be subject to	BOOK OF o: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) right y of Lincolnshire	s to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (A) Regulations 2009	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-006	D	Permanent Rights over 167 square metres drain (north of Younger's Lane)	J E Spence & Son Farms Ltd Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN (as assumed owner) Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (as assumed owner) Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner) Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (as assumed owner)	NONE	J E Spence & Son Farms Ltd Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 SAN (as assumed owner) Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (as assumed owner) Mark Leopold Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 SNS (as assumed owner) Maureen Teresa Caudwell Howlett House Slackholme End Howlett House Slackholme End Howlett House Slackholme End Howlett House Slackholme End Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 SNS (as assumed owner)	Unknown

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-006 cont'd			Unknown		Unknown		
09-007	L	square metres of agricultural land and copse (north of	J E Spence & Son Farms Ltd Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN	NONE	J E Spence & Son Farms Ltd Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead apparatus)	
09-008	L	square metres of agricultural land (north of Younger's Lane)	Andrew James Spence Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN Tracey Elizabeth Spence Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN	NONE	Andrew James Spence Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN Tracey Elizabeth Spence Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN	J E Spence & Son Farms Ltd Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN (in respect of a right to install water pipes granted in Deed dated 06 June 1961)	

	Land	which is proposed to be subject to	BOOK OF RI DOCK OF RI DOC (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-009	L	Permanent Rights over 30 square metres of verge (north of Younger's Lane)	Andrew James Spence Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN (as assumed owner) Tracey Elizabeth Spence Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN (as assumed owner) Unknown	NONE	Andrew James Spence Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN (as assumed owner) Tracey Elizabeth Spence Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN (as assumed owner) Unknown	Unknown
09-010	L	Permanent Rights over 7 square metres of verge (north of Younger's Lane)	Andrew James Spence Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN	NONE	Andrew James Spence Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN	L.J. Fairburn & Son Limited Ivy House Farm Farlesthorpe Road Bilsby ALFORD Lincolnshire LN13 9PL (in respect of rights of way and covenants listed in a Conveyance dated 30 November 2000 & in relation to rights granted in a Conveyance dated 02 March 1981)

	Land	which is proposed to be subject to	BOOK OF RE DOCK OF RE compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
09-010 cont'd			Tracey Elizabeth Spence Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN		Tracey Elizabeth Spence Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN	
09-011	D	Permanent Rights over 16026 square metres of agricultural land and copse (north of Younger's Lane)	J E Spence & Son Farms Ltd Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 SAN	NONE	J E Spence & Son Farms Ltd Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN	NONE
09-012	D	Permanent Rights over 269 square metres of hedgerow and copse (north of Younger's Lane)	J E Spence & Son Farms Ltd Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN (as assumed owner) Julie Anne Mason New Field Farm Youngers Lane Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JQ (as assumed owner)	NONE	J E Spence & Son Farms Ltd Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN (as assumed owner) Julie Anne Mason New Field Farm Youngers Lane Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JQ (as assumed owner)	Unknown

	Land v	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights ry of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	lications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-012 cont'd			Unknown		Unknown	
09-013	D	Permanent Rights over 14195 square metres of agricultural land, copse and scrubland (north of Younger's Lane)	Julie Anne Mason New Field Farm Youngers Lane Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JQ	NONE	Julie Anne Mason New Field Farm Youngers Lane Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JQ	Welton Aggregates Limited Welton Quarry Bluestone Heath Road Welton-le-Marsh SPILSBY PE23 5SG (as beneficiary of unilateral notice dated 17 May 2004 & 15 July 2004)
09-014	K	Temporary Rights over 1965 square metres of agricultural land, access track, hardstanding and outbuildings (north of Younger's Lane)	Julie Anne Mason New Field Farm Youngers Lane Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JQ	NONE	Julie Anne Mason New Field Farm Youngers Lane Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JQ	Welton Aggregates Limited Welton Quarry Bluestone Heath Road Welton-le-Marsh SPILSBY PE23 5SG (as beneficiary of unilateral notice dated 17 May 2004 & 15 July 2004)
09-015	K	Temporary Rights over 29 square metres of access track (north of Younger's Lane)	Julie Anne Mason New Field Farm Youngers Lane Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JQ (as assumed owner) Unknown	NONE	Julie Anne Mason New Field Farm Youngers Lane Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JQ (as assumed owner) Unknown	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
09-016	D	Permanent Rights over 223 square metres of verge and hedgerow (North of Younger's Lane)	Julie Anne Mason New Field Farm Youngers Lane Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JQ (as assumed owner) Unknown	NONE	Julie Anne Mason New Field Farm Youngers Lane Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JQ (as assumed owner) Unknown	Unknown			
09-017	D	Permanent Rights over 718 square metres of public road and verge (Younger's Lane)	Julie Anne Mason New Field Farm Youngers Lane Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JQ (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Land Plans		Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
09-017 cont'd			Samuel Kinning 20 West End Burgh Le Marsh SKEGNESS PE24 5EY (assumed in respect of subsoil beneath public highway) Unknown			Unknown			
09-018	D	square metres of agricultural land (south of Younger's Lane)	Samuel Kinning 20 West End Burgh Le Marsh SKEGNESS PE24 5EY	NONE	Ritson Riggall Croft House Bratoft SKEGNESS Lincolnshire PE24 5AP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			

	Land	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights ty of Lincolnshire	to carry out protective works (Regulation 7(1)	i(a))
Number on	Extent of acquisition or use		Qualifying persons under Regulation	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-019	К	Temporary Rights over 729 square metres of agricultural land and verge (south of Younger's Lane)	Samuel Kinning 20 West End Burgh Le Marsh SKEGNESS PE24 5EY	NONE	Ritson Riggall Croft House Bratoft SKEGNESS Lincolnshire PE24 5AP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (assumed in respect of rights of access)
09-020	K	Temporary Rights over 1011 square metres of agricultural land and copse (south of Younger's Lane)	John George Needham Rose Dene 1 Mill Lane Thimbleby HORNCASTLE Lincolnshire LN9 5JS	NONE	John George Needham Rose Dene 1 Mill Lane Thimbleby HORNCASTLE Lincolnshire LN9 5JS	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)
09-021	К	Temporary Rights over 747 square metres of agricultural land (south of Younger's Lane)	Henry Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU Unknown	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (trading as F Simpson & Son Ltd)	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS LincoInshire PE24 4TU (trading as F Simpson & Son Ltd) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
09-022	D	Permanent Rights over 16419 square metres of agricultural land and drains (south of Younger's Lane)	Henry Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU Unknown	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (trading as F Simpson & Son Ltd)	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (trading as F Simpson & Son Ltd) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown
10-001	К	Temporary Rights over 13692 square metres of agricultural land (south of Younger's Lane)	Graham Aliwell 73 Tower Road Boston Lincolnshire PE21 9AU (as assumed owner)	NONE	Graham Aliwell 73 Tower Road Boston Lincolnshire PE21 9AU (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
10-001 cont'd			Unknown		Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown
10-002	D	square metres of agricultural land, copse and drains (south of Younger's Lane)	Graham Aliwell 73 Tower Road Boston Lincolnshire PE21 9AU (as assumed owner) Unknown	NONE	Graham Aliwell 73 Tower Road Boston Lincolnshire PE21 9AU (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use		Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
10-003	K	Temporary Rights over 9204 square metres of agricultural land (west of Middlemarsh Road)	Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB Judith Marjorie Mackinder 4 Alexander Way Burgh Le Marsh SKEGNESS PE24 5JN	Claire Margaret Wilson The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (trading as Chestnut Beef) Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (trading as Chestnut Beef)	Claire Margaret Wilson The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (trading as Chestnut Beef) Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (trading as Chestnut Beef)	Annie Ruth Dalrymple The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (in respect of rights reserved by Transfer dated 22 December 2009) Claire Margaret Wilson The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (in respect of rights reserved by Transfer dated 22 December 2009) Luke Colin Martin Mackinder The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (in respect of rights reserved by Transfer dated 22 December 2009) Luke Colin Martin Mackinder The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (in respect of rights reserved by Transfer dated 22 December 2009)				

	Land w	hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) ri	nd Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights to f Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Extent of acquisition or use Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-003 cont'd						Mary Ann Mackinder-Jonas The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (in respect of rights reserved by Transfer dated 22 December 2009) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (in respect of rights reserved by Transfer dated 22 December 2009)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-003 cont'd						The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as the mortgagee of Skeg Grain Limited dated 28th August 2012)			
10-004	D	square metres of agricultural land, copse and drains (west of Middlemarsh Road)	Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB Judith Marjorie Mackinder 4 Alexander Way Burgh Le Marsh SKEGNESS PE24 5JN	Claire Margaret Wilson The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (trading as Chestnut Beef) Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (trading as Chestnut Beef)	Claire Margaret Wilson The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (trading as Chestnut Beef) Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (trading as Chestnut Beef)	Annie Ruth Dalrymple The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (in respect of rights reserved by Transfer dated 22 December 2009) Claire Margaret Wilson The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (in respect of rights reserved by Transfer dated 22 December 2009)			

	Land w	hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) r	nd Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights to f Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Extent of acquisition or use Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition or use		Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-004 cont'd						Luke Colin Martin Mackinder The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (in respect of rights reserved by Transfer dated 22 December 2009) Mary Ann Mackinder-Jonas The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (in respect of rights reserved by Transfer dated 22 December 2009) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB	

	Land	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-004 cont'd						SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (in respect of rights reserved by Transfer dated 22 December 2009) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as the mortgagee of Skeg Grain Limited dated 28th August 2012)
10-005	D	Permanent Rights over 212 square metres of copse and drain (west of Middlemarsh Road)	Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (as assumed owner) Judith Marjorie Mackinder 4 Alexander Way Burgh Le Marsh SKEGNESS PE24 5JN (as assumed owner)	NONE	Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (as assumed owner) Judith Marjorie Mackinder 4 Alexander Way Burgh Le Marsh SKEGNESS PE24 5JN (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	lications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-005 cont'd			Unknown		Unknown				
cont u									
10-006	D	Permanent Rights over 26361 square metres of agricultural land and drains (north of Skegness Road, A158)	Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB Judith Marjorie Mackinder 4 Alexander Way Burgh Le Marsh SKEGNESS PE24 5JN	Claire Margaret Wilson The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (trading as Chestnut Beef) Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (trading as Chestnut Beef)	Claire Margaret Wilson The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (trading as Chestnut Beef) Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (trading as Chestnut Beef)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (in respect of rights reserved by Conveyance dated 05 February 1898) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (in respect of rights reserved by			

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use	Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-006 cont'd						Ian George Read Meadow House Main Road Langton HORNCASTLE LincoInshire LN9 5JT (in respect of rights reserved by Conveyance dated 05 February 1898) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (in respect of riparian rights)
10-007	K	land, access tracks and drains (north of Skegness Road, A158)	Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB	Claire Margaret Wilson The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (trading as Chestnut Beef)	Claire Margaret Wilson The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (trading as Chestnut Beef)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (in respect of rights reserved by Conveyance dated 05 February 1898)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-007 cont'd			Judith Marjorie Mackinder 4 Alexander Way Burgh Le Marsh SKEGNESS PE24 5JN	Joseph Robert Martin Mackinder North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (trading as Chestnut Beef)	Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (trading as Chestnut Beef)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (in respect of rights reserved by Conveyance dated 05 February 1898) Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (in respect of rights reserved by Conveyance dated 05 February 1898)			

	Land v	vhich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquicition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans	Land Plans Extent of acquisition or use		Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-007 cont'd						Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)
10-008	Κ	Skegness Road, A158)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA LincoInshire LN10 GYH (as assumed owner) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA LincoInshire LN10 GYH (as assumed owner)	NONE	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-008 cont'd			Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (as assumed owner) Unknown		lan George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (as assumed owner) Unknown				
10-009	K	and drain (north of Skegness Road, A158)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner)	NONE	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown			

	Land w	hich is proposed to be subject	BOOK OF REF t to: (i) powers of compulsory acquisition, (ii) ri	nd Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights f Lincolnshire	to carry out protective works (Regulation 7(1)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Ap Regulations 2009	olications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-009 cont'd			Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (as assumed owner) Joseph Robert Martin Mackinder North Beck Tarm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (as assumed owner) Judith Marjorie Mackinder 4 Alexander Way Burgh Le Marsh SKEGNESS PE24 5JN (as assumed owner) Unknown		Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (as assumed owner) Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (as assumed owner) Judith Marjorie Mackinder 4 Alexander Way Burgh Le Marsh SKEGNESS PE24 5JN (as assumed owner) Unknown	

	Land	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights t y of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-010	К	Temporary Rights over 880 square metres of agricultural land (north of Skegness Road, A158) and public footpath (LL BurM 265/2)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT	NONE	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)

	Land v	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use	Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-010 cont'd					Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL BurM 265/2))	
10-011	D	Permanent Rights over 244 square metres of drain (north of Skegness Road, A158)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner)	NONE	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown

	Land w	hich is proposed to be subject	BOOK OF REF t to: (i) powers of compulsory acquisition, (ii) ri	nd Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights f Lincolnshire	to carry out protective works (Regulation 7(1)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-011 cont'd			Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (as assumed owner) Joseph Robert Martin Mackinder North Beck Tarm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (as assumed owner) Judith Marjorie Mackinder 4 Alexander Way Burgh Le Marsh SKEGNESS PE24 5JN (as assumed owner) Unknown		Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (as assumed owner) Joseph Robert Martin Mackinder Northbeck Farm North Beck Lane Hundleby SPILSBY Lincolnshire PE23 5NB (as assumed owner) Judith Marjorie Mackinder 4 Alexander Way Burgh Le Marsh SKEGNESS PE24 5JN (as assumed owner) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans		cquisition or use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
10-012	D	Permanent Rights over 35246 square metres of agricultural land and copse (north of Skegness Road, A158) and public footpath (LL BurM 265/2)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT	NONE	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT	NONE				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on		Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Land Plans Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-012 cont'd					Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL BurM 265/2))				
10-013	G	Temporary Rights over 9051 square metres of Skegness Road, A158)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH	NONE	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH	NONE			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acculation or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Land Plans Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
10-013 cont'd			Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT		lan George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT					
10-014	F	Temporary Rights over 65 square metres of Skegness Road, A158)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH	NONE	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH	NONE				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on		Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Land Plans Extent of acquisition or use [Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
10-014 cont'd			lan George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT		lan George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT					
10-015	G	of Skegness Road, A158)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner)	NONE	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner)	Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-015 cont'd			Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (as assumed owner) Unknown		lan George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (as assumed owner) Unknown				
10-016	D	of Skegness Road, A158)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner)	NONE	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-016 cont'd			Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (as assumed owner) Unknown		Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (as assumed owner) Unknown				
10-017		Permanent Rights over 2998 square metres of public road and verge (Skegness Road, A158)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (assumed in respect of subsoil beneath public highway) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers				
10-017 cont'd			Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG (assumed in respect of subsoil beneath public highway) Unknown			Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of apparatus) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-018	D	Permanent Rights over 23231 square metres of agricultural land (south of Skegness Road, A158)	The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG (as assumed owner)	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	Unknown			
			Unknown		Unknown				
10-019	G	Temporary Rights over 24721 square metres of agricultural land (south of Skegness Road, A158)	The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG (as assumed owner) Unknown	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA Unknown	Unknown			
10-020	К	Temporary Rights over 7 square metres of agricultural land (south of Skegness Road, A158)	The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG (as assumed owner) Unknown	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA Unknown	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-021	К	square metres of agricultural land (south of Skegness Road, A158) and public footpath (LL BurM 260/1)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Edward Martin Read Honeyhole	NONE	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH Edward Martin Read Honeyhole	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights relating to a water main) National Grid Electricity Distribution (East Midlands) PLC			
			Moor Lane Roughton WOODHALL SPA LincoInshire LN10 6YH		Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH	Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			
			Ian George Read Meadow House Main Road Langton HORNCASTLE LincoInshire LN9 5JT		Ian George Read Meadow House Main Road Langton HORNCASTLE LincoInshire LN9 5JT	Unknown (in respect of rights in Conveyance dated 11 October 1928)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Futorst of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Land Plans Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
10-021 cont'd					Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL BurM 260/1))				
10-022	K	land (south of Skegness Road, A158) and public footpath (LL BurM 260/1)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner)	NONE	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (as assumed owner)	NONE			

	Land	which is proposed to be subject to	BOOK OF R b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-022 cont'd			lan George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (as assumed owner) Unknown		Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL BurM 260/1)) Unknown	
11-001	К	Temporary Rights over 97 square metres of agricultural land (south of Skegness Road, A158)	The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG (as assumed owner) Unknown	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA Unknown	Unknown

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-002	Κ	Temporary Rights over 163 square metres of agricultural land and drain (south of Skegness Road, A158)	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 21 June 2018) Unknown
11-003	D	Permanent Rights over 20065 square metres of agricultural land and drains (south of Skegness Road, A158)	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 21 June 2018) Unknown

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order :FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-004	K	Temporary Rights over 958 square metres of agricultural land (south of Skegness Road, A158)	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 21 June 2018) Unknown
11-005	К	Temporary Rights over 1202 square metres of agricultural land (south of Skegness Road, A158)	The President And Scholars Of The College Of Saint Mary Magdalen In The University Of Oxford Magdalen College High Street OXFORD Oxfordshire OX1 4AU (as assumed owner) Unknown	Richard Dobson College Farm Gunby Road Candlesby SPILSBY Lincolnshire PE23 5SB	Richard Dobson College Farm Gunby Road Candlesby SPILSBY Lincolnshire PE23 5SB Unknown	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 2 July 2019) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
11-006	D	Permanent Rights over 7646 square metres of agricultural land and hedgerow (west of Middlemarsh Road) and public footpath (LL BurM 261/3)	The President And Scholars Of The College Of Saint Mary Magdalen In The University Of Oxford Magdalen College High Street OXFORD Oxfordshire OX1 4AU (as assumed owner) Unknown	Richard Dobson College Farm Gunby Road Candlesby SPILSBY Lincolnshire PE23 5SB	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL BurM 261/3)) Richard Dobson College Farm Gunby Road Candlesby SPILSBY Lincolnshire PE23 5SB Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 2 July 2019) Unknown				
11-007	D	Permanent Rights over 2825 square metres of agricultural land (west of Middlemarsh Road) and public footpath (LL BurM 261/3)	Janet Dodsworth The Hollies 38 West End Burgh Le Marsh SKEGNESS Lincolnshire PE24 5EF	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL BurM 261/3))	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Janet Dodsworth and Simon James Dodsworth)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
11-007 cont'd			Simon James Dodsworth The Hollies 38 West End Burgh Le Marsh SKEGNESS Lincolnshire PE24 5EF			National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			
11-008	D		Ian Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD (as assumed owner) The Executor of the Estate of the Late James Leonard Dodsworth The Hollies 38 West End Burgh Le Marsh SKEGNESS Lincolnshire PE24 5EF (as assumed owner) Unknown	NONE	Ian Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD (as assumed owner) The Executor of the Estate of the Late James Leonard Dodsworth The Hollies 38 West End Burgh Le Marsh SKEGNESS Lincolnshire PE24 5EF (as assumed owner) Unknown	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation						
Land Plans		Description of faild	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
11-009	К	Middlemarsh Road) and public	Janet Dodsworth The Hollies 38 West End Burgh Le Marsh SKEGNESS Lincolnshire PE24 5EF Simon James Dodsworth The Hollies 38 West End Burgh Le Marsh SKEGNESS Lincolnshire PE24 5EF	NONE	Janet Dodsworth The Hollies 38 West End Burgh Le Marsh SKEGNESS Lincolnshire PE24 5EF Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL BurM 261/3)) Simon James Dodsworth The Hollies 38 West End Burgh Le Marsh SKEGNESS Lincolnshire PE24 5EF	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Janet Dodsworth and Simon James Dodsworth)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-010	D	Permanent Rights over 18844 square metres of agricultural land and drain (north of Billgate Lane)	lan Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD	NONE	Ian Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Ian Chamberlain)			
11-011	D	Permanent Rights over 167 square metres of agricultural land (north of Billgate Lane)	lan Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD (as assumed owner) Unknown	NONE	Ian Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD (as assumed owner) Unknown	Unknown			

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use	Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-012	D	square metres of agricultural land and drain (north of Billgate Lane)	lan Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD	NONE	lan Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD	NONE
11-013	K	square metres of agricultural land (north of Billgate Lane) and public footpath (LL BurM 263/1)	lan Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD	NONE	Ian Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpaths (LL BurM I 263/1))	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Ian Chamberlain)

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-014	K	Temporary Rights over 1359 square metres of agricultural land (north of Billgate Lane)	lan Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD	NONE	Ian Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Ian Chamberlain)
11-015	D	Permanent Rights over 10 square metres of verge (north of Billgate Lane)	lan Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD (in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Ian Chamberlain)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
11-015 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)							
11-016	D	Lane)	Ian Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-017	D	Permanent Rights over 260 square metres of agricultural land (south of Billgate Lane)	Ian Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD (as assumed owner) Unknown	NONE	lan Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD (as assumed owner) Unknown	Unknown			
11-018	D	Permanent Rights over 15337 square metres of agricultural land and drain (south of Billgate Lane)	lan Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD	NONE	lan Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Ian Chamberlain)			
11-019	D	Permanent Rights over 7460 square metres of agricultural land (north of Middlemarsh Road) and public footpath (LL BurM 263/2)	lan Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD	NONE	Ian Chamberlain Jenkenis Middlemarsh Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AD	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning						
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
11-019 cont'd					Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL BurM 263/2))	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Ian Chamberlain)				
11-020	D	square metres of public road and verges (Middlemarsh Road)	Jenkenis	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Land Plans		Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
11-020 cont'd			SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (assumed in respect of subsoil beneath public highway) The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG (assumed in respect of subsoil beneath public highway) Unknown			Unknown			
11-021	D	5	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (as assumed owner) Unknown	NONE	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (as assumed owner) Unknown	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
11-022	D	Permanent Rights over 110 square metres of access track (south of Middlemarsh Road)	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (as assumed owner) Unknown	NONE	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (as assumed owner) Unknown	Unknown			
11-023	D	Permanent Rights over 20163 square metres of agricultural land (east of Middlemarsh Road)	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF	NONE	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			

	Land v	which is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
11-023 cont'd						The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as the mortgagee of Skeg Grain Limited dated 8th December 2009) Unknown (in respect of rights granted by Deed dated 04 August 1981)
12-001	K	Temporary Rights over 20 square metres of agricultural land (east of Middlemarsh Road)	The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG (as assumed owner) Unknown	NONE	The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus)

	Land v	which is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights y of Lincolnshire	to carry out protective works (Regulation 7(1)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 '(1)(a) of the Infrastructure Planning (Ap Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans	Extent of acquisition of use	Description of hard	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
12-001 cont'd						Unknown
12-002	K	Temporary Rights over 1500 square metres of agricultural land (east of Middlemarsh Road)	The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground and overhead electricity apparatus)
12-003	K	Temporary Rights over 43 square metres of agricultural land and drain (east of Middlemarsh Road)	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (as assumed owner)	NONE	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-003 cont'd			The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG (as assumed owner) Unknown		The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG (as assumed owner) Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown			
12-004	K	land (east of Middlemarsh	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF	NONE	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
12-004 cont'd						The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as the mortgagee of Skeg Grain Limited dated 8th December 2009) Unknown (in respect of rights granted by Deed dated 04 August 1981)			
12-005	D	Permanent Rights over 304 square metres of agricultural land (east of Middlemarsh Road)	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (as assumed owner) Unknown	NONE	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-005 cont'd						Unknown			
12-006	D	Permanent Rights over 1271 square metres of agricultural land (east of Middlemarsh Road)	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF	NONE	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Skeg Grain Limited)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-007	D	square metres of drain (east of Middlemarsh Road)	SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (as assumed owner) The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG (as assumed owner) Unknown	NONE	PE24 4UF (as assumed owner) The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown			
12-008	D	square metres of agricultural	The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-009	D	Permanent Rights over 133 square metres of drain (east of Middlemarsh Road)	Low Road Croft SKEGNESS LincoInshire PE24 4SA (as assumed owner) The Guttridge Family Foundation 16 Algitha Road SKEGNESS LincoInshire PE25 2AG (as assumed owner) Unknown	NONE	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA (as assumed owner) The Guttridge Family Foundation 16 Algitha Road SKEGNESS Lincolnshire PE25 2AG (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown			
12-010	D	Permanent Rights over 33143 square metres of agricultural land and drains (north of Low Road)	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	NONE	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	June Smith Park Farm Main Road Roughton WOODHALL SPA LincoInshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-010 cont'd						Lindsey Marsh Drainage Board Wellington House Manby Park Manby Park LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of deed of grant for underground and overhead electricity apparatus dated 6 May 1939) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Stephen Chamberlain)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
12-010 cont'd						The Executor of the Estate of the Late Andrew Leslie Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)			
12-011	K	square metres of private road, verges, agricultural land and access tracks (north of Low Road)	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	NONE	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	June Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-011 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Stephen Chamberlain) The Executor of the Estate of the Late Andrew Leslie Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-012	D	Permanent Rights over 7488 square metres of agricultural land (north of Low Road)	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	NONE	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	NONE			
12-013	D	Permanent Rights over 9271 square metres of private road, verges and agricultural land (north of Low Road)	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	NONE	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	June Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-013 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Stephen Chamberlain) The Executor of the Estate of the Late Andrew Leslie Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)			

	Land v	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		t of acquisition or use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-014	K	square metres of private road and verges (north of Low Road)	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	NONE	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	June Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-014 cont'd						The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Stephen Chamberlain) The Executor of the Estate of the Late Andrew Leslie Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)			
12-015	K	Temporary Rights over 12 square metres of private road and verge (north of Low Road)	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA (as assumed owner)	NONE	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans	Land Plans		Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
12-015 cont'd			Unknown		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecoms apparatus) Unknown				
12-016	F	square metres of agricultural land, drain and verge (north of Low Road)	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	NONE	SKEGNESS Lincolnshire PE24 4SA	June Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Stephen Chamberlain)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-017	K	Low Road)	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	NONE	Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA	June Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Stephen Chamberlain)			
12-018	D	Permanent Rights over 705 square metres of public road and verges (Low Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)			

	Land wl	hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) r	ind Development Consent Order FERENCE - PART 1 ight to use the land, and/or (iii) rights to f Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-018 cont'd			Sarah Louise Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY (assumed in respect of subsoil beneath public highway) Stephen Chamberlain Rookery Farm Low Road Croft SKEGNESS Lincolnshire PE24 4SA (assumed in respect of subsoil beneath public highway) Terence Reginald Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY (assumed in respect of subsoil beneath public highway) Uncloshire PE24 4RY (assumed in respect of subsoil beneath public highway) Unknown			Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
12-019	D	Permanent Rights over 167 square metres of agricultural land (south of Low Road)	Sarah Louise Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY (as assumed owner) Terence Reginald Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY (as assumed owner) Unknown	NONE	Sarah Louise Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY (as assumed owner) Terence Reginald Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown			
12-020	D	Permanent Rights over 25718 square metres of agricultural land and drains (west of Pinchbeck Lane)	Sarah Louise Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY	NONE	Andy Limb 2 Mill Cottage Croft Lane Croft SKEGNESS PE24 4PA	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
12-020 cont'd			Terence Reginald Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY		Stephen Limb 2 Mill Cottage Croft Lane Croft SKEGNESS PE24 4PA	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			
13-001	D	square metres of verge (north of Pinchbeck Lane)	Sarah Louise Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY (as assumed owner) Terence Reginald Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY (as assumed owner) Unknown	NONE	Sarah Louise Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY (as assumed owner) Terence Reginald Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY (as assumed owner) Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-002	D	Permanent Rights over 581 square metres of public road, verges and drains (Pinchbeck Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Sarah Louise Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown			

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-002 cont'd			Terence Reginald Drury Bridge House Pinchbeck Lane Croft SKEGNESS Lincolnshire PE24 4RY (assumed in respect of subsoil beneath public highway) Unknown			
13-003	D	Permanent Rights over 209 square metres of drain (south of Pinchbeck Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of drainage rights) Unknown
13-004	D	Permanent Rights over 52355 square metres of agricultural land and drains (south of Pinchbeck Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)

	Land v	which is proposed to be subject to	BOOK OF RI b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-004 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of rights granted in Wayleave Consent dated 23 June 1969)	
13-005	K	Temporary Rights over 339 square metres of agricultural land and drain (south of Pinchbeck Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
13-006	К	Temporary Rights over 931 square metres of private road and verges, agricultural land and access track (Gutheram Drove)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
13-007	K	Temporary Rights over 1601 square metres of access track (east of Gutheram Drove)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			
13-008	Κ	Temporary Rights over 1702 square metres of agricultural land and access track (north of Gutheram Drove)	Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner) Roy Sanderson Drumour Croft Road Croft SKEGNESS Lincolnshire PE24 4RP (as assumed owner) Unknown	NONE	Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner) Roy Sanderson Drumour Croft Road Croft SKEGNESS Lincolnshire PE24 4RP (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown			

	Land	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to y of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans	Extent of acquisition of use	Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-008 cont'd			William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner)		William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner)	
13-009	K	Temporary Rights over 637 square metres of agricultural land and drain (south of Gutheram Drove)	Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU	C. W. Parker (Wainfleet) Limited The Farm Office Pinchbeck Farm Sea Lane Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4BA	C. W. Parker (Wainfleet) Limited The Farm Office Pinchbeck Farm Sea Lane Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4BA	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)
			William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU			
13-010	D	Permanent Rights over 433 square metres of agricultural land (north of Croft Bank, A52)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner)	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-010 cont'd			Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner) Unknown William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner)		Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner) Unknown William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner)	Unknown			
13-011	D	Permanent Rights over 37136 square metres of agricultural land, hedgerow and drains (north of Croft Bank, A52)	Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU	C. W. Parker (Wainfleet) Limited The Farm Office Pinchbeck Farm Sea Lane Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4BA	C. W. Parker (Wainfleet) Limited The Farm Office Pinchbeck Farm Sea Lane Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4BA	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		tent of acquisition of use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-011 cont'd			William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU						
13-012	F	Temporary Rights over 1501 square metres of agricultural land, access splays and drain (north of Croft Bank, A52)	Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU	C. W. Parker (Wainfleet) Limited The Farm Office Pinchbeck Farm Sea Lane Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4BA	C. W. Parker (Wainfleet) Limited The Farm Office Pinchbeck Farm Sea Lane Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4BA	NONE			
13-013	G	Temporary Rights over 22086 square metres of agricultural land, access splays and drain (north of Croft Bank, A52)	Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU	C. W. Parker (Wainfleet) Limited The Farm Office Pinchbeck Farm Sea Lane Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4BA	C. W. Parker (Wainfleet) Limited The Farm Office Pinchbeck Farm Sea Lane Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4BA	NONE			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acculation or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans	Land Plans Extent of acquisition or use Descriptio	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
13-013 cont'd			William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU							
13-014	G	square metres of agricultural land and verge (north of Croft Bank, A52)	Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner) Unknown William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner)	NONE	Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 SRU (as assumed owner) Unknown William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 SRU (as assumed owner)	Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-015	К	Temporary Rights over 670 square metres of agricultural land (south of Gutheram Drove)	Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU	C. W. Parker (Wainfleet) Limited The Farm Office Pinchbeck Farm Sea Lane Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4BA		Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			
13-016	K	Temporary Rights over 33 square metres of agricultural land (south of Gutheram Drove)	Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN LincoInshire LN2 5RU (as assumed owner)	NONE	Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition of use	Description of hand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
13-016 cont'd			Gregory Wilkinson Autumn Heights Brewster Lane Wainfleet SKEGNESS Lincolnshire PE24 4LS (as assumed owner) Unknown William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner)		Gregory Wilkinson Autumn Heights Brewster Lane Wainfleet SKEGNESS Lincolnshire PE24 4LS (as assumed owner) Unknown William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus) Unknown			
13-017	К	Temporary Rights over 645 square metres of agricultural land and drains (south of Gutheram Drove) and public footpath (LL Crof 264/1)	Gregory Wilkinson Autumn Heights Brewster Lane Wainfleet SKEGNESS Lincolnshire PE24 4LS	NONE	Gregory Wilkinson Autumn Heights Brewster Lane Wainfleet SKEGNESS Lincolnshire PE24 4LS	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
13-017 cont'd					Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Crof 264/1))	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			
13-018	D		Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner) Gregory Wilkinson Autumn Heights Brewster Lane Wainfleet SKEGNESS Lincolnshire PE24 4LS (as assumed owner) Unknown	NONE	Alison Diane Philippa Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner) Gregory Wilkinson Autumn Heights Brewster Lane Wainfleet SKEGNESS Lincolnshire PE24 4LS (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown			

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
13-018 cont'd			William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner)		William John Epton Cliff House Eastcliff Road LINCOLN Lincolnshire LN2 5RU (as assumed owner)	
13-019	D	(south of Gutheram Drove) and public footpath (LL Crof 264/1)		NONE	Gregory Wilkinson Autumn Heights Brewster Lane Wainfleet SKEGNESS Lincolnshire PE24 4LS Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Crof 264/1))	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)
13-020	D	Permanent Rights over 253 square metres of agricultural land and drain (south of Gutheram Drove)	Gregory Wilkinson Autumn Heights Brewster Lane Wainfleet SKEGNESS Lincolnshire PE24 4LS (as assumed owner)	NONE	Gregory Wilkinson Autumn Heights Brewster Lane Wainfleet SKEGNESS Lincolnshire PE24 4LS (as assumed owner)	Unknown

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-020 cont'd			Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW (as assumed owner) Unknown		Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW (as assumed owner) Unknown	
13-021	D	Permanent Rights over 17907 square metres of agricultural land (east of Church Lane)	Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW	NONE	Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)
14-001	F	Temporary Rights over 66 square metres of agricultural land (south of Gutheram Drove)	Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW (as assumed owner) Unknown	NONE	Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW (as assumed owner) Unknown	Unknown

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-002	F	Temporary Rights over 554 square metres of agricultural land (east of Church Lane)	Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW	NONE	Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW	NONE	
14-003	F	Temporary Rights over 21 square metres of agricultural land (east of Church Lane)	Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW (as assumed owner) Unknown	NONE	Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW (as assumed owner) Unknown	Unknown	
14-004	D	Permanent Rights over 76 square metres of agricultural land (east of Church Lane)	Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW (as assumed owner) Unknown	NONE	Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW (as assumed owner) Unknown	Unknown	

	Land	which is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-005	D	Permanent Rights over 60 square metres of agricultural land (east of Church Lane)	Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW (as assumed owner) Unknown	NONE	Louise Jane Brooks Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW (as assumed owner) Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown
14-006	D	Permanent Rights over 755 square metres of public road and verges (Church Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Louise Jane Brooks Ruebury Cottage Ruebury Cottage Ruebury Lane Osmotherley NORTHALLERTON North Yorkshire DL6 3AW (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
		2000, pilon on and	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
14-006 cont'd			Roy Sanderson Drumour Croft Road Croft SKEGNESS Lincolnshire PE24 4RP (assumed in respect of subsoil beneath public highway) Unknown			National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown			
14-007	D	Permanent Rights over 207 square metres of hedgerow (west of Church Lane)	Roy Sanderson Drumour Croft Road Croft SKEGNESS Lincolnshire PE24 4RP (as assumed owner)	NONE	Roy Sanderson Drumour Croft Road Croft SKEGNESS Lincolnshire PE24 4RP (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans		isition of use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
14-007 cont'd			Unknown		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown			
14-008	D	square metres of agricultural land and drains (west of Church Lane)	Roy Sanderson Drumour Croft Road Croft SKEGNESS Lincolnshire PE24 4RP	NONE	Roy Sanderson Drumour Croft Road Croft SKEGNESS Lincolnshire PE24 4RP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)			
14-009	D	square metres of agricultural		NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpaths (LL Crof 276/2, LL Crof 276/3 and LL Crof 276/4))	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)			

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans	d Plans Extent of acquisition or use Description of land	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
14-009 cont'd			The Executor of the Estate of the Late Jean Elizabeth Sanderson Croft Road Croft SKEGNESS Lincolnshire PE24 4RP		Roy Sanderson Drumour Croft Road Croft SKEGNESS Lincolnshire PE24 4RP The Executor of the Estate of the Late Jean Elizabeth Sanderson Croft Road Croft SKEGNESS Lincolnshire PE24 4RP	Unknown (in respect of a restrictive covenant as stated in Deed dated 18 April 1977)
14-010	К	Temporary Rights over 2617 square metres of agricultural land and access track (west of Church Lane) and public footpath (LL Crof 276/4)	Roy Sanderson Drumour Croft Road Croft SKEGNESS Lincolnshire PE24 4RP	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpaths (LL Crof 276/4))	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
14-010 cont'd			The Executor of the Estate of the Late Jean Elizabeth Sanderson Croft Road SKEGNESS Lincolnshire PE24 4RP		Roy Sanderson Drumour Croft Road Croft SKEGNESS Lincolnshire PE24 4RP The Executor of the Estate of the Late Jean Elizabeth Sanderson Croft Road Croft SKEGNESS Lincolnshire PE24 4RP	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown (in respect of a restrictive covenant as stated in Deed dated 18 April 1977)			
14-011	D	-	Elaine Patricia Bradley Green Acres East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire							
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009		
14-011 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Roy Sanderson Drumour Croft Road Croft SKEGNESS Lincolnshire PE24 4RP (assumed in respect of subsoil beneath public highway) Stephen John Bradley Green Acres East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (assumed in respect of subsoil beneath public highway) Unclinshire PE24 4PQ (assumed in respect of subsoil beneath public highway) Unknown			Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
14-012	D	Permanent Rights over 318 square metres of watercourse (The Lymm)	Elaine Patricia Bradley Green Acres East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Stephen John Bradley Green Acres East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	NONE	Elaine Patricia Bradley Green Acres East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Stephen John Bradley Green Acres East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown			
14-013	D	square metres of scrubland and	Elaine Patricia Bradley Green Acres East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	NONE	Elaine Patricia Bradley Green Acres East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
14-013 cont'd			Stephen John Bradley Green Acres East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ		Stephen John Bradley Green Acres East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)				
14-014	D	Permanent Rights over 777 square metres of public roads and verges (East End)	Elaine Patricia Bradley Green Acres East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (assumed in respect of subsoil beneath public highway) J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
14-014 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Stephen John Bradley Green Acres East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (assumed in respect of subsoil beneath public highway) Unknown			National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus) Unknown			
14-015	D	Permanent Rights over 62761 square metres of agricultural land and drains (east of Wainfleet Road, B1195)	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			

	Land	which is proposed to be subject t	BOOK OF F co: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t y of Lincolnshire	o carry out protective works (Regulation 7(1)	ı(a))
Number on	Extent of accusition or use	Deceription of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
14-015 cont'd			Unknown		Unknown	Unknown
14-016	К	Temporary Rights over 317 square metres of agricultural land (west of East End)	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown
15-001	К	Temporary Rights over 1286 square metres of agricultural land (east of Wainfleet Road, B1195)	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-002	К	metres of agricultural land (east of Wainfleet Road, B1195)	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Unknown	Unknown			
15-003	K	square metres of agricultural land (east of Wainfleet Road, B1195)	Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)			

	Land	which is proposed to be subject to	BOOK OF RI DOCK OF RI DOC (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
15-004	D	Permanent Rights over 10351 square metres of agricultural land (east of Wainfleet Road, B1195)	Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)
15-005	К	Temporary Rights over 1901 square metres of access track, agricultural land and drain (north of Wainfleet Road, B1195)	Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
15-006	D		Catherine Mary Everard 86 Osterley Park View Road LONDON W7 2HH (as assumed owner) Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Giles Everard 18 Orby Road Burgh Le Marsh SKEGNESS PE24 5JH (as assumed owner) Graham Cecil Everard Grove House Farm Burgh Lane Bratoft SKEGNESS Lincolnshire PE24 5AR (as assumed owner)	NONE	Catherine Mary Everard 86 Osterley Park View Road LONDON W7 2HH (as assumed owner) Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) Giles Everard 18 Orby Road Burgh Le Marsh SKEGNESS PE24 5JH (as assumed owner) Graham Cecil Everard Grove House Farm Burgh Lane Bratoft SKEGNESS Lincolnshire PE24 5AR (as assumed owner)	Unknown				

	Land	which is proposed to be subject to	BOOK OF RE DOCK OF RE BOOK OF RE	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-006 cont'd			Jill Everard Grove House Farm Burgh Lane Bratoft SKEGNESS Lincolnshire PE24 5AR (as assumed owner) John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner) Unknown		Jill Everard Grove House Farm Burgh Lane Bratoft SKEGNESS Lincolnshire PE24 5AR (as assumed owner) John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner) Unknown		
15-007	D	Permanent Rights over 14642 square metres of agricultural land (east of Wainfleet Road, B1195)	Catherine Mary Everard 86 Osterley Park View Road LONDON W7 2HH Giles Everard 18 Orby Road Burgh Le Marsh SKEGNESS PE24 5JH	NONE	Giles Everard 18 Orby Road Burgh Le Marsh SKEGNESS PE24 5JH (trading as G & K Everard) Kay Dawson Everard 18 Orby Road Burgh Le Marsh SKEGNESS PE24 5JH (trading as G & K Everard)	Unknown (in respect of terms of Deed of Appointment dated 7 February 2023)	

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
15-007 cont'd			Graham Cecil Everard Grove House Farm Burgh Lane Bratoft SKEGNESS Lincolnshire PE24 5AR Jill Everard Grove House Farm Burgh Lane Bratoft SKEGNESS Lincolnshire PE24 5AR			
15-008	D	Permanent Rights over 42 square metres of verge (north of Wainfleet Road, B1195)	Catherine Mary Everard 86 Osterley Park View Road LONDON W7 2HH (as assumed owner) Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner)	NONE	Catherine Mary Everard 86 Osterley Park View Road LONDON W7 2HH (as assumed owner) Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown

	Land w	hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) r	nd Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights f Lincolnshire	to carry out protective works (Regulation 7(1))	a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Sesenption on talla	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-008 cont'd			Giles Everard 18 Orby Road Burgh Le Marsh SKEGNESS PE24 5JH (as assumed owner) Graham Cecil Everard Grove House Farm Burgh Lane Bratoft SKEGNESS Lincolnshire PE24 5AR (as assumed owner) Jill Everard Grove House Farm Burgh Lane Bratoft SKEGNESS Lincolnshire PE24 5AR (as assumed owner) John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner)		Giles Everard 18 Orby Road Burgh Le Marsh SKEGNESS PE24 5JH (as assumed owner) Graham Cecil Everard Grove House Farm Burgh Lane Bratoft SKEGNESS Lincolnshire PE24 5AR (as assumed owner) Jill Everard Grove House Farm Burgh Lane Bratoft SKEGNESS Lincolnshire PE24 5AR (as assumed owner) John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner)		

	Land	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to • of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-008 cont'd			Unknown		Unknown		
15-009	D	Permanent Rights over 73 square metres of access track (north of Wainfleet Road, B1195)	Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	
15-010	D	Permanent Rights over 6 square metres of verge (Wainfleet Road, B1195)	Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner)	NONE	Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
15-010 cont'd			John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner) Unknown		John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner) Unknown	Unknown				
15-011	D	square metres of public road, verges and drains (Wainfleet Road, B1195)	Catherine Mary Everard 86 Osterley Park View Road LONDON W7 2HH (assumed in respect of subsoil beneath public highway) Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of street furniture)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers				
15-011 cont'd			Giles Everard 18 Orby Road Burgh Le Marsh SKEGNESS PE24 5JH (assumed in respect of subsoil beneath public highway) Graham Cecil Everard Grove House Farm Burgh Lane Bratoft SKEGNESS Lincolnshire PE24 5AR (assumed in respect of subsoil beneath public highway) Jill Everard Grove House Farm Burgh Lane Bratoft SKEGNESS Lincolnshire PE24 5AR (assumed in respect of subsoil beneath Burgh Lane Bratoft SKEGNESS Lincolnshire PE24 5AR (assumed in respect of subsoil beneath public highway)			Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-011 cont'd			John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown						
15-012	D	Permanent Rights over 204 square metres of agricultural land and drain (west of Wainfleet Road, B1195)	Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner)	NONE	Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)			

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order :FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-012 cont'd			John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner) Unknown		John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner) Unknown	Unknown	
15-013	D	square metres of agricultural land (south of Wainfleet Road, B1195)	Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	

	Land	which is proposed to be subject to	BOOK OF F o: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to r of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-014	K	Temporary Rights over 49 square metres of agricultural land (south of Wainfleet Road, B1195)	Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)
15-015	D	Permanent Rights over 293 square metres of drain (south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner)	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner)	Unknown

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	tent of acquisition or use Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
15-015 cont'd			John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner) Unknown		John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner) Unknown	
15-016	D	Permanent Rights over 18594 square metres of agricultural land (south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A. E. Lenton (Estates) Limited) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)
15-017	G	Temporary Rights over 11326 square metres of agricultural land (south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A. E. Lenton (Estates) Limited)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
15-017 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)				
15-018	F	square metres of access track and drain (south of Wainfleet	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus and proposed underground telecommunication apparatus) Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
15-019	μ	square metres of access track (south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner) Unknown	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecoms apparatus) Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
15-020	F			NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner) John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecoms apparatus) Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
15-021	F	Temporary Rights over 49 square metres of access track and drain (south of Wainfleet Road, B1195)	Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ Unknown	NONE	Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT				
15-022	F	Temporary Rights over 211 square metres of agricultural land (south of Wainfleet Road, B1195)	Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	NONE	J F Rowson & Sons (Farmers) Limited Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)				

	Land	which is proposed to be subject to	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
15-022 cont'd			John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ			National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)
15-023	F	Temporary Rights over 148 square metres of agricultural land (south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner)	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Frederic James Rowson Manor House Farm East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning						
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
15-023 cont'd			John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner) Unknown		John Michael Rowson Thorpe House East End Thorpe St. Peter SKEGNESS PE24 4PQ (as assumed owner) Unknown	Unknown				
15-024	F	square metres of agricultural land and access track (south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A. E. Lenton (Estates) Limited)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
15-024 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			
15-025	L	square metres of agricultural land and access track (south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP		A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-025 cont'd						Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A. E. Lenton (Estates) Limited) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
15-026	F	square metres of agricultural land and access track (south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A. E. Lenton (Estates) Limited)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	F. 4 4	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-026 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			
15-027		. ,	Hagnaby Priory		A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A. E. Lenton (Estates) Limited)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Future of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition or use	Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-027 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			
15-028	Ι	(south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A. E. Lenton (Estates) Limited)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
15-029		square metres of agricultural land and access track (south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby SPILSBY Lincolnshire PE23 4BP	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 &UU (assumed in respect of rights of access) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A. E. Lenton (Estates) Limited)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Future of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition or use Description of land	Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-029 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			
15-030	Ι		A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A. E. Lenton (Estates) Limited)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans	Land Plans Extent of acquisition or use Description of land		Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-030 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			
15-031	I	square metres of agricultural land, access track (south of Wainfleet Road, B1195) and drain (Wainfleet Relief Channel)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)			
15-032	I	square metres of access track (south of Wainfleet Road, B1195) and drain (Wainfleet Relief Channel)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-033	I	Temporary Rights over 354 square metres of agricultural land, access track (south of Wainfleet Road, B1195) and drain (Wainfleet Relief Channel)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)			
15-034	I	Temporary Rights over 15 square metres of agricultural land (north of Brewster Lane and south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown			

	Land v	which is proposed to be subject to	BOOK OF F o: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-035	Ι	Permanent Rights over 4 square metres of agricultural land (north of Brewster Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown
15-036	1	Permanent Rights over 29 square metres of agricultural land and drain (north of Brewster Lane and south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)

	Land	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to r of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
15-037	G	Temporary Rights over 865 square metres of agricultural land (north of Brewster Lane and south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Unknown
15-038	L	Permanent Rights over 2 square metres of agricultural land (north of Brewster Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Extent of acquisition of use	ion or use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
15-039	L	Permanent Rights over 439 square metres of agricultural land and drain (north of Brewster Lane and south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)				
15-040	G	Temporary Rights over 1965 square metres of agricultural land (north of Brewster Lane and south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-041	D	Wainfleet Road, B1195) and drains (Wainfleet Relief	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			
15-042	D	square metres of agricultural land and drain (north of Brewster Lane and south of Wainfleet Road, B1195)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)			

	Land	which is proposed to be subject to	BOOK OF RI bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
15-042 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)
15-043	D	Permanent Rights over 17872 square metres of agricultural land (north of Brewster Lane)	Richard John Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner) Unknown	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU Unknown	Unknown
15-044	D	Permanent Rights over 515 square metres of agricultural land and drain (north of Brewster Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
15-045	F	Temporary Rights over 363 square metres of agricultural land (north of Brewster Lane)	Richard John Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner) Unknown	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU Unknown	Unknown			
15-046	D	Permanent Rights over 929 square metres of agricultural land and drain (north of Brewster Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Unknown			
15-047	D	Permanent Rights over 1355 square metres of public road and verges (Brewster Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)			

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
15-047 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Richard John Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (assumed in respect of subsoil beneath public highway) Unknown			Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown
15-048	D	Permanent Rights over 82 square metres of drain (south of Brewster Lane)	Richard John Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner)	NONE	Richard John Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)

	Land	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-048 cont'd			Unknown		Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
15-049	D	Permanent Rights over 92 square metres of agricultural land (south of Brewster Lane)	Richard John Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
15-050	D	square metres of agricultural	Richard John Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner) Unknown	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus)	

	Land	which is proposed to be subject to	BOOK OF F bo: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on		Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-050 cont'd						Unknown
15-051	F	Temporary Rights over 1035 square metres of agricultural land (south of Brewster Lane)	Richard John Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner) Unknown	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown
15-052	G	Temporary Rights over 4600 square metres of agricultural land and drain (south of Brewster Lane and north of Collison Gate)	Richard John Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner) Unknown	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU Unknown	Unknown
15-053	D	Permanent Rights over 1975 square metres of railway (Wainfleet and Boston) and works	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (as assumed owner) Unknown	NONE	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (as assumed owner) Unknown	Unknown

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	F. 4 4	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-054		Permanent Rights over 10247 square metres of agricultural land (north of Collison Gate)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus)	
15-055	D	Permanent Rights over 172 square metres of railway (Wainfleet and Boston) and works	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Richard John Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Richard John Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner) Unknown	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown	

	Land	which is proposed to be subject to	BOOK OF I b: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t y of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-056	D	Permanent Rights over 6543 square metres of agricultural land (north of Collison Gate)	Richard John Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus)
15-057	D	Permanent Rights over 6 square metres of agricultural land (north of Collison Gate)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE
16-001	F	Temporary Rights over 6 square metres of access splay and drain (south of Collison Gate)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner)	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)

	Land v	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-001 cont'd			Unknown		Unknown	Unknown
16-002	F	Temporary Rights over 45 square metres of access splay and drain (south of Collison Gate)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)
16-003	D	Permanent Rights over 1706 square metres of public road (Collison Gate)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of street furniture)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-003 cont'd			Richard John Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (assumed in respect of subsoil beneath public highway)			National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus)			
16-004	D	Permanent Rights over 172 square metres of drain (south of Collison Gate)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown			
16-005	D	Permanent Rights over 19985 square metres of agricultural land (south of Collison Gate and east of Crow's Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)			

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))		
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans	Extent of acquisition of use	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
16-006	D	Permanent Rights over 126 square metres of drain (south of Collison Gate)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Unknown		
16-007	D	Permanent Rights over 22061 square metres of agricultural land (north of Mill Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Unknown (in respect of water rights as stated in Conveyance dated 28 May 1986)		
16-008	K	Temporary Rights over 242 square metres of agricultural land (north of Mill Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Unknown (in respect of water rights as stated in Conveyance dated 28 May 1986)		
16-009	D	Permanent Rights over 1197 square metres of public road (Mill Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
16-009 cont'd			Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown			Unknown			
16-010	D		Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)			

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
16-011	D	Permanent Rights over 2231 square metres of river (Steeping River)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Unknown	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Unknown	Unknown
16-012	D	Permanent Rights over 647 square metres of river (Steeping River), foreshore, bed and banks thereof and verge (north of Mill Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
16-013	D	Permanent Rights over 561 square metres of public road and access track (Mill Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
16-014	D	Permanent Rights over 252 square metres of verge (south of Mill Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)
16-015	D	Permanent Rights over 246 square metres of verge (south of Mill Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)

	Land	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) right ty of Lincolnshire	s to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (Ap Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-016	D	Permanent Rights over 48443 square metres of agricultural land and drain (south of Mill Lane and west of Washdike Lane)	Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown (in respect of rights contained in Conveyance dated 22 July 1919) Unknown (in respect of a right of way contained in Conveyance dated 06 April 1907) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
16-017	К	Temporary Rights over 3088 square metres of agricultural land and drain (north of Church Lane and west of Washdike Lane)	Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	Unknown (in respect of rights contained in Conveyance dated 22 July 1919)

	Land	which is proposed to be subject to	BOOK OF R b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-017 cont'd 16-018	К	Temporary Rights over 701 square metres of agricultural	Pamela Mary Smith Homelands	P K Farms Willowdene Farm	P K Farms Willowdene Farm	Unknown (in respect of a right of way contained in Conveyance dated 06 April 1907) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights) Unknown (in respect of rights contained in	
		land (north of Church Lane and west of Washdike Lane)	Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD	Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	Conveyance dated 22 July 1919) Unknown (in respect of a right of way contained in Conveyance dated 06 April 1907)	
16-019	К	land (north of Church Lane and west of Washdike Lane)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner)	NONE	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner)	Unknown	

	Land	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
16-019 cont'd			John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner) Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner) Unknown		John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner) Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner) Unknown	
16-020	D	Permanent Rights over 39 square metres of drain (north of Church Lane)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner) John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner)	NONE	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner) John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-020 cont'd			Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner) Unknown		Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner) Unknown				
16-021	D	square metres of agricultural land and drain (north of Church Lane)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD	NONE	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD	NONE			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition of use	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-022	D	Permanent Rights over 182 square metres of drain (north of Church Lane)	Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD (as assumed owner) Unknown	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR Unknown	Unknown			
16-023	D	Permanent Rights over 120 square metres of drain (north of Church Lane)	Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	Unknown (in respect of rights contained in Conveyance dated 22 July 1919) Unknown (in respect of a right of way contained in Conveyance dated 06 April 1907)			
16-024	D	Permanent Rights over 13592 square metres of agricultural land (north of Church Lane)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner)	NONE	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner)	Unknown			

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		on or use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-024 cont'd			John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner) Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner) Unknown		John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner) Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner) Unknown	
16-025	D	Permanent Rights over 1638 square metres of public road, verge and copse (Church Lane)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-025 cont'd			John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (assumed in respect of subsoil beneath public highway) Unknown			Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			

	Land	which is proposed to be subject t	BOOK OF R co: (i) powers of compulsory acquisition, (ii)	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to [,] of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-026	K	Temporary Rights over 33 square metres of agricultural land (south of Church Lane)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner) John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner) Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner) Unknown	NONE	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner) John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner) Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)

	Land	which is proposed to be subject to	BOOK OF RI b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
16-027	D	Permanent Rights over 8061 square metres of agricultural land and drain (south of Church Lane)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner) John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner) Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner) Unknown	NONE	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner) John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner) Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
16-028	K	Temporary Rights over 212 square metres of agricultural land (south of Church Lane)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ	NONE	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-028 cont'd			John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD The Executor of the Estate of the Late Mary Elizabeth Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD		John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD	
16-029	D	Permanent Rights over 6389 square metres of agricultural land (south of Church Lane)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ	NONE	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
16-029 cont'd			John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD The Executor of the Estate of the Late Mary Elizabeth Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD		John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD				
16-030	D	Permanent Rights over 436 square metres of drain (south of Church Lane)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner)	NONE	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land wh	nich is proposed to be subject	BOOK OF REFE t to: (i) powers of compulsory acquisition, (ii) rig	d Development Consent Order RENCE - PART 1 ht to use the land, and/or (iii) rights Lincolnshire	s to carry out protective works (Regulation 7(1)((a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
16-030 cont'd			John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner) John Walter Rowson Lucerne House Croft Lane Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA (as assumed owner) Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner) Unknown		John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner) John Walter Rowson Lucerne House Croft Lane Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA (as assumed owner) Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner) Unknown	

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on		Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	nt of acquisition or use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-031	D		John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA	NONE	John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA	Unknown (in respect of rights relating to sewers, drains and water pipes listed in Deed of Gift dated 10 April 1992) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-001	D	Permanent Rights over 1021 square metres of access track (north of Scald Gate) and public footpath (LL WStM 371/1)	Gail Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-001 cont'd			John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown		Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			
17-002	D	Permanent Rights over 234 square metres of drain (south of Church Lane)	Gail Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner)	NONE	Gail Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans	1 Plans	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
17-002 cont'd			Unknown		Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
17-003	D	square metres of agricultural land (south of Church Lane)	Gail Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
17-004	К	, , , , , , , , , , , , , , , , , , ,	John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA (as assumed owner)	NONE	John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation					
Land Plans	Land Plans	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-004 cont'd			Unknown		Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
17-005	К		John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA	NONE	John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA	Unknown (in respect of rights relating to sewers, drains and water pipes listed in Deed of Gift dated 10 April 1992)			
						Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
17-006	D	Permanent Rights over 437 square metres of drain (south of Church Lane)	Gail Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner) John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA (as assumed owner) Unknown	NONE	Gail Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner) John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			
17-007	D	Permanent Rights over 9897 square metres of agricultural land (south of Church Lane)	John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA	NONE	John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA	Unknown (in respect of rights relating to sewers, drains and water pipes listed in Deed of Gift dated 10 April 1992)			

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-008	D	Permanent Rights over 28 square metres of drain (south of Church Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Gail Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Gail Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
17-009	D	Permanent Rights over 250 square metres of drain (south of Church Lane)	Gail Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner)	NONE	Gail Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
17-009 cont'd			John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA (as assumed owner) Unknown		John Walter Rowson Lucerne House Croft Lane SKEGNESS Lincolnshire PE24 4PA (as assumed owner) Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
17-010	D	square metres of agricultural land (west of Hall Gate)	Gail Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU	Paul Willoughby White House Farm Spilsby Road Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PU	Unknown (in respect of rights of entry listed in Conveyance dated 31 December 1919) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			

	Land	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	F . 4 . 4 . 6	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-011	D	Permanent Rights over 535 square metres of drain (south of Church Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) John Walter Rowson Lucerne House Croft Lane Croft SKEGNESS Lincolnshire PE24 4PA (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
17-012	D	Permanent Rights over 81 square metres of drain (west of Hall Gate)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner)	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-012 cont'd			Gail Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner) Unknown		Gail Shaw Colindale Farm Matt Pits Lane Wainfleet SKEGNESS Lincolnshire PE24 4LY (as assumed owner) Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			
17-013	D	square metres of agricultural	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of drainage rights) Unknown (in respect of rights reserved by Conveyance dated 12 July 1919) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights t y of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-014	D	Permanent Rights over 10843 square metres of agricultural land (east of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A. E. Lenton (Estates) Limited) Sarah Elizabeth Scupholm 18 Rumbold Lane Wainfleet SKEGNESS PE24 4DS (in respect of restrictive covenants as imposed by restriction dated 18 August 2022) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-015	D	Permanent Rights over 283 square metres of drain (north of Scald Gate)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner)	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-015 cont'd			Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Unknown		Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			
17-016	D	Permanent Rights over 12993 square metres of agricultural land and drain (north of Scald Gate)	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN	NONE	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation					
Land Plans	Land Plans	Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-016 cont'd			Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN		Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN				
17-017	K	Temporary Rights over 6306 square metres of drain and copse (north of Scald Gate)	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN	NONE	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-018	D	Permanent Rights over 133 square metres of drain (north of Scald Gate)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Unknown	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to of Lincolnshire	o carry out protective works (Regulation 7(1)	ı(a))	
Number on	F. 4	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-019	К	Temporary Rights over 395 square metres of agricultural land (north of Scald Gate)	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN	NONE	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-020	К	Temporary Rights over 27 square metres of agricultural land (north of Scald Gate)	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner)	NONE	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-020 cont'd			Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Unknown		Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
17-021	D	Permanent Rights over 165 square metres of drain (north of Scald Gate)	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Unknown	NONE	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land v	which is proposed to be subject to	BOOK OF RE DOCK OF RE (i) powers of compulsory acquisition, (ii)	'ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on		Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-022	D	Permanent Rights over 731 square metres of public road and verge (Scald Gate)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (assumed in respect of subsoil beneath public highway) Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE		Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
17-022 cont'd			Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (assumed in respect of subsoil beneath public highway) Unknown						
17-023	D	Permanent Rights over 241 square metres of drain (south of Scald Gate)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE		Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land v	which is proposed to be subject to	BOOK OF F bo: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-024	D	Permanent Rights over 15081 square metres of agricultural land (south of Scald Gate)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Unknown (in respect of right of entry as stated by an Indenture dated 09 October 1919)
						Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
17-025	K	Temporary Rights over 10 square metres of access splay (north of Scald Gate)	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner)	NONE	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)

	Land	which is proposed to be subject to	BOOK OF BOOK OF I	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t y of Lincolnshire	o carry out protective works (Regulation 7(1))(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-025 cont'd			Unknown		Unknown	
17-026	D	Permanent Rights over 384 square metres of drain (south of Scald Gate)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of drainage rights) Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
17-027	D	Permanent Rights over 35018 square metres of agricultural (east of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)

	Land	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
17-028	D	Permanent Rights over 89 square metres of agricultural land (east of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Unknown
17-029	D	Permanent Rights over 203 square metres of agricultural land (east of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE
17-030	D	Permanent Rights over 198 square metres of drain and copse (east of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD (as assumed owner)	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
17-030 cont'd			Unknown		Unknown					
17-031	К	square metres of agricultural land (east of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE				
17-032	K	square metres of drain (east of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY LincoInshire PE23 4BP (as assumed owner) Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS LincoInshire PE24 5JD (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 SJD (as assumed owner) Unknown	Unknown				

	Land	which is proposed to be subject to	BOOK OF o: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights ty of Lincolnshire	to carry out protective works (Regulation 7(1))(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-033	К	Temporary Rights over 248 square metres of agricultural land (east of Burgh Road)	Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of rights contained in Conveyance dated 16 February 1948)	
17-034	D	Permanent Rights over 11589 square metres of agricultural land (east of Burgh Road)	Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of rights contained in Conveyance dated 16 February 1948)	

	Land	which is proposed to be subject to	BOOK OF RI b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
17-035	D	Permanent Rights over 100 square metres of agricultural land (east of Burgh Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE
17-036	Η	Temporary Rights over 30 square metres of drain (east of Scald Gate)	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner)	NONE	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner) Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN (as assumed owner)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
17-036 cont'd			Unknown		Unknown	
17-037	Н	Temporary Rights over 27 square metres of drain (east of Scald Gate)	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN	NONE	Graham Frederick Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN Marian Jeanette Shaw Pepperthorpe Hall Low Road Wainfleet St. Mary SKEGNESS Lincolnshire PE24 4HN	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus and proposed underground telecommunication apparatus)
18-001	D	Permanent Rights over 759 square metres of public road (Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)

	Land v	vhich is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	arry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-001 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD (assumed in respect of subsoil beneath public highway) Unknown			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	
18-002	D	Permanent Rights over 163 square metres of drain (west of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	

	Land v	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-003	D	Permanent Rights over 11109 square metres of agricultural land (west of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY LincoInshire PE23 4BP	NONE
18-004	D	Permanent Rights over 2998 square metres of agricultural land (west of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY LincoInshire PE23 4BP	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of a right of way and drainage rights) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A. E. Lenton (Estates) Limited)
18-005	D	Permanent Rights over 345 square metres of drain (west of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner)	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY LincoInshire PE23 4BP (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-005 cont'd			Jonathan Harman Lynwood Burgh Road Friskney BOSTON Lincolnshire PE22 8NS (as assumed owner) Ruth Harman Lynwood Burgh Road Friskney BOSTON Lincolnshire PE22 8NS (as assumed owner) Unknown		Jonathan Harman Lynwood Burgh Road Friskney BOSTON Lincolnshire PE22 8NS (as assumed owner) Ruth Harman Lynwood Burgh Road Friskney BOSTON Lincolnshire PE22 8NS (as assumed owner) Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			
18-006	D	Permanent Rights over 5614 square metres of agricultural land (west of Burgh Road)	Jonathan Harman Lynwood Burgh Road Friskney BOSTON Lincolnshire PE22 8NS	NONE	Jonathan Harman Lynwood Burgh Road Friskney BOSTON Lincolnshire PE22 8NS	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Ruth Mary Harman and Jonathan Peter Harman)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
18-006 cont'd			Ruth Harman Lynwood Burgh Road Friskney BOSTON Lincolnshire PE22 8NS		Ruth Harman Lynwood Burgh Road Friskney BOSTON Lincolnshire PE22 8NS	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
18-007	D	square metres of agricultural land (west of Burgh Road)	Tower Pension Trustees Limited Dunn's House St Paul's Road SAILSBURY Wiltshire SP2 7BF	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	John Charles Thomas Winslow Moorfield Burgh Road Friskney BOSTON PE22 8NS (in respect of rights to run services, use of land, entry, support and others listed in Conveyance dated 25 August 1995) Susan Helen Winslow Moorfield Burgh Road Friskney BOSTON PE22 8NS (in respect of rights to run services, use of land, entry, support and others listed in Conveyance dated 25 August 1995)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
18-007 cont'd						Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
18-008	К	square metres of agricultural land (west of Burgh Road)	SAILSBURY Wiltshire SP2 7BF	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	John Charles Thomas Winslow Moorfield Burgh Road Friskney BOSTON PE22 8NS (in respect of rights to run services, use of land, entry, support and others listed in Conveyance dated 25 August 1995) Susan Helen Winslow Moorfield Burgh Road Friskney BOSTON PE22 8NS (in respect of rights to run services, use of land, entry, support and others listed in Conveyance dated 25 August 1995)			

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-008 cont'd						Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)
18-009	D	land (west of Burgh Road)	Tower Pension Trustees Limited Dunn's House St Paul's Road SAILSBURY Wiltshire SP2 7BF	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	NONE
18-010	D	square metres of drain (west of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Tower Pension Trustees Limited Dunn's House St Paul's Road SAILSBURY Wiltshire SP2 7BF (as assumed owner)	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner) Tower Pension Trustees Limited Dunn's House St Paul's Road SAILSBURY Wiltshire SP2 7BF (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of drainage rights) Unknown

	Land	which is proposed to be subject to	BOOK OF I b: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-010 cont'd			Unknown		Unknown	
18-011	D	Permanent Rights over 9164 square metres of agricultural land (west of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Gary Brunt Ty Bach Burgh Road Friskney BOSTON PE22 8NT (in respect of covenants relating to maintenance of a boundary hedge listed in Conveyance dated 25 August 1983)
18-012	К	Temporary Rights over 3057 square metres of agricultural land and access track (west of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	Gary Brunt Ty Bach Burgh Road Friskney BOSTON PE22 8NT (in respect of covenants relating to maintenance of a boundary hedge listed in Conveyance dated 25 August 1983)
18-013	D	Permanent Rights over 174 square metres of drain (west of Burgh Road)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner)	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of drainage rights)

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-013 cont'd			Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD (as assumed owner) Unknown		Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD (as assumed owner) Unknown	Unknown
18-014	D	Permanent Rights over 28313 square metres of agricultural land (north of Fen Bank)	Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JD	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of rights contained in Conveyance dated 16 February 1948)
18-015	K	Temporary Rights over 3368 square metres of access track (north of Fen Bank and west of Burgh Road)	Pamela Mary Smith Homelands Orby Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 SJD	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	P K Farms Willowdene Farm Armstrongs Lane Friskney BOSTON Lincolnshire PE22 8NR	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of rights contained in Conveyance dated 16 February 1948)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
18-016	D	Permanent Rights over 548 square metres of watercourse (Fodder Dike Bank)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP	NONE	Alan J Moore Marsh Farm Woad Lane Fishtoft BOSTON PE22 ORA	Unknown (in respect of unknown restrictive covenants which may have been imposed prior to 6 July 2021) Boston & District Angling Association 13 Brand End Road Butterwick BOSTON Lincolnshire PE22 0ET (in respect of fishing rights)			
18-017	D	Permanent Rights over 371 square metres of drain (north of Fen Bank)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Unknown (in respect of unknown restrictive covenants which may have been imposed prior to 6 July 2021)			

	Land v	which is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order :FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-018	D	Permanent Rights over 1517 square metres of public road (Fen Bank)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (assumed in respect of subsoil beneath public highway) John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown

	Land	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Land Plans Extent of acquisition or use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-018 cont'd			Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (assumed in respect of subsoil beneath public highway) Unknown			
18-019	D	Permanent Rights over 9573 square metres of agricultural land (south of Fen Bank)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS	NONE	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
18-019 cont'd			Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD		Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD				
18-020	G	Temporary Rights over 6871 square metres of agricultural land (south of Fen Bank)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD	NONE	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD	NONE			

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights r of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-021	D	Permanent Rights over 215 square metres of drain (south of Fen Bank)	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner) Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner) Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner)	NONE	David Thomas Holland Ellesmere Port Lade Bank Old Leake BOSTON PE22 9RJ (as assumed owner) Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) John Leo Holland East View Brewster Lane Wainfleet SKEGNESS PE24 4LS (as assumed owner) Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD (as assumed owner)	Unknown	

	Land v	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-021 cont'd			Unknown		Unknown	
18-022	К	Temporary Rights over 2181 square metres of agricultural land (west of Burgh Road)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR	NONE
18-023	D	Permanent Rights over 7557 square metres of agricultural land (west of Burgh Road)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR	NONE
18-024	D	Permanent Rights over 211 square metres of agricultural land (east of Burgh Road)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Unknown	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Unknown	Unknown

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to r of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
18-025	D	Permanent Rights over 4703 square metres of agricultural land (west of Burgh Road)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR	NONE
18-026	D	Permanent Rights over 206 square metres of agricultural land (west of Burgh Road)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Unknown	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Unknown	Unknown
18-027	К	Temporary Rights over 51 square metres of access splay (west of Burgh Road)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Unknown	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Unknown	Unknown

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-028	D	Permanent Rights over 27320 square metres of agricultural land and drain (west of Burgh Road)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR	NONE
18-029	К	Temporary Rights over 2458 square metres of agricultural land (west of Burgh Road)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR	NONE
18-030	D	Permanent Rights over 199 square metres of drain (east of Cranberry Lane)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Friskney United Charities Sigtoft Farm Low Road South Friskney BOSTON Lincolnshire PE22 8QH (as assumed owner)	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Friskney United Charities Sigtoft Farm Low Road South Friskney BOSTON Lincolnshire PE22 8QH (as assumed owner)	Unknown

	Land v	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to r of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-030 cont'd			Unknown		Unknown		
18-031	К		Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Unknown	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Unknown	Unknown	
18-032	K	square metres of access splay	Nigel Smith 18 Holmes Road Stickney BOSTON Lincolnshire PE22 8AZ (as assumed owner) Unknown	NONE	Nigel Smith 18 Holmes Road Stickney BOSTON Lincolnshire PE22 8AZ (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (App Regulations 2009	blications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of hard	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
18-033	D	Permanent Rights over 48 square metres of agricultural land (east of Cranberry Lane)	Friskney United Charities Sigtoft Farm Low Road South Friskney BOSTON Lincolnshire PE22 8QH	Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD	Peter James Holland The Hawthorns Church Road Friskney BOSTON PE22 8RD	Unknown (in respect of rights and restrictive covenants stated in an unknown document as may have been imposed before 5 December 2011)			
18-034	D	Permanent Rights over 10876 square metres of agricultural land (east of Cranberry Lane)	Friskney United Charities Sigtoft Farm Low Road South Friskney BOSTON Lincolnshire PE22 8QH	Graham Lawrence Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	Graham Lawrence Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	Unknown (in respect of rights and restrictive covenants stated in an unknown document as may have been imposed before 5 December 2011)			
18-035	К	Temporary Rights over 124 square metres of agricultural land and access track (east of Cranberry Lane)	Friskney United Charities Sigtoft Farm Low Road South Friskney BOSTON Lincolnshire PE22 8QH	Graham Lawrence Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	Graham Lawrence Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	Unknown (in respect of rights and restrictive covenants stated in an unknown document as may have been imposed before 5 December 2011)			
18-036	К	Temporary Rights over 543 square metres of agricultural land (west of Cranberry Lane)	Nigel Smith 18 Holmes Road Stickney BOSTON Lincolnshire PE22 8AZ	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	Colin Campbell 1 Hudson Crescent LEIGH-ON-SEA SS9 5XT (in respect of provisions and covenants listed in Transfer dated 03 September 2004)			

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-036 cont'd						HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Nigel Smith) Pamela Ann Campbell 1 Hudson Crescent LEIGH-ON-SEA SS9 5XT (in respect of provisions and covenants listed in Transfer dated 03 September 2004) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
18-037	D	Permanent Rights over 1013 square metres of public road and verge (Cranberry Lane)	Friskney United Charities Sigtoft Farm Low Road South Friskney BOSTON LincoInshire PE22 8QH (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to c of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-037 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Nigel Smith 18 Holmes Road Stickney BOSTON Lincolnshire PE22 8AZ (assumed in respect of subsoil beneath public highway) Unknown			Unknown
18-038	D	square metres of drain (west of	Nigel Smith 18 Holmes Road Stickney BOSTON Lincolnshire PE22 8AZ (as assumed owner) Unknown		Nigel Smith 18 Holmes Road Stickney BOSTON Lincolnshire PE22 8AZ (as assumed owner) Unknown	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
19-001	D	Permanent Rights over 9533 square metres of agricultural land (west of Cranberry Lane)	Nigel Smith 18 Holmes Road Stickney BOSTON Lincolnshire PE22 8AZ	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	Colin Campbell 1 Hudson Crescent LEIGH-ON-SEA SS9 5XT (in respect of provisions and covenants listed in Transfer dated 03 September 2004) HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Nigel Smith) Pamela Ann Campbell 1 Hudson Crescent LEIGH-ON-SEA SS9 5XT (in respect of provisions and covenants listed in Transfer dated 03 September 2004) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
19-002	D	Permanent Rights over 168 square metres of drain (west of Cranberry Lane)	Graham Lawrence Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD (as assumed owner) Janet Rose Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD (as assumed owner) Nigel Smith 18 Holmes Road Stickney BOSTON Lincolnshire PE22 8AZ (as assumed owner) Unknown	NONE	Graham Lawrence Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD (as assumed owner) Janet Rose Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD (as assumed owner) Nigel Smith 18 Holmes Road Stickney BOSTON Lincolnshire PE22 8AZ (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
19-003	D	Permanent Rights over 17041 square metres of agricultural land and access track (north of Mill Hill)	Graham Lawrence Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	NONE	Graham Lawrence Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	Unknown (in respect of unknown restrictive covenants which may have been imposed prior to 3 April 1996)			
			Janet Rose Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD		Janet Rose Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	Unknown (in respect of right to use conduits under the property together with a right of entry to clean, repair and replace as stated in transfer dated 29 March 1996)			
						Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of a right of access and a right to lay, maintain and repair electricity apparatus as stated in Agreement dated 26 April 2024)			

	Land	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights / of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on		Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-004	D	Permanent Rights over 194 square metres of copse (north of Mill Hill) (excluding all interests of the Crown)	Graham Lawrence Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD (as assumed owner) Janet Rose Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	Graham Lawrence Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD (as assumed owner) Janet Rose Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
19-005	К	Temporary Rights over 2608 square metres of agricultural land (north of Mill Hill)	Graham Lawrence Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	NONE	Graham Lawrence Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	Unknown (n respect of unknown restrictive covenants which may have been imposed prior to 3 April 1996)			
			Janet Rose Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD		Janet Rose Lenton Mill Lodge Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	Unknown (in respect of right to use conduits under the property together with a right of entry to clean, repair and replace as stated in transfer dated 29 March 1996)			
						Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of a right of access and a right to lay, maintain and repair electricity apparatus as stated in Agreement dated 26 April 2024)			

	Land v	which is proposed to be subject to	BOOK OF I o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights t y of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-006	D	Permanent Rights over 25424 square metres of agricultural land (north of Mill Hill) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)
				William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	SG Kleinwort Hambros Bank Limited One Bank Street Canary Wharf LONDON E14 4SG (in respect of rights listed in Conveyance dated 29 September 1987)
19-007	K	Temporary Rights over 60 square metres of agricultural land (north of Mill Hill) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	SG Kleinwort Hambros Bank Limited One Bank Street Canary Wharf LONDON E14 4SG (in respect of rights listed in Conveyance dated 29 September 1987)
				William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
19-008	D	square metres of drain (north	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Unknown				
19-009	D	square metres of public road (Mill Hill) (excluding all interests of the Crown)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (assumed in respect of subsoil beneath public highway) Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)				

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-009 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (assumed in respect of subsoil beneath public highway) Unknown			Unknown
19-010	D	Permanent Rights over 1047 square metres of agricultural land (south of Mill Hill and north of Church Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP	NONE
19-011	D	Permanent Rights over 246 square metres of drain (south of Mill Hill and north of Church Lane)	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner)	NONE	A. E. Lenton (Estates) Limited Hagnaby Priory Hagnaby SPILSBY Lincolnshire PE23 4BP (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
19-011 cont'd			Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Unknown		Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Unknown					
19-012	D	square metres of agricultural land (south of Mill Hill and east of Small End Road)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
19-013	D	Permanent Rights over 17657 square metres of agricultural land (south of Mill Hill and east of Small End Road)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR	NONE				
19-014	D	Permanent Rights over 249 square metres of drain and copse (south of Mill Hill and east of Small End Road)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Ronald Leslie Elvin The Laurels Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD (as assumed owner) Unknown	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Ronald Leslie Elvin The Laurels Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD (as assumed owner) Unknown	Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acculation or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
19-015	D	Permanent Rights over 2199 square metres of agricultural land and copse (north of Church Lane and east of Small End Road)	Ronald Leslie Elvin The Laurels Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	NONE	Ronald Leslie Elvin The Laurels Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD Beryl Elvin The Laurels Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD James Richard Elvin The Laurels Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	NONE				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on		Descriptions (the d	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
19-016	D	square metres of drain and	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Eileen Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner) John Timothy Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner) John Timothy Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner)	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Eileen Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner) John Timothy Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner) John Timothy Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner)	Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
19-016 cont'd			Ronald Leslie Elvin The Laurels Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD (as assumed owner) Unknown		Ronald Leslie Elvin The Laurels Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD (as assumed owner) Unknown					
19-017	D	square metres of drain (south of Mill Hill and east of Small End Road)	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Eileen Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner)	NONE	Dyson Farming Limited The Estate Office Cyclone Way Nocton LINCOLN Lincolnshire LN4 2GR (as assumed owner) Eileen Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner)	Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
19-017 cont'd			John Timothy Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner) Unknown		John Timothy Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner) Unknown					
19-018	D	square metres of agricultural land and copse (north of Church Lane and east of Small End Road)	Eileen Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG John Timothy Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG	NONE	Eileen Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG John Timothy Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
19-019	D	Permanent Rights over 153 square metres of drain and copse (south of Mill Hill and east of Small End Road)	David George Robinson Willoughby Farm Small End Friskney BOSTON Lincolnshire PE22 8PE (as assumed owner) Eileen Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner) John Timothy Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner) Lincolnshire PE22 8PG (as assumed owner) Unknown	NONE	David George Robinson Willoughby Farm Small End Friskney BOSTON Lincolnshire PE22 8PE (as assumed owner) Eileen Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner) John Timothy Ryland Bleak House Church Lane Friskney BOSTON Lincolnshire PE22 8PG (as assumed owner) Lincolnshire PE22 8PG (as assumed owner) Lincolnshire PE22 8PG (as assumed owner) Lincolnshire	Unknown			

	Land v	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to ^o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on		Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
19-020	D	Permanent Rights over 4818 square metres of agricultural land (east of Small End Road and north of Church Lane)	David George Robinson Willoughby Farm Small End Friskney BOSTON Lincolnshire PE22 8PE	NONE	David George Robinson Willoughby Farm Small End Friskney BOSTON Lincolnshire PE22 8PE	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
19-021	D	Permanent Rights over 174 square metres of agricultural land (east of Small End Road and north of Church Lane)	David George Robinson Willoughby Farm Small End Friskney BOSTON Lincolnshire PE22 8PE	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for David Robinson)
			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)			Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
19-022	D	Permanent Rights over 1367 square metres of public road (Small End Road)	David George Robinson Willoughby Farm Small End Friskney BOSTON Lincolnshire PE22 &PE (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Terra Estate Gmbh & Co Kg Riehler Straße 190 Köln 50735 GERMANY (assumed in respect of subsoil beneath public highway) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
19-023	К	Temporary Rights over 877 square metres of agricultural land (west of Small End Road)	Terra Estate Gmbh & Co Kg Riehler Straße 190 Köln 50735 GERMANY	J & V Casey and Son Limited Willow Farm Coningsby Road Dogdyke Lincolnshire LN4 4UY	J & V Casey and Son Limited Willow Farm Coningsby Road Dogdyke LincoInshire LN4 4UY	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)			
19-024	D	Permanent Rights over 37859 square metres of agricultural land and drain (west of Small End Road)	Terra Estate Gmbh & Co Kg Riehler Straße 190 Köln 50735 GERMANY	J & V Casey and Son Limited Willow Farm Coningsby Road Dogdyke Lincolnshire LN4 4UY	J & V Casey and Son Limited Willow Farm Coningsby Road Dogdyke Lincolnshire LN4 4UY	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	
19-025	D	Permanent Rights over 16703 square metres of agricultural land (north of Skirmore Road)	Dorothy Clow Goodacre Eaudyke Road Friskney BOSTON PE22 & RT (as assumed owner) John Clow Goodacre Eaudyke Road Friskney BOSTON PE22 & RT (as assumed owner) Unknown	NONE	Staples Brothers Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE Unknown	Unknown
20-001	F	Temporary Rights over 5695 square metres of agricultural land (north of Skirmore Road)	Dorothy Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT (as assumed owner)	Staples Brothers Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Staples Brothers Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans		Description on tune	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
20-001 cont'd			John Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT (as assumed owner) Unknown		Unknown	Unknown			
20-002	G	Temporary Rights over 42 square metres of agricultural land and drain (north of Skirmore Road)	Dorothy Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT (as assumed owner) John Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT (as assumed owner) Unknown	Staples Brothers Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Staples Brothers Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown			

	Land	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
20-003	G	Temporary Rights over 14409 square metres of agricultural land (north of Skirmore Road)	Dorothy Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT John Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT	NONE	Dorothy Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT John Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)
20-004	D	Permanent Rights over 12067 square metres of agricultural land and drain (north of Skirmore Road)	Dorothy Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT	NONE	Dorothy Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)

	Land v	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
20-004 cont'd			John Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT		John Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT	
20-005	D	Permanent Rights over 1140 square metres of public road, verges and drain (Skirmore Road)	Dorothy Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT (assumed in respect of subsoil beneath public highway) John Clow Goodacre Eaudyke Road Friskney BOSTON PE22 8RT (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	nd Plans Extent of acquisition or use Description of land	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
20-005 cont'd			Terra Estate Gmbh & Co Kg Riehler Straße 190 Köln 50735 GERMANY (assumed in respect of subsoil beneath public highway) Unknown						
20-006	D	Permanent Rights over 161 square metres of drain (south of Skirmore Road)	Terra Estate Gmbh & Co Kg Riehler Straße 190 Köln 50735 GERMANY (as assumed owner) Unknown	NONE	Terra Estate Gmbh & Co Kg Riehler Straße 190 Köln 50735 GERMANY (as assumed owner) Unknown	Unknown			
20-007	K	Temporary Rights over 149 square metres of access splay (south of Skirmore Road)	Terra Estate Gmbh & Co Kg Riehler Straße 190 Köln 50735 GERMANY	J & V Casey and Son Limited Willow Farm Coningsby Road Dogdyke Lincolnshire LN4 4UY	J & V Casey and Son Limited Willow Farm Coningsby Road Dogdyke Lincolnshire LN4 4UY	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to / of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
20-008	D	Permanent Rights over 51881 square metres of agricultural land (south of Skirmore Road and east of Patman's Lane)	Terra Estate Gmbh & Co Kg Riehler Straße 190 Köln 50735 GERMANY	J & V Casey and Son Limited Willow Farm Coningsby Road Dogdyke Lincolnshire LN4 4UY	J & V Casey and Son Limited Willow Farm Coningsby Road Dogdyke Lincolnshire LN4 4UY	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
20-009	Н	Temporary Rights over 99 square metres of agricultural land (south of Howgarth Lane)	Terra Estate Gmbh & Co Kg Riehler Straße 190 Köln 50735 GERMANY (as assumed owner)	NONE	Terra Estate Gmbh & Co Kg Riehler Straße 190 Köln 50735 GERMANY (as assumed owner)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus and proposed underground telecommunication apparatus)	
21-001	K	Temporary Rights over 32 square metres of access splay (west of Patman's Lane)	Unknown Unknown	NONE	Unknown Unknown	Unknown Unknown	
			Ver Limited The Manor House 3 Chalkpit Road Holme Next the Sea HUNSTANTON PE36 6LW (as assumed owner)		Ver Limited The Manor House 3 Chalkpit Road Holme Next the Sea HUNSTANTON PE36 6LW (as assumed owner)		

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
21-002	K	Temporary Rights over 1123 square metres of agricultural land (west of Patman's Lane)	Ver Limited The Manor House 3 Chalkpit Road Holme Next the Sea HUNSTANTON PE36 6LW	NONE	Ver Limited The Manor House 3 Chalkpit Road Holme Next the Sea HUNSTANTON PE36 6LW	Roughton Farms Limited The Grange Eaudyke Road Friskney BOSTON Lincolnshire PE22 8RT (in respect of the rights reserved by conveyance dated 27 November 1975 and right to use drainage systems as stated in conveyance dated 28 October 2009)	
21-003	D	Permanent Rights over 1418 square metres of public road and verges (Patman's Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Terra Estate Gmbh & Co Kg Riehler Straße 190 Köln 50735 GERMANY (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers				
21-003 cont'd			Unknown Ver Limited The Manor House 3 Chalkpit Road Holme Next the Sea HUNSTANTON PE36 6LW (assumed in respect of subsoil beneath public highway)			Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (assumed in respect of rights of access)			
21-004	D	square metres of drain (west of Patman's Lane)	Unknown Ver Limited The Manor House 3 Chalkpit Road Holme Next the Sea HUNSTANTON PE36 6LW (as assumed owner)		Unknown Ver Limited The Manor House 3 Chalkpit Road Holme Next the Sea HUNSTANTON PE36 6LW (as assumed owner)	Unknown			

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description on tand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
21-005	D	Permanent Rights over 35215 square metres of agricultural land (west of Patman's Lane) Permanent Rights over 21128 square metres of agricultural	Ver Limited The Manor House 3 Chalkpit Road Holme Next the Sea HUNSTANTON PE36 6LW Kathleen Helen Dowlman Applecroft House	NONE M. Leggate & Sons (Produce) Limited Main Road	Ver Limited The Manor House 3 Chalkpit Road Holme Next the Sea HUNSTANTON PE36 6LW M. Leggate & Sons (Produce) Limited Main Road	Roughton Farms Limited The Grange Eaudyke Road Friskney BOSTON Lincolnshire PE22 8RT (in respect of the rights reserved by conveyance dated 27 November 1975 and right to use drainage systems as stated in conveyance dated 28 October 2009) Unknown
		land (east of Ivery Lane)	Mill Lane Friskney BOSTON Lincolnshire PE22 8SF (as assumed owner) Unknown	Wrangle BOSTON Lincolnshire PE22 9AA	Wrangle BOSTON Lincolnshire PE22 9AA Unknown	
21-007	К	Temporary Rights over 11 square metres of access splay (east of Ivery Lane)	Kathleen Helen Dowlman Applecroft House Mill Lane Friskney BOSTON Lincolnshire PE22 8SF (as assumed owner)	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
21-007 cont'd			Unknown		Unknown				
21-008	D	Permanent Rights over 1162 square metres of public road, verges and drain (Ivery Lane)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (assumed in respect of subsoil beneath public highway) John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (assumed in respect of subsoil beneath public highway) Kathleen Helen Dowlman Applecroft House Mill Lane Friskney BOSTON Lincolnshire PE22 8SF (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
21-008 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown						
21-009	D	square metres of agricultural land, access track and drain (west of Ivery Lane)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG		George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons) John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons)	NONE			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on		Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent at acquisition or use Description	Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
21-009 cont'd					Pearl Eleanor Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons) Christine Roderick Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons)				
21-010	К	Temporary Rights over 269 square metres of agricultural land (west of Ivery Lane)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG	NONE	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons)	NONE			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition or use Descrip	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
21-010 cont'd			John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG		John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons) Pearl Eleanor Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons) Christine Roderick Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons) Lincolnshire PE22 9BG				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
21-011	D	Permanent Rights over 19862 square metres of agricultural road and drain (west of Ivery Lane)	Janet Codd Hawthorn Farm Ivery Lane Wrangle BOSTON Lincolnshire PE22 9BB	Staples Brothers Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Staples Brothers Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			
22-001	K	Temporary Rights over 6053 square metres of agricultural land and drain (west of Ivery Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	Unknown (in respect of rights granted by Conveyance dated 29 August 1987) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			
22-002	К	Temporary Rights over 12 square metres of drain (west of Ivery Lane) (excluding all interests of the Crown)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner)	NONE	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner)	Unknown			

	Land	which is proposed to be subject to	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
22-002 cont'd			John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown		John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)
22-003	К	Temporary Rights over 32 square metres of agricultural land and drain (west of Ivery Lane)	Janet Codd Hawthorn Farm Ivery Lane Wrangle BOSTON Lincolnshire PE22 9BB	Staples Brothers Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Staples Brothers Limited Station Farm Station Road Sibsey BOSTON LincoInshire PE22 OSE	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)
22-004	D	Permanent Rights over 7797 square metres of agricultural land (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	Unknown (in respect of rights granted by Conveyance dated 29 August 1987)

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
22-004 cont'd				William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ		
22-005	D	Permanent Rights over 117 square metres of drain (west of lvery Lane and south of Love Lane) (excluding all interests of the Crown)	Goldfen Bank	NONE	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Unknown	

	Land	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
22-006	К	Temporary Rights over 29 square metres of agricultural land (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	Unknown (in respect of rights granted by Conveyance dated 29 August 1987)
22-007	К	Temporary Rights over 9 square metres of drain (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner)	NONE	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
22-007 cont'd			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown		The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown				
22-008	Κ	Temporary Rights over 41 square metres of agricultural land (west of Ivery Lane and south of Love Lane)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG	NONE	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons) John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons)	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 0DW (in respect of rights reserved in Transfer dated 22 February 1995) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			

	Land	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights t y of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-008 cont'd					Pearl Eleanor Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons) Christine Roderick Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons)	
22-009	D	Permanent Rights over 9803 square metres of agricultural land and copse (south of Love Lane)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG	NONE	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons)	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 0DW (in respect of rights reserved in Transfer dated 22 February 1995)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Land Plans Extent of acquisition or use Description of land	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
22-009 cont'd			John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG		John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons) Pearl Eleanor Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons) Christine Roderick Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			

	Land v	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights y of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))	
Number on		Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
22-010	K	Temporary Rights over 1644 square metres of agricultural land (east of Broad Gate)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG	NONE	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons) John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons) Pearl Eleanor Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons)	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 0DW (in respect of rights reserved in Transfer dated 22 February 1995) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	

	Land v	which is proposed to be subject to	BOOK OF RI DOCK OF RI DOC (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
22-010 cont'd					Christine Roderick Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (trading as S. G. Danby & Sons)		
22-011	D	of Love Lane)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) The Consolidated Charities Plum Cottage East Davids Lane Benington BOSTON PE22 0DA (as assumed owner)	NONE	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) The Consolidated Charities Plum Cottage East Davids Lane Benington BOSTON PE22 0DA (as assumed owner)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
22-011 cont'd			Unknown		Unknown				
22-012	D	Permanent Rights over 2114 square metres of agricultural land (south of Love Lane)	The Consolidated Charities Plum Cottage East Davids Lane Benington BOSTON PE22 0DA	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 0DW	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 0DW	Unknown (in respect of restrictive covenants as may have been imposed thereon before 7 June 2011)			
22-013	D	Permanent Rights over 381 square metres of drain (south of Love Lane)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner)	NONE	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
22-013 cont'd			The Consolidated Charities Plum Cottage East Davids Lane Benington BOSTON PE22 0DA (as assumed owner) Unknown		The Consolidated Charities Plum Cottage East Davids Lane Benington BOSTON PE22 0DA (as assumed owner) Unknown				
22-014	К	land (south of Love Lane)	The Consolidated Charities Plum Cottage East Davids Lane Benington BOSTON PE22 0DA	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 0DW	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 0DW	Unknown (in respect of restrictive covenants as may have been imposed thereon before 7 June 2011)			
22-015	K	Temporary Rights over 15 square metres of drain (south of Love Lane)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner)	NONE	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner)	Unknown			

	Land	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	
22-015 cont'd			John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) The Consolidated Charities Plum Cottage East Davids Lane Benington BOSTON PE22 0DA (as assumed owner) Unknown		John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) The Consolidated Charities Plum Cottage East Davids Lane Benington BOSTON PE22 0DA (as assumed owner) Unknown	
22-016	К	Temporary Rights over 57 square metres of agricultural land (south of Love Lane)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG	T. & B. Containers Ltd Brenton Villa Wrangle Bank Wrangle BOSTON PE22 9DL	T. & B. Containers Ltd Brenton Villa Wrangle Bank Wrangle BOSTON PE22 9DL	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	and Plans	Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
22-016 cont'd			John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG						
22-017	Κ	Temporary Rights over 494 square metres of agricultural land (south of Love Lane)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG	Brenton Villa Wrangle Bank Wrangle BOSTON	T. & B. Containers Ltd Brenton Villa Wrangle Bank Wrangle BOSTON PE22 9DL	NONE			

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Vind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to ^o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	F 1 - 1 - f	Descriptions (local	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-018	D	Permanent Rights over 4243 square metres of agricultural land (south of Love Lane)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG	T. & B. Containers Ltd Brenton Villa Wrangle Bank BOSTON PE22 9DL	T. & B. Containers Ltd Brenton Villa Wrangle Bank BOSTON PE22 9DL	NONE
22-019	K	Temporary Rights over 8 square metres of drain (east of Broad Gate)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner)	NONE	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
22-019 cont'd			John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) Unknown		John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) Unknown				
22-020	K	square metres of agricultural land (east of Broad Gate)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG	T. & B. Containers Ltd Brenton Villa Wrangle Bank Wrangle BOSTON PE22 9DL	T. & B. Containers Ltd Brenton Villa Wrangle Bank Wrangle BOSTON PE22 9DL	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights y of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))		
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
22-021	D	Permanent Rights over 825 square metres of drain (east of Broad Gate)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner)	NONE	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Lincolnshire PE22 9AA (as assumed owner)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)		

	Land	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of hand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-021 cont'd			Unknown		Unknown	
22-022	D	Permanent Rights over 948 square metres of agricultural land (east of Broad Gate)	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Unknown	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	Unknown
22-023	D	Permanent Rights over 29011 square metres of agricultural land (east of Broad Gate)	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	NONE

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-023 cont'd			Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA			
22-024	К	Temporary Rights over 814 square metres of agricultural land (east of Broad Gate)	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)
			Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA			
22-025	K	Temporary Rights over 15 square metres of access track (east of Broad Gate)	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner)	NONE	George Henry Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner)	Unknown

	Land wl	hich is proposed to be subject	BOOK OF REFE t to: (i) powers of compulsory acquisition, (ii) rig	nd Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) right: Lincolnshire	s to carry out protective works (Regulation 7(1)	ı(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-025 cont'd			John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Lincolnshire PE22 9AA (as assumed owner) Lincolnshire		John Arthur Danby Brickyard Farm Goldfen Bank Wrangle BOSTON Lincolnshire PE22 9BG (as assumed owner) Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Lincolnshire PE22 9AA (as assumed owner) Lincolnshire PE22 9AA (as assumed owner)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on		Descriptions (the d	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-026	D		Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Unknown	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA Unknown	Unknown
22-027	D	square metres of public road and verges (Broad Gate)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
22-027 cont'd			Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (assumed in respect of subsoil beneath public highway) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (assumed in respect of subsoil beneath public highway) Unknown			Unknown			
22-028	D	Permanent Rights over 13791 square metres of agricultural land (west of Broad Gate)	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner)	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
22-028 cont'd			Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Unknown		Unknown	Unknown			
22-029	K	Temporary Rights over 2011 square metres of agricultural land (south of Cragmire Lane and west of Broad Gate)	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Unknown	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown			

	Land	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to / of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
22-030	К	Temporary Rights over 1198 square metres of agricultural land (south of Cragmire Lane)	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	NONE
22-031	D	Permanent Rights over 2601 square metres of agricultural land (south of Cragmire Lane)	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
22-032	D	Permanent Rights over 2201 square metres of agricultural land and access track (south of Cragmire Lane)	The Alenson and Erskine Educational Foundation Crooks Cottage Wrangle Bank Wrangle BOSTON Lincolnshire PE22 9DL	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	Unknown (in respect of restrictive covenants as may have been imposed thereon before 20 July 2010)	
23-001	D	Permanent Rights over 3839 square metres of agricultural land (east of Cragmire Lane)	Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	NONE	
23-002	D	Cragmire Lane)	Christopher William Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) Jane Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner)	NONE	Christopher William Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) Jane Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Sesenption of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
23-002 cont'd			John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Robert John Edwards The Mallards Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) Unknown		John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Robert John Edwards The Mallards Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) Unknown				

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-003	К	Temporary Rights over 1172 square metres of agricultural land (east of Cragmire Lane)	Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	NONE
23-004	К	Temporary Rights over 13 square metres of drain (east of Cragmire Lane)	Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Unknown	NONE	Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Unknown	Unknown
23-005	D	square metres of agricultural land (east of Cragmire Lane)	Christopher William Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT	J F & J Edwards & Sons Unlimited New Farm Soulby Lane Wrangle BOSTON PE22 9BT	J F & J Edwards & Sons Unlimited New Farm Soulby Lane Wrangle BOSTON PE22 9BT	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
23-005 cont'd			Jane Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT Robert John Edwards The Mallards Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT						
23-006	K	Temporary Rights over 881 square metres of agricultural land (south of Cragmire Lane)	Christopher William Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT	J F & J Edwards & Sons Unlimited New Farm Soulby Lane Wrangle BOSTON PE22 9BT	J F & J Edwards & Sons Unlimited New Farm Soulby Lane Wrangle BOSTON PE22 9BT	NONE			

	Land	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-006 cont'd			Jane Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT Robert John Edwards The Mallards Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT			
23-007	K	Temporary Rights over 20 square metres of drain (south of Cragmire Lane)	Christopher William Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner)	NONE	Christopher William Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
23-007 cont'd			Jane Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) Robert John Edwards The Mallards Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) Robert John Edwards		Jane Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) Robert John Edwards The Mallards Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) Robert John Edwards				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
23-008	D	square metres of drain (east of Cragmire Lane)	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 ODW	NONE	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 0DW	NONE			
23-009	D		Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	Christopher William Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (in respect of rights reserved by Transfer dated 10 November 1995) Jane Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (in respect of rights reserved by Transfer dated 10 November 1995)			

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-009 cont'd						John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (in respect of rights reserved by Transfer dated 10 November 1995) Robert John Edwards The Mallards Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (in respect of rights reserved by Transfer dated 10 November 1995)
23-010	D	Permanent Rights over 141 square metres of drain (east of Cragmire Lane)	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner)	NONE	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner)	Unknown

	Land v	which is proposed to be subject to	BOOK OF RE DOCK OF RE DOCK OF RE	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-010 cont'd			Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Unknown		Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Unknown	
23-011	D	Permanent Rights over 847 square metres of public road and verges (Cragmire Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunications apparatus) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
23-011 cont'd			Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (assumed in respect of subsoil beneath public highway) Unknown						
23-012	D	square metres of agricultural land (west of Cragmire Road)	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Unknown	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-013	D	Permanent Rights over 12266 square metres of agricultural land (north of Cragmire Lane and east of Double Bank) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	Unknown (in respect of rights granted by Conveyance dated 29 September 1987) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP
23-014	К	Temporary Rights over 1482 square metres of agricultural land (north of Cragmire Lane)(excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	Phillip Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	(assumed in respect of rights of access) Unknown (in respect of rights granted by Conveyance dated 29 September 1987)
				William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	William Wrisdale Bethlem Farm Wainfleet St. Mary SKEGNESS PE24 4JZ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on		Description of land	Qualifying persons under Regulation 7(:	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition or use	Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-015	D	interests of the Crown)	John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Unknown
23-016	D	square metres of agricultural land (east of Common Road)	John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT	J F & J Edwards & Sons Unlimited New Farm Soulby Lane Wrangle BOSTON PE22 9BT	J F & J Edwards & Sons Unlimited New Farm Soulby Lane Wrangle BOSTON PE22 9BT	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Frank Edwards)

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-016 cont'd 23-017	K	Temporary Rights over 4615 square metres of drain (east of Double Bank)	John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT	J F & J Edwards & Sons Unlimited New Farm Soulby Lane Wrangle BOSTON PE22 9BT	J F & J Edwards & Sons Unlimited New Farm Soulby Lane Wrangle BOSTON PE22 9BT	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for John Frank Edwards)	
						Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
23-018	К	square metres of access splay	John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner)	NONE	John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
23-018 cont'd			Unknown		Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
23-019	D	Permanent Rights over 583 square metres of drain (east of Common Road)	John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) Unknown	NONE	John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (as assumed owner) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunications apparatus) Unknown			
23-020	D	Permanent Rights over 1138 square metres of public road, drain and verges (Common Road)	John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
23-020 cont'd			John Robert Mowbray Windermere Low Road Friskney BOSTON Lincolnshire PE22 8NJ (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Margaret Elizabeth Allison Bromby Lodge Low Road Friskney BOSTON Lincolnshire PE22 8NW (assumed in respect of subsoil beneath public highway) Unknown			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
23-021	D	square metres of drain and verges (west of Common Road)	John Robert Mowbray Windermere Low Road Friskney BOSTON Lincolnshire PE22 8NJ Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Margaret Elizabeth Allison Bromby Lodge Low Road Friskney BOSTON Lincolnshire PE22 8NW	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
23-022	D	Permanent Rights over 10657 square metres of agricultural land (west of Common Road)	John Robert Mowbray Windermere Low Road Friskney BOSTON Lincolnshire PE22 8NJ Margaret Elizabeth Allison Bromby Lodge Low Road Friskney BOSTON Lincolnshire PE22 8NW	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			
23-023	G	Temporary Rights over 12779 square metres of agricultural land (west of Double Bank)	John Robert Mowbray Windermere Low Road Friskney BOSTON Lincolnshire PE22 8NJ	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			

	Land v	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))		
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
23-023 cont'd			Margaret Elizabeth Allison Bromby Lodge Low Road Friskney BOSTON Lincolnshire PE22 8NW					
23-024	D	Permanent Rights over 13221 square metres of agricultural land (west of Common Road)	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Unknown	NONE	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Unknown M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)		

	Land	which is proposed to be subject t	BOOK OF o: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights ty of Lincolnshire	to carry out protective works (Regulation 7(1))(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-025	К	Temporary Rights over 604 square metres of agricultural land (west of Common Road)	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (co. asumed owner)	NONE	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
			(as assumed owner) Unknown		(as assumed owner) Unknown M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	

	Land v	which is proposed to be subject to	BOOK OF R DOGK OF R (ii) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on		Description of load	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-026	D	Permanent Rights over 34 square metres of agricultural land (west of Common Road)	Castlegate Trustees Limited 8 Castlegate GRANTHAM Lincolnshire NG31 6SE (as assumed owner) Edith Alice Baker Wyberton House Wainfleet Road Old Leake BOSTON Lincolnshire PE22 9HT (as assumed owner) Ian Michael Baker The Gables Ings Lane Leverton BOSTON PE22 0AX (as assumed owner) Unknown	NONE	Castlegate Trustees Limited 8 Castlegate GRANTHAM Lincolnshire NG31 6SE (as assumed owner) Edith Alice Baker Wyberton House Wainfleet Road Old Leake BOSTON Lincolnshire PE22 9HT (as assumed owner) Ian Michael Baker The Gables Ings Lane Leverton BOSTON PE22 0AX (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-027	D	Permanent Rights over 9538 square metres of agricultural land (west of Common Road)	Castlegate Trustees Limited 8 Castlegate GRANTHAM Lincolnshire NG31 6SE (as trustee of M Baker (Produce) Ltd Pension Fund) Edith Alice Baker Wyberton House Wainfleet Road Old Leake BOSTON Lincolnshire PE22 9HT (as trustee of M Baker (Produce) Ltd Pension Fund) Ian Michael Baker The Gables Ings Lane Leverton BOSTON PE22 0AX (as trustee of M Baker (Produce) Ltd Pension Fund)	M.Baker Produce Limited Wyberton House Wainfleet Road Old Leake BOSTON Lincolnshire PE22 9HT	M.Baker Produce Limited Wyberton House Wainfleet Road Old Leake BOSTON Lincolnshire PE22 9HT	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)

	Land v	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights t [,] of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on		Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-02623- 028, 23-027,	K	Temporary Rights over 31 square metres of agricultural land (west of Common Road)	Castlegate Trustees Limited 8 Castlegate GRANTHAM Lincolnshire NG31 6SE (as trustee of M Baker (Produce) Ltd Pension Fund) Edith Alice Baker Wyberton House Wainfleet Road Old Leake BOSTON Lincolnshire PE22 9HT (as trustee of M Baker (Produce) Ltd Pension Fund) Ian Michael Baker The Gables Ings Lane Leverton BOSTON PE22 0AX(as trustee of M Baker (Produce) Ltd Pension Fund)	M.Baker Produce Limited Wyberton House Wainfleet Road Old Leake BOSTON Lincolnshire PE22 9HT	M.Baker Produce Limited Wyberton House Wainfleet Road Old Leake BOSTON Lincolnshire PE22 9HT	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
23-029	К	Temporary Rights over 83 square metres of access splay (west of Common Road)	John Robert Mowbray Windermere Low Road Friskney BOSTON Lincolnshire PE22 8NJ Margaret Elizabeth Allison Bromby Lodge Low Road Friskney BOSTON Lincolnshire PE22 8NW	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)				
23-030	D	Permanent Rights over 28270 square metres of agricultural land and drain (east of Manor Lane)	John Robert Mowbray Windermere Low Road Friskney BOSTON Lincolnshire PE22 8NJ	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 0SE	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans	Land Plans Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
23-030 cont'd			Margaret Elizabeth Allison Bromby Lodge Low Road Friskney BOSTON Lincolnshire PE22 8NW							
23-031	K	Temporary Rights over 469 square metres of agricultural land (east of Manor Lane)	John Robert Mowbray Windermere Low Road Friskney BOSTON Lincolnshire PE22 8NJ Margaret Elizabeth Allison Bromby Lodge Low Road Friskney BOSTON Lincolnshire PE22 8NW	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	NONE				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
23-032	D	Permanent Rights over 396 square metres of public road and verges (Manor Lane)	John Robert Mowbray Windermere Low Road Friskney BOSTON Lincolnshire PE22 8NJ (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Margaret Elizabeth Allison Bromby Lodge Low Road Friskney BOSTON Lincolnshire PE22 8NW (assumed in respect of subsoil beneath public highway) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans		Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
23-033	D	Permanent Rights over 328 square metres of public road and verges (Manor Lane)	John Robert Mowbray Windermere Low Road Friskney BOSTON Lincolnshire PE22 8NJ Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Margaret Elizabeth Allison Bromby Lodge Low Road Friskney BOSTON Lincolnshire PE22 8NW	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground apparatus)				

	Land v	which is proposed to be subject to	BOOK OF F bo: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	5	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
23-034	D	Permanent Rights over 7342 square metres of agricultural land and drain (west of Manor Lane)	John Robert Mowbray Windermere Low Road Friskney BOSTON Lincolnshire PE22 8NJ Margaret Elizabeth Allison Bromby Lodge Low Road Friskney BOSTON Lincolnshire PE22 8NW	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	NONE
23-035	К	Temporary Rights over 177 square metres of agricultural land (west of Manor Lane)	John Robert Mowbray Windermere Low Road Friskney BOSTON Lincolnshire PE22 8NJ	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	NONE

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
23-035 cont'd			Margaret Elizabeth Allison Bromby Lodge Low Road Friskney BOSTON Lincolnshire PE22 8NW			
24-001	К	Temporary Rights over 723 square metres of agricultural land (west of Manor Lane)	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner) Unknown	NONE	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
24-002	D	Permanent Rights over 25766 square metres of agricultural land (east of Sea Dyke and west of Manor Lane)	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner)	NONE	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)

	Land	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
24-002 cont'd			Unknown		Unknown	Unknown
24-003	D	Permanent Rights over 969 square metres of public road, drain and verges (Sea Dyke)	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown
24-004	K	Temporary Rights over 179 square metres of access splay (west of Sea Dyke)	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner)	NONE	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
24-004 cont'd			Unknown		Unknown	Unknown				
24-005	D	Permanent Rights over 52453 square metres of agricultural land, hedgerow, access track and drain (west of Sea Dyke)	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner) Unknown	NONE	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)				
24-006	K	Temporary Rights over 1253 square metres of access track and drain (west of Sea Dyke)	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner) Unknown	NONE	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner) Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground electricity apparatus) Unknown				

	Land	which is proposed to be subject to	BOOK OF RE b: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
24-006 cont'd						Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)
24-007	К	Temporary Rights over 2588 square metres of agricultural land and access track (west of Seadyke)	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner) Unknown	NONE	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
24-008	К	Temporary Rights over 705 square metres of agricultural land (east of Church Road)	Andrew John Grant Leake House Hobhole Bank Old Leake BOSTON Lincolnshire PE22 9RT (as trustee of the JW Grant Pension and Life Assurance Fund)	NONE	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans	Land Plans	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
24-008 cont'd			The Executor of the Estate of the Late James Edward Grant Leake House Hobhole Bank Old Leake BOSTON Lincolnshire PE22 9RT (as trustee of the JW Grant Pension and Life Assurance Fund)							
24-009	K	(east of Church Road)	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner) Unknown		J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on		Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
24-010	D	Permanent Rights over 20630 square metres of agricultural land, access track and drain (south of Fold Hill)	Dorothy May Holiday Rose Villa Church Road Old Leake BOSTON Lincolnshire PE22 9PD Robert Holiday Rose Villa Church Road Old Leake BOSTON Lincolnshire PE22 9PD	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
24-011	К	Temporary Rights over 335 square metres of access track (south of Fold Hill)	Dorothy May Holiday Rose Villa Church Road Old Leake BOSTON Lincolnshire PE22 9PD	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
24-011 cont'd			Robert Holiday Rose Villa Church Road Old Leake BOSTON Lincolnshire PE22 9PD						
24-012	K	and drain (south of Fold Hill)	Carole June Mackinder 2 Vine Farmyard Collingham NEWARK NG23 7NF (as trustee of The David Mackinder Will Trust) Jonathan David Mackinder 9 Chippendale Road LINCOLN LN6 3PP (as trustee of The David Mackinder Will Trust) Louise May Allen 14 Drove Lane Coddington NEWARK NG24 2RA (as trustee of The David Mackinder Will Trust)	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
24-013	D	Permanent Rights over 24351 square metres of agricultural land and drain (east of Church Road)	Carole June Mackinder 2 Vine Farmyard Collingham NEWARK NG23 7NF (as trustee of The David Mackinder Will Trust) Jonathan David Mackinder 9 Chippendale Road LINCOLN LN6 3PP (as trustee of The David Mackinder Will Trust) Louise May Allen 14 Drove Lane Coddington NEWARK NG24 2RA (as trustee of The David Mackinder Will Trust)	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus)
25-001	D	Permanent Rights over 8 square metres of agricultural land (east of Church Road)	Carole June Mackinder 2 Vine Farmyard Collingham NEWARK NG23 7NF (as assumed owner)	NONE	Carole June Mackinder 2 Vine Farmyard Collingham NEWARK NG23 7NF (as assumed owner)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-001 cont'd			Jonathan David Mackinder 9 Chippendale Road LINCOLN LN6 3PP (as assumed owner) Louise May Allen 14 Drove Lane Coddington NEWARK NG24 2RA (as assumed owner) Stephanie Marie Slater 20 Lodge Hill Road OSSETT West Yorkshire WF5 9RU (as assumed owner) Unknown		Jonathan David Mackinder 9 Chippendale Road LINCOLN LN6 3PP (as assumed owner) Louise May Allen 14 Drove Lane Coddington NEWARK NG24 2RA (as assumed owner) Stephanie Marie Slater 20 Lodge Hill Road OSSETT West Yorkshire WF5 9RU (as assumed owner) Unknown	Unknown			
25-002	D	Permanent Rights over 205 square metres of public road and verge (Church Road)	Carole June Mackinder 2 Vine Farmyard Collingham NEWARK NG23 7NF (as trustee of The David Mackinder Will Trust)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-002 cont'd			Jonathan David Mackinder 9 Chippendale Road LINCOLN LN6 3PP (as trustee of The David Mackinder Will Trust) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Louise May Allen 14 Drove Lane Coddington NEWARK NG24 2RA (as trustee of The David Mackinder Will Trust)			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)			
25-003	D	Permanent Rights over 627 square metres of public road (Church Road)	Andrew John Grant Leake House Hobhole Bank Old Leake BOSTON Lincolnshire PE22 9RT (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
25-003 cont'd			Carole June Mackinder 2 Vine Farmyard Collingham NEWARK NG23 7NF (assumed in respect of subsoil beneath public highway) Jonathan David Mackinder 9 Chippendale Road LINCOLN LN6 3PP (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Louise May Allen 14 Drove Lane Coddington NEWARK NG24 2RA (assumed in respect of subsoil beneath public highway)			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-003 cont'd			The Executor of the Estate of the Late James Edward Grant Leake House Hobhole Bank Old Leake BOSTON Lincolnshire PE22 9RT (assumed in respect of subsoil beneath public highway) The Executor of the Estate of the Late William Edward Grant Leake House Hobhole Bank Old Leake BOSTON Lincolnshire PE22 9RT (in respect of subsoil beneath public highway) Unknown						
25-004	D	Permanent Rights over 4516 square metres of agricultural land (west of Church Road)	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner)	NONE	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
25-004 cont'd			Unknown		Unknown				
25-005	D	Permanent Rights over 26572 square metres of agricultural land (west of Church Road)	Andrew John Grant Leake House Hobhole Bank Old Leake BOSTON Lincolnshire PE22 9RT The Executor of the Estate of the Late James Edward Grant Leake House Hobhole Bank Old Leake BOSTON Lincolnshire PE22 9RT The Executor of the Estate of the Late William Edward Grant Leake House Hobhole Bank Old Leake BOSTON Lincolnshire PE22 9RT	NONE	J W Grant Co Fold Hill Old Leake BOSTON Lincolnshire PE22 9PJ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)			

	Land	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order :FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
25-006	D	Permanent Rights over 1218 square metres of public road (Sibsey Road, B1184)	Andrew John Grant Leake House Hobhole Bank Old Leake BOSTON Lincolnshire PE22 9RT (assumed in respect of subsoil beneath public highway) Elsie May Young Beechcroft Southfields Lane Old Leake BOSTON Lincolnshire PE22 9LW (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-006 cont'd			The Executor of the Estate of the Late James Edward Grant Leake House Hobhole Bank Old Leake BOSTON Lincolnshire PE22 9RT (assumed in respect of subsoil beneath public highway) The Executor of the Estate of the Late William Edward Grant Leake House Hobhole Bank Old Leake BOSTON Lincolnshire PE22 9RT (in respect of subsoil beneath public highway) Unknown			Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
25-007	D	Permanent Rights over 158 square metres of drain (west of Sibsey Road, B1184)	Elsie May Young Beechcroft Southfields Lane Old Leake BOSTON Lincolnshire PE22 9LW (as assumed owner)	NONE	Elsie May Young Beechcroft Southfields Lane Old Leake BOSTON Lincolnshire PE22 9LW (as assumed owner)	Unknown			

	Land	which is proposed to be subject t	BOOK OF R co: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to r of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
25-007 cont'd			Unknown			Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
25-008	D	Permanent Rights over 1042 square metres of grassland (west of Sibsey Road, B1184)	Elsie May Young Beechcroft Southfields Lane Old Leake BOSTON Lincolnshire PE22 9LW	NONE	Staples Brothers Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	NONE
25-009	D	Permanent Rights over 784 square metres of public road and verge (Cowbroads Lane)	Elsie May Young Beechcroft Southfields Lane Old Leake BOSTON Lincolnshire PE22 9LW (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-009 cont'd			Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (assumed in respect of subsoil beneath public highway) Unknown						
25-010	D	Permanent Rights over 13742 square metres of agricultural land (south of The Gride)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Unknown	Chas Wright Farms Ltd Charmae Main Road Old Leake BOSTON Lincolnshire PE22 9LH	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques) Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques)	Unknown			

	Land	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	Planning (Applications: Prescribed Forms and Procedures) 2009 Qualifying per	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire
25-010 cont'd					Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques) Unknown C. Wright (Farmers) Limited Church End Old Leake BOSTON Lincolnshire PE22 9NU	
25-011	D	Permanent Rights over 2073 square metres of agricultural land (south of The Gride)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT	Chas Wright Farms Ltd Charmae Main Road Old Leake BOSTON Lincolnshire PE22 9LH	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques) Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques)	Keens House Anton Mill Road ANDOVER

	Land	which is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-011 cont'd					Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques) C. Wright (Farmers) Limited Church End Old Leake BOSTON Lincolnshire PE22 9NU		
25-012	D	Permanent Rights over 6222 square metres of agricultural land (south of The Gride)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT		Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques)	NONE	

	Land	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	Category 2 Qualifying persons under Regulation		
Land Plans		Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	0 1
25-012 cont'd					Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques) M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	
25-013	D	Permanent Rights over 2096 square metres of agricultural land and drain (south of The Gride)	Alan Richard Daubney Fair View Old Main Road Old Leake BOSTON PE22 9HR John Henry Daubney Bayholme Farm Wainfleet Road Old Leake BOSTON Lincolnshire PE22 9HT	NONE	Alan Richard Daubney Fair View Old Main Road Old Leake BOSTON PE22 9HR (trading as F Daubney & Sons) John Henry Daubney Bayholme Farm Wainfleet Road Old Leake BOSTON Lincolnshire PE22 9HT (trading as F. Daubney & Sons)	NONE

Reputed Owners	Category 1 le Infrastructure Planning (Applications: I Regulations 2009 Lessees or Tenants	: Prescribed Forms and Procedures) Occupiers	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
ibney Alan Richa		Occupiers	(Applications: Prescribed Forms and
hweikhardt Alan Richa			
Old Main Old Leake BOSTON PE22 9HR (trading as In John Henr nue Bayholme Bayholme INT Wainfleet Old Leake BOSTON Lincolnshi PE22 9HT (trading as ite Richard Jo age The Cherr Church En Wrangle	wFair Vn RoadOld MkeOld LeJBOSTCIRPE22as F Daubney & Sons)(tradinnry DaubneyJohn Hhe FarmBayhoet RoadWainfkeOld LeJBOSTChireLincolITPE22as F. Daubney & Sons)(tradinJohn DaubneyRichauITPE22as F. Daubney & Sons)(tradinJohn DaubneyRichaurriesThe CEndChurceWran	View Main Road Leake TON 2 9HR ling as F Daubney & Sons) Henry Daubney nolme Farm nfleet Road Leake TON Jolnshire 2 9HT ling as F. Daubney & Sons) ard John Daubney Cherries rch End ngle	NONE
it	Je Bayholn IT Wainfle Old Leal BOSTON Lincolns PE22 9- (trading e Richard ge Richard ge The Che Church Wrangle BOSTON PE22 9E	JeeBayholme FarmBayhITWainfleet RoadWainOld LeakeOld IBOSTONBOSTLincolnshireLincolPE22 9HTPE22(trading as F. Daubney & Sons)(trading as F. Daubney & Sons)geRichard John DaubneyRichgeThe CherriesTheChurch EndChurWrangleWraBOSTONBOSTPE22 9EWPE22	JeBayholme FarmBayholme FarmITWainfleet RoadWainfleet RoadOld LeakeOld LeakeBOSTONBOSTONLincolnshireLincolnshirePE22 9HTPE22 9HT(trading as F. Daubney & Sons)(trading as F. Daubney & Sons)eRichard John DaubneyRichard John DaubneygeChurch EndChurch EndWrangleWrangleBOSTONBOSTONBOSTONBOSTONPE22 9EWPE22 9EW

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on		Development	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-015	K	Temporary Rights over 1238 square metres of agricultural land (south of The Gride)	Elizabeth Amy Schweikhardt The Moorings Sea End Benington BOSTON Lincolnshire PE22 0DN Rosanna Skelham 17 Lyndham Avenue BURTON-ON-TRENT DE15 9BQ Victoria Jane White Three Ways Cottage Outgate Leverton BOSTON Lincolnshire PE22 0AA	Alan Richard Daubney Fair View Old Main Road Old Leake BOSTON PE22 9HR (trading as F. Daubney & Sons) John Henry Daubney Bayholme Farm Wainfleet Road Old Leake BOSTON Lincolnshire PE22 9HT (trading as F. Daubney & Sons) Richard John Daubney The Cherries Church End Wrangle BOSTON PE22 9EW (trading as F. Daubney & Sons)	Alan Richard Daubney Fair View Old Main Road Old Leake BOSTON PE22 9HR (trading as F. Daubney & Sons) John Henry Daubney Bayholme Farm Wainfleet Road Old Leake BOSTON Lincolnshire PE22 9HT (trading as F. Daubney & Sons) Richard John Daubney The Cherries Church End Wrangle BOSTON PE22 9EW (trading as F. Daubney & Sons)	NONE			

	Land	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Regulations 2009	Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures)		Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
25-016	K	Temporary Rights over 1431 square metres of agricultural land (south of The Gride)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT		Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques) M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on		Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition or use		Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-017	D	Permanent Rights over 3320 square metres of agricultural land (north of Pode Lane)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT		Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques) M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	NONE			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-018	D	Permanent Rights over 234 square metres of agricultural land (north of Pode Lane)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Unknown	NONE	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques) Unknown	Unknown			

	Land	which is proposed to be subject t	BOOK OF RE co: (i) powers of compulsory acquisition, (ii)	ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights t of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on		Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-018 cont'd					M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	
25-019	D	Permanent Rights over 5592 square metres of agricultural land (north of Pode Lane)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT		Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)
					Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques)	

	Land v	which is proposed to be subject to	BOOK OF F o: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights y of Lincolnshire	s to carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-019 cont'd					M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA		
25-020	D	square metres of agricultural	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner)	NONE	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques)	Unknown	

	Land	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii) r	ind Development Consent Order FERENCE - PART 1 ight to use the land, and/or (iii) rights f Lincolnshire	to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (App Regulations 2009	lications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans	Land Plans Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-020 cont'd			Unknown		Unknown M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	
25-021	D	Permanent Rights over 6748 square metres of agricultural land (north of Pode Lane)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT Susan Mary Jaques Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT		Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON LincoInshire PE22 9QT (trading as F & MP Jaques Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques)	NONE

	Land	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
25-021 cont'd					Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques) M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	
25-022	D	Permanent Rights over 836 square metres of public road and verge (Pode Lane)	Elsie May Young Beechcroft Southfields Lane Old Leake BOSTON Lincolnshire PE22 9LW (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus) Unknown

	Land wh	nich is proposed to be subject	BOOK OF REFI t to: (i) powers of compulsory acquisition, (ii) rig	nd Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
25-022 cont'd			Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (assumed in respect of subsoil beneath public highway) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (assumed in respect of subsoil beneath public highway) The Alenson and Erskine Educational Foundation Crooks Cottage Wrangle Bank Wrangle BOSTON Lincolnshire PE22 9DL (assumed in respect of subsoil beneath public highway) Unknown			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-023	K	Temporary Rights over 408 square metres of agricultural land (south of Pode Lane)	Elsie May Young Beechcroft Southfields Lane Old Leake BOSTON Lincolnshire PE22 9LW	NONE	Staples Brothers Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
25-024	D	square metres of agricultural	Elsie May Young Beechcroft Southfields Lane Old Leake BOSTON Lincolnshire PE22 9LW	NONE	Staples Brothers Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			
25-025	D	square metres of agricultural	Elsie May Young Beechcroft Southfields Lane Old Leake BOSTON Lincolnshire PE22 9LW	NONE	Staples Brothers Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-026	D		Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) The Alenson and Erskine Educational Foundation Crooks Cottage Wrangle Bank Wrangle BOSTON Lincolnshire PE22 9DL (as assumed owner) Unknown	NONE	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) The Alenson and Erskine Educational Foundation Crooks Cottage Wrangle Bank Wrangle BOSTON Lincolnshire PE22 9DL (as assumed owner) Unknown	Unknown			

	Land	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-026 cont'd					M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	
25-027	D	Permanent Rights over 11 square metres of agricultural land (south of Pode Lane)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner)	NONE	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques) Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-027 cont'd			Unknown		Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques) Unknown M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA				
25-028	K	Temporary Rights over 68 square metres of access track (south of Pode Lane)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner)	NONE	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-028 cont'd			Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Unknown		Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques) Unknown M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to ¹ of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faild	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-029	K	Temporary Rights over 18 square metres of agricultural land (south of Pode Lane)	The Alenson and Erskine Educational Foundation Crooks Cottage Wrangle Bank Wrangle BOSTON Lincolnshire PE22 9DL	Colin Frank Daubney Daubneys Bungalow Chapel Road Old Leake BOSTON PE22 9PN	Colin Frank Daubney Daubneys Bungalow Chapel Road Old Leake BOSTON PE22 9PN	Unknown (in respect of restrictive covenants as may have been imposed thereon before 20 July 2010) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)
25-030	D	Permanent Rights over 21731 square metres of agricultural land (south of Pode Lane)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner)	Chas Wright Farms Ltd Charmae Main Road Old Leake BOSTON Lincolnshire PE22 9LH	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	ent of acquisition or use Description of land	Qualifying persons under Regulation 7(1)(Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-030 cont'd			Unknown		Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques) Unknown C. Wright (Farmers) Limited Church End Old Leake BOSTON Lincolnshire PE22 9NU	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to ¹ of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faild	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
25-031	D	Permanent Rights over 12131 square metres of agricultural land and drain (south of Pode Lane and east of Skipmarsh Lane)	The Alenson and Erskine Educational Foundation Crooks Cottage Wrangle Bank Wrangle BOSTON Lincolnshire PE22 9DL	Colin Frank Daubney Daubneys Bungalow Chapel Road Old Leake BOSTON PE22 9PN	Colin Frank Daubney Daubneys Bungalow Chapel Road Old Leake BOSTON PE22 9PN	Unknown (in respect of restrictive covenants as may have been imposed thereon before 20 July 2010) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
25-032	D	Permanent Rights over 7515 square metres of agricultural land and drain (east of Skipmarsh Lane)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner)	NONE	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-032 cont'd			Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Unknown		Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques) Unknown M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
25-033	D		Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT		Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques) Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques) Susan Mary Jaques Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground sewerage apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			

	Land v	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order :FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-033 cont'd					M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA		
26-001	К	Temporary Rights over 271 square metres of agricultural land (east of Skipmarsh Lane)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner)	NONE	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques) Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground sewerage apparatus) Unknown	

	Land v	which is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-001 cont'd			Unknown		Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques) Unknown M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	
26-002	K	Temporary Rights over 45 square metres of agricultural land (east of Skipmarsh Lane)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT		Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques)	NONE

	Land	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-002 cont'd			Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT		Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques) Susan Mary Jaques Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques) M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	
26-003	D	Permanent Rights over 758 square metres of public road and verges (Skipmarsh Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)

	Land	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-003 cont'd			Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (assumed in respect of subsoil beneath public highway) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (assumed in respect of subsoil beneath public highway) Unknown			National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown
26-004	К	Temporary Rights over 153 square metres of agricultural land (west of Skipmarsh Lane)	Stanley David Codd Will Trust Sleepy Hollow Chapel Lane Wrangle BOSTON PE22 9AP	NONE	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE	Unknown

	Land	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
26-004 cont'd			Unknown		Unknown		
26-005	D	Permanent Rights over 24344 square metres of agricultural land (east of Southfields and west of Skipmarsh Lane)	Stanley David Codd Will Trust Sleepy Hollow Chapel Lane Wrangle BOSTON PE22 9AP Unknown	NONE	Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 OSE Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	
26-006	D	Permanent Rights over 1067 square metres of public road and verges (Southfields)	Carole Jane Mitcham Burton Lea Main Road Leverton BOSTON Lincolnshire PE22 OAZ (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	

	Land	which is proposed to be subject to	BOOK OF RI DOCK OF RI DOC (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
26-006 cont'd			The Poors Land Charity Honey Lodge Sea End Benington BOSTON PE22 ODN (assumed in respect of subsoil beneath public highway) Unknown			Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)
26-007	D		Carole Jane Mitcham Burton Lea Main Road Leverton BOSTON Lincolnshire PE22 0AZ (as assumed owner) Unknown	NONE	Unknown T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	Unknown

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights t of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
26-007 cont'd					M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA		
26-008	D	and drain (west of Southfields)	Carole Jane Mitcham Burton Lea Main Road Leverton BOSTON Lincolnshire PE22 0AZ (as assumed owner) The Poors Land Charity Honey Lodge Sea End Benington BOSTON PE22 0DN (as assumed owner) Unknown	NONE	Carole Jane Mitcham Burton Lea Main Road Leverton BOSTON Lincolnshire PE22 0AZ (as assumed owner) The Poors Land Charity Honey Lodge Sea End Benington BOSTON PE22 0DN (as assumed owner) Unknown T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	Unknown	

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-008 cont'd					M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	
26-009	D	Permanent Rights over 3900 square metres of agricultural land (west of Southfields)	The Poors Land Charity Honey Lodge Sea End Benington BOSTON PE22 0DN	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT	Unknown
			Unknown		Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
26-010	D	Permanent Rights over 318 square metres of drain (west of Southfields)	Carole Jane Mitcham Burton Lea Main Road Leverton BOSTON Lincolnshire PE22 OAZ (as assumed owner)	NONE	Carole Jane Mitcham Burton Lea Main Road Leverton BOSTON Lincolnshire PE22 OAZ (as assumed owner)	Unknown

	Land	which is proposed to be subject to	BOOK OF RI DOCK OF RI DOC (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-010 cont'd			The Poors Land Charity Honey Lodge Sea End Benington BOSTON PE22 0DN (as assumed owner) Unknown		The Poors Land Charity Honey Lodge Sea End Benington BOSTON PE22 ODN (as assumed owner) Unknown T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	
26-011	D	Permanent Rights over 16555 square metres of agricultural land and drain (west of Southfields)	Carole Jane Mitcham Burton Lea Main Road Leverton BOSTON Lincolnshire PE22 0AZ	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (in respect of riparian rights)

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
26-011 cont'd					T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	
26-012	D	Permanent Rights over 18074 square metres of agricultural land (east of Ings Drove)	Andrew Charles Wright Charmae Main Road Old Leake BOSTON Lincolnshire PE22 9LH (trading as Chas Wright Farms Limited)	NONE	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT Chas Wright Farms Ltd Charmae Main Road Old Leake BOSTON Lincolnshire	NONE
26-013	D	Permanent Rights over 827 square metres of public road and verges (Ings Drove)	Andrew Charles Wright Charmae Main Road Old Leake BOSTON Lincolnshire PE22 9LH (assumed in respect of subsoil beneath public highway)	NONE	PE22 9LH Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)

	Land wł	nich is proposed to be subject	t to: (i) powers of compulsory acquisition, (ii) rig	RENCE - PART 1	arry out protective works (Regulation 7(1))	a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(a	Category 1) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
26-013 cont'd			Christopher Ashley Clements 191 London Road BOSTON Lincolnshire PE21 7HG (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (assumed in respect of subsoil beneath public highway) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (assumed in respect of subsoil beneath public highway) Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (assumed in respect of subsoil beneath public highway)			Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
26-013 cont'd			Unknown						
26-014	D	Permanent Rights over 143 square metres of drain (south of Ings Drove)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	B.S. Pension Fund Trustee Limited 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (in respect of covenants relating to use of land listed in Transfer dated 07 April 1998) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of rights stated in Conveyance dated 18 December 1981)			

	Land v	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-015	К	Temporary Rights over 321 square metres of agricultural land (south of Ings Drove)	Christopher Ashley Clements 191 London Road BOSTON Lincolnshire PE21 7HG	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
26-016	D	Permanent Rights over 389 square metres of agricultural land (south of Ings Drove)	Christopher Ashley Clements 191 London Road BOSTON Lincolnshire PE21 7HG	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
26-017	D	Permanent Rights over 56 square metres of drain (south of Ings Drove)	Christopher Ashley Clements 191 London Road BOSTON Lincolnshire PE21 7HG (as assumed owner) Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner)	NONE	Christopher Ashley Clements 191 London Road BOSTON Lincolnshire PE21 7HG (as assumed owner) Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
26-017 cont'd			Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Unknown		Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA (as assumed owner) Unknown				
26-018	D	square metres of agricultural land (south of Ings Drove)	Malcolm Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA Neville Leggate Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	M. Leggate & Sons (Produce) Limited Main Road Wrangle BOSTON Lincolnshire PE22 9AA	B.S. Pension Fund Trustee Limited 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (in respect of covenants relating to use of land listed in Transfer dated 07 April 1998) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of rights stated in Conveyance dated 18 December 1981)			

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Land Plans Extent of acquisition or use Description of land	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-018 cont'd						Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
26-019	K	land (east of Ings Road)	Isaac Henry Roy Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 0EE Unknown	Bell Brothers Nurseries Limited The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	Bell Brothers Nurseries Limited The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA Unknown	Unknown
26-020	К			NONE	Isaac Henry Roy Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 OEE (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
26-020 cont'd			Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Unknown		Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Unknown				
26-021	К	Temporary Rights over 291 square metres of agricultural land (east of Ings Road)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Unknown	Chas Wright Farms Ltd Charmae Main Road Old Leake BOSTON Lincolnshire PE22 9LH	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques) Unknown Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques)	Unknown			

	Land	which is proposed to be subject to	BOOK OF RI bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-021 cont'd					Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques) C. Wright (Farmers) Limited Church End Old Leake BOSTON Lincolnshire PE22 9NU	
26-022	D	Permanent Rights over 6766 square metres of agricultural land (east of Ings Road)	Isaac Henry Roy Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 OEE (as assumed owner) Unknown	Bell Brothers Nurseries Limited The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	Bell Brothers Nurseries Limited The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA Unknown	Unknown

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to • of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-023	D	Permanent Rights over 184 square metres of drain (east of Ings Road)	Isaac Henry Roy Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 OEE (as assumed owner) Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Unknown	NONE	Isaac Henry Roy Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 OEE (as assumed owner) Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Unknown	Unknown
26-024	D	Permanent Rights over 18921 square metres of agricultural land and drain (east of Ings Road)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Unknown	Chas Wright Farms Ltd Charmae Main Road Old Leake BOSTON Lincolnshire PE22 9LH	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques) Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown

	Land v	which is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1))	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applic) Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
26-024 cont'd					Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques) C. Wright (Farmers) Limited Church End Old Leake BOSTON Lincolnshire PE22 9NU	
26-025	G	Temporary Rights over 9084 square metres of agricultural land (east of Ings Road)	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner) Unknown	Chas Wright Farms Ltd Charmae Main Road Old Leake BOSTON Lincolnshire PE22 9LH	Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (as assumed owner and trading as F & MP Jaques) Unknown	Unknown

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
26-025 cont'd					Simon Jaques 109 Spilsby Road BOSTON PE21 9PE (trading as F & MP Jaques) Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (trading as F & MP Jaques) C. Wright (Farmers) Limited Church End Old Leake BOSTON Lincolnshire PE22 9NU	
27-001	D	Permanent Rights over 898 square metres of public road and verge (Ings Road)	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
27-001 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Michael Peter Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT (assumed in respect of subsoil beneath public highway) Unknown			National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown			
27-002	D	Permanent Rights over 1536 square metres of agricultural land (west of Ings Road)	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (trading as A. E. Vere & Sons)	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	NONE			

	Land v	which is proposed to be subject to	BOOK OF R b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
27-003	G	Temporary Rights over 22350 square metres of agricultural land (west of Ings Road)	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (trading as A. E. Vere & Sons)	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	NONE
27-004	G	Temporary Rights over 9777 square metres of agricultural land (west of Ings Road)	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (trading as A. E. Vere & Sons)	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	NONE
27-005	D	Permanent Rights over 24498 square metres of agricultural land (west of Ings Road)	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (trading as A. E. Vere & Sons)	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	NONE
27-006	D	Permanent Rights over 18004 square metres of agricultural land (west of Ings Road)	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (trading as A. E. Vere & Sons)	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	NONE

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on		Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
27-007	D	square metres of drain (west of Ings Road)	Fishtoft BOSTON Lincolnshire PE22 0RH (as assumed owner) Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (as assumed owner) Unknown	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 0RH (as assumed owner) Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (as assumed owner) Unknown	Unknown
27-008	D	square metres of drain (west of Ings Road)	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (as assumed owner)	NONE	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
27-008 cont'd			Jonathan Darcy Bell Bell Brothers Nurseries West End BoSTON Lincolnshire PE22 OEE (as assumed owner) Robert Anthony Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 OEE (as assumed owner) Unknown		Jonathan Darcy Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 0EE (as assumed owner) Robert Anthony Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 0EE (as assumed owner) Unknown				
27-009	D	Permanent Rights over 581 square metres of drain (west of Ings Road)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH (as assumed owner)	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
27-009 cont'd			Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (as assumed owner) Jonathan Darcy Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 OEE (as assumed owner) Robert Anthony Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 OEE (as assumed owner) Uncolnshire PE22 OEE (as assumed owner)		Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (as assumed owner) Jonathan Darcy Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 0EE (as assumed owner) Robert Anthony Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 0EE (as assumed owner) Robert Anthony Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 0EE (as assumed owner) Unknown				

	Land	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
27-010	D	Permanent Rights over 2028 square metres of agricultural land (west of Ings Road)	Jonathan Darcy Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 0EE Robert Anthony Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 0EE	NONE	Bell Brothers Farms Limited The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Jonathan Darcy Bell and Robert Anthony Bell)
27-011	D	Permanent Rights over 6902 square metres of agricultural land and drain (west of Ings Road and east of Ings Drove)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 0RH	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
27-012	K	Temporary Rights over 1197 square metres of agricultural land (west of Ings Road)	Jonathan Darcy Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 0EE	NONE	Bell Brothers Farms Limited The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for Jonathan Darcy Bell and Robert Anthony Bell)

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
27-012 cont'd			Robert Anthony Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 OEE			National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
27-013	К	Temporary Rights over 2480 square metres of agricultural land (east of Ings Drove)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)
27-014	К	Temporary Rights over 8 square metres of access splay (east of Ings Drove)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 0RH (as assumed owner)	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
27-014 cont'd			Unknown		Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
27-015	D	Permanent Rights over 8030 square metres of agricultural land and drain (east of Ings Drove)	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (trading as A. E. Vere & Sons)	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	Unknown (in respect of easements as affect the same by virtue of Section 15(1)(b) of the Endowments and Glebe Measure 1976) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			
27-016	К	Temporary Rights over 52 square metres of drain (east of Ings Drove)	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (as assumed owner)	NONE	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition of use	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
27-016 cont'd			Jonathan Darcy Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 OEE (as assumed owner) Robert Anthony Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 OEE (as assumed owner) Unknown		Jonathan Darcy Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 0EE (as assumed owner) Robert Anthony Bell Bell Brothers Nurseries West End Benington BOSTON Lincolnshire PE22 0EE (as assumed owner) Unknown Staples (Vegetables) Limited Station Farm Station Road Sibsey BOSTON Lincolnshire PE22 0SE	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land	which is proposed to be subject t	BOOK OF BOOK OF I	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights y of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (App Regulations 2009	blications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
27-017	K	Temporary Rights over 25 square metres of agricultural land (east of Ings Drove)	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (trading as A. E. Vere & Sons)	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown (in respect of easements as affect the same by virtue of Section 15(1)(b) of the Endowments and Glebe Measure 1976) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
27-018	K	Temporary Rights over 1406 square metres of agricultural land and drain (east of Ings Drove)	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (trading as A. E. Vere & Sons)	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)

	Land v	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on		Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
27-019	D	Permanent Rights over 6579 square metres of agricultural land and drain (east of Ings Drove)	Frank Derek Vere Stone Cottage Main Road Anwick SLEAFORD NG34 9SJ (trading as A. E. Vere & Sons)	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)
27-020	К	Temporary Rights over 1789 square metres of agricultural land and access track (north of Ings Bank)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 0RH	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to • of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
27-021	D	Permanent Rights over 13605 square metres of agricultural land (north of Ings Bank)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 0RH	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
27-022	К	Temporary Rights over 12 square metres of access track (north of Ings Bank)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH (as assumed owner) Unknown	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 0RH (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
27-023	D	Permanent Rights over 169 square metres of drain (north of Ings Bank)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 0RH (as assumed owner) Unknown	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH (as assumed owner) Unknown	Unknown
27-024	D	Permanent Rights over 1211 square metres of public road (Ings Bank)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 0RH (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown

	Land	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
27-025	К	Temporary Rights over 21 square metres of drain (south of Ings Bank and west of Double Bank)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 0RH (as assumed owner) Unknown	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH (as assumed owner) Unknown	Unknown
27-026	К	Temporary Rights over 773 square metres of agricultural land (south of Ings Bank and west of Double Bank)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	NONE
27-027	D	Permanent Rights over 15624 square metres of agricultural land and drain (south of Ings Bank)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	NONE
27-028	К	Temporary Rights over 50 square metres of access track (west of Double Bank)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	NONE

	Land	which is proposed to be subject to	BOOK OF R c (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to r of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation		
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
27-029	K	Temporary Rights over 1558 square metres of agricultural land and drain (west of Double Bank)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	NONE
27-030	D	Permanent Rights over 13635 square metres of agricultural land (west of Double Bank)	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH	NONE	Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 0RH	NONE
28-001	D	Permanent Rights over 257 square metres of drain (west of Double Bank)	A W Phoenix & Sons Limited Ivy House Farm Toft Tunnel Brothertoft BOSTON Lincolnshire PE20 3SH (as assumed owner) Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 ORH (as assumed owner)	NONE	A W Phoenix & Sons Limited Ivy House Farm Toft Tunnel Brothertoft BOSTON Lincolnshire PE20 3SH (as assumed owner) Charles Crunkhorn Long Hedges Fishtoft BOSTON Lincolnshire PE22 0RH (as assumed owner)	Unknown

	Land v	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-001 cont'd			Unknown		Unknown	
28-002	D	Permanent Rights over 10486 square metres of agricultural land (west of Double Bank)	A W Phoenix & Sons Limited Ivy House Farm Toft Tunnel Brothertoft BOSTON Lincolnshire PE20 3SH	NONE	A W Phoenix & Sons Limited Ivy House Farm Toft Tunnel Brothertoft BOSTON Lincolnshire PE20 3SH	NONE
28-003	D	square metres of agricultural land (west of Double Bank)	A W Phoenix & Sons Limited Ivy House Farm Toft Tunnel Brothertoft BOSTON Lincolnshire PE20 3SH	NONE	A W Phoenix & Sons Limited Ivy House Farm Toft Tunnel Brothertoft BOSTON Lincolnshire PE20 3SH	Unknown (in respect of rights that granted by Conveyance dated 27 April 1979) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
28-004	K	Temporary Rights over 6019 square metres of agricultural land (west of Double Bank)	A W Phoenix & Sons Limited Ivy House Farm Toft Tunnel Brothertoft BOSTON Lincolnshire PE20 3SH	NONE	A W Phoenix & Sons Limited Ivy House Farm Toft Tunnel Brothertoft BOSTON Lincolnshire PE20 3SH	Unknown (in respect of rights that granted by Conveyance dated 27 April 1979) Witham Fourth District Internal Drainage Board 47 Norfolk Street
						BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)
28-005	D	Permanent Rights over 211 square metres of drain (west of Double Bank)	A W Phoenix & Sons Limited Ivy House Farm Toft Tunnel Brothertoft BOSTON Lincolnshire PE20 3SH (as assumed owner)	NONE	A W Phoenix & Sons Limited Ivy House Farm Toft Tunnel Brothertoft BOSTON Lincolnshire PE20 3SH (as assumed owner)	Unknown
			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner)		Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
28-005 cont'd			Unknown		Unknown				
28-006	D	Permanent Rights over 2724 square metres of agricultural land (west of Double Bank)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	Marilyn Wright The Barn Ireland Farm Freiston Ings BOSTON PE22 OPX (trading as J W Maplethorpe) Richard Wright The Barn Ireland Farm Freiston Ings BOSTON PE22 OPX (trading as J W Maplethorpe)	Marilyn Wright The Barn Ireland Farm Freiston Ings BOSTON PE22 OPX (trading as J W Maplethorpe) Richard Wright The Barn Ireland Farm Freiston Ings BOSTON PE22 OPX (trading as J W Maplethorpe)	NONE			
28-007	D	Permanent Rights over 20289 square metres of agricultural land (east of Hobhole Bank)	Joanne McLeod 69 St. Nicholas Road BOSTON PE21 0HH (as assumed owner)	Marilyn Wright The Barn Ireland Farm Freiston Ings BOSTON PE22 0PX	Marilyn Wright The Barn Ireland Farm Freiston Ings BOSTON PE22 0PX	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
28-007 cont'd			Unknown	Richard Wright The Barn Ireland Farm Freiston Ings BOSTON PE22 OPX	Richard Wright The Barn Ireland Farm Freiston Ings BOSTON PE22 OPX Unknown	Unknown			
28-008	K	square metres of agricultural land (north of Lowfields Lane)	Joanne McLeod 69 St. Nicholas Road BOSTON PE21 0HH (as assumed owner) Unknown	Marilyn Wright The Barn Ireland Farm Freiston Ings BOSTON PE22 OPX (trading as J W Maplethorpe) Richard Wright The Barn Ireland Farm Freiston Ings BOSTON PE22 OPX (trading as J W Maplethorpe)	Marilyn Wright The Barn Ireland Farm Freiston Ings BOSTON PE22 OPX (trading as J W Maplethorpe) Richard Wright The Barn Ireland Farm Freiston Ings BOSTON PE22 OPX (trading as J W Maplethorpe) Unknown	Unknown			

	Land	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights ry of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-009	К	Temporary Rights over 497 square metres of agricultural land (north of Lowfields Lane)	Stephen Richard Belton 118 Horncastle Road BOSTON Lincolnshire PE21 9HX (as assumed owner) Unknown	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ Unknown	Unknown
28-010	K	Temporary Rights over 2464 square metres of agricultural land (north of Lowfields Lane)	Stephen Richard Belton 118 Horncastle Road BOSTON Lincolnshire PE21 9HX	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	NONE
28-011	D	Permanent Rights over 14771 square metres of agricultural land (north of Lowfields Lane and east of Hobhole Bank)	Stephen Richard Belton 118 Horncastle Road BOSTON Lincolnshire PE21 9HX	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	NONE
28-012	D	Permanent Rights over 185 square metres of drain (north of Lowfields Lane)	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY (as assumed owner)	NONE	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY (as assumed owner)	Unknown

	Land	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-012 cont'd			Stephen Richard Belton 118 Horncastle Road BOSTON Lincolnshire PE21 9HX (as assumed owner) Unknown		Stephen Richard Belton 118 Horncastle Road BOSTON Lincolnshire PE21 9HX (as assumed owner) T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ Unknown	
28-013	К	Temporary Rights over 1284 square metres of agricultural land (north of Lowfields Lane)	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY	NONE	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY	NONE
28-014	D	Permanent Rights over 7741 square metres of agricultural land and drain (north of Lowfields Lane and east of Hobhole Bank)	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY	NONE	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY	NONE

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to a r of Lincolnshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
28-015	K	Temporary Rights over 274 square metres of agricultural land (north of Lowfields Lane)	Doreen Ann Belton Oak House Farm Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ	NONE Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA	D. & S. Belton Limited Oaknott House Oak House Lane Haltoft Freiston BOSTON PE22 OPJ	NONE
28-016	D	Permanent Rights over 8240 square metres of agricultural land and drain (north of Lowfields Lane)	Doreen Ann Belton Oak House Farm Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ	NONE Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA	D. & S. Belton Limited Oaknott House Oak House Lane Haltoft Freiston BOSTON PE22 OPJ	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
28-017	D	Permanent Rights over 629 square metres of public road and verge (Lowfields Lane)	Doreen Ann Belton Oak House Farm Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
28-018	D	Permanent Rights over 167 square metres of hedgerow and drain (south of Lowfields Lane)	Doreen Ann Belton Oak House Farm Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ (as assumed owner)	Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA	Doreen Ann Belton Oak House Farm Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ (as assumed owner)	Unknown			

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
28-018 cont'd			Unknown		Unknown D. & S. Belton Limited Oaknott House Oak House Lane Haltoft Freiston BOSTON PE22 OPJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)
28-019	D	Permanent Rights over 74304 square metres of agricultural land (Oak House Farm)	Doreen Ann Belton Oak House Farm Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ	Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA	D. & S. Belton Limited Oaknott House Oak House Lane Haltoft Freiston BOSTON PE22 OPJ	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
29-001	К	Temporary Rights over 327 square metres of access track (Oak House Farm)	Doreen Ann Belton Oak House Farm Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ	Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA	D. & S. Belton Limited Oaknott House Oak House Lane Haltoft Freiston BOSTON PE22 OPJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
29-002	К	Temporary Rights over 41 square metres of access splay and drain (west of Swandyke Lane)	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY (as assumed owner) Unknown	NONE	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
29-003	К	Temporary Rights over 676 square metres of agricultural land (Swandyke Farm)	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY	NONE	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
29-004	D	Permanent Rights over 515 square metres of hedgerow and drain (west of Swandyke Lane)	Doreen Ann Belton Oak House Farm Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ (as assumed owner) G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY (as assumed owner) Unknown	Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA	Doreen Ann Belton Oak House Farm Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ (as assumed owner) G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land	which is proposed to be subject to	BOOK OF RI DOCK OF RI BOOK OF RI BOOK OF RI BOOK OF RI	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
29-004 cont'd					D. & S. Belton Limited Oaknott House Oak House Lane Haltoft Freiston BOSTON PE22 0PJ	
29-005	D	Permanent Rights over 54368 square metres of agricultural land, hedgerow and access track (west of Swanhole Lane)	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY	NONE	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a surface water drain) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of electricity apparatus)
29-006	K	Temporary Rights over 1000 square metres of agricultural land (west of Swanhole Lane)	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY	NONE	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY	NONE

	Land	which is proposed to be subject to	BOOK OF RI b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
29-007	G	Temporary Rights over 6699 square metres of agricultural land and hedgerow (north of Wainfleet Road, A52)	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY	NONE	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a surface water drain)	
29-008	D	Permanent Rights over 91 square metres of footways and verge (Wainfleet Road, A52)	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunications apparatus)	
29-009	D	Permanent Rights over 864 square metres of public road, verges and footways (Wainfleet Road, A52)	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)	

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
29-009 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ (assumed in respect of subsoil beneath public highway) Unknown			Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus) Unknown	
29-010	D	Permanent Rights over 512 square metres of hedgerow (south of Wainfleet Road, A52)	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ (as assumed owner) Unknown	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ (as assumed owner) Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
29-011	G	Temporary Rights over 250 square metres of hedgerow (south of Wainfleet Road, A52)	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ (as assumed owner) Unknown	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ (as assumed owner) Unknown	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown			
29-012	G	Temporary Rights over 5604 square metres of agricultural land (south of Wainfleet Road, A52)	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for T H Clements & Son Limited) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on		Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
29-013	D	Permanent Rights over 5370 square metres of agricultural land (south of Wainfleet Road, A52)	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for T H Clements & Son Limited)
30-001	G	Temporary Rights over 1270 square metres of agricultural land (west of Foxhole Lane)	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for T H Clements & Son Limited)
30-002	D	Permanent Rights over 35257 square metres of agricultural land (west of Foxhole Lane)	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for T H Clements & Son Limited)
30-003	K	Temporary Rights over 904 square metres of agricultural land and access track (west of Foxhole Lane)	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for T H Clements & Son Limited)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
30-004	K	Temporary Rights over 28 square metres of access splay (west of Foxhole Lane)	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ (as assumed owner) Unknown	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ (as assumed owner) Unknown	Unknown			
30-005	D	Permanent Rights over 1599 square metres of agricultural land and drain (west of Foxhole Lane)	Christopher Ashley Clements 191 London Road BOSTON Lincolnshire PE21 7HG	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	NONE			
30-006	D	Permanent Rights over 207 square metres of access splay and drain (west of Foxhole Lane)	T H Clements & Son Limited West End Benington BOSTON LincoInshire PE22 0EJ (as assumed owner) Unknown	NONE	T H Clements & Son Limited West End Benington BOSTON LincoInshire PE22 0EJ (as assumed owner) Unknown	Unknown			

	Land	which is proposed to be subject t	BOOK OF RI DOCK OF RI DOCK OF RI BOOK OF RI BOOK OF RI	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	F . 4 . 4 . 6	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
30-007	D	Permanent Rights over 135 square metres of access splay and drain (west of Foxhole Lane)	Christopher Ashley Clements 191 London Road BOSTON Lincolnshire PE21 7HG (as assumed owner) Unknown	NONE	Christopher Ashley Clements 191 London Road BOSTON Lincolnshire PE21 7HG (as assumed owner) Unknown	Unknown
30-008	D	Permanent Rights over 915 square metres of public road and verge (Foxhole Lane)	Christopher Ashley Clements 191 London Road BOSTON Lincolnshire PE21 7HG (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
30-008 cont'd			T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ (assumed in respect of subsoil beneath public highway) Unknown						
30-009	D		T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ		T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for T H Clements & Son Limited) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
30-010	D	Permanent Rights over 4449 square metres of agricultural land (east of Foxhole Lane)	T H Clements & Son Limited West End Benington BOSTON LincoInshire PE22 OEJ	NONE	T H Clements & Son Limited West End Benington BOSTON LincoInshire PE22 OEJ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for T H Clements & Son Limited)				
30-011	К	Temporary Rights over 190 square metres of agricultural land (east of Foxhole Lane)	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for T H Clements & Son Limited)				
30-012	K	Temporary Rights over 170 square metres of agricultural land (east of Foxhole Lane)	Keith Pearson 50 Brand End Road Butterwick BOSTON Lincolnshire PE22 0JD	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	NONE				
30-013	D	Permanent Rights over 10025 square metres of agricultural land (east of Foxhole Lane)	Keith Pearson 50 Brand End Road Butterwick BOSTON Lincolnshire PE22 0JD	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	T H Clements & Son Limited West End Benington BOSTON LincoInshire PE22 OEJ	NONE				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
30-014	D	square metres of agricultural land (east of Foxhole Lane)	David Pearson 7 Daybrook Close Harlaxton GRANTHAM NG32 1AF Roy Pearson Church End Farm Butterwick Road Freiston BOSTON Lincolnshire PE22 0LG	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	NONE				
30-015	D	square metres of agricultural land (north of Butterwick Road)	David Pearson 7 Daybrook Close Harlaxton GRANTHAM NG32 1AF Roy Pearson Church End Farm Butterwick Road Freiston BOSTON Lincolnshire PE22 0LG	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	NONE				

	Land	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
30-016	D	Permanent Rights over 168 square metres of drain (north of Butterwick Road)	David Pearson 7 Daybrook Close Harlaxton GRANTHAM NG32 1AF (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Roy Pearson Church End Farm Butterwick Road Freiston BOSTON Lincolnshire PE22 0LG (as assumed owner) Unknown	NONE	David Pearson 7 Daybrook Close Harlaxton GRANTHAM NG32 1AF (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Roy Pearson Church End Farm Butterwick Road Freiston BOSTON Lincolnshire PE22 0LG (as assumed owner) Unknown	Unknown
30-017	D	Permanent Rights over 7797 square metres of agricultural land and access track (north of Butterwick Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	Christopher Victor Flatters 107 Sea Lane Butterwick BOSTON Lincolnshire PE22 0HN	Christopher Victor Flatters 107 Sea Lane Butterwick BOSTON Lincolnshire PE22 0HN	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
30-018	К	Temporary Rights over 831 square metres of agricultural land and access track (north of Butterwick Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	Christopher Victor Flatters 107 Sea Lane Butterwick BOSTON Lincolnshire PE22 OHN	Christopher Victor Flatters 107 Sea Lane Butterwick BOSTON Lincolnshire PE22 OHN	NONE				
30-019	К	Temporary Rights over 73 square metres of access track and drain (north of Butterwick Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	Unknown				
30-020	D	Permanent Rights over 145 square metres of drain (north at Butterwick Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
30-021	D	Permanent Rights over 724 square metres of public road and verges (Butterwick Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown				
30-022	D	Permanent Rights over 315 square metres of access track and drain (south of Butterwick Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	Unknown				

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
30-023	D	Permanent Rights over 20273 square metres of agricultural land (south of Butterwick Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	David Pearson 7 Daybrook Close Harlaxton GRANTHAM NG32 1AF	David Pearson 7 Daybrook Close Harlaxton GRANTHAM NG32 1AF	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)
30-024	D	Permanent Rights over 25374 square metres of agricultural land (north of Shore Road)	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as trustee of the Fred Grant Settlement Trust) Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as trustee of the Fred Grant Settlement Trust)	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
31-001	D	Permanent Rights over 217 square metres of access splay and drain (north of Shore Road)	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner)	NONE	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
31-001 cont'd			Simon Martin Scriven Williams Croxby Croxby MARKET RASEN LN7 6BW (as assumed owner) Unknown		Simon Martin Scriven Williams Croxby Croxby MARKET RASEN LN7 6BW (as assumed owner) Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
31-002	D	Permanent Rights over 725 square metres of public road, verges and footways (Shore Road)	David James Hamshaw 100 Wide Bargate BOSTON Lincolnshire PE21 6SE (assumed in respect of subsoil beneath public highway) Graham Clive Holdich Smith 5 Eltisley Avenue CAMBRIDGE CB3 9JG (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers				
31-002 cont'd			Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as trustee of the Fred Grant Settlement Trust) (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Mavis Stebbings Jabbeke Wainfleet Road BOSTON Lincolnshire PE21 9RW (assumed in respect of subsoil beneath public highway)			Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
31-002 cont'd			Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as trustee of the Fred Grant Settlement Trust) (assumed in respect of subsoil beneath public highway) Unknown						
31-003	D	Permanent Rights over 12569 square metres of agricultural land (south of Shore Road)	David James Hamshaw 100 Wide Bargate BOSTON Lincolnshire PE21 6SE Graham Clive Holdich Smith 5 Eltisley Avenue CAMBRIDGE CB3 9JG	Nigel Clarence Marshall Mill House Mill Lane Freiston BOSTON Lincolnshire PE22 ONB (trading as FW Marshall & Sons) Barbara Marshall Mill House Mill Lane Freiston BOSTON Lincolnshire PE22 ONB (trading as FW Marshall & Sons)	Nigel Clarence Marshall Mill House Mill Lane Freiston BOSTON Lincolnshire PE22 ONB (trading as FW Marshall & Sons) Barbara Marshall Mill House Mill Lane Freiston BOSTON Lincolnshire PE22 ONB (trading as FW Marshall & Sons)	Julia Kay Johnson West View Shore Road Freiston BOSTON PE22 ONA (as trustee for the Frank Arnold Marshall Will Trust) (in respect of a restriction) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for Mavis Stebbings, Graham Clive Holditch Smith and David James Hamshaw)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	nt of acquisition or use Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
31-003 cont'd			Mavis Stebbings Jabbeke Wainfleet Road BOSTON Lincolnshire PE21 9RW		Mavis Stebbings Jabbeke Wainfleet Road BOSTON Lincolnshire PE21 9RW (trading as FW Marshall & Sons) Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Nigel Clarence Marshall Mill House Mill Lane Freiston BOSTON Lincolnshire PE22 ONB (as trustee for the Frank Arnold Marshall Will Trust) (in respect of a restriction) Unknown (in respect of restrictions relating to dispositions of the proprietors registered estate)			
31-004	D	Permanent Rights over 62198 square metres of agricultural land, hedgerow, access track and drain (east of Church End Road)	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as trustee of the Fred Grant Settlement Trust)	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewerage apparatus)			

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
31-004 cont'd			Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as trustee of the Fred Grant Settlement Trust)			National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
31-005	D	Permanent Rights over 6 square metres of agricultural land (east of Church End Road)	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner and as trustee of the Fred Grant Settlement Trust) Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as assumed owner and as trustee of the Fred Grant Settlement Trust)	NONE	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner and as trustee of the Fred Grant Settlement Trust) Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as assumed owner and as trustee of the Fred Grant Settlement Trust)	Unknown

	Land v	which is proposed to be subject t	BOOK OF RE co: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
31-005 cont'd			Unknown		Unknown	
31-006	D	Permanent Rights over 1065 square metres of public road and verge (Church End Road)	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as trustee of the Fred Grant Settlement Trust) (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as trustee of the Fred Grant Settlement Trust) (assumed in respect of subsoil beneath public highway) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applic) Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
31-007	D	Permanent Rights over 62 square metres of agricultural land (west of Church End Road)	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner and as trustee of the Fred Grant Settlement Trust) Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as assumed owner and as trustee of the Fred Grant Settlement Trust) Unknown	NONE	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner and as trustee of the Fred Grant Settlement Trust) Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as assumed owner and as trustee of the Fred Grant Settlement Trust) Unknown	Unknown			
31-008	D	Permanent Rights over 34988 square metres of agricultural land (west of Church End Road)	SPALDING Lincolnshire PE11 3YR	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water and sewerage apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
Land Plans	Land Plans	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
31-008 cont'd			Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as trustee of the Fred Grant Settlement Trust)							
31-009	D		Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as trustee of the Fred Grant Settlement Trust) (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)		Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
31-009 cont'd			Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as trustee of the Fred Grant Settlement Trust) (assumed in respect of subsoil beneath public highway) Unknown			Unknown			
31-010	D	-	SPALDING Lincolnshire PE11 3YR	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	NONE			

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
32-001	К	Temporary Rights over 89 square metres of agricultural land (west of Church End Road)	SPALDING Lincolnshire PE11 3YR	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water and sewerage apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground and overhead telecommunication apparatus)
32-002	Κ	Temporary Rights over 21 square metres of agricultural land (south of Clampgate Road)	SPALDING Lincolnshire PE11 3YR	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
32-003	D	Permanent Rights over 10676 square metres of agricultural land (east of Hobhole Drain)	Unknown W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF (as assumed owner)	NONE	Unknown W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 OSF (as assumed owner)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of drainage apparatus) Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			
32-004	D	Permanent Rights over 7717 square metres of agricultural land (east of Grovefield Lane)	Steven William Taylor Hathaway House Church Road Freiston BOSTON PE22 ONX	NONE	Steven William Taylor Hathaway House Church Road Freiston BOSTON PE22 0NX	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of drainage apparatus and street furniture)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
32-004 cont'd			The Executor of the Estate of the Late William Thomas Taylor Evadale House Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ Trevor Andrew Taylor Evadale House Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ		The Executor of the Estate of the Late William Thomas Taylor Evadale House Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ Trevor Andrew Taylor Evadale House Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			
32-005	D	Permanent Rights over 354 square metres of verge (Grovefield Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Steven William Taylor Hathaway House Church Road Freiston BOSTON PE22 ONX (trading as W T Taylor & Sons)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers				
32-005 cont'd			The Executor of the Estate of the Late William Thomas Taylor Evadale House Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ Trevor Andrew Taylor Evadale House Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ (trading as W T Taylor & Sons)						
32-006	D	Permanent Rights over 328 square metres of public road and verge (Grovefield Lane)	Barbara Clements Branwell House Barrs Lane Benington BOSTON Lincolnshire PE22 OED (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Tuse Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers				
32-006 cont'd			Christopher William Bradley Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 OSF (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Steven William Taylor Hathaway House Church Road Freiston BOSTON PE22 ONX (assumed in respect of subsoil beneath public highway)			Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
32-006 cont'd			The Executor of the Estate of the Late William Thomas Taylor Evadale House Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ (assumed in respect of subsoil beneath public highway) Trevor Andrew Taylor Evadale House Oak House Lane Freiston BOSTON Lincolnshire PE22 OPJ (assumed in respect of subsoil beneath public highway) Unknown						
32-007	D	Permanent Rights over 7 square metres of agricultural land and verge (Grovefield Lane)	Christopher William Bradley Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 OSF	W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 OSF	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE			

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
32-007 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)		W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	
32-008	D	Permanent Rights over 5401 square metres of agricultural land (west of Grovefield Lane)	Christopher William Bradley Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	Unknown (in respect of rights stated in Conveyance dated 26 March 1981)
						Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
32-009	D	Permanent Rights over 268 square metres of drain (west of Grovefield Lane)	Barbara Clements Branwell House Barrs Lane Benington BOSTON Lincolnshire PE22 0ED (as assumed owner)	NONE	Barbara Clements Branwell House Barrs Lane Benington BOSTON Lincolnshire PE22 0ED (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
32-009 cont'd			Christopher William Bradley Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF (as assumed owner) Unknown		Christopher William Bradley Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF (as assumed owner) Unknown	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			
32-010	D	square metres of drain (west of	Barbara Clements Branwell House Barrs Lane Benington BOSTON Lincolnshire PE22 0ED (as assumed owner) Unknown	NONE	Barbara Clements Branwell House Barrs Lane Benington BOSTON Lincolnshire PE22 OED (as assumed owner) Unknown	Unknown			
32-011	D	square metres of agricultural land and drain (west of	Barbara Clements Branwell House Barrs Lane Benington BOSTON Lincolnshire PE22 0ED	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
32-012	К	Temporary Rights over 646 square metres of agricultural land (west of Grovefield Lane)	Barbara Clements Branwell House Barrs Lane Benington BOSTON Lincolnshire PE22 0ED	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			
32-013	К	Temporary Rights over 11 square metres of access splay (west of Grovefield Lane)	Barbara Clements Branwell House Barrs Lane Benington BOSTON Lincolnshire PE22 0ED (as assumed owner) Unknown	NONE	Barbara Clements Branwell House Barrs Lane Benington BOSTON Lincolnshire PE22 OED (as assumed owner) Unknown	Unknown			
32-014	D	Permanent Rights over 6396 square metres of agricultural land (west of Grovefield Lane)	Barbara Clements Branwell House Barrs Lane Benington BOSTON Lincolnshire PE22 0ED	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	NONE			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
32-015	D	Permanent Rights over 987 square metres of agricultural land and drain (west of Grovefield Lane)	Barbara Clements Branwell House Barrs Lane Benington BOSTON Lincolnshire PE22 0ED	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	NONE			
32-016	D	Permanent Rights over 16035 square metres of agricultural land (west of Grovefield Lane)	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as trustee of the Fred Grant Settlement Trust) Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as trustee of the Fred Grant Settlement Trust)	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	NONE			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
32-017	D	Permanent Rights over 1 square metres of verge (off Grovefield Lane)	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as trustee of the Fred Grant Settlement Trust) (in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as trustee of the Fred Grant Settlement Trust) (in respect of subsoil beneath public highway)		Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE			

	Land	which is proposed to be subject to	BOOK OF RE DOCK OF RE BOOK OF RE	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on		Description of lond	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
32-018	D	Permanent Rights over 814 square metres of agricultural land (west of Grovefield Lane)	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as trustee of the Fred Grant Settlement Trust) Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as trustee of the Fred Grant Settlement Trust)	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	NONE
32-019	D	Permanent Rights over 507 square metres of public road and verge (west of Grovefield Lane)	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as trustee of the Fred Grant Settlement Trust) (in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highway authority)	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acculation or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans	Extent at acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
32-019 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highway authority) Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as trustee of the Fred Grant Settlement Trust) (in respect of subsoil beneath public highway)							
32-020	K	Temporary Rights over 276 square metres of agricultural land (west of Grovefield Lane)	Betsy Mary Ann Chapman Teola Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	NONE				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
32-021	K	Temporary Rights over 292 square metres of agricultural land (west of Grovefield Lane)	Betsy Mary Ann Chapman Teola Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	NONE			
32-022	K	Temporary Rights over 68 square metres of agricultural land (west of Grovefield Lane)	Betsy Mary Ann Chapman Teola Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF (as assumed owner) Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner and as trustee of the Fred Grant Settlement Trust) Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as assumed owner and as trustee of the Fred Grant Settlement Trust)	NONE	Betsy Mary Ann Chapman Teola Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF (as assumed owner) Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner and as trustee of the Fred Grant Settlement Trust) Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as assumed owner and as trustee of the Fred Grant Settlement Trust)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
32-022 cont'd			Unknown		Unknown				
32-023	K	e e e e e e e e e e e e e e e e e e e	Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	NONE			
32-024	К	Temporary Rights over 9 square metres of access splay (west of Grovefield Lane)		NONE	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner and as trustee of the Fred Grant Settlement Trust)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
32-024 cont'd			Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as assumed owner and as trustee of the Fred Grant Settlement Trust) Unknown		Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as assumed owner and as trustee of the Fred Grant Settlement Trust) Unknown				
32-025	D	Permanent Rights over 5636 square metres of agricultural land (west of Grovefield Lane)	Betsy Mary Ann Chapman Teola Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 OSF	W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	NONE			
32-026	D	-	Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as trustee of the Fred Grant Settlement Trust)	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	Freiston Farms Coupledyke Hall Church End Road Freiston BOSTON Lincolnshire PE22 OLL	NONE			

	Land	which is proposed to be subject to	BOOK OF RE DOCK OF RE BOOK OF RE	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
32-026 cont'd			Simon Martin Scriven Williams Croxby Hall Croxby MARKET RASEN LN7 6BW (as trustee of the Fred Grant Settlement Trust)			
33-001	D	square metres of agricultural	Betsy Mary Ann Chapman Teola Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF	NONE
33-002	D	Permanent Rights over 261 square metres of drain (north of Cut End Road)	Betsy Mary Ann Chapman Teola Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner)	NONE	Betsy Mary Ann Chapman Teola Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 OSF (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner)	Unknown

	Land	which is proposed to be subject to	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
33-002 cont'd			Unknown		Unknown	
33-003	D	Permanent Rights over 24317 square metres of agricultural land and access track (east of Cut End Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	John Frederick Mason Maple Farm Armtree Road Langrick BOSTON Lincolnshire PE22 7AQ	John Frederick Mason Maple Farm Armtree Road Langrick BOSTON Lincolnshire PE22 7AQ	Beverley Elizabeth Lister 2 Marsh Farm Cut End Road Fishtoft BOSTON PE22 0QZ (in respect of service media rights and a right to enter as stated in Transfers dated 20 August 1999 and 14 June 2001) Robert John Lister c/o Beverley Elizabeth Lister 2 Marsh Farm Cut End Road Fishtoft BOSTON Lincolnshire PE22 0QZ (in respect of service media rights and a right to enter as stated in Transfers dated 20 August 1999 and 14 June 2001)
33-004	D	Permanent Rights over 4398 square metres of agricultural land (east of Cut End Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	John Frederick Mason Maple Farm Armtree Road Langrick BOSTON Lincolnshire PE22 7AQ	John Frederick Mason Maple Farm Armtree Road Langrick BOSTON Lincolnshire PE22 7AQ	NONE

	Land	which is proposed to be subject to	BOOK OF BOOK OF I	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to y of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-005	D	Permanent Rights over 2361 square metres of agricultural land and access track (east of Cut End Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	John Frederick Mason Maple Farm Armtree Road Langrick BOSTON Lincolnshire PE22 7AQ	John Frederick Mason Maple Farm Armtree Road Langrick BOSTON Lincolnshire PE22 7AQ	Unknown (in respect of a restriction in unknown document dated 19 December 1984)	
33-006	К	Temporary Rights over 1591 square metres of agricultural land and access track (east of Cut End Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	John Frederick Mason Maple Farm Armtree Road Langrick BOSTON Lincolnshire PE22 7AQ	John Frederick Mason Maple Farm Armtree Road Langrick BOSTON Lincolnshire PE22 7AQ	Unknown (in respect of a restriction in unknown document dated 19 December 1984)	
33-007	К	Temporary Rights over 48 square metres of access splay (east of Cut End Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	Unknown	
33-008	D	Permanent Rights over 8781 square metres of agricultural land (east of Cut End Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	John Frederick Mason Maple Farm Armtree Road Langrick BOSTON Lincolnshire PE22 7AQ	John Frederick Mason Maple Farm Armtree Road Langrick BOSTON Lincolnshire PE22 7AQ	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of proposed overhead telecommunication apparatus)	

	Land v	which is proposed to be subject to	BOOK OF RI b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
33-009	D	Permanent Rights over 134 square metres of verge (east of Cut End Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of proposed overhead telecommunication apparatus) Unknown
33-010	D	Permanent Rights over 677 square metres of public road, verge and drain (Cut End Road)	Alec Coney (Farms) Limited Woad Farm Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QP (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus and proposed overhead telecommunication apparatus) Unknown

	Land	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-011	G	Temporary Rights over 4507 square metres of agricultural land (east of Woad Lane and west of Cut End Road)	Alec Coney (Farms) Limited Woad Farm Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QP	NONE	Alec Coney (Farms) Limited Woad Farm Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QP	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
33-012	D	Permanent Rights over 10986 square metres of agricultural land (east of Woad Lane)	Alec Coney (Farms) Limited Woad Farm Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QP	NONE	Alec Coney (Farms) Limited Woad Farm Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QP	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)

	Land v	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-012 cont'd						Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
33-013	D	Permanent Rights over 1394 square metres of agricultural land (east of Woad Lane)	Alec Coney (Farms) Limited Woad Farm Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QP	NONE	Alec Coney (Farms) Limited Woad Farm Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QP	Unknown (in respect of restrictive covenants dated 13 April 2007 stated in Register dated 20 October 2009)	
33-014	D	Permanent Rights over 941 square metres of public road and verge (Woad Lane)	Alec Coney (Farms) Limited Woad Farm Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QP (assumed in respect of subsoil beneath public highway) Janice Norma Pettitt Franklyn Farm White House Lane BOSTON Lincolnshire PE21 0BH (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
33-014 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Richard Nelson Pettitt Franklyn Farm White House Lane BOSTON Lincolnshire PE21 0BH (assumed in respect of subsoil beneath public highway) Unknown						
33-015	D	Permanent Rights over 18164 square metres of agricultural land (west of Woad Lane)	Janice Norma Pettitt Franklyn Farm White House Lane BOSTON Lincolnshire PE21 0BH	NONE	Janice Norma Pettitt Franklyn Farm White House Lane BOSTON Lincolnshire PE21 0BH (trading as F Pettitt & Son	NONE			

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-015 cont'd			Richard Nelson Pettitt Franklyn Farm White House Lane BOSTON Lincolnshire PE21 0BH		Richard Nelson Pettitt Franklyn Farm White House Lane BOSTON Lincolnshire PE21 OBH (trading as F Pettitt & Son		
33-016	D	Permanent Rights over 15353 square metres of watercourse (Hobhole Drain) and copse (west of Woad Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP	NONE	Alan J Moore Marsh Farm Woad Lane Fishtoft BOSTON PE22 ORA	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown (in respect of unknown restrictive covenants which may have been imposed prior to 30 March 2021) Boston & District Angling Association 13 Brand End Road Butterwick BOSTON Lincolnshire PE22 0ET (in respect of fishing rights)	
33-017	Н	Temporary Rights over 82 square metres of drain (west of Pinfold Lane)	Unknown	NONE	Unknown	Unknown	

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-017 cont'd			W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF (as assumed owner)		W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF (as assumed owner)	
33-018	Н	Temporary Rights over 259 square metres of drain (north of Pinfold Lane)	Unknown W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 OSF (as assumed owner)	NONE	Unknown W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF (as assumed owner)	Unknown
33-019	F	Temporary Rights over 708 square metres of agricultural land, access track and verge (south of Pinfold Lane and east of Southfield Lane)	Unknown	NONE	Unknown	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
33-019 cont'd			W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF (as assumed owner) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (as assumed owner)		W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 0SF (as assumed owner) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (as assumed owner)				
33-020	F		John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highway authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highway authority)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
33-020 cont'd			Unknown W Bradley & Sons (Freiston) Limited Fishtoft Grange Gaysfield Road Fishtoft BOSTON Lincolnshire PE21 OSF (assumed in respect of subsoil beneath public highway)						
33-021	F	square metres of agricultural land (south of Pinfold Lane and east of Southfield Lane)	John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner) Unknown	NONE	John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner) Unknown	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-021 cont'd					T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ		
33-022	F	Temporary Rights over 45 square metres of agricultural land (south of Pinfold Lane)	John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	Mark Bolland The White House Pinfold Lane Fishtoft BOSTON Lincolnshire PE21 OSL (in respect of drainage rights granted by a Transfer dated 24 November 2015) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
33-023	F	Temporary Rights over 575 square metres of agricultural land (south of Pinfold Lane and east of Southfield Lane)	John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	NONE	

	Land	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights t / of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-024	F	land (south of Pinfold Lane and east of Southfield Lane)	John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner) Unknown	NONE	John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 OQR (as assumed owner) Unknown T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	Unknown	
33-025	F	land and drain (east of Southfield Lane)	John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 OEJ	NONE	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
33-026	D	Permanent Rights over 557 square metres of agricultural land, drain and copse (east of Southfield Lane)	John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR	NONE	T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	NONE				
33-027	D	Permanent Rights over 239 square metres of drain (east of Southfield Lane)	Ann Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner) John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner) Unknown	NONE	Ann Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner) John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner) Unknown	Unknown				

	Land	which is proposed to be subject to	BOOK OF RE bo: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-027 cont'd			Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (as assumed owner)		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (as assumed owner)		
33-028	D	Permanent Rights over 27664 square metres of agricultural land (east of Southfield Lane)	Ann Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR John Thomas Woods Wombwell House Fishtoft Road Eichtoft	NONE	Ann Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR John Thomas Woods Wombwell House Fishtoft Road Eishtoft	Unknown	
			Fishtoft BOSTON Lincolnshire PE21 0QR		Fishtoft BOSTON Lincolnshire PE21 0QR		
33-029	D	Permanent Rights over 4652 square metres of agricultural land (east of Southfield Lane)	John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner)	NONE	John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner)		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures Regulations 2009			Category 2 Qualifying persons under Regulation				
Land Plans		or acquisition of use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
33-029 cont'd			Unknown		Unknown T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ					
33-030	G	square metres of agricultural land (east of Southfield Lane)	John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner) Unknown		John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner) Unknown T H Clements & Son Limited West End Benington BOSTON Lincolnshire PE22 0EJ	Unknown				

	Land v	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on		Description of load	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-031	G	Temporary Rights over 6074 square metres of agricultural land and drain (east of Southfield Lane)	Ann Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR	NONE	Ann Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR	
33-032	К	Temporary Rights over 2345 square metres of agricultural land, access track and drain (east of Southfield Lane)	Fishtoft Parish Council 24 Granville Avenue Wyberton BOSTON PE21 7BY	Mark William Skipworth Old Orchard Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HX	Mark William Skipworth Old Orchard Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HX	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
33-033	K			NONE	Betty Skipworth Nevadun Burton Croft Road Fishtoft BOSTON Lincolnshire PE21 0SA (as assumed owner) Fishtoft Parish Council 24 Granville Avenue Wyberton BOSTON PE21 7BY (as assumed owner) The Executor of the Estate of the Late William Garfield Skipworth Nevadun Burton Croft Road Fishtoft BOSTON Lincolnshire PE21 0SA (as assumed owner) Unknown	Unknown			

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
33-034	К	Temporary Rights over 12 square metres of agricultural land (east of Southfield Lane)	Betty Skipworth Nevadun Burton Croft Road Fishtoft BOSTON Lincolnshire PE21 0SA The Executor of the Estate of the Late William Garfield Skipworth Nevadun Burton Croft Road Fishtoft BOSTON Lincolnshire PE21 0SA	Mark Skipworth & Son Farming Limited Old Orchard Seadyke Road Old Leake BOSTON PE22 9HX	Mark Skipworth & Son Farming Limited Old Orchard Seadyke Road Old Leake BOSTON PE22 9HX	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of a right of access to maintain and repair drains)
33-035	D	Permanent Rights over 485 square metres of drain (east of Southfield Lane)	Ann Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner) Fishtoft Parish Council 24 Granville Avenue Wyberton BOSTON PE21 7BY (as assumed owner)	NONE	Ann Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner) Fishtoft Parish Council 24 Granville Avenue Wyberton BOSTON PE21 7BY (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
33-035 cont'd			John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner) Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (as assumed owner)		John Thomas Woods Wombwell House Fishtoft Road Fishtoft BOSTON Lincolnshire PE21 0QR (as assumed owner) Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (as assumed owner)				
33-036	D	square metres of agricultural	Fishtoft Parish Council 24 Granville Avenue Wyberton BOSTON PE21 7BY	Mark William Skipworth Old Orchard Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HX	Mark William Skipworth Old Orchard Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HX	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
33-037	D	Permanent Rights over 8666 square metres of agricultural land and hedgerow (east of Southfield Lane)	Betty Skipworth Nevadun Burton Croft Road Fishtoft BOSTON Lincolnshire PE21 0SA The Executor of the Estate of the Late William Garfield Skipworth Nevadun Burton Croft Road Fishtoft BOSTON Lincolnshire PE21 0SA	Mark Skipworth & Son Farming Limited Old Orchard Seadyke Road Old Leake BOSTON PE22 9HX	Mark Skipworth & Son Farming Limited Old Orchard Seadyke Road Old Leake BOSTON PE22 9HX	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of a right of access to maintain and repair drains)
33-038	D		Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA	NONE	Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
33-039	D	Permanent Rights over 9165 square metres of agricultural land and access track (south of Southfield Lane)	Pinchbeck's School, Butterwick Poynton Lodge Shore Road Freiston BOSTON PE22 ONH	R. Hardy (Vegetables) Limited Willoughby House Fishtoft BOSTON Lincolnshire PE21 9SA	R. Hardy (Vegetables) Limited Willoughby House Fishtoft BOSTON Lincolnshire PE21 9SA	Betty Skipworth Nevadun Burton Croft Road Fishtoft BOSTON Lincolnshire PE21 0SA (in respect of covenants and rights reserved by Transfer dated 11 December 1995) Mark William Skipworth Old Orchard Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HX (in respect of covenants and rights reserved by Transfer dated 11 December 1995) Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA (in respect of covenants and rights reserved by Transfer dated 11 December 1995)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
33-039 cont'd						The Executor of the Estate of the Late William Garfield Skipworth Nevadun Burton Croft Road Fishtoft BOSTON Lincolnshire PE21 0SA (in respect of covenants and rights reserved by Transfer dated 11 December 1995) Unknown (in respect of a right of way as stated in Conveyance dated 09 May 1968)			
33-040	D	Permanent Rights over 58974 square metres of agricultural land and access track (east of Scalp Road) and watercourse (The Graft)	Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA	NONE	Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of rights stated in Wayleave Agreement dated 17 September 1956) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)			

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
34-001	К	square metres of access track (east of Scalp Road)	Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA	NONE	Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
34-002	K	square metres of copse (east of Scalp Road)	Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA (as assumed owner) Unknown	NONE	Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA (as assumed owner) Unknown	Unknown
34-003	D	square metres of private road,	Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA (as assumed owner)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Fish 13/11))	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
34-003 cont'd			Unknown		Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA (as assumed owner) Unknown	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of street furniture) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus) Unknown			

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on		Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-004	D	Permanent Rights over 7342 square metres of grassed area, shrubbery (west of Scalp Road) and public footpath (LL Fish 13/11)	Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Fish 13/11)) Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Unknown
34-005	D	, ,	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Unknown (in respect of rights reserved by Deed of Grant dated 25 July 1973)

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
34-006	D	Permanent Rights over 19799 square metres of river (The Haven), foreshore, bed and banks thereof (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Boston Borough Council Municipal Buildings West Street BOSTON Lincolnshire PE21 &QR (in respect of rights granted by Deed of Grant dated 30 October 1936) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Deed of Grant dated 04 September 1964)

	Land v	which is proposed to be subject to	BOOK OF I c: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights y of Lincolnshire	to carry out protective works (Regulation 7(1)(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-007	D	Permanent Rights over 104860 square metres of agricultural land and drain (Bleak House Farm)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of drainage apparatus) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A.E. Lenton Limited)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
34-007 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Conveyance dated 15 April 1966) The Royal Society For The Protection Of Birds The Lodge Potton Road SANDY Bedfordshire SG19 2DL (in respect of rights granted by Deed of Grant dated 14 March 2008)			

	Land	which is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	arry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-007 cont'd						Unknown (in respect of restrictive convenant not to build or erect anything that might damage the jetty work of the Haven of Boston or the ground or foreshore in favour of the Borough of Boston as stated in conveyance dated 6 August 1862) Unknown (in respect of all rights as now exist over the outmarsh and adjoining bank as stated in conveyance dated 1 March 1965)	
34-008	К	Temporary Rights over 1296 square metres of agricultural land, access track, copse and outbuildings (Bleak House Farm)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)	

	Land w	hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) ri	nd Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to c Lincolnshire	arry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-008 cont'd						Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A.E. Lenton Limited) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground and overhead telecommunication apparatus) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Conveyance dated 15 April 1966) Unknown (in respect of restrictive convenant not to build or erect anything that might damage the jetty work of the Haven of Boston or the ground or foreshore in favour of the Borough of Boston as stated in conveyance dated 6 August 1862)	

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
34-008 cont'd						Unknown (in respect of all rights as now exist over the outmarsh and adjoining bank as stated in conveyance dated 1 March 1965)
34-009	G	Temporary Rights over 5554 square metres of agricultural land (Bleak House Farm)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A.E. Lenton Limited) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Conveyance dated 15 April 1966) Unknown (in respect of restrictive convenant not to build or erect anything that might damage the jetty work of the Haven of Boston or the ground or foreshore in favour of the Borough of Boston as stated in conveyance dated 6 August 1862)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
34-009 cont'd						Unknown (in respect of all rights as now exist over the outmarsh and adjoining bank as stated in conveyance dated 1 March 1965)			
34-010	F	Temporary Rights over 6824 square metres of agricultural land and access track (Bleak House Farm)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights granted by Deed of Grant dated 14 March 2008) Lloyds Bank PLC 25 Gresham Street LONDON EC2V 7HN (as mortgagee for A.E. Lenton Limited) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Conveyance dated 15 April 1966)			

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-010 cont'd						The Royal Society For The Protection Of Birds The Lodge Potton Road SANDY Bedfordshire SG19 2DL (in respect of rights granted by Deed of Grant dated 14 March 2008) Unknown (in respect of restrictive convenant not to build or erect anything that might damage the jetty work of the Haven of Boston or the ground or foreshore in favour of the Borough of Boston as stated in conveyance dated 6 August 1862) Unknown (in respect of all rights as now exist over the outmarsh and adjoining bank as stated in conveyance dated 1 March 1965)
34-011	F	Temporary Rights over 600 square metres of access track and copse (east of Wyberton Road)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (in respect of rights to an accessway listed in Deed of Easement dated 18 June 2013)

	Land w	/hich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii)	'ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
34-011 cont'd						Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights stated in Transfer Scheme document dated 1 September 1989) Lincolnshire Wildlife Trust Banovallum House Manor House Street HORNCASTLE Lincolnshire LN9 5HF (in respect of a right to pass and repass stated in Conveyance dated 12 March 1986) Unknown (in respect of a right to pass and repass stated in Conveyance dated 30 July 1884)

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Vind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights • of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (App Regulations 2009	lications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-012	F	Temporary Rights over 133 square metres of access track, verge and copse (east of Wyberton Road) and public footpath (LL Wybe 2/4)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Unknown	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpaths (LL I Wybe I 2/4)) Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown
34-013	F	Temporary Rights over 120 square metres of copse (east of Wyberton Road)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Unknown	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
34-014	D	Permanent Rights over 1259 square metres of agricultural land and copse (east of Wyberton Road)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Unknown	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Unknown	Unknown			
34-015	D	Permanent Rights over 1105 square metres of public road and verges (Wyberton Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (assumed in respect of subsoil beneath public highway) B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown			

	Land v	vhich is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-015 cont'd			E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown				
34-016	D	Permanent Rights over 31 square metres of public road and verge (Wyberton Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	

	Land v	which is proposed to be subject to	BOOK OF RI b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-016 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)			Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)
34-017	К	Temporary Rights over 51 square metres of agricultural land (west of Wyberton Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-017 cont'd						Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)
34-018	D	Permanent Rights over 2116 square metres of agricultural land (west of Wyberton Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
34-018 cont'd						Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)			
34-019	D	Permanent Rights over 147 square metres of drain and copse (west of Wyberton Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (as assumed owner) B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner)	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (as assumed owner) B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans		2.000 pilon on and	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
34-019 cont'd			E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown		E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown				
34-020	D	land and access track (west of Wyberton Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (as assumed owner) B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner)	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (as assumed owner) B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
34-020 cont'd			E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown		E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown				
34-021	G	square metres of agricultural land (west of Wyberton Road)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 7(1)(a) of the Infrastructure Planning (Ap Regulations 2009	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
34-022	G	Temporary Rights over 6870 square metres of agricultural land (west of Wyberton Road)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	Kim Lesley Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Lesley Jean Sermon The Ship Wyberton Roads Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Nicholas Alexander Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 CON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
34-022 cont'd						The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for B Bush & Sons Limited and E M Bush Limited)			
34-023	D	square metres of agricultural land, hedgerows, copse and drains (north of Frampton Road)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	Kim Lesley Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Lesley Jean Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
34-023 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of rights granted by Deed dated 19 August 2004) Nicholas Alexander Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)			

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
34-023 cont'd						The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for B Bush & Sons Limited and E M Bush Limited) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown (in respect of rights stated in Conveyance dated 11 January 1984)
34-024	К	Temporary Rights over 7288 square metres of agricultural land (west of Wyberton Road)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	Kim Lesley Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
34-024 cont'd			E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU		E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	Lesley Jean Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of rights granted by Deed dated 19 August 2004) Nicholas Alexander Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
34-024 cont'd						The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for B Bush & Sons Limited and E M Bush Limited) Unknown (in respect of rights stated in Conveyance dated 11 January 1984)			
34-025	H	Temporary Rights over 26 square metres of copse (north of Wyberton Road)	Colin Brotherton 72 Station Road Kirton BOSTON PE20 1LQ (as assumed owner) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Unknown	NONE	Colin Brotherton 72 Station Road Kirton BOSTON PE20 1LQ (as assumed owner) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
35-001	н	Temporary Rights over 64 square metres of access track (east of Wyberton Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE			
35-002	Н	Temporary Rights over 49 square metres of access track (east of Wyberton Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (as assumed owner) Unknown	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (as assumed owner) Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown			
35-003	Н	Temporary Rights over 39 square metres of agricultural land and copse (north of Wyberton Road)	Colin Brotherton 72 Station Road Kirton BOSTON PE20 1LQ	NONE	Colin Brotherton 72 Station Road Kirton BOSTON PE20 1LQ	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
35-004	K	Temporary Rights over 897 square metres of agricultural land (east of Wiley's Lane)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	Kim Lesley Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Lesley Jean Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Michelle Wright Marsh Farm Cottage Frampton BOSTON Lincolnshire PE20 1AY (in respect of rights granted by Transfer dated 19 February 2003)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
35-004 cont'd									

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
35-004 cont'd						The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)			
35-005	K	Temporary Rights over 7 square metres of access track and drain (south of Wiley's Lane)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
35-006	К	square metres of access track (Wiley's Lane)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (assumed in respect of rights of access) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)			
35-007	К	Temporary Rights over 15 square metres of access track (south of Wiley's Lane)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner)	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
35-007 cont'd			E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown		E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown	Unknown			
35-008	K	Temporary Rights over 1347 square metres of agricultural land and drain (west of Wiley's Lane)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	Kim Lesley Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Lesley Jean Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
35-008 cont'd									

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
35-008 cont'd						The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)			
36-002	G	square metres of agricultural	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT (as assumed owner) Unknown	NONE	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT (as assumed owner) Unknown	Unknown			
36-003	G		Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	NONE	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	National Westminster Bank Public Limited Company 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Pengethley Potatoes Limited)			

	Land	which is proposed to be subject to	BOOK OF Rf c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
36-003 cont'd						W Dennis & Sons Ltd Kirton House Kirton BOSTON PE20 1JD (in respect of rights and restrictive covenants stated in Conveyance dated 13th April 1948)	
36-004	G	Temporary Rights over 289 square metres of agricultural land (north of Millfield Lane East)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	National Westminster Bank Public Limited Company 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Pengethley Potatoes Limited) W Dennis & Sons Ltd Kirton House Kirton BOSTON PE20 1JD (in respect of rights and restrictive covenants stated in Conveyance dated 13th April 1948)	
36-005	Н	Temporary Rights over 34 square metres of agricultural land (west of Low Road)	Lincoln Diocesan Trust And Board Of Finance Limited Edward King House Minster Yard LINCOLN Lincolnshire LN2 1PU	NONE	Lincoln Diocesan Trust And Board Of Finance Limited Edward King House Minster Yard LINCOLN Lincolnshire LN2 1PU	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of restrictions with the Endowments and Glebe Measure 1976 or some other Measure or authority)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
36-006	Н	square metres of drain (west of Low Road)	Lincoln Diocesan Trust And Board Of Finance Limited Edward King House Minster Yard LINCOLN Lincolnshire LN2 1PU (as assumed owner) Unknown	NONE	Lincoln Diocesan Trust And Board Of Finance Limited Edward King House Minster Yard LINCOLN Lincolnshire LN2 1PU (as assumed owner) Unknown	Unknown			
36-007	Η	square metres of access track (south of Millfield Lane East)	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT (as assumed owner) Unknown	NONE	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT (as assumed owner) Unknown	Unknown			
36-008	Η	square metres of access track (south of Millfield Lane East)	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	NONE	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	National Westminster Bank Public Limited Company 250 Bishopsgate LONDON EC2M 4AA (as mortgagee for Pengethley Potatoes Limited)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
36-008 cont'd						W Dennis & Sons Ltd Kirton House Kirton BOSTON PE20 1JD (in respect of rights and restrictive covenants stated in Conveyance dated 13th April 1948)			
37-001	D	square metres of public road (Frampton Road)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (assumed in respect of subsoil beneath public highway) E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
37-001 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown			Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown			
37-002	D	Permanent Rights over 264 square metres of copse (south of Frampton Road)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner)	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
37-002 cont'd			E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown		E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown	Unknown			
37-003	D	Permanent Rights over 10408 square metres of agricultural land and access track (south of Frampton Road)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for B Bush & Sons Limited and E M Bush Limited) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
37-003 cont'd						Unknown (in respect of rights granted by Conveyance dated 11 January 1984)			
37-004	D	Permanent Rights over 196 square metres of drain (south of Frampton Road)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (as assumed owner) Unknown	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (as assumed owner) Unknown	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown			

	Land	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
37-005	K	Temporary Rights over 41 square metres of access splay (south of Frampton Road)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) Unknown	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown
37-006	K	Temporary Rights over 1486 square metres of access track (south of Frampton Road)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON LincoInshire PE22 7NU	The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for B Bush & Sons Limited and E M Bush Limited)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
37-006 cont'd			E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU		E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)				
37-007	K	Temporary Rights over 11 square metres of agricultural land and drain (south of Frampton Road)	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner)	NONE	B Bush & Sons Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner) E M Bush Limited Home Farm Main Road Revesby BOSTON Lincolnshire PE22 7NU (as assumed owner)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
37-007 cont'd			Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (as assumed owner) Unknown		Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (as assumed owner) Unknown				
37-008	K	Temporary Rights over 200 square metres of agricultural land (south of Frampton Road)	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX	NONE	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (trading as J Fowler & Son) Sarah Jane Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (trading as J Fowler & Son)	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Jonathan Gordon Fowler) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Jonathan Gordon Fowler)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
37-008 cont'd						Jacqueline Alison Bowens 1 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018) James Andrew Brock 2 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018)			
37-009	D	square metres of agricultural land, access track and drains (Manor Farm)	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX	NONE	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (trading as J Fowler & Son)	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Jonathan Gordon Fowler)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
37-009 cont'd					PE20 1AX (trading as J Fowler & Son)	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Jonathan Gordon Fowler) Jacqueline Alison Bowens 1 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018) James Andrew Brock 2 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
37-009 cont'd						The Royal Society For The Protection Of Birds The Lodge Potton Road SANDY Bedfordshire SG19 2DL (in respect of rights granted by Conveyance dated 26 September 1986)			
37-010	K	Farm)	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX		Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (trading as J Fowler & Son) Sarah Jane Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (trading as J Fowler & Son)	Barclays Bank PLC 1 Churchill Place LONDON E14 5HP (as mortgagee for Jonathan Gordon Fowler) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Jonathan Gordon Fowler)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
37-010 cont'd						Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and			

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
37-011	D	square metres of drain (east of Sandholme Lane) (excluding all interests of the Crown)	Church End Frampton BOSTON Lincolnshire PE20 1AX (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Unknown
37-012	D	Permanent Rights over 54857 square metres of agricultural land, access track and drain (east of Sandholme Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	F . 4 4	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
37-012 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)				
38-001	К	Temporary Rights over 2663 square metres of agricultural land and access track (east of Sandholme Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)				
38-002	К	Temporary Rights over 37 square metres of access track (north of Sandholme Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning						
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009				
38-003	F	square metres of agricultural land and drain (north of Sandholme Lane) (excluding all	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)				
38-004	F	square metres of verge (north of Sandholme Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Unknown				
38-005	D	square metres of public road (Sandholme Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
38-006	D	square metres of public road (Sandholme Lane) (excluding all interests of the Crown)	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (assumed in respect of subsoil beneath public highway) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
38-007	D		Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Jonathan Gordon Fowler)			
38-008	D	land, copse and drain (north of Marsh Lane)	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX	NONE	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (trading as J Fowler & Son)	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Jonathan Gordon Fowler)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
38-008 cont'd					Sarah Jane Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (trading as J Fowler & Son)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)			
38-009	F	square metres of agricultural land and access track (south of Sandholme Lane)	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX	NONE	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (trading as J Fowler & Son) Sarah Jane Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (trading as J Fowler & Son)	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Jonathan Gordon Fowler) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)			

	Land v	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	ı(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
39-001	D	Permanent Rights over 201 square metres of access track and copse (north of Marsh Lane)	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (as assumed owner) Unknown	NONE	Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (as assumed owner) Unknown	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown
39-002	D	Permanent Rights over 19 square metres of access track and copse (north of Marsh Lane)	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (as assumed owner)	NONE	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (as assumed owner)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
39-002 cont'd			Unknown		Unknown				
39-003	D	Permanent Rights over 208 square metres of public road (Marsh Lane)	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (assumed in respect of subsoil beneath public highway) Jonathan Gordon Fowler Cotton Hall Church End Frampton BOSTON Lincolnshire PE20 1AX (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
39-004	D	Permanent Rights over 80 square metres of access track (east of Marsh Lane) and public footpath (LL Kirt 877/1)	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	NONE	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Kirt 877/1))	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)			
39-005	D	Permanent Rights over 5888 square metres of agricultural land and access track (south of Marsh Lane)	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Unknown	NONE	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Unknown	Unknown			

	Land	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-006	D	Permanent Rights over 31216 square metres of agricultural land, access track and drain (south of Marsh Lane)	Frank William Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU Margaret Ann Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU Paul Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to r of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-007	D	Permanent Rights over 22813 square metres of agricultural land, access track and drain (north of Marsh Road)	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	NONE	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)	
39-008	D	Permanent Rights over 3313 square metres of agricultural land, access track and drain (north of Marsh Road)	Paul Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Unknown	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU Unknown	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown	
39-009	D	Permanent Rights over 12 square metres of drain (north of Marsh Road)	Frank William Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner)	NONE	Frank William Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Land Plans		Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
39-009 cont'd			Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Margaret Ann Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Paul Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Paul Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Unknown		Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Margaret Ann Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Paul Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Lincolnshire PE20 1LU (as assumed owner) Lincolnshire PE20 1LU (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
39-010	K	Temporary Rights over 1752 square metres of access track (north of Marsh Road)	Frank William Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU Margaret Ann Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU Paul Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU Paul Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU		Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)			

	Land	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-011	K	Temporary Rights over 30 square metres of access track (north of Marsh Road)	Frank William Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Margaret Ann Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Margaret Ann Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner)	NONE	Frank William Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Margaret Ann Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Margaret Ann Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner)	The Black Sluice Internal Drainage Board North End Business Park Station Road BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
39-011 cont'd			Paul Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Unknown		Paul Cameron Holmes Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner) Unknown				
39-012	К	Temporary Rights over 1014 square metres of agricultural land (north of Marsh Road)	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	NONE	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)			
39-013	G	Temporary Rights over 21991 square metres of agricultural land (north of Marsh Road)	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	NONE	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	NONE			

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	Description of faild	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-014	D	Permanent Rights over 3168 square metres of agricultural land (north of Marsh Road)	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	NONE	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	NONE	
39-015	D	Permanent Rights over 623 square metres of public road (Marsh Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (assumed in respect of subsoil beneath public highway) Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground and overhead telecommunication apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
39-015 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown			Unknown				
39-016	D	square metres of agricultural land and drains (south of Marsh Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus) Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
39-016 cont'd						The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description on tand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
39-017	K	Temporary Rights over 613 square metres of agricultural land (south of Marsh Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)			

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
39-018	K	Marsh Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (as assumed owner) Parson Latham's Hospital in Oundle 24 North Street Oundle PETERBOROUGH Cambridgeshire PE8 4AL (as assumed owner) Unknown	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (as assumed owner) Parson Latham's Hospital in Oundle 24 North Street Oundle PETERBOROUGH Cambridgeshire PE8 4AL (as assumed owner) Unknown	Unknown
39-019	К	Temporary Rights over 6 square metres of access track (south of Marsh Road)		NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (as assumed owner) Parson Latham's Hospital in Oundle 24 North Street Oundle PETERBOROUGH Cambridgeshire PE8 4AL (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
39-019 cont'd			Unknown		Unknown				
39-021	G	Temporary Rights over 16906 square metres of agricultural land (east of Bucklegate Lane)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground and overhead telecommunication apparatus) Richard Paul Tunnard Sandholme House Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
39-021 cont'd						Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)			
39-022	Η		Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU Unknown	NONE	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown			
39-023	Н	Temporary Rights over 112 square metres of agricultural land (south of Nidd's Lane and west of Marsh Road)	T C Cheer Company Limited Southfields Farm Eleven Acre Lane Kirton BOSTON Lincolnshire PE20 1LS	NONE	T C Cheer Company Limited Southfields Farm Eleven Acre Lane Kirton BOSTON Lincolnshire PE20 1LS	James Edward Cheer Burton House Nidds Lane Kirton BOSTON Lincolnshire PE20 1LZ (in respect of rights reserved as stated in Transfer dated 13 July 2007)			

	Land w	hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) r	nd Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights to f Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Evtent of acquisition or use	Extent of acquisition or use Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-023 cont'd						Joanna Louise Cheer Burton House Nidds Lane Kirton BOSTON Lincolnshire PE20 1LZ (in respect of rights reserved as stated in Transfer dated 13 July 2007) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunications apparatus) The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as the mortgagee of T C Cheer Company Limited dated 26th June 2011)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
40-001	K	Temporary Rights over 1985 square metres of agricultural land (south of Marsh Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus) Richard Paul Tunnard Sandholme House Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
40-001 cont'd						Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)			
40-002	K	square metres of agricultural land (south of Marsh Road and east of Bucklegate Lane)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	NONE	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus) Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
40-002 cont'd						Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)			
40-003	K	square metres of access track	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner)		Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU (as assumed owner)	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND (in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
40-003 cont'd			Howard Stanley Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (as assumed owner) Melanie Clare Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (as assumed owner) Unknown		Howard Stanley Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Kirt 1/5)) Melanie Clare Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (as assumed owner) Unknown	James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND (in respect of rights of access) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown			

	Land v	which is proposed to be subject to	BOOK OF BOOK OF I c (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
40-006	К	and verge (east of Clough Lane) and public footpath (LL Kirt 1/5)	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre- Emption agreement dated 17 December 2018)	
				James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND	James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights to maintain sea defences)	
					Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Kirt 1/5))	Howard Stanley Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in Deed dated 27 May 2005)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
40-006 cont'd									

	Land w	hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) ri	nd Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights to o f Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-006 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Henry Tunnard Limited) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown (in respect of reservations, covenants and conditions as stated in Deed dated 23 June 1870)

	Land v	which is proposed to be subject to	BOOK OF F b: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to y of Lincolnshire	o carry out protective works (Regulation 7(1)(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
40-007	D	Permanent Rights over 46713 square metres of agricultural land, access track, copse and drain (east of Clough Lane) and public footpath (LL Kirt 1/5)	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre- Emption agreement dated 17 December 2018)	
				James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND	James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights to maintain sea defences)	
					Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Kirt 1/5))	Howard Stanley Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in Deed dated 27 May 2005)	

	Land w	hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) ri	nd Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to c f Lincolnshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
40-007 cont'd						James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre- Emption agreement dated 17 December 2018) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights listed in a Transfer dated 11 May 2018) Melanie Clare Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in respect of restrictive covenants stated in Deed dated 27 May 2005)

	Land w	hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) ri	nd Development Consent Order ERENCE - PART 1 ight to use the land, and/or (iii) rights to o f Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-007 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Henry Tunnard Limited) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown (in respect of reservations, covenants and conditions as stated in Deed dated 23 June 1870)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
40-007 cont'd						Unknown			
40-008	K	Temporary Rights over 263 square metres of agricultural land access track and drain (west of Clatterdyke Road)	Henry Tunnard Limited Blossom Hall Skeldyke Road Kirton BOSTON Lincolnshire PE20 1LU	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre- Emption agreement dated 17 December 2018)			
				James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND	James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights to maintain sea defences)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
40-008 cont'd						Howard Stanley Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in Deed dated 27 May 2005) James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre- Emption agreement dated 17 December 2018) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights listed in a Transfer dated 11 May 2018)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
40-008 cont'd						Melanie Clare Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in Deed dated 27 May 2005) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB The Agricultural Mortgage Corporation PLC Keens House Anton Mill Road ANDOVER Hampshire SP10 2NQ (as mortgagee for Henry Tunnard Limited)			

	Land	which is proposed to be subject to	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
40-008 cont'd						The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown (in respect of reservations, covenants and conditions as stated in Deed dated 23 June 1870)	
40-009	К	Temporary Rights over 4982 square metres of access track and copse (east of Low Mill Lane)	Andrew Peter Dennis Woodlands Farm Wash Road Algarkirk BOSTON Lincolnshire PE20 2DN	NONE	Andrew Peter Dennis Woodlands Farm Wash Road Algarkirk BOSTON Lincolnshire PE20 2DN	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of a right of access) Howard Stanley Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of a rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
40-009 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of a right to pass as stated in Conveyance dated 18 March 1953)			
40-010	D	Permanent Rights over 58240 square metres of agricultural land, drains and access track (east of Low Mill Lane)	Andrew Peter Dennis Woodlands Farm Wash Road Algarkirk BOSTON Lincolnshire PE20 2DN	NONE	Andrew Peter Dennis Woodlands Farm Wash Road Algarkirk BOSTON Lincolnshire PE20 2DN	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of a right of access)			

	Land v	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation		
Land Plans		or use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
40-010 cont'd						Howard Stanley Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of a rights of access) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of a right to pass as stated in Conveyance dated 18 March 1953)
41-001	K	Temporary Rights over 904 square metres of agricultural land and access track (east of Low Mill Lane)	Andrew Peter Dennis Woodlands Farm Wash Road Algarkirk BOSTON Lincolnshire PE20 2DN	NONE	Andrew Peter Dennis Woodlands Farm Wash Road Algarkirk BOSTON Lincolnshire PE20 2DN	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of a right of access)

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
41-001 cont'd						Howard Stanley Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of a rights of access) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of a right to pass as stated in Conveyance dated 18 March 1953)
41-002	D	square metres of drain (east of Low Mill Lane)	Andrew Peter Dennis Woodlands Farm Wash Road Algarkirk BOSTON Lincolnshire PE20 2DN (as assumed owner)	NONE	Andrew Peter Dennis Woodlands Farm Wash Road Algarkirk BOSTON Lincolnshire PE20 2DN (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on		Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
41-002 cont'd			Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 0NH (as assumed owner) Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW (as assumed owner) Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PW (as assumed owner)		Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 ONH (as assumed owner) Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW (as assumed owner) Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PW (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
41-002 cont'd			Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH (as assumed owner) Unknown		Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH (as assumed owner) Unknown				
41-003	D	land, access tracks and drains (east of Craven's Lane)	Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 ONH Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW	Oldershaws Of Moulton Limited The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP	Oldershaws Of Moulton Limited The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP	B.S. Pension Fund Trustee Limited 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (in respect of a restriction stated in Transfer dated 25 July 2007) Moulton Bulb Company Limited Long Lane Moulton SPALDING PE12 6PP (in respect of a restriction stated in Definitive Trust Deed dated 26 September 1986)			

	Land v	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans		se Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
41-003 cont'd			Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH			The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown (in respect of rights granted in Conveyance dated 28 January 1985)
42-001	К		Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 ONH (as assumed owner)	NONE	Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 ONH (as assumed owner)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
42-001 cont'd			Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW (as assumed owner) Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP (as assumed owner) Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH (as assumed owner) Unknown		Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW (as assumed owner) Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP (as assumed owner) Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH (as assumed owner) Unknown	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown			

	Land v	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights y of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))
Number on	5	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
42-002	K		Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 ONH Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP	Oldershaws Of Moulton Limited The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Fosd 8/1)) Oldershaws Of Moulton Limited The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) B.S. Pension Fund Trustee Limited 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (in respect of a restriction stated in Transfer dated 25 July 2007) Moulton Bulb Company Limited Long Lane Moulton SPALDING PE12 6PP (in respect of a restriction stated in Definitive Trust Deed dated 26 September 1986)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
42-002 cont'd			Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH			The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown (in respect of rights granted in Conveyance dated 28 January 1985)			
42-003	G	square metres of agricultural land, access track and hardstanding (south of Craven's Lane)	Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 0NH Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW	Oldershaws Of Moulton Limited The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP	Oldershaws Of Moulton Limited The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP	B.S. Pension Fund Trustee Limited 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (in respect of a restriction stated in Transfer dated 25 July 2007) Moulton Bulb Company Limited Long Lane Moulton SPALDING PE12 6PP (in respect of a restriction stated in Definitive Trust Deed dated 26 September 1986)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
42-003 cont'd			Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH			The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights granted in Conveyance dated 28 January 1985)			
42-004	F	square metres of access track, copse, drains and agricultural land (east of Craven's Lane) and public footpath (LL Fosd 8/1)	Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 ONH (as assumed owner)	Oldershaws Of Moulton Limited The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Fosd 8/1))	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	·		Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
42-004 cont'd			Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW (as assumed owner) Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP (as assumed owner) Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH (as assumed owner) Unknown		Oldershaws Of Moulton Limited The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP Unknown				

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
42-005	Н	Temporary Rights over 62 square metres of drain (south of Craven's Lane)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	NONE	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	Unknown
42-006	Н	Temporary Rights over 54 square metres of agricultural land (south of Craven's Lane)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Brian Douglas Naylor and Simon Brian Naylor)

	Land v	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation		
Land Plans	Extent of acquisition of use	Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
42-007	Н	Temporary Rights over 41 square metres of drain (north of Craven's Lane)	Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 ONH Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP	NONE	Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 ONH Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PW	B.S. Pension Fund Trustee Limited 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (in respect of a restriction stated in Transfer dated 25 July 2007) Moulton Bulb Company Limited Long Lane Moulton SPALDING PE12 6PP (in respect of a restriction stated in Definitive Trust Deed dated 26 September 1986) Unknown (in respect of rights granted in Conveyance dated 28 January 1985)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on		Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	and Plans Extent of acquisition or use Description of lar	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
42-007 cont'd			Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 GRH		Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH				
42-008	Н	Temporary Rights over 80 square metres of verge (Craven's Lane)	Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 0NH (as assumed owner) Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW (as assumed owner)	NONE	Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 ONH (as assumed owner) Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW (as assumed owner)	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of hand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
42-008 cont'd			Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP (as assumed owner) Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH (as assumed owner) Unknown		Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP (as assumed owner) Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH (as assumed owner) Unknown				
42-009	н	Temporary Rights over 59 square metres of drain (south of Craven's Lane)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	NONE	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	Unknown			

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans	Land Plans		Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
42-009 cont'd			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown		Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	
42-010	Н	Temporary Rights over 69 square metres of agricultural land (south of Craven's Lane)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Brian Douglas Naylor and Simon Brian Naylor)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	F. 4 4	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 			
42-011	К	square metres of access track (east of Wash Road)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	NONE	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Unknown			
42-012	K	square metres of agricultural land (east of Wash Road)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Brian Douglas Naylor and Simon Brian Naylor)			

	Land	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description on tand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
42-012 cont'd			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG			
42-013	Н	Temporary Rights over 52 square metres of agricultural land (east of Wash Road)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Brian Douglas Naylor and Simon Brian Naylor)
42-014	Н	Temporary Rights over 50 square metres of agricultural land (east of Wash Road)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	NONE	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
42-014 cont'd			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown		Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunications apparatus) Unknown			
42-015	D	square metres of access track, drains and woodland (south of Craven's Lane) and public footpath (LL Fosd 8/1)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 0NH (as assumed owner)	NONE	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 ONH (as assumed owner)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of a secondary flood defence) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans	Land Plans	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
42-015 cont'd			Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW (as assumed owner)		Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights to access and maintain the drain known as Roman Bank Drain)			
			Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP (as assumed owner) Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH (as assumed owner)		Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL I Fosd I 8/1))	Unknown			

	Land	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Land Plans	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
42-015 cont'd			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown		Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	
42-016	D	square metres of agricultural land (south of Craven's Lane)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Brian Douglas Naylor and Simon Brian Naylor)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
42-017	D	square metres of drain (east of Pullover Lane)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown Wendy Bell Bank House Pullover Lane Fosdyke BOSTON Lincolnshire PE20 2DF (as assumed owner)	NONE	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown Wendy Bell Bank House Pullover Lane Fosdyke BOSTON Lincolnshire PE20 2DF (as assumed owner)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunications apparatus) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans		Te of acquisition of use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
42-017 cont'd					Louise Bell Bank House Pullover Lane Fosdyke BOSTON Lincolnshire PE20 2DF Jeremy Bell Bank House Pullover Lane Fosdyke BOSTON Lincolnshire PE20 2DF				
42-018	D	square metres of drain (east of Pullover Lane)	Gerald William Hicks Suffolk House Old Inn Lane Fosdyke BOSTON Lincolnshire PE20 2DE Unknown	NONE	Gerald William Hicks Suffolk House Old Inn Lane Fosdyke BOSTON Lincolnshire PE20 2DE Unknown	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans	nd Plans	Owners or Reputed Owners	Lessees or Tenants	Occupiers					
42-018 cont'd			Wendy Bell Bank House Pullover Lane Fosdyke BOSTON Lincolnshire PE20 2DF		Wendy Bell Bank House Pullover Lane Fosdyke BOSTON Lincolnshire PE20 2DF				
42-019		square metres of private road and verges (Pullover Lane)	Wendy Bell Bank House Pullover Lane Fosdyke BOSTON Lincolnshire PE20 2DF		Wendy Bell Bank House Pullover Lane Fosdyke BOSTON Lincolnshire PE20 2DF Louise Bell Bank House Pullover Lane Fosdyke BOSTON Lincolnshire PE20 2DF	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)			

	Land v	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
42-019 cont'd					Jeremy Bell Bank House Pullover Lane Fosdyke BOSTON Lincolnshire PE20 2DF	
42-020	D	Permanent Rights over 4290 square metres of agricultural land and drains (east of Wash Road)	Gerald William Hicks Suffolk House Old Inn Lane Fosdyke BOSTON Lincolnshire PE20 2DE	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)
42-021	D	Permanent Rights over 1368 square metres of public road and verges (Wash Road)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
42-021 cont'd			Gerald William Hicks Suffolk House Old Inn Lane Fosdyke BOSTON Lincolnshire PE20 2DE (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (assumed in respect of subsoil beneath public highway) Unknown			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and			

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	
42-022	D	Permanent Rights over 478 square metres of drain (west of Wash Road)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	NONE	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Unknown
42-023	D	Permanent Rights over 16866 square metres of agricultural land and drains (north of Wash Road)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Brian Douglas Naylor and Simon Brian Naylor)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
42-023 cont'd			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG						
42-024	D	square metres of drain (north of Wash Road)	Alan Harold Naylor Denva House Fulney Lane South SPALDING Lincolnshire PE12 6FA (as assumed owner) Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner)	NONE	Alan Harold Naylor Denva House Fulney Lane South SPALDING Lincolnshire PE12 6FA (as assumed owner) Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner)	Unknown			

	Land	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii	Vind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to • of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
42-024 cont'd			Unknown		Unknown	
42-025	D	Permanent Rights over 34294 square metres of agricultural land (east of Main Road, A17)	Alan Harold Naylor Denva House Fulney Lane South SPALDING Lincolnshire PE12 6FA Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	NONE
42-026	К	Temporary Rights over 1067 square metres of agricultural land (west of Wash Road)	Alan Harold Naylor Denva House Fulney Lane South SPALDING Lincolnshire PE12 6FA Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	NONE

	Land v	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use	Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
43-001	D	Permanent Rights over 134 square metres of agricultural land (east of Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL Unknown	NONE
43-002	D	Permanent Rights over 1185 square metres of public roads, verges, copse and drains (Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunications apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-003	D	Permanent Rights over 495 square metres of public roads and verges (Surfleet Bank)	John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully and in respect of street furniture) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus)			

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to c of Lincolnshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
43-003 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown
43-004	D	square metres of agricultural land (west of Main Road, A17)	John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN (as assumed owner) Unknown		BOSTON Lincolnshire PE20 2BN (as assumed owner) Unknown	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of drainage apparatus and gully) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus)

	Land	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
43-004 cont'd						Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown
43-005	D	Permanent Rights over 4512 square metres of agricultural land (west of Main Road, A17)	John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN	NONE	John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a gully)
43-006	D	Permanent Rights over 44 square metres of agricultural land (east of Five Towns Drain)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD	NONE	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD	Unknown (in respect of covenants contained in Conveyance dated 04 November 1957)
43-007	D		John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN (as assumed owner)	NONE	John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN (as assumed owner)	Unknown

	Land v	which is proposed to be subject to	BOOK OF RI DOCK OF RI DOC (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on		Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	t of acquisition or use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-007 cont'd			The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (as assumed owner) Unknown		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (as assumed owner) Unknown		
43-008	D	banks (Five Towns Drain)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD	NONE	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD	Unknown (in respect of covenants contained in Conveyance dated 04 November 1957)	
43-009	D	3	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE	

	Land v	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to ^o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faild	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
43-010	D	Permanent Rights over 1652 square metres of public road, verges and footways (Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus)
43-011	D	Permanent Rights over 952 square metres of public road, verges and footways (Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)

	Land w	hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) r	ind Development Consent Order FERENCE - PART 1 ight to use the land, and/or (iii) rights to o f Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
43-011 cont'd			Timothy James Gratton Graves Farm Main Road Fosdyke BOSTON PE20 2DB (assumed in respect of subsoil beneath public highway) Unknown			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of drainage apparatus and street furniture) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus)

	Land	which is proposed to be subject to	BOOK OF I c: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to y of Lincolnshire	o carry out protective works (Regulation 7(1)	ı(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
43-011 cont'd						Unknown
43-012	D	Permanent Rights over 1992 square metres of agricultural land (south of Graves Farm, west of Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	Christopher Moore 1 Graves Farm Main Road Fosdyke BOSTON Lincolnshire PE20 2DB (trading as E Moore & Sons)	Christopher Moore 1 Graves Farm Main Road Fosdyke BOSTON Lincolnshire PE20 2DB (trading as E Moore & Sons)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus)
43-013	D	Permanent Rights over 132 square metres of access track, grassed area and shrubbery (Graves Farm)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	Unknown
43-014	D	Permanent Rights over 2697 square metres of agricultural land, access track and grassed area (west of Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	Christopher Moore 1 Graves Farm Main Road Fosdyke BOSTON Lincolnshire PE20 2DB (trading as E Moore & Sons)	Christopher Moore 1 Graves Farm Main Road Fosdyke BOSTON Lincolnshire PE20 2DB (trading as E Moore & Sons)	Timothy James Gratton Graves Farm Main Road Fosdyke BOSTON PE20 2DB (in respect of a right to service media and right to enter as stated in Transfer dated 27 June 1997)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-015	D	Permanent Rights over 2656 square metres of agricultural land (south of Graves Farm, west of Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	Christopher Moore 1 Graves Farm Main Road Fosdyke BOSTON Lincolnshire PE20 2DB (trading as E Moore & Sons)	Christopher Moore 1 Graves Farm Main Road Fosdyke BOSTON Lincolnshire PE20 2DB (trading as E Moore & Sons)	Unknown (in respect of rights in connection to the Dive Towns Drain contained in an undated document)			
43-016	D	Permanent Rights over 930 square metres of watercourse and banks (Five Towns Drain, west of Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (as assumed owner) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (as assumed owner) Unknown	Unknown			

	Land v	which is proposed to be subject to	BOOK OF RI b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-017	D	Permanent Rights over 288 square metres of watercourse banks (Five Towns Drain)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD	NONE	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD	Unknown (in respect of covenants contained in Conveyance dated 04 November 1957)	
43-018	D	Permanent Rights over 35764 square metres of agricultural land, access track and drain (north of River Welland and west of Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	Christopher Moore 1 Graves Farm Main Road Fosdyke BOSTON Lincolnshire PE20 2DB (trading as E Moore & Sons)	Christopher Moore 1 Graves Farm Main Road Fosdyke BOSTON Lincolnshire PE20 2DB (trading as E Moore & Sons)	NONE	
43-019	D	Permanent Rights over 2301 square metres of copse and drain (west of Main Road, A17) and public bridleway (LL Fosd 2/2)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public bridleway (LL Fosd 2/2)) Unknown	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown	

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
43-020	D	Permanent Rights over 55956 square metres of agricultural land (north of River Welland and west of Moulton Washway, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN (trading as JP & S Ulyatt & Son)	John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN (trading as JP & S Ulyatt & Son)	NONE
43-021	D	Permanent Rights over 239 square metres of agricultural land and access track (south of Smeeton's Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	Unknown
43-022	Н	Temporary Rights over 58 square metres of verge and drain (south of Wash Road and east of Main Road, A17)	Colin Darren Fanthom Mayden Wash Road Fosdyke BOSTON PE20 2DD (as assumed owner) Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ (as assumed owner)	NONE	Colin Darren Fanthom Mayden Wash Road Fosdyke BOSTON PE20 2DD (as assumed owner) Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ (as assumed owner)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
43-022 cont'd			Jean Fanthom Mayden Wash Road Fosdyke BOSTON PE20 2DD (as assumed owner) Unknown		Jean Fanthom Mayden Wash Road Fosdyke BOSTON PE20 2DD (as assumed owner) Unknown				
43-023	F	square metres of grassed area (west of Main Road, A17 and east of Five Towns Drain)	Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ	NONE	Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ	Boston Borough Council Municipal Buildings West Street BOSTON Lincolnshire PE21 8QR (in respect of rights granted by Deed of Grant dated 16th June 1995) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
43-023 cont'd									

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-024	F	Temporary Rights over 1925 square metres of access track and verges (west of Main Road, A17) and public byway (LL Fosd 2/1)	Elaine Hilda Cox Bridge House Main Road Fosdyke BOSTON PE20 2DB (as assumed owner) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ (as assumed owner) James Bingham Sunkyst Main Road Fosdyke BOSTON	NONE	Elaine Hilda Cox Bridge House Main Road Fosdyke BOSTON PE20 2DB (as assumed owner) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ (as assumed owner) James Bingham Sunkyst Main Road Fosdyke BOSTON	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunications apparatus) Unknown			
			Lincolnshire PE20 2DB (as assumed owner)		Lincolnshire PE20 2DB (as assumed owner)				

	Land w	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Seachpiton of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
43-024 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Mark Cox Bridge House Main Road Fosdyke BOSTON PE20 2DB (as assumed owner) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (as assumed owner) Unknown		Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public byway (LL I Fosd I 2/1)) Mark Cox Bridge House Main Road Fosdyke BOSTON PE20 2DB (as assumed owner) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (as assumed owner)	

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
43-024 cont'd					Unknown	
43-025	K	square metres of access track and verges (west of Main Road, A17) and public byway (LL Fosd 2/1)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (as assumed owner) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (as assumed owner) Unknown	Unknown
43-026	G	Temporary Rights over 209 square metres of agricultural land (west of Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-027	G	Temporary Rights over 4818 square metres of agricultural land and access track (west of Five Towns Drain)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	Christopher Moore 1 Graves Farm Main Road Fosdyke BOSTON Lincolnshire PE20 2DB	Christopher Moore 1 Graves Farm Main Road Fosdyke BOSTON Lincolnshire PE20 2DB	Unknown (in respect of covenants contained in Conveyance dated 04 November 1957)			
43-028	G	Temporary Rights over 17163 square metres of agricultural land and access track (west of Five Towns Drain)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	Christopher Moore 1 Graves Farm Main Road Fosdyke BOSTON Lincolnshire PE20 2DB (trading as E Moore & Sons)	Christopher Moore 1 Graves Farm Main Road Fosdyke BOSTON Lincolnshire PE20 2DB (trading as E Moore & Sons)	NONE			
43-029	F	· ·	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights stated in Transfer Scheme document dated 1 September 1989)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-029 cont'd						Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of unknown rights listed in Deed dated 14 November 1989)			
43-030	F	square metres of grassed area (west of Main Road, A17)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus)			
43-031	F	square metres of access track (west of Moulton Washway, A17), public bridleway (LL Fosd 2/2) and public	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights stated in Transfer Scheme document dated 1 September 1989)			

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
43-031 cont'd					Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public bridleway (in respect of public bridleway (LL Fosd 2/2) and public byways (LL Fosd 2/1 and LL Fosd 3/1)))	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of unknown rights listed in Deed dated 14 November 1989) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)
43-032	F	Temporary Rights over 13 square metres of grassed area (north of River Welland and west of Main Road, A17)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
43-033	F	Temporary Rights over 75 square metres of access track (north of River Welland and west of Main Road, A17)	John Peter Ulyatt Mill House Mill Lane Fosdyke BOSTON Lincolnshire PE20 2BL John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN The Executor of the Estate of the Late Stella Ulyatt Mill House Mill Lane Fosdyke BOSTON Lincolnshire PE20 2BL	NONE	John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN	NONE
43-034	F	Temporary Rights over 263 square metres of agricultural land and access track (west of Main Road, A17) and public bridleway (LL Fosd 2/2)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner)	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-034 cont'd			John Peter Ulyatt Mill House Mill Lane Fosdyke BOSTON Lincolnshire PE20 2BL (as assumed owner) John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner)		John Peter Ulyatt Mill House Mill Lane Fosdyke BOSTON Lincolnshire PE20 2BL (as assumed owner) John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public bridleway (LL Fosd 2/2))	Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-034 cont'd			The Executor of the Estate of the Late Stella Ulyatt Mill House Mill Lane Fosdyke BOSTON Lincolnshire PE20 2BL (as assumed owner) Unknown		Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) The Executor of the Estate of the Late Stella Ulyatt Mill House Mill Lane Fosdyke BOSTON Lincolnshire PE20 2BL (as assumed owner) Unknown				
43-035	F	Temporary Rights over 1158 square metres of agricultural land and access track (west of Main Road, A17 and north of River Welland)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN (trading as JP & S Ulyatt & Son)	John William Ulyatt Villa Farm Fosdyke BOSTON LincoInshire PE20 2BN (trading as JP & S Ulyatt & Son)	NONE			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
43-036	К	square metres of access track (west of Moulton Washway, A17), public byways (LL Fosd 3/1) (excluding all	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public byway (LL I Fosd I 3/1)))	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights stated in Transfer Scheme document dated 1 September 1989) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of unknown rights listed in Deed dated 14 November 1989)			
43-038	G	square metres of hardstanding and copse (east of Moulton Washway, A17)	Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ	NONE	Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right of way listed in Transfer dated 31 December 1929)			

	Land	which is proposed to be subject to	BOOK OF RI bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))		
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
43-038 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)		Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)			
43-039	G	Temporary Rights over 2508 square metres of hardstanding (east of Moulton Washway, A17)	Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ	NONE	Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL ((in respect of a right of way listed in Transfer dated 31 December 1929) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus) Unknown (as Kings's Nominee in respect of bona vacantia land for Simpsons (Fosdyke) Limited, dissolved) (in respect of covenants relating to use of land listed in Deed dated 06 November 1996)		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-040	G	Temporary Rights over 42 square metres of access track (east of Main Road, A17)	Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Unknown			
43-041	G	Temporary Rights over 63 square metres of grassed area (east of Main Road, A17)	Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ (as assumed owner) Unknown	NONE	Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ (as assumed owner) Unknown	Unknown			

	Land v	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights y of Lincolnshire	to carry out protective works (Regulation 7(1)(a)) -	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-042	F	Temporary Rights over 94 square metres of access track (east of Main Road, A17) (excluding all interests of the Crown)	Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ	NONE	Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of covenants and rights to discharge water from an adjoining highway listed in a Deed of Exchange dated 06 November 1996) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown (in respect of restrictive covenants stated in unknown document dated 1 January 1926)	
43-043	F	Temporary Rights over 426 square metres of access track (east of Moulton Washway, A17) and public bridleway (LL Fosd 7/1)	Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ	NONE	Fosdyke Yacht Haven Limited Northgate Pinchbeck SPALDING Lincolnshire PE11 3SQ	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right of way listed in Transfer dated 31 December 1929)	

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights t v of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-043 cont'd					Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public bridleway (LL Fosd 7/1))	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground and overhead telecommunication apparatus) Unknown (in respect of a right of way stated in an undated unknown document) Unknown (in respect of manorial rights as stated in Conveyance dated 30 June 1917) Unknown (in respect of manorial rights as stated in conveyance dated 31 December 1929)	
43-044	F	Temporary Rights over 17 square metres of access splay (west of Main Road, A17 and south of River Welland)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-044 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunications apparatus)			
43-045	F	River Welland)		NONE	Alan Mowton Shady Nook 5 Old Main Road Fosdyke BOSTON Lincolnshire PE20 2BU (as assumed owner) George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (as assumed owner)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-045 cont'd			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Robin Twigg 16 Cook Road Gosberton SPALDING Lincolnshire PE11 4PD (as assumed owner) Unknown		Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Robin Twigg 16 Cook Road Gosberton SPALDING Lincolnshire PE11 4PD (as assumed owner) Unknown				
43-046	F	•	Catherine Elizabeth Phyllis Ross Welland House Moulton Washway Fosdyke Bridge SPALDING Lincolnshire PE12 6LH (as assumed owner)	NONE	Catherine Elizabeth Phyllis Ross Welland House Moulton Washway Fosdyke Bridge SPALDING Lincolnshire PE12 6LH (as assumed owner)	Alan Mowton Shady Nook 5 Old Main Road Fosdyke BOSTON Lincolnshire PE20 2BU (in respect of assumed access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-046 cont'd			George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown		George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public bridleway (LL Fosd 6/1)) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner)	Alexander James Hay Wragg Marsh Farm Wragg Marsh Farm Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of assumed access) Andrew George Hay Wragg Marsh Farm Wragg Marsh Farm Wragg Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of assumed access) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 (a) of the Infrastructure Planning (App Regulations 2009	lications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
43-046 cont'd					Unknown	Belinda Elizabeth Needham Sunny View Moulton Washway Fosdyke Bridge SPALDING PE12 6LH (in respect of assumed access) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Hazel Rose Needham Tumbleweed Moulton Marsh SPALDING PE12 6LJ (in respect of assumed access) James Needham Sunny View Moulton Washway Fosdyke Bridge SPALDING PE12 6LH (in respect of assumed access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-046 cont'd						Mary Ellenor Hay Wragg Marsh Farm Wragg Marsh Farm Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of assumed access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Robin Twigg 16 Cook Road Gosberton SPALDING Lincolnshire PE11 4PD (in respect of assumed access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-046 cont'd						Unknown			
43-047	F	square metres of access track (east of Moulton Washway, A17)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public bridleway (LL Fosd 6/1))	Alan Mowton Shady Nook 5 Old Main Road Fosdyke BOSTON Lincolnshire PE20 2BU (in respect of assumed access) Alexander James Hay Wragg Marsh Farm Wragg Marsh Farm Wragg Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of assumed access) Andrew George Hay Wragg Marsh Farm Wragg Marsh Farm Wragg Marsh Farm Wragg Marsh Farm Wragg Marsh Farm Mragg Marsh Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of assumed access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
43-047 cont'd						Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights stated in Transfer Scheme document dated 1 September 1989) Belinda Elizabeth Needham Sunny View Moulton Washway Fosdyke Bridge SPALDING PE12 6LH (in respect of assumed access) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights of entry and covenants listed in Transfer dated 03 March 2021)			

	Land w	hich is proposed to be subject	BOOK OF REFI to: (i) powers of compulsory acquisition, (ii) rig	nd Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to c Lincolnshire	arry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-047 cont'd						Hazel Rose Needham Tumbleweed Moulton Marsh SPALDING PE12 6LJ (in respect of assumed access) James Needham Sunny View Moulton Washway Fosdyke Bridge SPALDING PE12 6LH (in respect of assumed access) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Deed dated 14 November 1989) Mary Ellenor Hay Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of assumed access)	

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
43-047 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Robin Twigg 16 Cook Road Gosberton SPALDING Lincolnshire PE11 4PD (in respect of assumed access)
44-001	D	Permanent Rights over 2038 square metres of agricultural land (south of Smeeton's Lane)	John Peter Ulyatt Mill House Mill Lane Fosdyke BOSTON Lincolnshire PE20 2BL	NONE	John Peter Ulyatt Mill House Mill Lane Fosdyke BOSTON Lincolnshire PE20 2BL	NONE

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
44-001 cont'd			John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN The Executor of the Estate of the Late Stella Ulyatt Mill House Mill Lane Fosdyke BOSTON Lincolnshire PE20 2BL		John William Ulyatt Villa Farm Fosdyke BOSTON Lincolnshire PE20 2BN The Executor of the Estate of the Late Stella Ulyatt Mill House Mill Lane Fosdyke BOSTON Lincolnshire PE20 2BL		
44-002	D	Permanent Rights over 42847 square metres of agricultural land and access track (south of Smeeton's Lane)	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	NONE	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for R. Bratley (Quadring) Limited) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (in respect of rights reserved by Transfer dated 26 March 2010)	

	Land v	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
44-003	D	Permanent Rights over 25 square metres of banks (south of Smeeton's Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	
44-004	D	Permanent Rights over 141 square metres of watercourse (Risegate Eau) and banks thereof	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD	NONE	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD	NONE	
44-005	D	Permanent Rights over 2282 square metres of watercourse (Risegate Eau) and banks thereof	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner)	NONE	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner)	Unknown	

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
44-005 cont'd			John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (as assumed owner) Unknown		John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (as assumed owner) Unknown	
44-006	D	Permanent Rights over 17226 square metres of 17226 agricultural land, access track and drain (south of Sea Bank)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	NONE

	Land w	hich is proposed to be subject	BOOK OF R to: (i) powers of compulsory acquisition, (ii)	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to r of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Evtopt of acquirition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Land Plans Extent of acquisition or use Des	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
44-006 cont'd			John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB				
44-007			The Executor of the Estate of the Late Geoffrey Elijah Jessop Clayton Sunrise Mill Lane Fosdyke BOSTON Lincolnshire PE20 2BL (as registered owner) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner)	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (in respect of a right of way as contained in an unknown undated document) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (in respect of a right of way as contained in an unknown undated document)	

	Land	which is proposed to be subject to	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
44-008	D	Permanent Rights over 80 square metres of agricultural land (south of Sea Bank)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) Unknown	NONE	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) Unknown	Unknown
44-009	D	Permanent Rights over 149 square metres of agricultural land and drain (south of Smeeton's Lane)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner)	NONE	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner)	Unknown

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 I right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
44-009 cont'd			Unknown		Unknown	
44-010	D	Permanent Rights over 1473 square metres of agricultural land (south of Smeeton's Lane)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	NONE
44-011	D	Permanent Rights over 8 square metres of drain (south of Sea Bank)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner)	NONE	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner)	Unknown

	Land	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
44-011 cont'd			Unknown		Unknown	
44-012	D	Permanent Rights over 10368 square metres of agricultural land, access track and drain (south of Sea Bank)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	NONE
44-013	D	Permanent Rights over 196 square metres of drain (south of Sea Bank)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	NONE	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
44-013 cont'd			John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) Unknown		John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) Unknown				
44-014	D	Permanent Rights over 32339 square metres of agricultural land (south of Marsh Lane)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	A.H. Worth (Fosdyke) Limited Fleet Estate Office Manor Farm Holbeach Hurn Holbeach SPALDING Lincolnshire PE12 8LR	NONE			
44-015	K	Temporary Rights over 85 square metres of access track (west of Smeeton's Lane)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	NONE			

	Land w	hich is proposed to be subject	BOOK OF I to: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Evtopt of acquirition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Land Plans Extent of acquisition or use Descri	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
44-015 cont'd			John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB				
44-016			NONE	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (in respect of a right of way as contained in an unknown undated document) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (in respect of a right of way as contained in an unknown undated document)	

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
44-017	K	Temporary Rights over 1377 square metres of access track (west of Smeeton's Lane)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	NONE
44-018	К	Temporary Rights over 4740 square metres of agricultural land and access track (west of Smeeton's Lane)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	NONE

	Land v	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights y of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	F	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
44-019	К	Temporary Rights over 102 square metres of banks (south of Risegate Eau)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) Unknown	NONE	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (as assumed owner) Unknown	Unknown

	Land	which is proposed to be subject to	BOOK OF R b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
44-020	К	Temporary Rights over 1463 square metres of agricultural land (south of Sea Bank)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	A.H. Worth (Fosdyke) Limited Fleet Estate Office Manor Farm Holbeach Hurn Holbeach SPALDING Lincolnshire PE12 8LR	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971)
44-021	К	Temporary Rights over 5067 square metres of access track (south of Smeeton's Lane) and public byways (LL Alga 9/1 and LL Fosd 3/1)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public byways (LL Alga 9/1 and LL Fosd 3/1))	NONE
44-022	К	Temporary Rights over 508 square metres of access track (south of Smeeton's Lane) and public byway (LL Alga 9/1)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public byway (LL Alga 9/1))	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
44-022 cont'd					The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD		
44-023	K	Temporary Rights over 574 square metres of agricultural land, access track (south of Smeeton's Lane) and public byway (LL Alga 9/1)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public byway (LL Alga 9/1))	NONE	
44-024	K	Temporary Rights over 20 square metres of grassed area (south of Smeeton's Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner)	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner)	Unknown	

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
44-024 cont'd			The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (as assumed owner) Unknown		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (as assumed owner) Unknown	
44-025	К	Temporary Rights over 128 square metres of agricultural land (south of Smeeton's Lane)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	NONE	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	NONE

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to ¹ of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
44-026	К	Temporary Rights over 1492 square metres of access track (south of Sea Bank)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	R. Bratley (Quadring) Limited Quadring Eaudyke SPALDING Lincolnshire PE11 4QB	NONE
44-027	F	Temporary Rights over 61 square metres of access track and verge (south of River Welland)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	Alexander James Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access)

	Land w	hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) ri	nd Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to a Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
44-027 cont'd						Andrew George Hay Wragg Marsh Farm Wragg Marsh Farm Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
44-027 cont'd						Mary Ellenor Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground and overhead electricity apparatus)			
44-028	F	Temporary Rights over 17 square metres of access track and verge (south of River Welland)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner)	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner)	Alexander James Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
44-028 cont'd			Unknown			Andrew George Hay Wragg Marsh Farm Wragg Marsh Farm Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (in respect of rights of access)			

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
44-028 cont'd						Mary Ellenor Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access) Unknown	
44-029	F	square metres of private accessway and scrubland (south of River Welland)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	Alexander James Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access) Andrew George Hay Wragg Marsh Farm Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access)	

	Land w	hich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) ri	nd Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
44-029 cont'd						Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (in respect of rights of access) Mary Ellenor Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ Lincolnshire	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
44-029 cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus)			
45-001	Freehold Acquisition	metres of agricultural land and access track (south of Sea Bank and west of Marsh Lane)	-	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)			
45-002	J, L	-	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
45-002 cont'd						Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-003	Υ	Permanent Rights over 3692 square metres of agricultural land, access track and drain (east of A16 and south of Sea Bank) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of gas apparatus) Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-004	J, L	Permanent Rights over 28 square metres of access splay (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of gas apparatus) Unknown			
45-005	Freehold Acquisition	Freehold over 4235 square metres of agricultural land and access track (west of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-005 cont'd						National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of gas apparatus) Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)			
45-006	J	square metres of agricultural land, access track and drain	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972)			

	Land v	which is proposed to be subject to	BOOK OF RI b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-006 cont'd						National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of gas apparatus) Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
45-007	J, L	Permanent Rights over 3 square metres of verge (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE Unknown	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	

	Land	which is proposed to be subject to	BOOK OF F b: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
45-008	Freehold Acquisition	Freehold over 1647 square metres of agricultural land (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)
45-009	J	Permanent Rights over 1464 square metres of agricultural land, access track and drain (east of A16 and south of Sea Bank) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)

	Land v	which is proposed to be subject to	BOOK OF RE b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-009 cont'd						The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-010	J	Permanent Rights over 94 square metres of watercourse (Risegate Eau) and access track (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
45-011	J, L	° .	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-012	J	Permanent Rights over 71 square metres of watercourse (Risegate Eau) and access track (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown			
45-013	J	Permanent Rights over 657 square metres of agricultural land, access track and drain (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-014	J, L	Permanent Rights over 2483 square metres of agricultural land, access track and drain (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE A.H. Worth (Fosdyke) Limited Fleet Estate Office Manor Farm Holbeach Hurn Holbeach SPALDING Lincolnshire PE12 8LR	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)			
45-015	Freehold Acquisition	Freehold over 2454 square metres of agricultural land (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
45-015 cont'd						The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			
45-016	J	square metres of agricultural land, access track and drain (south of Marsh Lane)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-017	J	Permanent Rights over 15 square metres of drain (south of Marsh Lane) (excluding all interests of the Crown)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown			
45-018	J	Permanent Rights over 72 square metres of drain (south of Marsh Lane) (excluding all interests of the Crown)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation						
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
45-018 cont'd			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown		The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Unknown				
45-019	J	square metres of agricultural land, drain and access track	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)				
45-020	J, L	square metres of agricultural land, drain and access track	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	NONE				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
45-021	Freehold Acquisition	Freehold over 3667 square metres of agricultural land (south of Marsh Lane)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			
45-022	J	Permanent Rights over 25 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE			
45-023	K	Temporary Rights over 1 square metres of drain (south of Marsh Lane)		NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-024	К	Temporary Rights over 106 square metres of access track (south of Marsh Lane)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	A.H. Worth (Fosdyke) Limited Fleet Estate Office Manor Farm Holbeach Hurn Holbeach SPALDING Lincolnshire PE12 8LR	NONE			
45-025	D	Permanent Rights over 6163 square metres of agricultural land (south of Marsh Lane)	Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY	Matthew James William Naylor The Old Pack House Sycamore Farm Common Road Moulton Seas End SPALDING Lincolnshire PE12 6LF (trading as Naylor Flowers Limited)	Matthew James William Naylor The Old Pack House Sycamore Farm Common Road Moulton Seas End SPALDING Lincolnshire PE12 6LF (trading as Naylor Flowers Limited)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			
45-026	D	Permanent Rights over 816 square metres of drain (south of Marsh Lane)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(:	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
45-026 cont'd			Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY (as assumed owner) Unknown		Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Unknown
45-027	D	Permanent Rights over 268 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	NONE
45-028	D	Permanent Rights over 14356 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-028 cont'd						National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of underground gas apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			
45-029	D	Permanent Rights over 718 square metres of drain (south of Marsh Lane) (excluding all interests of the Crown)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of underground gas apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-029 cont'd			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown		The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown			
45-030	D	Permanent Rights over 27105 square metres of agricultural land (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of gas pipeline apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans		Sition of use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-030 cont'd						Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)			
45-031	D	square metres of drain (east of A16) (excluding all interests of the Crown)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown		John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Unknown			

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to • of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
45-032	K	Permanent Rights over 43864 square metres of agricultural land, drain and access track (east of The Reservoir, A16) Temporary Rights over 3472 square metres of agricultural land (south of Risegate Eau) (excluding all interests of the Crown)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	NONE Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	A.H. Worth (Fosdyke) Limited Fleet Estate Office Manor Farm Holbeach Hurn Holbeach SPALDING Lincolnshire PE12 8LR Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	NONE Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as
45-034	J	Permanent Rights over 1036 square metres of drain (east of A16) (excluding all interests of the Crown)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	stated in a Lease dated 27 January 2022) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
45-034 cont'd			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown		The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown				
45-035	J, L	Permanent Rights over 1809 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE			
45-036	J, L	Permanent Rights over 1125 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	NONE			
45-037	J	Permanent Rights over 5999 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE			

	Land	which is proposed to be subject to	BOOK OF RI b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
45-038	J	Permanent Rights over 1272 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	NONE
45-039	К	Temporary Rights over 632 square metres of agricultural land and access track (east of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE
45-040	G	Temporary Rights over 17840 square metres of agricultural land and access track (east of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE
45-041	Freehold Acquisition	Freehold over 259554 square metres of agricultural land, access track and drain (east of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted in Transfer dated 29 April 1998)

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
45-041 cont'd						The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)
45-042	J	Permanent Rights over 7351 square metres of drain and banks (east of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE
45-043	J		John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown

	Land	which is proposed to be subject to	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
45-043 cont'd			Unknown		Unknown	
45-044	J	square metres of watercourse (Risegate Eau) (excluding all interests of the Crown)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown
45-045	J	Permanent Rights over 493 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-046	J	Permanent Rights over 334 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			
45-046a	E	Permanent Rights over 8 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and covenants granted by Deed dated 31 December 1971) Joanne Lesley Davis Welland Farm House Surfleet Marsh Surfleet SPALDING Lincolnshire PE11 4DW (in respect of rights granted by Transfer dated 07 November 2013)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Futort of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-046a cont'd						Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of drainage apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted in Transfer dated 29 April 1998) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of underground gas apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition of use		Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-046a cont'd						National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of rights granted by Deed dated 21 February 2007) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of rights granted by Deed dated 21 February 2007) Patricia Mary Marguerite Bealby 24 Catherine Place Pine Gardens MELTON MOWBRAY LE13 1FH (in respect of rights granted by Deed dated 22 March 2000)			

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
45-046a cont'd						Russell Lee Davis Welland Farm House Surfleet Marsh Surfleet SPALDING Lincolnshire PE11 4DW (in respect of rights granted by Transfer dated 07 November 2013) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)
45-047	J	Permanent Rights over 417 square metres of drain (east of A16) (excluding all interests of the Crown)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-047 cont'd			Unknown		Unknown				
45-048	J, L	Permanent Rights over 528 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	NONE			
45-049	J	Permanent Rights over 6604 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	A.H. Worth (Fosdyke) Limited Fleet Estate Office Manor Farm Holbeach Hurn Holbeach SPALDING Lincolnshire PE12 8LR Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (in respect of riparian rights)			

	Land	which is proposed to be subject to	BOOK OF R b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
45-050	Freehold Acquisition	Freehold over 11140 square metres of agricultural land (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	A.H. Worth (Fosdyke) Limited Fleet Estate Office Manor Farm Holbeach Hurn Holbeach SPALDING Lincolnshire PE12 8LR	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)
45-051	J, L	Permanent Rights over 5486 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)
45-052	J	Permanent Rights over 2371 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)
45-053	К	Temporary Rights over 2566 square metres of agricultural land and access track (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009						
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
45-054	Ε	Permanent Rights over 22177 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	A.H. Worth (Fosdyke) Limited Fleet Estate Office Manor Farm Holbeach Hurn Holbeach SPALDING Lincolnshire PE12 8LR	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)				
45-055	J	Permanent Rights over 302 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)				
45-056	J	Permanent Rights over 1695 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-057	J, L	Permanent Rights over 2434 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)			
45-058	E	Permanent Rights over 17760 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			
45-059	J	Permanent Rights over 494 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE			
45-060	J, L	Permanent Rights over 444 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE			

	Land	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on		Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
45-061	J	Permanent Rights over 227 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE
45-062	E	Permanent Rights over 191 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE
45-063	J	Permanent Rights over 1048 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE
45-064	J	Permanent Rights over 2079 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-065	J, L	square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of underground gas apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use		Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
45-066	К	square metres of access track and verge (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			
45-067	J	square metres of agricultural land, drain and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of underground gas apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans	and Plans Extent of acquisition or use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers					
45-067 cont'd						The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			
45-068	Freehold Acquisition	metres of agricultural land (east of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	PE12 6LG	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of underground gas apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans	Land Plans Extent of acquisition or use Description of land	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
45-068 cont'd						The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			
45-069	J	square metres of drain (east of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of underground gas apparatus)			

	Land	which is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
45-069 cont'd						The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)
45-070	K	Temporary Rights over 101 square metres of access track and verge (east of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE
45-071	К	Temporary Rights over 19 square metres of agricultural land (north of River Welland)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)
			Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY (as assumed owner)		Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY (as assumed owner)	Unknown

	Land v	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
45-071 cont'd			Unknown		Unknown	
45-072	K	Temporary Rights over 2315 square metres of agricultural land (north of River Welland)	Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY	Matthew James William Naylor The Old Pack House Sycamore Farm Common Road Moulton Seas End SPALDING Lincolnshire PE12 6LF (trading as Naylor Flowers Limited)	Matthew James William Naylor The Old Pack House Sycamore Farm Common Road Moulton Seas End SPALDING Lincolnshire PE12 6LF (trading as Naylor Flowers Limited)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)
45-073	J	Permanent Rights over 946 square metres of drain (south of Marsh Lane)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	NONE	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown
46-001	J, L	Permanent Rights over 22 square metres of access track (east of A16) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	NONE	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Unknown

	Land	which is proposed to be subject to	BOOK OF F b: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
46-002	Freehold Acquisition	metres of agricultural land (east	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE Unknown	Unknown
46-003	Freehold Acquisition	Freehold over 480 square metres of agricultural land (east of A16) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)
46-004	J, L	Permanent Rights over 1232 square metres of agricultural land (east of A16 and north of Bicker Creek Drain) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire									
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
46-005	J	Permanent Rights over 1680 square metres of agricultural land (east of A16 and north of Bicker Creek Drain) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)				
46-006	Freehold Acquisition	Freehold over 4031 square metres of agricultural land, access track and drain (east of A16) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)				
46-007	J	Permanent Rights over 85 square metres of drain (Bicker Creek) (excluding all interests of the Crown)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	Unknown				

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
46-007 cont'd			The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown		The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner) Unknown				
46-008	Freehold Acquisition	metres of agricultural land and drain (east of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE			
46-009	J, L	square metres of agricultural land and access track (east of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			

	Land v	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to v of Lincolnshire	carry out protective works (Regulation 7(1)	ı(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
46-010	J	Permanent Rights over 2522 square metres of agricultural land and access track (east of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)
46-011	J, L	Permanent Rights over 15 square metres of access track (east of A16) and banks (Risegate Eau)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (in respect of assumed rights of access) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights and obligations listed in a Conveyance dated 30 November 1988)

	Land	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights ty of Lincolnshire	to carry out protective works (Regulation 7(1)(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
46-012	J	Permanent Rights over 37 square metres of access track (east of A16) and banks (Risegate Eau)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (in respect of assumed rights of access) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights and obligations listed in a Conveyance dated 30 November 1988)	
46-013	J	Permanent Rights over 200 square metres of banks (Risegate Eau)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted in Transfer dated 22 May 2001)	

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
46-013 cont'd			Unknown		Unknown	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (assumed in respect of rights of access) Unknown
46-014	J, L	Permanent Rights over 2760 square metres of agricultural land and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE
46-015	1	Permanent Rights over 9166 square metres of agricultural land and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE
46-016	E	Permanent Rights over 26354 square metres of agricultural land and access track (east of The Reservoir, A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
46-017	Freehold Acquisition	metres of agricultural land (east of The Reservoir, A16)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Brian Douglas Naylor and Simon Brian Naylor) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000)			
46-018	J, L	track (east of The Reservoir, A16) (excluding all interests of the Crown)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Brian Douglas Naylor and Simon Brian Naylor) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of underground cable apparatus)			

	Land	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
46-018 cont'd						Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)
46-019	J	Permanent Rights over 296 square metres of agricultural land, hedgerow and access track (east of The Reservoir, A16) (excluding all interests of the Crown)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Brian Douglas Naylor and Simon Brian Naylor)

	Land	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
46-019 cont'd			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG		Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of underground cable apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus)	
46-020	J	square metres of agricultural land, hedgerow and access track (east of The Reservoir,	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Brian Douglas Naylor and Simon Brian Naylor)	

	Land w	hich is proposed to be subject	BOOK OF RE to: (i) powers of compulsory acquisition, (ii) r	ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on Land Plans	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Lanu Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
46-020 cont'd			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of underground cable apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
46-020 cont'd						The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			
46-021	Freehold Acquisition	metres of agricultural land and access track (east of A16)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Brian Douglas Naylor and Simon Brian Naylor) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of underground cable apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000)			

	Land	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights y of Lincolnshire	to carry out protective works (Regulation 7(1)(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Ap Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
46-022	L	Permanent Rights over 190 square metres of agricultural land, hedgerow and access track (east of The Reservoir, A16) (excluding all interests of the Crown)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Brian Douglas Naylor and Simon Brian Naylor) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of underground cable apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000)
46-023	E	Permanent Rights over 455 square metres of drain (north of Marsh Drove)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	NONE	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	Unknown

	Land	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
46-023 cont'd			Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown		Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	
46-024	E	Permanent Rights over 453 square metres of drain (north of Marsh Drove)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	NONE	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	Unknown

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	F. 4	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition or use Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
46-024 cont'd			Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown		Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	
46-025	E	Permanent Rights over 10257 square metres of agricultural land (north of Marsh Drove)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor)

	Land	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
46-025 cont'd			Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right to enter as stated in Transfer dated 25 March 2003) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain) Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)
46-026	K	Temporary Rights over 1404 square metres of access track (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	A.H. Worth (Fosdyke) Limited Fleet Estate Office Manor Farm Holbeach Hurn Holbeach SPALDING Lincolnshire PE12 8LR	NONE

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Vind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to r of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
46-027	J	Permanent Rights over 22 square metres of agricultural land and access track (west of A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights and obligations listed in a Conveyance dated 30 November 1988)
46-028	J, L	Permanent Rights over 24 square metres of agricultural land and access track (west of A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (in respect of assumed rights of access) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)

	Land v	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
46-028 cont'd						Unknown (in respect of rights and obligations listed in a Conveyance dated 30 November 1988)
46-029]	Permanent Rights over 105 square metres of agricultural land, access track and banks (west of A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (in respect of assumed rights of access) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights and obligations listed in a Conveyance dated 30 November 1988)

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	arry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
46-030	J	Permanent Rights over 4 square metres of access track (west of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner)	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted in Transfer dated 22 May 2001) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
46-031	J	Permanent Rights over 5236 square metres of agricultural land (west of A16) and banks (Risegate Eau)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation				
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
46-031 cont'd						The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			
46-032	J, L	square metres of agricultural land (west of A16) and banks (Risegate Eau)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE			
46-033	Freehold Acquisition	metres of watercourse (Risegate Eau)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			

	Land	which is proposed to be subject to	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
46-034	J	Permanent Rights over 6284 square metres of drain and banks (west of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner) Unknown	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted in Transfer dated 22 May 2001) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	

	Land	which is proposed to be subject	BOOK OF to: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights y of Lincolnshire	to carry out protective works (Regulation 7(1))(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans		Description of Iana	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
46-035	J	Permanent Rights over 83 square metres of drain and banks (west of A16)	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP (as assumed owner) Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL (as assumed owner)	NONE	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP (as assumed owner) Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights t of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
46-036	J	Permanent Rights over 1029 square metres of drain and copse (east of Gosberton Bank)	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP (as assumed owner) Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner) Judith East The Gables Gosberton Bank Gosberton SPALDING PE11 4PB (as assumed owner) Unknown William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL (as assumed owner)	NONE	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP (as assumed owner) Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner) Judith East The Gables Gosberton Bank Gosberton SPALDING PE11 4PB (as assumed owner) Unknown William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL (as assumed owner)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
46-037	Freehold Acquisition	Freehold over 2248 square metres of agricultural land and copse (east of Gosberton Bank)		Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	NONE			
46-038	J	square metres of agricultural	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
46-038 cont'd			Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL						
46-039	J, L	-	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 .)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
46-039 cont'd			William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL						
46-040	J	Permanent Rights over 1696 square metres of agricultural land and access track (east of Gosberton Road)	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Applic Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
46-041	L	Permanent Rights over 172 square metres of agricultural land and access track (east of Gosberton Road)	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)			
46-042	J	Permanent Rights over 31 square metres of agricultural land and access track (east of Gosberton Road)	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
46-042 cont'd			Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL						
46-043	J	square metres of drain and copse (east of Gosberton Bank)	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP (as assumed owner) Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner)	NONE	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP (as assumed owner) Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR (as assumed owner)	Unknown			

	Land	which is proposed to be subject	BOOK OF R to: (i) powers of compulsory acquisition, (ii)	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of Iana	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
46-043 cont'd			Judith East The Gables Gosberton Bank Gosberton SPALDING PE11 4PB (as assumed owner) Unknown William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL (as assumed owner)		Judith East The Gables Gosberton Bank Gosberton SPALDING PE11 4PB (as assumed owner) Unknown William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL (as assumed owner)	
46-044	Freehold Acquisition	Freehold over 4730 square metres of copse and agricultural land (north of Gosberton Bank)	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	NONE

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
46-044 cont'd			William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL				
46-045	J, L	square metres of agricultural land and access track (east of Gosberton Road)	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
46-046	J	Permanent Rights over 2762 square metres of agricultural land and access track (east of Gosberton Road)	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)
46-047	J	Permanent Rights over 159 square metres of agricultural land and access track (east of Gosberton Road)	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	NONE

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
46-047 cont'd			Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL						
46-048	L	square metres of agricultural land and access track (east of Gosberton Road)	Davina Lynette Fillingham 12 Platinum Close Gosberton SPALDING PE11 4PP Jarred Thomas Wright Enterprise Way Pinchbeck SPALDING Lincolnshire PE11 3YR	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	Walter Smith (Gosberton) Ltd Tull Grange Cawood Lane Gosberton SPALDING PE11 4HB	NONE			

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
46-048 cont'd			William Eric Creasey 29 Birch Grove SPALDING Lincolnshire PE11 2HL			
47-001	К	Temporary Rights over 3 square metres of copse (east of A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE
47-002	К	A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)

	Land v	which is proposed to be subject to	BOOK OF o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights y of Lincolnshire	s to carry out protective works (Regulation 7(1))(a))	
Number on	F 1	Descriptions (local	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-003	К	, , ,	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner)	NONE	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as assumed owner)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
47-003 cont'd			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown		Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	Unknown			
47-004	G	square metres of agricultural land and access track (east of A16)	PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of underground cable apparatus and street furniture)			

	Land	which is proposed to be subject to	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation	
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
47-004 cont'd			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG			Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right of way as stated in Deed dated 16 March 1999) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain) Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)
47-005	Freehold Acquisition	Freehold over 4 square metres of private road (North of Surfleet Bank)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	NONE	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on		Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Extent of acquisition or use Description	Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
47-005 cont'd			Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown		Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	Unknown			
47-006	Freehold Acquisition	Freehold over 4722 square metres of agricultural land, access track and hedgerow (Woad Farm)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor)			

	Land wh	nich is proposed to be subject	BOOK OF REF to: (i) powers of compulsory acquisition, (ii) ri	nd Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to o f Lincolnshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 (a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use	Description of fand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
47-006 cont'd			Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG			Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to r of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans	Extent of acquisition of use	Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
47-007	J	Permanent Rights over 3507 square metres of agricultural land, access track and drain (east of A16)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor) Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)
47-008	Freehold Acquisition	Freehold over 742 square metres of drain (east of A16)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	NONE	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)

	Land	which is proposed to be subject	BOOK OF R to: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-008 cont'd			Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown		Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	Unknown	
47-009	Freehold Acquisition	Freehold over 4252 square metres of access track and verge (east of A16)	Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Brian Douglas Naylor and Simon Brian Naylor) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (in respect of a rights stated in Conveyance dated 05 October 1984)	

	Land	which is proposed to be subject t	BOOK OF RE o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-009 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
47-010	J	Permanent Rights over 679 square metres of drain and copse (east of A16)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	NONE	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
47-010 cont'd			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown		Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown				
47-011	Freehold Acquisition	metres of agricultural land (east of A16)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)			

	Land v	which is proposed to be subject to	BOOK OF RI DOCK OF RI DOC (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans	Land Plans	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
47-011 cont'd			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG			Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)
47-012	J, L	Permanent Rights over 2267 square metres of agricultural land, access track and drain (east of A16)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor) Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
47-013	J	Permanent Rights over 3340 square metres of agricultural land, access track and drain (east of A16)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor) Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)			
47-014	L	Permanent Rights over 9109 square metres of agricultural land, access track and drain (east of A16)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Alan Burdall Aston Leigh Surfleet Marsh Surfleet SPALDING PE11 4DP (in respect of a right of way as stated in Transfer dated 08 June 2001)			

	Land w	hich is proposed to be subject	BOOK OF REFI to: (i) powers of compulsory acquisition, (ii) rig	nd Development Consent Order ERENCE - PART 1 ght to use the land, and/or (iii) rights to o Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1)(Category 1 a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
47-014 cont'd			Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG			Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor) Gloria June Burdall Aston Leigh Surfleet Marsh Surfleet Marsh Surfleet SPALDING PE11 4DP (in respect of a right of way as stated in Transfer dated 08 June 2001) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 2 Qualifying persons under Regulation					
Land Plans	Extent of acquisition of use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
47-014 cont'd						Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)			
47-015	J	Permanent Rights over 448 square metres of agricultural land, access track and drain (east of A16)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)			

	Land	which is proposed to be subject to	BOOK OF F o: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1 i) right to use the land, and/or (iii) right: y of Lincolnshire	s to carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans	Extent of acquisition of use	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-016	J	Permanent Rights over 325 square metres of drain (north of Marsh Drove)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	NONE	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
			Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) John Grant (Donington)		Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) John Grant (Donington)	Unknown	
			The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)		The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)		
			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner)		Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner)		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
47-016 cont'd			Unknown		Unknown				
47-017	Freehold Acquisition	Freehold over 2938 square metres of agricultural land (north of Marsh Drove)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain) Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)			

	Land	which is proposed to be subject to	BOOK OF F bo: (i) powers of compulsory acquisition, (i	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights to y of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-018	J	Permanent Rights over 3360 square metres of agricultural land, access track and drain (east of A16)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor) Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)	
47-019	J	Permanent Rights over 417 square metres of drain (north of Marsh Drove)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	NONE	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquicition or use	Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans	Land Plans Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
47-019 cont'd			Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown		Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner) Unknown	Unknown			

	Land	which is proposed to be subject to	BOOK OF R bo: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights t r of Lincolnshire	o carry out protective works (Regulation 7(1))(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-020	J	Permanent Rights over 576 square metres of drain (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON LincoInshire PE21 6NA (as assumed owner) Unknown	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
47-021	J	Permanent Rights over 32 square metres of drain (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON LincoInshire PE21 6NA	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
47-022	J, L	Permanent Rights over 4 square metres of drain (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	

	Land	which is proposed to be subject t	BOOK OF F o: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) right y of Lincolnshire	s to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Aj Regulations 2009	oplications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
47-023	J, L	Permanent Rights over 31 square metres of drain (north of Marsh Drove)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	NONE	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)
			Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner)		Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner)	Unknown

	Land v	which is proposed to be subject to	BOOK OF R o: (i) powers of compulsory acquisition, (ii	Vind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights • of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Ap Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
47-023 cont'd			Unknown		Unknown	
47-024]	Permanent Rights over 61 square metres of drain (north of Marsh Drove)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner)	NONE	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG (as assumed owner)	Unknown

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
47-024 cont'd			Unknown		Unknown	
47-025	J	Permanent Rights over 53 square metres of agricultural land, access track and drain (east of A16)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor) Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)
47-026	Freehold Acquisition	Freehold over 3678 square metres of agricultural land (north of Marsh Drove)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor)

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
47-026 cont'd			Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG			The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain) Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)
47-027	J	Permanent Rights over 2189 square metres of agricultural land, access track and drain (east of A16)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor) Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			Category 2 Qualifying persons under Regulation			
Land Plans	Land Plans	Description of faild	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
47-027 cont'd			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG						
47-028	K	square metres of agricultural land (north of Marsh Drove)	Ann Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA Brian Douglas Naylor 1-4 London Road SPALDING Lincolnshire PE11 2TA	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Naylor Farms Roman Bank Moulton Seas End Spalding Lincolnshire PE12 6LG	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for Ann Naylor, Brian Douglas Naylor and Simon Brian Naylor) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans	Land Plans		Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
47-028 cont'd			Simon Brian Naylor Saltings Roman Bank Moulton Seas End SPALDING Lincolnshire PE12 6LG			Unknown (in respect of restrictive covenants as may have been imposed before 02 April 1940)			
47-029	G	square metres of agricultural	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	A.H. Worth (Fosdyke) Limited Fleet Estate Office Manor Farm Holbeach Hurn Holbeach SPALDING Lincolnshire PE12 8LR	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			
47-030	K		John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)			

	Land	which is proposed to be subject t	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights t of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 1)(a) of the Infrastructure Planning (Appl Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of Iana	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
47-031	Freehold Acquisition	Freehold over 4404 square metres of agricultural land (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	Pengethley Potatoes Limited 5 Resolution Close BOSTON Lincolnshire PE21 7TT	NONE
47-032	E	Permanent Rights over 581 square metres of public road and verges (Marsh Road)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY (assumed in respect of subsoil beneath public highway)	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus)

	Land	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights ty of Lincolnshire	s to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 7(1)(a) of the Infrastructure Planning (Ap Regulations 2009	plications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
47-032 cont'd			Unknown			Unknown
47-033	E	square metres of agricultural	Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY	Jason Palmer 29 Coalbeach Lane South Surfleet SPALDING PE11 4DE Keith Palmer High Toft Birds Drove Surfleet SPALDING PE11 4BE	Jason Palmer 29 Coalbeach Lane South Surfleet SPALDING PE11 4DE Keith Palmer High Toft Birds Drove Surfleet SPALDING PE11 4BE	NONE
47-034	К	Temporary Rights over 210 square metres of agricultural land and access track (south of Marsh Drove)	Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY	Jason Palmer 29 Coalbeach Lane South Surfleet SPALDING PE11 4DE Keith Palmer High Toft Birds Drove Surfleet SPALDING PE11 4BE	Jason Palmer 29 Coalbeach Lane South Surfleet SPALDING PE11 4DE Keith Palmer High Toft Birds Drove Surfleet SPALDING PE11 4BE	NONE

	Land	which is proposed to be subject to	BOOK OF F o: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights / of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))		
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009				
Land Plans		Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
47-035	К	Temporary Rights over 45 square metres of agricultural land (south of Marsh Drove)	Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY	Martyn J Templeman Esq. Fairview Newlands Road Surfleet SPALDING PE11 4DT	Martyn J Templeman Esq. Fairview Newlands Road Surfleet SPALDING PE11 4DT	NONE		
47-036	J	Permanent Rights over 30 square metres of drain (north of Marsh Drove)	Esmond Herbert Gadd Big Tree Farm Surfleet Marsh Surfleet SPALDING Lincolnshire PE11 4DW (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Judith Gadd Big Tree Farm Surfleet Marsh Surfleet SPALDING Lincolnshire PE11 4DW (as assumed owner)	NONE	Esmond Herbert Gadd Big Tree Farm Surfleet Marsh Surfleet SPALDING Lincolnshire PE11 4DW (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Judith Gadd Big Tree Farm Surfleet Marsh Surfleet SPALDING Lincolnshire PE11 4DW (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown		

	Land	which is proposed to be subject t	BOOK OF R o: (i) powers of compulsory acquisition, (ii	Vind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
47-036 cont'd			Unknown		Unknown	
47-037	J	Permanent Rights over 1793 square metres of agricultural land, hedgerow and drain (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
47-038	J, L	Permanent Rights over 1081 square metres of agricultural land, hedgerow and drain (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)			
47-039	Freehold Acquisition	Freehold over 1991 square metres of agricultural land (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)			

	Land	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights t ry of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
48-001	J	Permanent Rights over 17 square metres of verge (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
			Unknown		Unknown	Unknown	
48-002	Freehold Acquisition	Freehold over 471 square metres of agricultural land (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-003	J	Permanent Rights over 1047 square metres of agricultural land, access track and hedgerow (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	

	Land	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
48-004	J, L	Permanent Rights over 655 square metres of agricultural land, access track and hedgerow (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-005	Freehold Acquisition	Freehold over 1312 square metres of agricultural land and grassed area (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-006	Freehold Acquisition	Freehold over 22 square metres of agricultural land (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-007	Freehold Acquisition	Freehold over 885 square metres of agricultural land, grassed area and shrubbery (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	
48-008	J, L	Permanent Rights over 458 square metres of agricultural land (south of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
48-009	J	Permanent Rights over 688 square metres of agricultural land (south of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE
48-010	J	Permanent Rights over 622 square metres of agricultural land (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	Unknown
48-011	J, L	Permanent Rights over 484 square metres of agricultural land (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
48-012	Freehold Acquisition	Freehold over 905 square metres of agricultural land (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	Unknown			
48-013	J	Permanent Rights over 297 square metres of agricultural land (north of Marsh Drove)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	Unknown			
48-014	E	Permanent Rights over 714 square metres of agricultural land and access track (south of Marsh Drove)	Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY	Martyn J Templeman Esq. Fairview Newlands Road Surfleet SPALDING PE11 4DT	Martyn J Templeman Esq. Fairview Newlands Road Surfleet SPALDING PE11 4DT	NONE			

	Land	which is proposed to be subject to	BOOK OF c: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights ty of Lincolnshire	to carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 2 Qualifying persons under Regulation		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
48-015	E	Permanent Rights over 18958 square metres of agricultural land (north of River Welland)	Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY	Steven John Mayfield Esq. Glen Willows 18 Stock House Lane Surfleet SPALDING Lincolnshire PE11 4AR	Steven John Mayfield Esq. Glen Willows 18 Stock House Lane Surfleet SPALDING Lincolnshire PE11 4AR	NONE
48-016	К	Temporary Rights over 523 square metres of agricultural land (north of River Welland)	Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY	Steven John Mayfield Esq. Glen Willows 18 Stock House Lane Surfleet SPALDING Lincolnshire PE11 4AR	Steven John Mayfield Esq. Glen Willows 18 Stock House Lane Surfleet SPALDING Lincolnshire PE11 4AR	NONE
48-017	E	Permanent Rights over 264 square metres of agricultural land (south of Marsh Drove)	Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY	Jason Palmer 29 Coalbeach Lane South Surfleet SPALDING PE11 4DE Keith Palmer High Toft Birds Drove Surfleet SPALDING PE11 4BE	Jason Palmer 29 Coalbeach Lane South Surfleet SPALDING PE11 4DE Keith Palmer High Toft Birds Drove Surfleet SPALDING PE11 4BE	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)

	Land	which is proposed to be subject t	BOOK OF co: (i) powers of compulsory acquisition, (Wind Development Consent Order REFERENCE - PART 1 ii) right to use the land, and/or (iii) rights t y of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))	
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans		Beschption on land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
48-018	E	Permanent Rights over 5136 square metres of hedgerow, grassed area and shrubbery (north of River Welland) and public byway (LL Surf 3/4)	Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY	Tom French 84 Hallgate Holbeach SPALDING PE12 7HZ	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public byways (LL Surf 3/4)) Tom French 84 Hallgate Holbeach SPALDING PE12 7HZ	Unknown (in respect of restrictive covenants as may have been imposed thereon before 16 January 2018 and are still subsisting and capable of being enforced)	
48-019	Ε	Permanent Rights over 6564 square metres of river (River Welland), foreshore, bed and banks thereof	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY (as assumed owner) Unknown	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (as assumed owner) Surfleet United Charities 2 Farthing Close BOSTON Lincolnshire PE21 7LY (as assumed owner) Unknown	Unknown	

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
48-020	Ε	square metres of agricultural land, scrubland and hedgerow (north of Marsh Road) and public bridleway (LL Wstn 6/2)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public bridleways (LL Surf I 8/2 and LL Wstn 6/2))	The Dean And Chapter Of The Cathedral Church Of Christ In Oxford Of The Foundation Of King Henry The Eighth Christ Church St. Aldates Oxford Oxfordshire OX1 1DP (in respect of personal covenants contained in Conveyance dated 25 July 1949) Unknown (in respect of rights reserved in Transfer dated 01 September 1989)
48-021	E	square metres of agricultural	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE

	Land v	which is proposed to be subject to	BOOK OF RI c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
48-021 cont'd					Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public bridleways (LL Surf 8/2))	
48-022	E	square metres of agricultural land (north of Marsh Road)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE
48-023	E	square metres of agricultural land (north of Marsh Road)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for George Hay & Sons Limited) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
48-023 cont'd 48-024	Ε	Permanent Rights over 17109 square metres of agricultural land (north of Marsh Road)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of rights of access) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Unknown (in respect of restrictive covenants that
48-025	E	Permanent Rights over 14133	George Hay & Sons Limited	NONE	George Hay & Sons Limited	may have been imposed thereon before 11 December 2013) Barclays Security Trustee Limited
		square metres of agricultural land (north of Marsh Road)	Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE		Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	1 Churchill Place LONDON E14 5HP (as mortgagee for George Hay & Sons Limited)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
48-025 cont'd						Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground and overhead electricity apparatus) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of rights of access)			

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning	
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
49-001	F	Temporary Rights over 18 square metres of scrubland (south of River Welland)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH	NONE
49-002	F	square metres of access track (north of Marsh Road) and	Alexander James Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (as trustees of the George Hay & Sons retirement benefit scheme) Andrew George Hay Wragg Marsh Farm Wragg Marsh Farm Wragg Marsh Spalding Marsh Spalding Marsh SPALDING Lincolnshire PE12 6HQ (as trustees of the George Hay & Sons retirement benefit scheme)	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public bridleway (LL Surf 9/1))	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) TM Trustees Limited Dunn's House St Paul's Road SAILSBURY Wiltshire SP2 7BF (in respect of a restriction in an unknown document dated 28 April 2020)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
49-002 cont'd			Mary Ellenor Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (as trustees of the George Hay & Sons retirement benefit scheme)			Unknown (in respect of restrictive covenants that may have been imposed thereon before 11 December 2013)			
49-003	F	Temporary Rights over 509 square metres of private road and verges (West of Wragg Marsh Farm) and public bridleways (LL Wstn 4/1 and LL Wstn 5/1)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public bridleway (LL Wstn /1 and LL Wstn 5/1))	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for George Hay & Sons Limited) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
49-003 cont'd						Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)			
49-004	F	and verge (Marsh Road)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE South Holland Internal Drainage Board Marsh Reeves Foxes Lowe Road Holbeach SPALDING Lincolnshire PE12 7PA (in respect of Lord's Drain)	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE South Holland Internal Drainage Board Marsh Reeves Foxes Lowe Road Holbeach SPALDING Lincolnshire PE12 7PA (in respect of Lord's Drain)	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for George Hay & Sons Limited) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access)			

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
49-004 cont'd						Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005) Unknown (in respect of restrictive covenants that may have been imposed thereon before 21 March 2000) Unknown (in respect of a restriction stated in an unknown document dated 1 July 2002)
49-005	G	Temporary Rights over 2286 square metres of agricultural land (north of Marsh Road and south of River Welland)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for George Hay & Sons Limited) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
49-005 cont'd						InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground and overhead electricity apparatus) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of rights of access)			
49-006	G	Temporary Rights over 5104 square metres of agricultural land (Wragg Marsh Farm)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for George Hay & Sons Limited)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
49-006 cont'd						Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
49-007	E	of Marsh Road)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of electricity apparatus)			

	Land	which is proposed to be subject to	BOOK OF RI o: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of failu	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
49-007 cont'd						Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of rights of access)
49-008	E	Permanent Rights over 261 square metres of public road (Marsh Road)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (assumed in respect of subsoil beneath public highway) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority) Unknown	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
49-009	E	square metres of verge (south of Marsh Road)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	NONE	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (as highways authority)	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
49-009 cont'd						Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of rights of access)			
49-010	E	Permanent Rights over 38389 square metres of agricultural land and private road (Wragg Marsh Farm)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for George Hay & Sons Limited) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground and overhead electricity apparatus)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning			
Land Plans		Description of fund	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
49-010 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)			
49-011	К	Temporary Rights over 1194 square metres of agricultural land (Wragg Marsh Farm)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for George Hay & Sons Limited) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation			
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
49-011 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)			
50-001	К		St John's College Cambridge St. John's College CAMBRIDGE Cambridgeshire CB2 1TP	H.C.C.Tinsley & Son Limited 14 St. Marys Street STAMFORD PE9 2DF	H.C.C.Tinsley & Son Limited 14 St. Marys Street STAMFORD PE9 2DF	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of legal easements granted by Lease dated 29 October 2004)			

	Land	which is proposed to be subject to	BOOK OF R c: (i) powers of compulsory acquisition, (ii	Nind Development Consent Order EFERENCE - PART 1) right to use the land, and/or (iii) rights to r of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
50-002	E	Permanent Rights over 506159 square metres of building and hardstanding (Crowtree Farm), agricultural land and access track (east of Marsh Road), electricity cables and pylons and public footpath (LL Wstn 7/1)	St John's College Cambridge St. John's College CAMBRIDGE Cambridgeshire CB2 1TP	H.C.C.Tinsley & Son Limited 14 St. Marys Street STAMFORD PE9 2DF	H.C.C.Tinsley & Son Limited 14 St. Marys Street STAMFORD PE9 2DF Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL I Wstn I 7/1))	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of legal easements granted by Lease dated 29 October 2004)

	Land	which is proposed to be subject to	BOOK OF RI DOCK OF RI BOOK OF RI BOOK OF RI	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	arry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
50-003	E	Permanent Rights over 203228 square metres of agricultural land and access track (east of Marsh Road), electricity cable, pylons and public footpath (LL Wstn 7/1)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of public footpath (LL Wstn 7/1))	Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
50-004	E	Permanent Rights over 1958 square metres of agricultural land and drain (east of Marsh Road)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of subsoil rights)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for George Hay & Sons Limited) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning				
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009			
50-004 cont'd						Spalding Energy Expansion Limited 30 Crown Place LONDON EC2A 4ES (in respect of the rights granted by the Deed dated 26 February 2018)			
50-005	E	land, access track, electricity cable and pylons (east of Marsh Road)	Wykeham Abbey The Chase	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	Barclays Security Trustee Limited 1 Churchill Place LONDON E14 5HP (as mortgagee for George Hay & Sons Limited) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground and overhead electricity apparatus)			

	Land v	which is proposed to be subject to	BOOK OF RE DOCK OF RE DOC (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 2 Qualifying persons under Regulation		
Land Plans		Description of faile	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
50-005 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)
50-006	E	square metres of agricultural land, access and drain (east of	H Pennington & Son Welland House Farm Spalding Marsh SPALDING PE12 6HF (as assumed owner) Unknown	NONE	H Pennington & Son Welland House Farm Spalding Marsh SPALDING PE12 6HF (as assumed owner) Unknown	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)

	Land v	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
50-007	Ε	Permanent Rights over 40886 square metres of agricultural land, access track and drain (east of Marsh Road)	Mary Lake 6 Orchard Close Ramsey HARWICH Essex CO12 5HG (as assumed owner) Unknown	NONE	H Pennington & Son Welland House Farm Spalding Marsh SPALDING PE12 6HF Unknown	Unknown
51-001	E	Permanent Rights over 39873 square metres of agricultural land and drain (east side of Marsh Road)	H Pennington & Son Welland House Farm Spalding Marsh SPALDING PE12 6HF (as assumed owner) Unknown	NONE	H Pennington & Son Welland House Farm Spalding Marsh SPALDING PE12 6HF (as assumed owner) Unknown	NONE
51-002	E	Permanent Rights over 22267 square metres of agricultural land (east Marsh Road)	Ian Fred Pennington Welland House Farm Spalding Marsh SPALDING Lincolnshire PE12 6HF	NONE	Ian Fred Pennington Welland House Farm Spalding Marsh SPALDING Lincolnshire PE12 6HF	NONE

	Land	which is proposed to be subject to	BOOK OF R b: (i) powers of compulsory acquisition, (ii)	Vind Development Consent Order EFERENCE - PART 1 right to use the land, and/or (iii) rights to of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Appli Regulations 2009	cations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
51-003	E	Permanent Rights over 521 square metres of drain (east of Marsh Road)	lan Fred Pennington Welland House Farm Spalding Marsh SPALDING Lincolnshire PE12 6HF (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	NONE	lan Fred Pennington Welland House Farm Spalding Marsh SPALDING Lincolnshire PE12 6HF (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	Unknown
51-004	E	Permanent Rights over 128568 square metres of agricultural land (east of Marsh Road)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	NONE	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA	HSBC UK Bank PLC 1 Centenary Square BIRMINGHAM B1 1HQ (as mortgagee for John Grant as contained in the Charge dated 01 December 1993)
51-005	E	Permanent Rights over 17 square metres of drain (east of Marsh Road)	H Pennington & Son Welland House Farm Spalding Marsh SPALDING PE12 6HF (as assumed owner)	NONE	H Pennington & Son Welland House Farm Spalding Marsh SPALDING PE12 6HF (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire								
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009					
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
51-005 cont'd			John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown		John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown				
51-006	E	square metres of drain (east of Marsh Road)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (as assumed owner) John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (as assumed owner) Unknown	Unknown			

	Land	which is proposed to be subject to	BOOK OF F c: (i) powers of compulsory acquisition, (ii	Wind Development Consent Order REFERENCE - PART 1) right to use the land, and/or (iii) rights t / of Lincolnshire	o carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7	Category 1 (1)(a) of the Infrastructure Planning (Appl Regulations 2009	ications: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
51-007	E	Permanent Rights over 159921 square metres of agricultural land, drain and access track (east of Marsh Road), electricity cables and pylons	Wykeham Abbey The Chase	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE PE12 6HE	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights and restrictive covenants listed in a Deed dated 08 May 2003) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of the legal easements granted by the Lease dated 14 January 2005)
51-008	E	Permanent Rights over 2157 square metres of agricultural land and drain (east of Marsh Road)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of subsoil rights)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of apparatus)

	Land	which is proposed to be subject to	BOOK OF RE c: (i) powers of compulsory acquisition, (ii)	/ind Development Consent Order FERENCE - PART 1 right to use the land, and/or (iii) rights to o of Lincolnshire	carry out protective works (Regulation 7(1)	(a))
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	tions: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
51-008 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights and restrictive covenants listed in a Deed dated 08 May 2003) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of the legal easements granted by the Lease dated 14 January 2005 and 26 February 2018) Spalding Energy Expansion Limited 30 Crown Place LONDON EC2A 4ES (in respect of the rights that are granted by the Deed dated 26 February 2018)
51-009	E	Permanent Rights over 189781 square metres of agricultural land, drain and access track (east of Marsh Road), electricity cables and pylons	Wykeham Abbey The Chase	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire					
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 L)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans		Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
51-009 cont'd						National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights and restrictive covenants listed in a Deed dated 08 May 2003) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of the legal easements granted by the Lease dated 14 January 2005) Unknown (in respect to rights relating to nine inch sanitary pipes detailed in a deed dated 29 April 1943)
51-010	E	square metres of drain (east of Marsh Road)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (as assumed owner)	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (as assumed owner)	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire						
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(2	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
51-010 cont'd			St John's College Cambridge St. John's College CAMBRIDGE Cambridgeshire CB2 1TP (as assumed owner) Unknown		St John's College Cambridge St. John's College CAMBRIDGE Cambridgeshire CB2 1TP (as assumed owner) Unknown	Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of rights of access) Unknown	
51-011	E		St John's College Cambridge St. John's College CAMBRIDGE Cambridgeshire CB2 1TP	H.C.C.Tinsley & Son Limited 14 St. Marys Street STAMFORD PE9 2DF Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of subsoil rights)	H.C.C.Tinsley & Son Limited 14 St. Marys Street STAMFORD PE9 2DF	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of legal easements granted by the Lease dated 29 October 2004) Spalding Energy Expansion Limited 30 Crown Place LONDON EC2A 4ES (in respect of rights and a restriction stated in Deed dated 26 February 2018)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire					
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(Category 1 1)(a) of the Infrastructure Planning (Applic Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
51-012	E	Permanent Rights over 27076 square metres of agricultural land (east of Marsh Road)	St John's College Cambridge St. John's College CAMBRIDGE Cambridgeshire CB2 1TP	H.C.C.Tinsley & Son Limited 14 St. Marys Street STAMFORD PE9 2DF	H.C.C.Tinsley & Son Limited 14 St. Marys Street STAMFORD PE9 2DF	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of apparatus) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING PE11 2BB (in respect of legal easements granted by Lease dated 29 October 2004)
51-013	Ε	Permanent Rights over 5 square metres of drain (east of Marsh Road)	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 GHE (as assumed owner) Unknown	NONE	George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (as assumed owner) Unknown	Unknown
51-014	E	Permanent Rights over 308 square metres of drain (east of Marsh Road)	Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 ONH (as assumed owner)	NONE	Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 ONH (as assumed owner)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire					
Number on		Description of land	Qualifying persons under Regulation 7(1)	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans	Extent of acquisition or use	Description of land	Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
51-014 cont'd			Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW (as assumed owner) George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (as assumed owner) Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP (as assumed owner)		Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW (as assumed owner) George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (as assumed owner) Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP (as assumed owner)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of Lincolnshire					
Number on	Extent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
Land Plans			Owners or Reputed Owners	Lessees or Tenants	Occupiers	7(1)(a) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
51-014 cont'd			Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH (as assumed owner) Unknown		Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH (as assumed owner) Unknown	
51-015	E	Permanent Rights over 22693 square metres of agricultural land (east of Marsh Road), electricity cables and pylons	Bridgett Lucy Posey Priors Lodge Shore Road Freiston BOSTON Lincolnshire PE22 ONH Denis John Grimwood The Gatehouse Broad Lane Moulton SPALDING Lincolnshire PE12 6PW	Oldershaws Of Moulton Limited The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP	Oldershaws Of Moulton Limited The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP	Gordon Albert Mills St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH (in respect of rights granted in Transfer dated 22 April 2003) Helen Juliette Mills St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH (in respect of rights granted in Transfer dated 22 April 2003)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 1 Land which is proposed to be subject to: (i) powers of compulsory acquisition, (ii) right to use the land, and/or (iii) rights to carry out protective works (Regulation 7(1)(a)) County of LincoInshire					
Number on	Evtent of acquisition or use	Description of land	Qualifying persons under Regulation 7(1	Category 1)(a) of the Infrastructure Planning (Applica Regulations 2009	ations: Prescribed Forms and Procedures)	Category 2 Qualifying persons under Regulation 7(1)(a) of the Infrastructure Planning
Land Plans	Extent of acquisition or use Description of land	Description of rand	Owners or Reputed Owners	Lessees or Tenants	Occupiers	(Applications: Prescribed Forms and Procedures) Regulations 2009
51-015 cont'd			Jane Roberta Ashby Cooke The Gables Broad Lane Moulton SPALDING Lincolnshire PE12 6PP Robert Henry Oldershaw St. Lamberts Hall Hallgate Weston SPALDING Lincolnshire PE12 6RH			Moulton Bulb Company Limited Long Lane Moulton SPALDING PE12 6PP (in respect of restrictions listed in a Definitive Trust Deed dated 26 September 1986 and rights granted in Transfer dated 22 April 2003) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights relating to electricity lines and pylons listed in a Deed of Grant dated 16 April 2003) Prudential Financial Planning Limited 10 Fenchurch Avenue LONDON EC3M 5AG (in respect of restrictions listed in a Definitive Trust Deed dated 26 September 1986) Unknown (in respect of restrictions listed in a Definitive Trust Deed dated 26 September 1986)

	Outer Dowsing Offshore Wind Development Consent Order				
	BOOK OF REFERENCE - PART 2 County of Lincolnshire				
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
01-006	93177 square metres of agricultural land, drain, ponds and path (East of Roman Bank)	Alco Estates Limited Boathouse Sandhurst Road Sandilands MABLETHORPE Lincolnshire LN12 2RH (in respect of rights and restrictive covenants stated in Tranfer dated 30 September 2009) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted by Deed dated 14 January 1941)			
01-007	553 square metres of verge (Roman Bank)	Unknown (in respect of rights and restrictive covenants stated in Tranfer dated 30 September 2009)			
01-008	184 square metres of access track and verge (east of Roman Bank)	Unknown			
01-009	2521 square metres of public road and verges (Roman Bank)	Unknown			
01-010	2154 square metres of verge (Roman Bank)	Unknown			
01-011	177 square metres of public road, verge and drain (Roman Bank)	Unknown			
01-012	569 square metres of agricultural land (west of Roman Bank)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown			

	C	Duter Dowsing Offshore Wind Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of Lincolnshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-013	22897 square metres of agricultural land (west of Roman Bank)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown
01-014	8249 square metres of agricultural land (west of Roman Bank)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown
01-015	20085 square metres of agricultural land (west of Roman Bank)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown
01-016	34738 square metres of agricultural land (west of Roman Bank)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)

	Outer Dow	rsing Offshore Wind Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of Lincolnshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
01-016 cont'd		Unknown
01-017	43296 square metres of agricultural land (west of Roman Bank)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown
01-018	6314 square metres of agricultural land, drain and access track (west of Roman Bank)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of rights reserved by Conveyance dated 16 March 1965)

	Outer Dowsing Offshore Wind Development Consent Order			
		BOOK OF REFERENCE - PART 2		
		County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
01-019	812 square metres of agricultural land and drain (west of Roman Bank)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown		
02-001	26863 square metres of access track and drain (west of Roman Bank)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown		
02-002	13744 square metres of agricultural land and access track (west of Roman Bank)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)		

	Outer Dows	sing Offshore Wind Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of Lincolnshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
02-002 cont'd		Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown
02-003	719 square metres of agricultural land and drain (west of Roman Bank)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown
	2278 square metres of agricultural land and hedgerow (west of Roman Bank) and public footpath (LL Chap 19/5)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)

1	Outer Do	wsing Offshore Wind Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of Lincolnshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
02-004 cont'd		Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown
02-005	491 square metres of agricultural land (north of Ember Lane) and public footpath (LL Chap 19/5)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown
02-006	36239 square metres of agricultural land (north of Ember Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown
02-007	1386 square metres of public road and verges (Ember Lane) and public footpath (LL Chap 21/4)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-008	751 square metres of agricultural land (south of Ember Lane)	Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL (in respect of a rights stated in Transfer dated 28 May 2009 and Conveyance dated 22 October 1979) Unknown (in respect of rights granted by Conveyance dated 11 October 1993)	
02-011	178 square metres of agricultural land (south of Ember Lane)	Unknown	
02-015	17 square metres of hedgerow (south of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
02-016	1073 square metres of agricultural land (south of Ember Lane) and public footpath (LL Chap 27/3)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-017	1 square metres of agricultural land (south of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
02-018	1 square metres of agricultural land (south of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (assumed in respect of rights of access) Unknown	
02-019	28 square metres of agricultural land (south of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
02-020	2153 square metres of agricultural land (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of reservation of mines and minerals and rights to work the same)	
02-021	16 square metres of agricultural land (south of Ember Lane)	Unknown	
03-001	47 square metres of agricultural land (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
03-002	485 square metres of agricultural land (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-003	30 square metres of agricultural land (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
03-004	3 square metres of agricultural land and drain (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of reservation of mines and minerals and rights to work the same)	
03-005	61 square metres of agricultural land and drain (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of reservation of mines and minerals and rights to work the same)	
03-006	248 square metres of agricultural land (east of Ember Lane)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of a rights as stated in Deed dated 28 February 1980 and and in Transfer dated 22 December 1997)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number			
Number on	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans		of the Planning Act 2008	
03-006		Lindsey Marsh Drainage Board	
cont'd		Wellington House	
		Manby Park	
		Manby	
		LOUTH	
		Lincolnshire	
		LN11 8UU	
		(assumed in respect of rights of access)	
03-007	106 square metres of agricultural land (east of Ember Lane)	Allan Walter Clarke	
		Hill View Farm	
		Wigg Lane	
		Chapel St. Leonards	
		SKEGNESS	
		Lincolnshire	
		PE24 5RJ	
		(in respect of a right to enter and service media rights stated in Transfer dated 28 May 2008)	
		Catherine Ellen Hill	
		Highfield	
		Authorpe	
		LOUTH	
		Lincolnshire	
		LN11 8PG	
		(in respect of rights and restrictive covenants as stated in Transfer dated 28 May 2009)	
		Lindsey Marsh Drainage Board	
		Wellington House	
		Manby Park	
		Manby	
		LOUTH	
		Lincolnshire	
		LN11 8UU	
		(assumed in respect of rights of access)	
		Richard John Hill	
		Highfield	
		Authorpe	
		LOUTH	
		Lincolnshire	
		LN11 8PG	
		(in respect of rights and restrictive covenants as stated in Transfer dated 28 May 2009)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-008	1044 square metres of agricultural land (east of Ember Lane)	Allan Waiter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 SRJ (in respect of a right to enter and service media rights stated in Transfer dated 28 May 2008) Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire LN11 8PG (in respect of rights and restrictive covenants as stated in Transfer dated 28 May 2009) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Richard John Hill Highfield Authorpe LOUTH Lincolnshire LN11 8FG LN11 8F	
03-009	172 square metres of watercourse (Wigg Drain)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-009 cont'd		Unknown	
03-010	58924 square metres of agricultural land, hedgerows, access track (east of Ember Lane), watercourse (Wigg Drain) and public footpath (LL Hogs 34/4, LL Hogs 28/1)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
03-011	1023 square metres of agricultural land and access track (east of Ember Lane)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of a rights as stated in Deed dated 28 February 1980 and and in Transfer dated 22 December 1997) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
03-012	8 square metres of access track (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (assumed in respect of rights of access) Unknown	
03-015	890 square metres of public road and verge (Langham Road)	Unknown	

r	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
03-024	194 square metres of agricultural land (north of Lowgate Road)	Unknown	
03-026	675 square metres of public road and verges (Lowgate Road)	Unknown	
03-027	2424 square metres of agricultural land and drain (south of Lowgate Road)	Unknown (in respect of rights of drainage reserved by a Transfer dated 5 September 2000)	
03-028	23668 square metres of agricultural land and drain (west of Lowgate Road)	Unknown (in respect of rights of drainage reserved by a Transfer dated 5 September 2000)	
03-029	20779 square metres of agricultural land and access track (south of Lowgate Road) and public footpaths (LL Hogs 58/2, LL Hogs 57/1)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 19 April 2019)	
03-030	4889 square metres of agricultural land (north of Bracken Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Lease dated 12 December 2018)	
03-031	651 square metres of agricultural land and access track (north of Bracken Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 19 April 2019)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-001	242 square metres of agricultural land (west of Lowgate Road)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights granted by the Deed of Grant dated 12 December 2018)	
04-002	621 square metres of agricultural land (north of Bracken Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Lease dated 12 December 2018)	
04-003	202 square metres of agricultural land and access track (west of Lowgate Road)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 19 April 2019)	
	3291 square metres of agricultural land (north of Bracken Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in a Lease dated 26 October 2018)	
04-005	586 square metres of hedgerow, access track and drain (north of Bracken Lane)	Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of an assumed right of access) Unknown	

1	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-006	22399 square metres of agricultural land and electricity cable and pylon (north of Bracken Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 26 October 2018)	
04-007	513 square metres of agricultural land and access track (north of Bracken Lane)	Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of an assumed right of access) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Caution dated 12 August 2021) Unknown	
04-008	16 square metres of agricultural land and access track (north of Bracken lane)	Unknown	
04-009	370 square metres of agricultural land (north of Bracken Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 26 October 2018)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-010	131 square metres of access track and drain (north of Bracken Lane)	Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 SPB (in respect of an assumed right of access) Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of an assumed right of access) Unknown	
04-011	2886 square metres of agricultural land (north of Bracken Lane)	L H Ward & Son Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of rights granted by Deed dated 09 April 2019) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deeds dated 9 April 2019 and 26 October 2018)	
04-012	22 square metres of drain (north of Mumby Road, A52)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	County of Lincolnshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-013	714 square metres of agricultural land, access track and copse (north of Mumby Road, A52)	L H Ward & Son Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of rights granted by Deed dated 09 April 2019) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deeds dated 9 April 2019 and 26 October 2018)	
04-014	7467 square metres of agricultural land, access track and copse (north of Mumby Road, A52)	L H Ward & Son Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of rights granted by Deed dated 09 April 2019) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deeds dated 9 April 2019 and 26 October 2018)	
04-015	483 square metres of agricultural land, access track and copse (north of Mumby Road, A52)	L H Ward & Son Hillcrest Cumberworth ALFORD LincoInshire LN13 9LB (in respect of rights granted by Deed dated 09 April 2019)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-015 cont'd		Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deeds dated 9 April 2019 and 26 October 2018)	
04-017	1499 square metres of public road, verges and hedgerow (Mumby Road, A52)	Unknown	
04-018	29900 square metres of agricultural land and access track (south of Mumby Road, A52)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Agreement dated 31 December 1973) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
04-019	219 square metres of drain (south of Bracken Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	County of Lincolnshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-020	3647 square metres of agricultural land and drain (north of Listoft Lane)	Andrew Michael Snell The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 SPG (in respect of rights granted in Conveyance dated 21 January 1993) Gillian Mary Sharpe The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 SPG (in respect of rights granted in Conveyance dated 21 January 1993) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB	
04-021	17206 square metres of agricultural land and drain (north of Listoft Lane)	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (in respect of rights granted by Deed of Grant dated 12 December 2018)	
04-022	1269 square metres of agricultural land and access splay (north of Listoft Lane)	Andrew Michael Snell The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 5PG (in respect of rights granted in Conveyance dated 21 January 1993)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
04-022 cont'd		Gillian Mary Sharpe The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 5PG (in respect of rights granted in Conveyance dated 21 January 1993) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by Deed of Grant dated 12 December 2018)	
04-023	3964 square metres of agricultural land and drain (north of Listoft Lane)	Andrew Michael Snell The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 5PG (in respect of rights granted in Conveyance dated 21 January 1993) Gillian Mary Sharpe The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 5PG (in respect of rights granted in Conveyance dated 21 January 1993)	
05-001	407 square metres of agricultural land (south of Listoft Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by Deed of Grant dated 12 December 2018)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-002	1379 square metres of public road (Listoft Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Unknown	
05-003	16258 square metres of agricultural land and drain (south of Listoft Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by Deed of Grant dated 12 December 2018)	
05-004	222 square metres of drain (Willoughby High Drain, south of Listoft Lane)	Unknown	
05-005	1375 square metres of agricultural land and copse (west of Sloothby High Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-005 cont'd		Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Deed dated 26 October 2018)	
05-006		Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Deed dated 26 October 2018)	
05-007	1064 square metres of public road and verges (Sloothby High Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-007 cont'd		Unknown	
05-008	19597 square metres of agricultural land (east of Sloothby High Lane) and drain (Wyche Drain)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
05-009	106 square metres of drain (Wyche Drain, east of Sloothby Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
05-010	5883 square metres of agricultural land and copse (east of Sloothby High Lane)	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	County of Lincolnshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-011	4416 square metres of agricultural land and drain (east of Sloothby High Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
05-013	50 square metres of agricultural land (east of Sloothby High Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (in respect of right to lay a copper pipe as stated in a Deed dated the 30 October 1961. Also in respect of right to pass as stated in a Transfer dated the 13 May 2004) British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) David Robert Norman Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (in respect of right to lay a copper pipe as stated in a Deed dated the 30 October 1961. Also in respect of right to pass as stated in a Transfer dated the 13 May 2004)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-013 cont'd		Maxine Hayley Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (in respect of right to lay a copper pipe as stated in a Deed dated the 30 October 1961. Also in respect of a right to pass as stated in a Transfer dated the 13 May 2004)	
05-014	784 square metres of public road (south Ings Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
05-016	3009 square metres of agricultural land and copse (east of Sloothby High Lane)	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989)	
05-018	32131 square metres of agricultural land, copse and drains (east of South Ings Lane)	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989)	
05-019	1827 square metres of agricultural land (east of South Ings Lane)	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
05-020	841 square metres of access track (east of South Ings Lane)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of rights pursuant to transfer dated 5 January 2021)	
05-021	19541 square metres of agricultural land, access track, drain (east of South Ings Lane) and public footpath (LL Hogs 48/1)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of rights pursuant to transfer dated 5 January 2021)	
06-001	41 square metres of drain (east of South Ings Lane)	Unknown	
06-004	18 square metres of drain (east of South Ings Lane)	Unknown	
06-005	360 square metres of drain (east of South Ings Lane)	Unknown	
06-008	38463 square metres of agricultural land and drain (east of South Ings Lane)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of transfer dated 5 January 2021)	
06-014	4239 square metres of agricultural land and access track (east South Ings Lane)	Unknown (in respect of subjections as stated in conveyance dated 29 September 1989)	
06-016	2219 square metres of agricultural land and drain (east of South Ings Lane)	Unknown (in respect of subjections as stated in conveyance dated 29 September 1989)	
06-017	34903 square metres of agricultural land (east of South Ings Lane)	Unknown (in respect of subjections as stated in conveyance dated 29 September 1989)	
06-018	847 square metres of public road and verges (South Ings Lane)	Unknown	

1	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
06-019	1583 square metres of agricultural land (west of South Ings Lane)	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989)	
06-020	641 square metres of agricultural land and access splay (east of South Ings Lane)	Unknown (in respect of subjections as stated in conveyance dated 29 September 1989)	
07-002	1283 square metres of agricultural land and drain (west of South Ings Lane)	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
	61514 square metres of agricultural land and drain (west of South Ings Lane and north of Marsh Lane)	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Do	wsing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-003 cont'd		Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Deed dated 28 January 2019)	
07-004	2282 square metres of agricultural land (north of Marsh Lane)	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Deed dated 28 January 2019)	
	199 square metres of drain (north of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (in respect of riparian rights) Unknown	
07-006	54064 square metres of agricultural land and copse (north of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-006 cont'd		Unknown (in respect of the rights granted by the Deed dated 07 December 1966)	
07-007	205 square metres of agricultural land and copse (north of Marsh Lane)	Unknown (in respect of the rights granted by the Deed dated 07 December 1966)	
07-008	8330 square metres of agricultural land (north of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of the rights granted by the Deed dated 07 December 1966)	
07-009	120 square metres of agricultural land and copse (north of Marsh Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of personal covenants stated in Transfer dated 5 January 1994) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LINCOLN LINCOLN LINCOLN (assumed in respect of rights of access)	
07-010	240 square metres of access splay (south of Marsh Lane)	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (in respect of an assumed right of access)	

		Outer Dowsing Offshore Wind Development Consent Order
		BOOK OF REFERENCE - PART 2
County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
07-010		Julia Clare Pound
cont'd		Ayshford Court
		Westleigh
		TIVERTON
		EX16 7HL
		(in respect of an assumed right of access)
		Lindsey Marsh Drainage Board
		Wellington House
		Manby Park
		Manby
		LOUTH
		Lincolnshire
		LN11 8UU
		(assumed in respect of rights of access)
		Mark Caudwell Limited
		Howlett House Farm
		Hogsthorpe
		Skegness
		Lincolnshire
		PE24 5NS
		(in respect of an assumed right of access)
		Mark Leopold Caudwell
		Howlett House
		Slackholme End
		Hogsthorpe
		SKEGNESS
		Lincolnshire
		PE24 5NS
		(in respect of an assumed right of access)
		Unknown
		(in respect of covenants contained in Conveyance dated 22 February 1991)

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-011	1026 square metres of agricultural land (south of Marsh Lane)	David Mark Simpson	
		Whitehouse Farm	
		Addlethorpe	
		SKEGNESS	
		Lincolnshire	
		PE24 4TU	
		(in respect of an assumed right of way)	
		Julia Clare Pound	
		Ayshford Court	
		Westleigh	
		TIVERTON	
		EX16 7HL	
		(in respect of an assumed right of access)	
		Lindsey Marsh Drainage Board	
		Wellington House	
		Manby Park	
		Manby	
		LOUTH	
		Lincolnshire	
		(in respect of riparian rights)	
		Mark Caudwell Limited	
		Howlett House Farm	
		Hogsthorpe	
		Skegness	
		Lincolnshire PE24 5NS	
		(in respect of an assumed right of access)	
		Mark Leopold Caudwell	
		Howlett House	
		Slackholme End	
		Hogsthorpe	
		SKEGNESS	
		Lincolnshire	
		PE24 5NS	
		(in respect of an assumed right of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
07-011 cont'd		Maureen Teresa Caudwell Howlett House Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (in respect of an assumed right of access) Unknown	
07-012	591 square metres of agricultural land (south of Marsh Lane)	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (in respect of a right of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
08-001	17 square metres of access splay (north of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-001 cont'd		Unknown	
08-002	4 square metres of access splay (north of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of an option agreement dated 29 March 2019) Unknown	
08-003	626 square metres of agricultural land and access track (north of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants granted by Deed dated 29 March 2019) Unknown (in respect of the rights granted by the Deed dated 07 December 1966)	

	Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	County of Lincolnshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
08-004	1 square metres of drain (north of Marsh Lane)	David Peter Hand Stone Arches Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (in respect of rights as stated in Conveyance dated 18 February 1991) Graham Kenneth Hand Grange Farm Chapel Lane Addlethorpe SKEGNESS Lincolnshire PE24 4TG (in respect of rights as stated in Conveyance dated 18 February 1991)		
08-005	844 square metres of drain (north of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown		
08-006	420 square metres of agricultural land (south of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown		

-	Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	County of Lincolnshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
08-007	17857 square metres of agricultural land (south of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) RLS House Limited Alex House 260-268 Chapel Street SALFORD Greater Manchester M3 5JZ (in respect of informal rights agreement for car parking) Unknown (in respect of a right of way as contained in Conveyance dated 06 April 1962)		
08-008	39 square metres of agricultural land (south of Marsh Lane)	RLS House Limited Alex House 260-268 Chapel Street SALFORD Greater Manchester M3 5JZ (in respect of informal rights agreement for car parking) Unknown (in respect of a right of way as contained in Conveyance dated 06 April 1962)		
08-009	7624 square metres of agricultural land (south of Marsh Lane)	RLS House Limited Alex House 260-268 Chapel Street SALFORD Greater Manchester M3 5JZ (in respect of informal rights agreement for car parking) Unknown (in respect of a right of way as contained in Conveyance dated 06 April 1962)		

	Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	County of Lincolnshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
08-010	48 square metres of agricultural land (south of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) RLS House Limited Alex House 260-268 Chapel Street SALFORD Greater Manchester M3 5JZ (in respect of informal rights agreement for car parking) Unknown (in respect of a right of way as contained in Conveyance dated 06 April 1962)		
08-011	7 square metres of access track and drain (east of Skegness Stadium, Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown		
08-012	77 square metres of access track (east of Skegness Stadium)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)		

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-012 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of a Wayleave Agreement relating to the construction and maintenance of an electric line dated 01 April 1992)	
08-013	99 square metres of drain (east of Skegness Stadium)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
08-014	29191 square metres of agricultural land (south of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of a Wayleave Agreement relating to the construction and maintenance of an electric line dated 01 April 1992)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
Number	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-015	123 square metres of access track (south of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH	
		Lincolnshire LN11 8UU (in respect of riparian rights)	
		Unknown	
		Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (In respect of rights as stated in deed dated 28 January 2019) Unknown (in respect of rights contained in Conveyance dated 08 September 1948)	
08-017	763 square metres of access track (north of Ingoldmells Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
08-017 cont'd		Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (In respect of rights as stated in deed dated 28 January 2019) Unknown (in respect of rights contained in Conveyance dated 08 September 1948)	
09-001	26 square metres of access splay (south of Ingoldmells Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
09-002	976 square metres of public road and drain (Ingoldmells Road)	Unknown	
	152 square metres of drain (south of Ingoldmells Road)	Unknown	
	59860 square metres of agricultural land and drains (south of Ingoldmells Road and north of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of right of way for a drift road reserved by a conveyance dated 8 September 1948)	

	Outer Dov	wsing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-005	9375 square metres of access track and drain (north of Younger's Lane and south of Ingoldmells Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of right of way for a drift road reserved by a conveyance dated 8 September 1948)	
09-006	167 square metres drain (north of Younger's Lane)	Unknown	
09-008	2633 square metres of agricultural land (north of Younger's Lane)	J E Spence & Son Farms Ltd Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN (in respect of a right to install water pipes granted in Deed dated 06 June 1961)	
09-009	30 square metres of verge (north of Younger's Lane)	Unknown	
09-010	7 square metres of verge (north of Younger's Lane)	L.J. Fairburn & Son Limited Ivy House Farm Farlesthorpe Road Bilsby ALFORD Lincolnshire LN13 9PL (in respect of rights of way and covenants listed in a Conveyance dated 30 November 2000 & in relation to rights granted in a Conveyance dated 02 March 1981)	
09-012	269 square metres of hedgerow and copse (north of Younger's Lane)	Unknown	
09-015	29 square metres of access track (north of Younger's Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-016	223 square metres of verge and hedgerow (North of Younger's Lane)	Unknown	
09-017	718 square metres of public road and verge (Younger's Lane)	Unknown	
09-018	7608 square metres of agricultural land (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
09-019	729 square metres of agricultural land and verge (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
09-020	1011 square metres of agricultural land and copse (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
09-021	747 square metres of agricultural land (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
09-021 cont'd		Unknown	
09-022	16419 square metres of agricultural land and drains (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
10-001	13692 square metres of agricultural land (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (assumed in respect of rights of access) Unknown	
10-002	20400 square metres of agricultural land, copse and drains (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on		Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
10-003	9204 square metres of agricultural land (west of Middlemarsh Road)	Annie Ruth Dalrymple	
		The Chestnuts	
		Wainfleet Road	
		Burgh Le Marsh	
		SKEGNESS	
		Lincolnshire	
		PE24 5AH	
		(in respect of rights reserved by Transfer dated 22 December 2009)	
		Claire Margaret Wilson	
		The Chestnuts	
		Wainfleet Road	
		Burgh Le Marsh	
		SKEGNESS	
		Lincolnshire	
		PE24 5AH	
		(in respect of rights reserved by Transfer dated 22 December 2009)	
		Luke Colin Martin Mackinder	
		The Chestnuts	
		Wainfleet Road	
		Burgh Le Marsh	
		SKEGNESS	
		Lincolnshire	
		PE24 5AH	
		(in respect of rights reserved by Transfer dated 22 December 2009)	
		Mary Ann Mackinder-Jonas	
		The Chestnuts	
		Wainfleet Road	
		Burgh Le Marsh	
		SKEGNESS	
		Lincolnshire	
		PE24 5AH	
		(in respect of rights reserved by Transfer dated 22 December 2009)	
		SKEG Grain Limited	
		Burgh Road West	
		SKEGNESS	
		Lincolnshire	
		PE24 4UF	
		(in respect of rights reserved by Transfer dated 22 December 2009)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on		Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
10-004	9307 square metres of agricultural land, copse and drains (west of Middlemarsh Road)	Annie Ruth Dalrymple	
		The Chestnuts	
		Wainfleet Road	
		Burgh Le Marsh	
		SKEGNESS	
		Lincolnshire	
		PE24 5AH	
		(in respect of rights reserved by Transfer dated 22 December 2009)	
		Claire Margaret Wilson	
		The Chestnuts	
		Wainfleet Road	
		Burgh Le Marsh	
		SKEGNESS	
		Lincolnshire	
		PE24 5AH	
		(in respect of rights reserved by Transfer dated 22 December 2009)	
		Luke Colin Martin Mackinder	
		The Chestnuts	
		Wainfleet Road	
		Burgh Le Marsh	
		SKEGNESS	
		Lincolnshire	
		PE24 5AH	
		(in respect of rights reserved by Transfer dated 22 December 2009)	
		Mary Ann Mackinder-Jonas	
		The Chestnuts	
		Wainfleet Road	
		Burgh Le Marsh	
		SKEGNESS	
		Lincolnshire	
		PE24 5AH	
		(in respect of rights reserved by Transfer dated 22 December 2009)	
		SKEG Grain Limited	
		Burgh Road West	
		SKEGNESS	
		Lincolnshire	
		PE24 4UF	
		(in respect of rights reserved by Transfer dated 22 December 2009)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-005	212 square metres of copse and drain (west of Middlemarsh Road)	Unknown	
10-006	26361 square metres of agricultural land and drains (north of Skegness Road, A158)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (in respect of rights reserved by Conveyance dated 05 February 1898) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (in respect of rights reserved by Conveyance dated 05 February 1898) Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LNS 5TT (in respect of rights reserved by Conveyance dated 05 February 1898) Lindsey Marsh Drainage Board Wellington House Manby Lundsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of rights negrites)	

	Outer Dows	ing Offshore Wind Development Consent Order	
		BOOK OF REFERENCE - PART 2	
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-007	3803 square metres of agricultural land, access tracks and drains (north of Skegness Road,	Andrew Charles Read	
	A158)	Roughton House Farm	
		Moor Lane	
		Roughton	
		WOODHALL SPA	
		Lincolnshire	
		LN10 6YH	
		(in respect of rights reserved by Conveyance dated 05 February 1898)	
		Edward Martin Read	
		Honeyhole	
		Moor Lane	
		Roughton	
		WOODHALL SPA	
		Lincolnshire	
		LN10 6YH	
		(in respect of rights reserved by Conveyance dated 05 February 1898)	
		lan George Read	
		Meadow House	
		Main Road	
		Langton	
		HORNCASTLE	
		Lincolnshire	
		LN9 5JT	
		(in respect of rights reserved by Conveyance dated 05 February 1898)	
		Lindsey Marsh Drainage Board	
		Wellington House	
		Manby Park	
		Manby	
		LOUTH	
		Lincolnshire	
		LN11 8UU	
		(in respect of riparian rights)	
I			

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	County of Lincolnshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-008	2 square metres of drain (north of Skegness Road, A158)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
10-009	29 square metres of access track and drain (north of Skegness Road, A158)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (assumed in respect of rights of access) Unknown	
10-010	880 square metres of agricultural land (north of Skegness Road, A158) and public footpath (LL BurM 265/2)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
10-011	244 square metres of drain (north of Skegness Road, A158)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

r	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
10-011 cont'd		Unknown	
10-015	31 square metres of copse (north of Skegness Road, A158)	Unknown	
10-016	92 square metres of copse (north of Skegness Road, A158)	Unknown	
10-017	2998 square metres of public road and verge (Skegness Road, A158)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
10-018	23231 square metres of agricultural land (south of Skegness Road, A158)	Unknown	
10-019	24721 square metres of agricultural land (south of Skegness Road, A158)	Unknown	
10-020	7 square metres of agricultural land (south of Skegness Road, A158)	Unknown	
10-021	2360 square metres of agricultural land (south of Skegness Road, A158) and public footpath (LL BurM 260/1)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights relating to a water main) Unknown (in respect of rights in Conveyance dated 11 October 1928)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-001	97 square metres of agricultural land (south of Skegness Road, A158)	Unknown	
11-002	163 square metres of agricultural land and drain (south of Skegness Road, A158)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 21 June 2018) Unknown	
11-003	20065 square metres of agricultural land and drains (south of Skegness Road, A158)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 21 June 2018) Unknown	
11-004	958 square metres of agricultural land (south of Skegness Road, A158)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 21 June 2018) Unknown	

	Outer Dow	vsing Offshore Wind Development Consent Order	
		BOOK OF REFERENCE - PART 2	
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-005	1202 square metres of agricultural land (south of Skegness Road, A158)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 2 July 2019) Unknown	
11-006	7646 square metres of agricultural land and hedgerow (west of Middlemarsh Road) and public footpath (LL BurM 261/3)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 2 July 2019) Unknown	
11-008	3 square metres of drain and hedgerow (west of Middlemarsh Road)	Unknown	
11-011	167 square metres of agricultural land (north of Billgate Lane)	Unknown	
11-016	571 square metres of public road and verge (north of Billgate Lane)	Unknown	
11-017	260 square metres of agricultural land (south of Billgate Lane)	Unknown	
11-020	881 square metres of public road and verges (Middlemarsh Road)	Unknown	
11-021	698 square metres of agricultural land and access track (south of Middlemarsh Road)	Unknown	
11-022	110 square metres of access track (south of Middlemarsh Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
11-023	20163 square metres of agricultural land (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of rights granted by Deed dated 04 August 1981)	
12-001	20 square metres of agricultural land (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
12-002	1500 square metres of agricultural land (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
12-003	43 square metres of agricultural land and drain (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-003 cont'd		Unknown	
12-004	240 square metres of agricultural land (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of rights granted by Deed dated 04 August 1981)	
12-005	304 square metres of agricultural land (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
12-006	1271 square metres of agricultural land (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-007	481 square metres of drain (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
12-008	11156 square metres of agricultural land and drains (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
12-009	133 square metres of drain (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
12-010	33143 square metres of agricultural land and drains (north of Low Road)	June Smith Park Farm Main Road Roughton WOODHALL SPA LincoInshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-010 cont'd		Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) The Executor of the Estate of the Late Andrew Leslie Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 GYJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)	
	4704 square metres of private road, verges, agricultural land and access tracks (north of Low Road)	June Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	County of Lincolnshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
12-011 cont'd		The Executor of the Estate of the Late Andrew Leslie Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 GYJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)	
12-013	9271 square metres of private road, verges and agricultural land (north of Low Road)	June Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 GYJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) The Executor of the Estate of the Late Andrew Leslie Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire	

	Outer Dowsing Offshore Wind Development Consent Order				
	BOOK OF REFERENCE - PART 2				
	County of Lincolnshire				
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
12-014	657 square metres of private road and verges (north of Low Road)	June Smith Park Farm Main Road Roughton WOODHALL SPA LincoInshire LN10 GYJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957) Lindsey Marsh Drainage Board Wellington House Manby Park Manby Park Manby LOUTH LincoInshire LN11 &UU (assumed in respect of rights of access) The Executor of the Estate of the Late Andrew Leslie Smith Park Farm Main Road Roughton WOODHALL SPA LincoInshire LN10 GYJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)			
12-015	12 square metres of private road and verge (north of Low Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	County of Lincolnshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
12-016	1295 square metres of agricultural land, drain and verge (north of Low Road)	June Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 GYJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)		
12-017	863 square metres of agricultural land, drain and verge (north of Low Road)	June Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 GYJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)		
12-018	705 square metres of public road and verges (Low Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown		
12-019	167 square metres of agricultural land (south of Low Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown		

	Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	County of Lincolnshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
12-020	25718 square metres of agricultural land and drains (west of Pinchbeck Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)		
13-001	144 square metres of verge (north of Pinchbeck Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown		
13-002	581 square metres of public road, verges and drains (Pinchbeck Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown		
13-003	209 square metres of drain (south of Pinchbeck Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of drainage rights)		

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-003 cont'd		Unknown	
13-004	52355 square metres of agricultural land and drains (south of Pinchbeck Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (in respect of riparian rights) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of rights granted in Wayleave Consent dated 23 June 1969)	
13-005	339 square metres of agricultural land and drain (south of Pinchbeck Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
	931 square metres of private road and verges, agricultural land and access track (Gutheram Drove)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-007	1601 square metres of access track (east of Gutheram Drove)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
13-008	1702 square metres of agricultural land and access track (north of Gutheram Drove)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
13-009	637 square metres of agricultural land and drain (south of Gutheram Drove)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
13-010	433 square metres of agricultural land (north of Croft Bank, A52)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-011	37136 square metres of agricultural land, hedgerow and drains (north of Croft Bank, A52)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
13-014	143 square metres of agricultural land and verge (north of Croft Bank, A52)	Unknown	
13-015	670 square metres of agricultural land (south of Gutheram Drove)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
13-016	33 square metres of agricultural land (south of Gutheram Drove)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
13-017	645 square metres of agricultural land and drains (south of Gutheram Drove) and public footpath (LL Crof 264/1)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
13-018	303 square metres of agricultural land and drain (south of Gutheram Drove)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
13-019	14335 square metres of agricultural land, access track and drain (south of Gutheram Drove) and public footpath (LL Crof 264/1)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
13-020	253 square metres of agricultural land and drain (south of Gutheram Drove)	Unknown	
14-001	66 square metres of agricultural land (south of Gutheram Drove)	Unknown	
14-003	21 square metres of agricultural land (east of Church Lane)	Unknown	
14-004	76 square metres of agricultural land (east of Church Lane)	Unknown	
14-005	60 square metres of agricultural land (east of Church Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
14-006	755 square metres of public road and verges (Church Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
14-007	207 square metres of hedgerow (west of Church Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
14-008	10649 square metres of agricultural land and drains (west of Church Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
14-009	49828 square metres of agricultural land, drains and verge (east of Croft Lane) and public footpath (LL Crof 276/2, LL Crof 276/3 and LL Crof 276/4)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
14-009 cont'd		Unknown (in respect of a restrictive covenant as stated in Deed dated 18 April 1977)	
14-010	2617 square metres of agricultural land and access track (west of Church Lane) and public footpath (LL Crof 276/4)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of a restrictive covenant as stated in Deed dated 18 April 1977)	
14-011	758 square metres of public roads and verges (Croft Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Unknown	
14-012	318 square metres of watercourse (The Lymm)	Unknown	
14-013	1111 square metres of scrubland and hedgerow (west of Croft Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
14-014	777 square metres of public roads and verges (East End)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Unknown	
14-015	62761 square metres of agricultural land and drains (east of Wainfleet Road, B1195)	Unknown	
14-016	317 square metres of agricultural land (west of East End)	Unknown	
15-001	1286 square metres of agricultural land (east of Wainfleet Road, B1195)	Unknown	
15-002	1 square metres of agricultural land (east of Wainfleet Road, B1195)	Unknown	
15-003	1173 square metres of agricultural land (east of Wainfleet Road, B1195)	Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	
15-004	10351 square metres of agricultural land (east of Wainfleet Road, B1195)	Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	
15-005	1901 square metres of access track, agricultural land and drain (north of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	
15-006	227 square metres of drain (east of Wainfleet Road, B1195)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
15-008	42 square metres of verge (north of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
15-009	73 square metres of access track (north of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	
15-010	6 square metres of verge (Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
15-011	1551 square metres of public road, verges and drains (Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
15-012	204 square metres of agricultural land and drain (west of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 &UU (in respect of riparian rights) Unknown	
15-013	17709 square metres of agricultural land (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	

1	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
15-014	49 square metres of agricultural land (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	
15-015	293 square metres of drain (south of Wainfleet Road, B1195)	Unknown	
15-018	47 square metres of access track and drain (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
15-019	11 square metres of access track (south of Wainfleet Road, B1195)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
15-019 cont'd		Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
15-020	5 square metres of access track (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (assumed in respect of rights of access) Unknown	
15-021	49 square metres of access track and drain (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 2	
		County of Lincolnshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
15-022	211 square metres of agricultural land (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)
15-023	148 square metres of agricultural land (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown
15-024	1132 square metres of agricultural land and access track (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)
15-025	903 square metres of agricultural land and access track (south of Wainfleet Road, B1195)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
15-025 cont'd		Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH	
		Lincolnshire LN11 8UU (assumed in respect of rights of access)	
15-026	328 square metres of agricultural land and access track (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
	6 square metres of agricultural land and access track (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
15-028	9 square metres of agricultural land (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
15-029	18 square metres of agricultural land and access track (south of Wainfleet Road, B1195)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
15-030	32 square metres of agricultural land (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
15-031	476 square metres of agricultural land, access track (south of Wainfleet Road, B1195) and drain (Wainfleet Relief Channel)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
15-032	337 square metres of access track (south of Wainfleet Road, B1195) and drain (Wainfleet Relief Channel)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
	354 square metres of agricultural land, access track (south of Wainfleet Road, B1195) and drain (Wainfleet Relief Channel)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
15-034	15 square metres of agricultural land (north of Brewster Lane and south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
15-035	4 square metres of agricultural land (north of Brewster Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
15-036	29 square metres of agricultural land and drain (north of Brewster Lane and south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
15-037	865 square metres of agricultural land (north of Brewster Lane and south of Wainfleet Road, B1195)	Unknown	
15-038	2 square metres of agricultural land (north of Brewster Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
15-039	439 square metres of agricultural land and drain (north of Brewster Lane and south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
15-041	11110 square metres of agricultural land, access tracks (south of Wainfleet Road, B1195) and drains (Wainfleet Relief Channel)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	

	Outer Dows	sing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
15-042	23156 square metres of agricultural land and drain (north of Brewster Lane and south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
15-043	17872 square metres of agricultural land (north of Brewster Lane)	Unknown	
15-044	515 square metres of agricultural land and drain (north of Brewster Lane)	Unknown	
15-045	363 square metres of agricultural land (north of Brewster Lane)	Unknown	
15-046	929 square metres of agricultural land and drain (north of Brewster Lane)	Unknown	
15-047	1355 square metres of public road and verges (Brewster Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
15-048	82 square metres of drain (south of Brewster Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
15-048 cont'd		Unknown	
15-049	92 square metres of agricultural land (south of Brewster Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
15-050	18382 square metres of agricultural land (south of Brewster Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
15-051	1035 square metres of agricultural land (south of Brewster Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
15-052	4600 square metres of agricultural land and drain (south of Brewster Lane and north of Collison Gate)	Unknown	
15-053	1975 square metres of railway (Wainfleet and Boston) and works	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
15-055	172 square metres of railway (Wainfleet and Boston) and works	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown	
16-001	6 square metres of access splay and drain (south of Collison Gate)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
16-002	45 square metres of access splay and drain (south of Collison Gate)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
16-004	172 square metres of drain (south of Collison Gate)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-005	19985 square metres of agricultural land (south of Collison Gate and east of Crow's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
16-006	126 square metres of drain (south of Collison Gate)	Unknown	
16-009	1197 square metres of public road (Mill Lane)	Unknown	
16-011	2231 square metres of river (Steeping River)	Unknown	
	561 square metres of public road and access track (Mill Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
16-014	252 square metres of verge (south of Mill Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
16-015	246 square metres of verge (south of Mill Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
-	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-016	48443 square metres of agricultural land and drain (south of Mill Lane and west of Washdike Lane)	Unknown (in respect of rights contained in Conveyance dated 22 July 1919) Unknown (in respect of a right of way contained in Conveyance dated 06 April 1907) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (in respect of riparian rights)	
16-017	3088 square metres of agricultural land and drain (north of Church Lane and west of Washdike Lane)	Unknown (in respect of rights contained in Conveyance dated 22 July 1919) Unknown (in respect of a right of way contained in Conveyance dated 06 April 1907) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 GPP (in respect of riparian rights)	
16-018	701 square metres of agricultural land (north of Church Lane and west of Washdike Lane)	Unknown (in respect of rights contained in Conveyance dated 22 July 1919) Unknown (in respect of a right of way contained in Conveyance dated 06 April 1907)	
16-019	518 square metres of agricultural land (north of Church Lane and west of Washdike Lane)	Unknown	
16-020	39 square metres of drain (north of Church Lane)	Unknown	
16-022	182 square metres of drain (north of Church Lane)	Unknown	
16-023	120 square metres of drain (north of Church Lane)	Unknown (in respect of rights contained in Conveyance dated 22 July 1919)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-023 cont'd		Unknown (in respect of a right of way contained in Conveyance dated 06 April 1907)	
16-024	13592 square metres of agricultural land (north of Church Lane)	Unknown	
16-025	1638 square metres of public road, verge and copse (Church Lane)	Unknown	
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
	33 square metres of agricultural land (south of Church Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
16-027	8061 square metres of agricultural land and drain (south of Church Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
16-028	212 square metres of agricultural land (south of Church Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2 County of LincoInshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
16-029	6389 square metres of agricultural land (south of Church Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
16-030	436 square metres of drain (south of Church Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
16-031	14513 square metres of agricultural land and path (south of Church Lane)	Unknown (in respect of rights relating to sewers, drains and water pipes listed in Deed of Gift dated 10 April 1992) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (assumed in respect of rights of access)	
	1021 square metres of access track (north of Scald Gate) and public footpath (LL WStM 371/1)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (in respect of riparian rights)	
17-002	234 square metres of drain (south of Church Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-002 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-003	6428 square metres of agricultural land (south of Church Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-004	1 square metres of agricultural land (west of Hall Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-005	98 square metres of agricultural land (south of Church Lane)	Unknown (in respect of rights relating to sewers, drains and water pipes listed in Deed of Gift dated 10 April 1992) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-006	437 square metres of drain (south of Church Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-006 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-007	9897 square metres of agricultural land (south of Church Lane)	Unknown (in respect of rights relating to sewers, drains and water pipes listed in Deed of Gift dated 10 April 1992)	
	28 square metres of drain (south of Church Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
	250 square metres of drain (south of Church Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-010	1334 square metres of agricultural land (west of Hall Gate)	Unknown (in respect of rights of entry listed in Conveyance dated 31 December 1919) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-011	535 square metres of drain (south of Church Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-011 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-012	81 square metres of drain (west of Hall Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-013	11550 square metres of agricultural land (east of Burgh Road and south of Church Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of drainage rights) Unknown (in respect of rights reserved by Conveyance dated 12 July 1919) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-014	10843 square metres of agricultural land (east of Burgh Road)	Sarah Elizabeth Scupholm 18 Rumbold Lane Wainfleet SKEGNESS PE24 4DS (in respect of restrictive covenants as imposed by restriction dated 18 August 2022)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-014 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-015	283 square metres of drain (north of Scald Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-016	12993 square metres of agricultural land and drain (north of Scald Gate)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-017	6306 square metres of drain and copse (north of Scald Gate)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-018	133 square metres of drain (north of Scald Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-019	395 square metres of agricultural land (north of Scald Gate)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-020	27 square metres of agricultural land (north of Scald Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 GPP (assumed in respect of rights of access)	
17-021	165 square metres of drain (north of Scald Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 GPP (in respect of riparian rights)	
17-022	731 square metres of public road and verge (Scald Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (in respect of riparian rights)	
17-023	241 square metres of drain (south of Scald Gate)	Unknown	

1	Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	County of Lincolnshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
17-023 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)		
17-024	15081 square metres of agricultural land (south of Scald Gate)	Unknown (in respect of right of entry as stated by an Indenture dated 09 October 1919) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)		
17-025	10 square metres of access splay (north of Scald Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)		
17-026	384 square metres of drain (south of Scald Gate)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of drainage rights) Unknown		

-	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
17-026 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-028	89 square metres of agricultural land (east of Burgh Road)	Unknown	
17-030	198 square metres of drain and copse (east of Burgh Road)	Unknown	
17-032	17 square metres of drain (east of Burgh Road)	Unknown	
17-033	248 square metres of agricultural land (east of Burgh Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of rights contained in Conveyance dated 16 February 1948)	
17-034	11589 square metres of agricultural land (east of Burgh Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of rights contained in Conveyance dated 16 February 1948)	
17-036	30 square metres of drain (east of Scald Gate)	Unknown	
18-001	759 square metres of public road (Burgh Road)	Unknown	
18-002	163 square metres of drain (west of Burgh Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	County of Lincolnshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
18-004	2998 square metres of agricultural land (west of Burgh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of a right of way and drainage rights)		
18-005	345 square metres of drain (west of Burgh Road)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)		
18-006	5614 square metres of agricultural land (west of Burgh Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (assumed in respect of rights of access)		
18-007	3783 square metres of agricultural land (west of Burgh Road)	John Charles Thomas Winslow Moorfield Burgh Road Friskney BOSTON PE22 8NS (in respect of rights to run services, use of land, entry, support and others listed in Conveyance dated 25 August 1995) Susan Helen Winslow Moorfield Burgh Road Friskney BOSTON PE22 8NS (in respect of rights to run services, use of land, entry, support and others listed in Conveyance dated 25 August 1995)		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2				
	County of Lincolnshire				
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008			
18-007 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (assumed in respect of rights of access)			
18-008	1960 square metres of agricultural land (west of Burgh Road)	John Charles Thomas Winslow Moorfield Burgh Road Friskney BOSTON PE22 8NS (in respect of rights to run services, use of land, entry, support and others listed in Conveyance dated 25 August 1995) Susan Helen Winslow Moorfield Burgh Road Friskney BOSTON PE22 8NS (in respect of rights to run services, use of land, entry, support and others listed in Conveyance dated 25 August 1995) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)			
18-010	189 square metres of drain (west of Burgh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (in respect of drainage rights) Unknown			

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-011	9164 square metres of agricultural land (west of Burgh Road)	Gary Brunt Ty Bach Burgh Road Friskney BOSTON PE22 8NT (in respect of covenants relating to maintenance of a boundary hedge listed in Conveyance dated 25 August 1983)	
18-012	3057 square metres of agricultural land and access track (west of Burgh Road)	Gary Brunt Ty Bach Burgh Road Friskney BOSTON PE22 8NT (in respect of covenants relating to maintenance of a boundary hedge listed in Conveyance dated 25 August 1983)	
18-013	174 square metres of drain (west of Burgh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of drainage rights) Unknown	
18-014	28313 square metres of agricultural land (north of Fen Bank)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of rights contained in Conveyance dated 16 February 1948)	
18-015	3368 square metres of access track (north of Fen Bank and west of Burgh Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of rights contained in Conveyance dated 16 February 1948)	

		BOOK OF REFERENCE - PART 2		
	County of Lincolnshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
18-016 5	548 square metres of watercourse (Fodder Dike Bank)	Unknown (in respect of unknown restrictive covenants which may have been imposed prior to 6 July 2021)		
		Boston & District Angling Association 13 Brand End Road Butterwick BOSTON Lincolnshire PE22 OET (in respect of fishing rights)		
18-017 3	371 square metres of drain (north of Fen Bank)	Unknown (in respect of unknown restrictive covenants which may have been imposed prior to 6 July 2021)		
18-018 1	1517 square metres of public road (Fen Bank)	Unknown		
18-021 2	215 square metres of drain (south of Fen Bank)	Unknown		
18-024 2	211 square metres of agricultural land (east of Burgh Road)	Unknown		
18-026 2	206 square metres of agricultural land (west of Burgh Road)	Unknown		
18-027 5	51 square metres of access splay (west of Burgh Road)	Unknown		
18-030 1	199 square metres of drain (east of Cranberry Lane)	Unknown		
18-031 4	48 square metres of access splay (west of Burgh Road)	Unknown		
18-032 2	22 square metres of access splay (west of Cranberry Lane)	Unknown		

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-032 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
18-033	48 square metres of agricultural land (east of Cranberry Lane)	Unknown (in respect of rights and restrictive covenants stated in an unknown document as may have been imposed before 5 December 2011)	
18-034	10876 square metres of agricultural land (east of Cranberry Lane)	Unknown (in respect of rights and restrictive covenants stated in an unknown document as may have been imposed before 5 December 2011)	
18-035	124 square metres of agricultural land and access track (east of Cranberry Lane)	Unknown (in respect of rights and restrictive covenants stated in an unknown document as may have been imposed before 5 December 2011)	
18-036	543 square metres of agricultural land (west of Cranberry Lane)	Colin Campbell 1 Hudson Crescent LEIGH-ON-SEA SS9 5XT (in respect of provisions and covenants listed in Transfer dated 03 September 2004) Pamela Ann Campbell 1 Hudson Crescent LEIGH-ON-SEA SS9 5XT (in respect of provisions and covenants listed in Transfer dated 03 September 2004) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
18-037	1013 square metres of public road and verge (Cranberry Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
18-038	249 square metres of drain (west of Cranberry Lane)	Unknown	
19-001	9533 square metres of agricultural land (west of Cranberry Lane)	Colin Campbell I Hudson Crescent LEIGH-ON-SEA SS9 5XT (in respect of provisions and covenants listed in Transfer dated 03 September 2004) Pamela Ann Campbell I Hudson Crescent LEIGH-ON-SEA SS9 5XT (in respect of provisions and covenants listed in Transfer dated 03 September 2004) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
19-002	168 square metres of drain (west of Cranberry Lane)	Unknown	
19-003	17041 square metres of agricultural land and access track (north of Mill Hill)	Unknown (in respect of unknown restrictive covenants which may have been imposed prior to 3 April 1996) Unknown (in respect of right to use conduits under the property together with a right of entry to clean, repair and replace as stated in transfer dated 29 March 1996) Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of a right of access and a right to lay, maintain and repair electricity apparatus as stated in Agreement dated 26 April 2024)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
19-004	194 square metres of copse (north of Mill Hill) (excluding all interests of the Crown)	Unknown	
19-005	2608 square metres of agricultural land (north of Mill Hill)	Unknown (n respect of unknown restrictive covenants which may have been imposed prior to 3 April 1996)	
		Unknown (in respect of right to use conduits under the property together with a right of entry to clean, repair and replace as stated in transfer dated 29 March 1996)	
		Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA	
		(in respect of a right of access and a right to lay, maintain and repair electricity apparatus as stated in Agreement dated 26 April 2024)	
19-006	25424 square metres of agricultural land (north of Mill Hill) (excluding all interests of the Crown)	SG Kleinwort Hambros Bank Limited One Bank Street Canary Wharf LONDON E14 4SG (in respect of rights listed in Conveyance dated 29 September 1987)	
19-007	60 square metres of agricultural land (north of Mill Hill) (excluding all interests of the Crown)	SG Kleinwort Hambros Bank Limited One Bank Street Canary Wharf LONDON E14 4SG (in respect of rights listed in Conveyance dated 29 September 1987)	
19-008	181 square metres of drain (north of Mill Hill) (excluding all interests of the Crown)	Unknown	
19-009	901 square metres of public road (Mill Hill) (excluding all interests of the Crown)	Unknown	
19-011	246 square metres of drain (south of Mill Hill and north of Church Lane)	Unknown	
19-012	9 square metres of agricultural land (south of Mill Hill and east of Small End Road)	Unknown	

	Outer Dows	ing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
19-014	249 square metres of drain and copse (south of Mill Hill and east of Small End Road)	Unknown	
19-016	155 square metres of drain and copse (south of Mill Hill and east of Small End Road)	Unknown	
19-017	44 square metres of drain (south of Mill Hill and east of Small End Road)	Unknown	
19-019	153 square metres of drain and copse (south of Mill Hill and east of Small End Road)	Unknown	
19-020	4818 square metres of agricultural land (east of Small End Road and north of Church Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
19-021	174 square metres of agricultural land (east of Small End Road and north of Church Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
19-022	1367 square metres of public road (Small End Road)	Unknown	
19-025	16703 square metres of agricultural land (north of Skirmore Road)	Unknown	
20-001	5695 square metres of agricultural land (north of Skirmore Road)	Unknown	
20-002	42 square metres of agricultural land and drain (north of Skirmore Road)	Unknown	
20-005	1140 square metres of public road, verges and drain (Skirmore Road)	Unknown	
20-006	161 square metres of drain (south of Skirmore Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
20-007	149 square metres of access splay (south of Skirmore Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
20-008	51881 square metres of agricultural land (south of Skirmore Road and east of Patman's Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
20-009	99 square metres of agricultural land (south of Howgarth Lane)	Unknown	
21-001	32 square metres of access splay (west of Patman's Lane)	Unknown	
21-002	1123 square metres of agricultural land (west of Patman's Lane)	Roughton Farms Limited The Grange Eaudyke Road Friskney BOSTON Lincolnshire PE22 8RT (in respect of the rights reserved by conveyance dated 27 November 1975 and right to use drainage systems as stated in conveyance dated 28 October 2009)	
21-003	1418 square metres of public road and verges (Patman's Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 GPP (assumed in respect of rights of access)	
21-004	173 square metres of drain (west of Patman's Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
	35215 square metres of agricultural land (west of Patman's Lane)	Roughton Farms Limited The Grange Eaudyke Road Friskney BOSTON Lincolnshire PE22 8RT (in respect of the rights reserved by conveyance dated 27 November 1975 and right to use drainage systems as stated in conveyance dated 28 October 2009)	
21-006	21128 square metres of agricultural land (east of Ivery Lane)	Unknown	
21-007	11 square metres of access splay (east of Ivery Lane)	Unknown	
21-008	1162 square metres of public road, verges and drain (Ivery Lane)	Unknown	
21-011	19862 square metres of agricultural road and drain (west of Ivery Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
22-001	6053 square metres of agricultural land and drain (west of Ivery Lane) (excluding all interests of the Crown)	Unknown (in respect of rights granted by Conveyance dated 29 August 1987) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
22-002	12 square metres of drain (west of Ivery Lane) (excluding all interests of the Crown)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
22-002 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
22-003	32 square metres of agricultural land and drain (west of Ivery Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 GPP (assumed in respect of rights of access)	
22-004	7797 square metres of agricultural land (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	Unknown (in respect of rights granted by Conveyance dated 29 August 1987)	
22-005	117 square metres of drain (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	Unknown	
22-006	29 square metres of agricultural land (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	Unknown (in respect of rights granted by Conveyance dated 29 August 1987)	
22-007	9 square metres of drain (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	Unknown	
22-008	41 square metres of agricultural land (west of Ivery Lane and south of Love Lane)	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 0DW (in respect of rights reserved in Transfer dated 22 February 1995) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	County of Lincolnshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
22-009	9803 square metres of agricultural land and copse (south of Love Lane)	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 0DW (in respect of rights reserved in Transfer dated 22 February 1995) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
22-010	1644 square metres of agricultural land (east of Broad Gate)	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 0DW (in respect of rights reserved in Transfer dated 22 February 1995) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
22-011	150 square metres of drain (south of Love Lane)	Unknown	
22-013	381 square metres of drain (south of Love Lane)	Unknown	
22-015	15 square metres of drain (south of Love Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
22-019	8 square metres of drain (east of Broad Gate)	Unknown	
22-020	1306 square metres of agricultural land (east of Broad Gate)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
22-021	825 square metres of drain (east of Broad Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
22-022	948 square metres of agricultural land (east of Broad Gate)	Unknown	
22-024	814 square metres of agricultural land (east of Broad Gate)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
22-025	15 square metres of access track (east of Broad Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
22-026	12419 square metres of agricultural land (east of Broad Gate)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
22-027	538 square metres of public road and verges (Broad Gate)	Unknown	
22-028	13791 square metres of agricultural land (west of Broad Gate)	Unknown	
22-029	2011 square metres of agricultural land (south of Cragmire Lane and west of Broad Gate)	Unknown	
22-032	2201 square metres of agricultural land and access track (south of Cragmire Lane)	Unknown (in respect of restrictive covenants as may have been imposed thereon before 20 July 2010)	
23-002	179 square metres of drain (east of Cragmire Lane)	Unknown	
23-004	13 square metres of drain (east of Cragmire Lane)	Unknown	
23-007	20 square metres of drain (south of Cragmire Lane)	Unknown	
23-009	1940 square metres of agricultural land and access track (east of Cragmire Lane)	Christopher William Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (in respect of rights reserved by Transfer dated 10 November 1995) Jane Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (in respect of rights reserved by Transfer dated 10 November 1995)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
23-009 cont'd		John Frank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (in respect of rights reserved by Transfer dated 10 November 1995) Robert John Edwards The Mallards Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (in respect of rights reserved by Transfer dated 10 November 1995)	
23-010	141 square metres of drain (east of Cragmire Lane)	Unknown	
23-011	847 square metres of public road and verges (Cragmire Lane)	Unknown	
23-012	12013 square metres of agricultural land (west of Cragmire Road)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
23-013	12266 square metres of agricultural land (north of Cragmire Lane and east of Double Bank) (excluding all interests of the Crown)	Unknown (in respect of rights granted by Conveyance dated 29 September 1987)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
23-013 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
	1482 square metres of agricultural land (north of Cragmire Lane)(excluding all interests of the Crown)	Unknown (in respect of rights granted by Conveyance dated 29 September 1987) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 GPP (assumed in respect of rights of access)	
23-015	218 square metres of drain (east of Double Bank) (excluding all interests of the Crown)	Unknown	
23-016	14153 square metres of agricultural land (east of Common Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
23-017	4615 square metres of drain (east of Double Bank)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
23-018	62 square metres of access splay (east of Double Bank)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
23-018 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
23-019	583 square metres of drain (east of Common Road)	Unknown	
23-020	1138 square metres of public road, drain and verges (Common Road)	Unknown	
23-024	13221 square metres of agricultural land (west of Common Road)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
23-025	604 square metres of agricultural land (west of Common Road)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
23-026	34 square metres of agricultural land (west of Common Road)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
23-027	9538 square metres of agricultural land (west of Common Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
23-028	31 square metres of agricultural land (west of Common Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
23-029	83 square metres of access splay (west of Common Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
23-030	28270 square metres of agricultural land and drain (east of Manor Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
23-032	396 square metres of public road and verges (Manor Lane)	Unknown	
24-001	723 square metres of agricultural land (west of Manor Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	County of Lincolnshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
24-002	25766 square metres of agricultural land (east of Sea Dyke and west of Manor Lane)	Unknown	
24-003	969 square metres of public road, drain and verges (Sea Dyke)	Unknown	
24-004	179 square metres of access splay (west of Sea Dyke)	Unknown	
	52453 square metres of agricultural land, hedgerow, access track and drain (west of Sea Dyke)	Unknown	
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
24-006	1253 square metres of access track and drain (west of Sea Dyke)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 GPP (assumed in respect of rights of access)	
24-007	2588 square metres of agricultural land and access track (west of Seadyke)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
24-008	705 square metres of agricultural land (east of Church Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
24-009	705 square metres of access track (east of Church Road)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
24-010	20630 square metres of agricultural land, access track and drain (south of Fold Hill)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
24-011	335 square metres of access track (south of Fold Hill)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
24-012	1913 square metres of access track and drain (south of Fold Hill)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
25-001	8 square metres of agricultural land (east of Church Road)	Unknown	

BOOK OF REFERENCE - PART 2 County of Lincolnshire Number on Land Plans Description of Land Potential claims under \$10 Compulsor 25-003 627 square metres of public road (Church Road) Unknown 25-004 4516 square metres of agricultural land (west of Church Road) Unknown 25-006 1218 square metres of public road (Sibsey Road, B1184) Unknown 25-006 1218 square metres of public road (Sibsey Road, B1184) Unknown 25-007 158 square metres of drain (west of Sibsey Road, B1184) Unknown 25-007 158 square metres of drain (west of Sibsey Road, B1184) Unknown	Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
Number on Land Plans Description of Land Potential claims under S10 Compulsor 25-003 627 square metres of public road (Church Road) Unknown 25-004 4516 square metres of agricultural land (west of Church Road) Unknown 25-006 1218 square metres of public road (Sibsey Road, B1184) Unknown 25-006 1218 square metres of public road (Sibsey Road, B1184) Unknown 25-007 158 square metres of drain (west of Sibsey Road, B1184) Unknown 25-007 158 square metres of drain (west of Sibsey Road, B1184) Unknown	
Land Plans Description of Land 25-003 627 square metres of public road (Church Road) Unknown 25-004 4516 square metres of agricultural land (west of Church Road) Unknown 25-006 1218 square metres of public road (Sibsey Road, B1184) Unknown 25-006 1218 square metres of public road (Sibsey Road, B1184) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access) 25-007 158 square metres of drain (west of Sibsey Road, B1184) Unknown	
25-004 4516 square metres of agricultural land (west of Church Road) Unknown 25-006 1218 square metres of public road (Sibsey Road, B1184) Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access) 25-007 158 square metres of drain (west of Sibsey Road, B1184) Unknown	
25-006 1218 square metres of public road (Sibsey Road, B1184) Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access) 25-007 158 square metres of drain (west of Sibsey Road, B1184) Unknown Witham Fourth District Internal Drainage Board Witham Fourth District Internal Drainage Board	
25-007 158 square metres of drain (west of Sibsey Road, B1184) Witham Fourth District Internal Drainage Board	
47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access) 25-007 158 square metres of drain (west of Sibsey Road, B1184) Unknown Witham Fourth District Internal Drainage Board	
Witham Fourth District Internal Drainage Board	
BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
25-009 784 square metres of public road and verge (Cowbroads Lane) Unknown	
25-010 13742 square metres of agricultural land (south of The Gride) Unknown	
25-018 234 square metres of agricultural land (north of Pode Lane) Unknown	
25-020 368 square metres of agricultural land (north of Pode Lane) Unknown	
25-022 836 square metres of public road and verge (Pode Lane) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
25-022 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
25-023	408 square metres of agricultural land (south of Pode Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (assumed in respect of rights of access)	
25-024	3015 square metres of agricultural land (south of Pode Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
25-025	1376 square metres of agricultural land and drain (south of Pode Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
25-026	81 square metres of drain (south of Pode Lane)	Unknown	
25-027	11 square metres of agricultural land (south of Pode Lane)	Unknown	
25-028	68 square metres of access track (south of Pode Lane)	Unknown	

1	Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	County of Lincolnshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
25-028 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)		
25-029	18 square metres of agricultural land (south of Pode Lane)	Unknown (in respect of restrictive covenants as may have been imposed thereon before 20 July 2010) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)		
25-030	21731 square metres of agricultural land (south of Pode Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)		
	12131 square metres of agricultural land and drain (south of Pode Lane and east of Skipmarsh Lane)	Unknown (in respect of restrictive covenants as may have been imposed thereon before 20 July 2010) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)		
25-032	7515 square metres of agricultural land and drain (east of Skipmarsh Lane)	Unknown		
26-001	271 square metres of agricultural land (east of Skipmarsh Lane)	Unknown		

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
26-003	758 square metres of public road and verges (Skipmarsh Lane)	Unknown	
26-004	153 square metres of agricultural land (west of Skipmarsh Lane)	Unknown	
26-005	24344 square metres of agricultural land (east of Southfields and west of Skipmarsh Lane)	Unknown	
26-006	1067 square metres of public road and verges (Southfields)	Unknown	
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
26-007	88 square metres of access track and drain (west of Southfields)	Unknown	
26-008	14 square metres of access track and drain (west of Southfields)	Unknown	
26-009	3900 square metres of agricultural land (west of Southfields)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
26-010	318 square metres of drain (west of Southfields)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
26-011	16555 square metres of agricultural land and drain (west of Southfields)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (in respect of riparian rights)	
26-013	827 square metres of public road and verges (Ings Drove)	Unknown	
26-014	143 square metres of drain (south of Ings Drove)	 B.S. Pension Fund Trustee Limited 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (in respect of covenants relating to use of land listed in Transfer dated 07 April 1998) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of rights stated in Conveyance dated 18 December 1981) 	
26-015	321 square metres of agricultural land (south of Ings Drove)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (in respect of riparian rights)	
26-016	389 square metres of agricultural land (south of Ings Drove)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

1		Outer Dowsing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
26-017	56 square metres of drain (south of Ings Drove)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
26-018	37348 square metres of agricultural land (south of Ings Drove)	 B.S. Pension Fund Trustee Limited 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (in respect of covenants relating to use of land listed in Transfer dated 07 April 1998) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of rights stated in Conveyance dated 18 December 1981) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights) 	
26-019	80 square metres of agricultural land (east of Ings Road)	Unknown	
26-020	8 square metres of agricultural land (east of Ings Road)	Unknown	
26-021	291 square metres of agricultural land (east of Ings Road)	Unknown	
26-022	6766 square metres of agricultural land (east of Ings Road)	Unknown	
26-023	184 square metres of drain (east of Ings Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
26-024	18921 square metres of agricultural land and drain (east of Ings Road)	Unknown	
26-025	9084 square metres of agricultural land (east of Ings Road)	Unknown	
27-001	898 square metres of public road and verge (Ings Road)	Unknown	
27-007	10 square metres of drain (west of Ings Road)	Unknown	
27-008	71 square metres of drain (west of Ings Road)	Unknown	
27-009	581 square metres of drain (west of Ings Road)	Unknown	
27-011	6902 square metres of agricultural land and drain (west of Ings Road and east of Ings Drove)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-012	1197 square metres of agricultural land (west of Ings Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-013	2480 square metres of agricultural land (east of Ings Drove)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
27-014	8 square metres of access splay (east of Ings Drove)	Unknown	

2	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
27-014 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
27-015	8030 square metres of agricultural land and drain (east of Ings Drove)	Unknown (in respect of easements as affect the same by virtue of Section 15(1)(b) of the Endowments and Glebe Measure 1976) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-016	52 square metres of drain (east of Ings Drove)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 GPP (in respect of riparian rights)	
27-017	25 square metres of agricultural land (east of Ings Drove)	Unknown (in respect of easements as affect the same by virtue of Section 15(1)(b) of the Endowments and Glebe Measure 1976) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	County of Lincolnshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
27-018	1406 square metres of agricultural land and drain (east of Ings Drove)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-019	6579 square metres of agricultural land and drain (east of Ings Drove)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
27-020	1789 square metres of agricultural land and access track (north of Ings Bank)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-021	13605 square metres of agricultural land (north of Ings Bank)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-022	12 square metres of access track (north of Ings Bank)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-023	169 square metres of drain (north of Ings Bank)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
27-024	1211 square metres of public road (Ings Bank)	Unknown	
27-025	21 square metres of drain (south of Ings Bank and west of Double Bank)	Unknown	
28-001	257 square metres of drain (west of Double Bank)	Unknown	
28-003	11508 square metres of agricultural land (west of Double Bank)	Unknown (in respect of rights that granted by Conveyance dated 27 April 1979) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
28-004	6019 square metres of agricultural land (west of Double Bank)	Unknown (in respect of rights that granted by Conveyance dated 27 April 1979) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
28-005	211 square metres of drain (west of Double Bank)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
28-007	20289 square metres of agricultural land (east of Hobhole Bank)	Unknown	

1	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
28-008	1766 square metres of agricultural land (north of Lowfields Lane)	Unknown	
28-009	497 square metres of agricultural land (north of Lowfields Lane)	Unknown	
28-012	185 square metres of drain (north of Lowfields Lane)	Unknown	
28-017	629 square metres of public road and verge (Lowfields Lane)	Unknown	
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
28-018	167 square metres of hedgerow and drain (south of Lowfields Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
28-019	74304 square metres of agricultural land (Oak House Farm)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
29-001	327 square metres of access track (Oak House Farm)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
29-002	41 square metres of access splay and drain (west of Swandyke Lane)	Unknown	
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
29-003	676 square metres of agricultural land (Swandyke Farm)	Witham Fourth District Internal Drainage Board	
		47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
29-004	515 square metres of hedgerow and drain (west of Swandyke Lane)	Unknown	
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
29-009	864 square metres of public road, verges and footways (Wainfleet Road, A52)	Unknown	
29-010	512 square metres of hedgerow (south of Wainfleet Road, A52)	Unknown	
29-011	250 square metres of hedgerow (south of Wainfleet Road, A52)	Unknown	
30-004	28 square metres of access splay (west of Foxhole Lane)	Unknown	
30-006	207 square metres of access splay and drain (west of Foxhole Lane)	Unknown	

·	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
30-007	135 square metres of access splay and drain (west of Foxhole Lane)	Unknown	
30-008	915 square metres of public road and verge (Foxhole Lane)	Unknown	
30-016	168 square metres of drain (north of Butterwick Road)	Unknown	
30-019	73 square metres of access track and drain (north of Butterwick Road)	Unknown	
30-020	145 square metres of drain (north at Butterwick Road)	Unknown	
30-021	724 square metres of public road and verges (Butterwick Road)	Unknown	
30-022	315 square metres of access track and drain (south of Butterwick Road)	Unknown	
30-024	25374 square metres of agricultural land (north of Shore Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
31-001	217 square metres of access splay and drain (north of Shore Road)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 GPP (assumed in respect of rights of access)	
31-002	725 square metres of public road, verges and footways (Shore Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
31-002 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
31-004	62198 square metres of agricultural land, hedgerow, access track and drain (east of Church End Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
31-005	6 square metres of agricultural land (east of Church End Road)	Unknown	
31-006	1065 square metres of public road and verge (Church End Road)	Unknown	
31-007	62 square metres of agricultural land (west of Church End Road)	Unknown	
31-009	906 square metres of public road, verges and drain (Clampgate Road)	Unknown	
32-003	10676 square metres of agricultural land (east of Hobhole Drain)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
32-004	7717 square metres of agricultural land (east of Grovefield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
32-005	354 square metres of verge (Grovefield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
32-006	328 square metres of public road and verge (Grovefield Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 GPP (in respect of riparian rights)	
32-008	5401 square metres of agricultural land (west of Grovefield Lane)	Unknown (in respect of rights stated in Conveyance dated 26 March 1981) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
	268 square metres of drain (west of Grovefield Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
32-010	13 square metres of drain (west of Grovefield Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
32-011	8076 square metres of agricultural land and drain (west of Grovefield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
32-012	646 square metres of agricultural land (west of Grovefield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
32-013	11 square metres of access splay (west of Grovefield Lane)	Unknown	
32-022	68 square metres of agricultural land (west of Grovefield Lane)	Unknown	
32-024	9 square metres of access splay (west of Grovefield Lane)	Unknown	
33-002	261 square metres of drain (north of Cut End Road)	Unknown	
33-003	24317 square metres of agricultural land and access track (east of Cut End Road)	Beverley Elizabeth Lister 2 Marsh Farm Cut End Road Fishtoft BOSTON PE22 OQZ (in respect of service media rights and a right to enter as stated in Transfers dated 20 August 1999 and 14 June 2001)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	County of Lincolnshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
33-003 cont'd		Robert John Lister c/o Beverley Elizabeth Lister 2 Marsh Farm Cut End Road Fishtoft BOSTON Lincolnshire PE22 OQZ (in respect of service media rights and a right to enter as stated in Transfers dated 20 August 1999 and 14 June 2001)	
33-007	48 square metres of access splay (east of Cut End Road)	Unknown	
33-009	134 square metres of verge (east of Cut End Road)	Unknown	
33-010	677 square metres of public road, verge and drain (Cut End Road)	Unknown	
33-011	4507 square metres of agricultural land (east of Woad Lane and west of Cut End Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
33-012	10986 square metres of agricultural land (east of Woad Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
33-013	1394 square metres of agricultural land (east of Woad Lane)	Unknown (in respect of restrictive covenants dated 13 April 2007 stated in Register dated 20 October 2009)	
33-014	941 square metres of public road and verge (Woad Lane)	Unknown	
33-016	15353 square metres of watercourse (Hobhole Drain) and copse (west of Woad Lane)	Unknown (in respect of unknown restrictive covenants which may have been imposed prior to 30 March 2021)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
33-016 cont'd		Boston & District Angling Association 13 Brand End Road Butterwick BOSTON Lincolnshire PE22 0ET (in respect of fishing rights)	
33-017	82 square metres of drain (west of Pinfold Lane)	Unknown	
33-018	259 square metres of drain (north of Pinfold Lane)	Unknown	
	708 square metres of agricultural land, access track and verge (south of Pinfold Lane and east of Southfield Lane)	Unknown	
33-020	785 square metres of access track (south of Pinfold Lane and east of Southfield Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
33-021	1491 square metres of agricultural land (south of Pinfold Lane and east of Southfield Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

r	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
33-022	45 square metres of agricultural land (south of Pinfold Lane)	Mark Bolland The White House Pinfold Lane Fishtoft BOSTON Lincolnshire PE21 0SL (in respect of drainage rights granted by a Transfer dated 24 November 2015) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
33-024	262 square metres of agricultural land (south of Pinfold Lane and east of Southfield Lane)	Unknown	
33-027	239 square metres of drain (east of Southfield Lane)	Unknown	
33-029	4652 square metres of agricultural land (east of Southfield Lane)	Unknown	
33-030	859 square metres of agricultural land (east of Southfield Lane)	Unknown	
33-032	2345 square metres of agricultural land, access track and drain (east of Southfield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
33-033	3 square metres of agricultural land (east of Southfield Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
33-034	12 square metres of agricultural land (east of Southfield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of a right of access to maintain and repair drains)	
33-035	485 square metres of drain (east of Southfield Lane)	Unknown	
33-036	7456 square metres of agricultural land (east of Southfield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
33-037	8666 square metres of agricultural land and hedgerow (east of Southfield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of a right of access to maintain and repair drains)	
33-039	9165 square metres of agricultural land and access track (south of Southfield Lane)	Betty Skipworth Nevadun Burton Croft Road Fishtoft BOSTON Lincolnshire PE21 OSA (in respect of covenants and rights reserved by Transfer dated 11 December 1995) Mark William Skipworth Old Orchard Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HX (in respect of covenants and rights reserved by Transfer dated 11 December 1995)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
33-039 cont'd 33-040	58974 square metres of agricultural land and access track (east of Scalp Road) and watercourse (The Graft)	Richard Hardy (Fishtoft) Willoughby House Willoughby Hills Fishtoft BOSTON PE21 9SA (in respect of covenants and rights reserved by Transfer dated 11 December 1995) The Executor of the Estate of the Late William Garfield Skipworth Nevadun Burton Croft Road Fishtoft BOSTON Lincolnshire PE21 0SA (in respect of covenants and rights reserved by Transfer dated 11 December 1995) Unknown (in respect of covenants and rights reserved by Transfer dated 11 December 1995) Unknown (in respect of a right of way as stated in Conveyance dated 09 May 1968) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of rights stated in Wayleave Agreement dated 17 September 1956)	
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
34-001	2500 square metres of access track (east of Scalp Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
34-002	18 square metres of copse (east of Scalp Road)	Unknown	
	2217 square metres of private road, verge and hardstanding (Scalp Road) and public footpath (LL Fish 13/11)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Unknown	
	7342 square metres of grassed area, shrubbery (west of Scalp Road) and public footpath (LL Fish 13/11)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Unknown	
	1809 square metres of river (The Haven), foreshore, bed and banks thereof (excluding all interests of the Crown)	Unknown (in respect of rights reserved by Deed of Grant dated 25 July 1973)	
	19799 square metres of river (The Haven), foreshore, bed and banks thereof (excluding all interests of the Crown)	Boston Borough Council Municipal Buildings West Street BOSTON Lincolnshire PE21 8QR (in respect of rights granted by Deed of Grant dated 30 October 1936)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
34-006 cont'd		The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Deed of Grant dated 04 September 1964)	
34-007	104860 square metres of agricultural land and drain (Bleak House Farm)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 SAH (assumed in respect of rights of access) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Conveyance dated 15 April 1966) The Royal Society For The Protection Of Birds The Lodge Potton Road SANDY Bedfordshire SG19 2DL (in respect of rights granted by Deed of Grant dated 14 March 2008) Unknown (in respect of restrictive convenant not to build or erect anything that might damage the jetty work of the Haven of Boston or the ground or foreshore in favour of the Borough of Boston as stated in conveyance dated 1 March 1965)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
	1296 square metres of agricultural land, access track, copse and outbuildings (Bleak House Farm)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Conveyance dated 15 April 1966) Unknown (in respect of restrictive convenant not to build or erect anything that might damage the jetty work of the Haven of Boston or the ground or foreshore in favour of the Borough of Boston as stated in conveyance dated 1 March 1965)	
34-009	5554 square metres of agricultural land (Bleak House Farm)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Conveyance dated 15 April 1966) Unknown (in respect of restrictive convenant not to build or erect anything that might damage the jetty work of the Haven of Boston or the ground or foreshore in favour of the Borough of Boston as stated in conveyance dated 6 August 1862) Unknown (in respect of all rights as now exist over the outmarsh and adjoining bank as stated in conveyance dated 1 March 1965)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
34-010	6824 square metres of agricultural land and access track (Bleak House Farm)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights granted by Deed of Grant dated 14 March 2008) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Conveyance dated 15 April 1966) The Royal Society For The Protection Of Birds The Lodge Potton Road SANDY Bedfordshire SG19 2DL (in respect of rights granted by Deed of Grant dated 14 March 2008)	
34-011	600 square metres of access track and copse (east of Wyberton Road)	Unknown (in respect of restrictive convenant not to build or erect anything that might damage the jetty work of the Haven of Boston or the ground or foreshore in favour of the Borough of Boston as stated in conveyance dated 6 August 1862) Unknown (in respect of all rights as now exist over the outmarsh and adjoining bank as stated in conveyance dated 1 March 1965) A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (in respect of rights to an accessway listed in Deed of Easement dated 18 June 2013)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
34-011 cont'd		Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights stated in Transfer Scheme document dated 1 September 1989) Lincolnshire Wildlife Trust Banovallum House Manor House Street HORNCASTLE Lincolnshire Ling 5HF (in respect of a right to pass and repass stated in Conveyance dated 12 March 1986) Unknown (in respect of a right to pass and repass stated in Conveyance dated 30 July 1884)	
34-012	133 square metres of access track, verge and copse (east of Wyberton Road) and public footpath (LL Wybe 2/4)	Unknown	
34-013	120 square metres of copse (east of Wyberton Road)	Unknown	
34-014	1259 square metres of agricultural land and copse (east of Wyberton Road)	Unknown	
34-015	1105 square metres of public road and verges (Wyberton Road)	Unknown	
34-016	31 square metres of public road and verge (Wyberton Road)	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
34-016 cont'd		Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983)	
		Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	
34-017	51 square metres of agricultural land (west of Wyberton Road)	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right or other easements (if any) as stated in conveyance dated 8 April 1983)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
34-018	2116 square metres of agricultural land (west of Wyberton Road)	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	
34-019	147 square metres of drain and copse (west of Wyberton Road)	Unknown	
34-020	56 square metres of agricultural land and access track (west of Wyberton Road)	Unknown	
34-021	140 square metres of agricultural land (west of Wyberton Road)	Unknown	
34-022	6870 square metres of agricultural land (west of Wyberton Road)	Kim Lesley Sermon The Ship Wyberton Roads Wyberton BOSTON LincoInshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
34-022 cont'd		Lesley Jean Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Nicholas Alexander Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)	
	99628 square metres of agricultural land, hedgerows, copse and drains (north of Frampton Road)	Kim Lesley Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Lesley Jean Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
34-023 cont'd		National Grid Electricity Distribution (East Midlands) PLCAvonbankFeeder RoadBNISTOLAvonBS2 0TB(in respect of rights granted by Deed dated 19 August 2004)Nicholas Alexander SermonThe ShipWyberton RoadsWybertonBOSTONLincolnshirePE20 1BA(in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)The Black Sluice Internal Drainage BoardNorth End Business ParkStation RoadSwinesheadBOSTONLincolnshirePE20 3PW(in respect of riparian rights)Unknown(in respect of riphts stated in Conveyance dated 11 January 1984)	
34-024	7288 square metres of agricultural land (west of Wyberton Road)	Kim Lesley Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
34-024 cont'd		Lesley Jean Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of rights granted by Deed dated 19 August 2004) Nicholas Alexander Sermon The Ship Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987) Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987) Unknown (in respect of rights stated in Conveyance dated 24 June 1987)	
34-025	26 square metres of copse (north of Wyberton Road)	Unknown	
35-002	49 square metres of access track (east of Wyberton Road)	Unknown	

		Outer Dowsing Offshore Wind Development Consent Order	
		BOOK OF REFERENCE - PART 2	
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
35-004	897 square metres of agricultural land (east of Wiley's Lane)	Kim Lesley Sermon	
		The Ship	
		Wyberton Roads	
		Wyberton	
		BOSTON	
		Lincolnshire	
		PE20 1BA	
		(in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)	
		Lesley Jean Sermon	
		The Ship	
		Wyberton Roads	
		Wyberton	
		BOSTON	
		Lincolnshire	
		PE20 1BA	
		(in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)	
		Michelle Wright	
		Marsh Farm Cottage	
		Frampton Roads	
		Frampton	
		BOSTON	
		Lincolnshire	
		PE20 1AY	
		(in respect of rights granted by Transfer dated 19 February 2003)	
		Neil Wright	
		Marsh Farm Cottage	
		Frampton Roads	
		Frampton	
		BOSTON	
		Lincolnshire	
		PE20 1AY	
		(in respect of rights granted by Transfer dated 19 February 2003)	

·	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
35-004 cont'd		Nicholas Alexander Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)	
35-005	7 square metres of access track and drain (south of Wiley's Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown	
35-006	547 square metres of access track (Wiley's Lane)	A.E. Lenton Limited Church End Friskney BOSTON LincoInshire PE22 8NE (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
35-006 cont'd 35-007	15 square metres of access track (south of Wiley's Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) The Black Sluice Internal Drainage Board	
		North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown	
35-008	1347 square metres of agricultural land and drain (west of Wiley's Lane)	Kim Lesley Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Lesley Jean Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
35-008		Michelle Wright	
cont'd		Marsh Farm Cottage	
		Frampton Roads	
		Frampton	
		BOSTON	
		Lincolnshire	
		PE20 1AY	
		(in respect of rights granted by Transfer dated 19 February 2003)	
		Neil Wright	
		Marsh Farm Cottage	
		Frampton Roads	
		Frampton	
		BOSTON	
		Lincolnshire	
		PE20 1AY	
		(in respect of rights granted by Transfer dated 19 February 2003)	
		Nicholas Alexander Sermon	
		The Ship	
		Wyberton Roads	
		Wyberton	
		BOSTON	
		Lincolnshire	
		PE20 1BA	
		(in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)	
		The Black Sluice Internal Drainage Board	
		North End Business Park	
		Station Road	
		Swineshead	
		BOSTON	
		Lincolnshire	
		PE20 3PW	
		(in respect of riparian rights)	
36-002	433 square metres of agricultural land and access track (east of Millfield Lane East)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
36-003	33201 square metres of agricultural land (north of Millfield Lane East)	W Dennis & Sons Ltd Kirton House Kirton BOSTON PE20 1JD (in respect of rights and restrictive covenants stated in Conveyance dated 13th April 1948)	
36-004	289 square metres of agricultural land (north of Millfield Lane East)	W Dennis & Sons Ltd Kirton House Kirton BOSTON PE20 1JD (in respect of rights and restrictive covenants stated in Conveyance dated 13th April 1948)	
36-005	34 square metres of agricultural land (west of Low Road)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of restrictions with the Endowments and Glebe Measure 1976 or some other Measure or authority)	
36-006	44 square metres of drain (west of Low Road)	Unknown	
36-007	20 square metres of access track (south of Millfield Lane East)	Unknown	
36-008	92 square metres of access track (south of Millfield Lane East)	W Dennis & Sons Ltd Kirton House Kirton BOSTON PE20 1JD (in respect of rights and restrictive covenants stated in Conveyance dated 13th April 1948)	
37-001	625 square metres of public road (Frampton Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)	

-	Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	County of Lincolnshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
37-001 cont'd		Unknown		
37-002	264 square metres of copse (south of Frampton Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON LincoInshire PE20 3PW (assumed in respect of rights of access) Unknown		
37-003	10408 square metres of agricultural land and access track (south of Frampton Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown (in respect of rights granted by Conveyance dated 11 January 1984)		
37-004	196 square metres of drain (south of Frampton Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown		

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
37-005	41 square metres of access splay (south of Frampton Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown	
37-006	1486 square metres of access track (south of Frampton Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)	
37-007	11 square metres of agricultural land and drain (south of Frampton Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown	
37-008	200 square metres of agricultural land (south of Frampton Road)	Jacqueline Alison Bowens 1 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
37-008 cont'd		James Andrew Brock 2 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018)	
37-009	51823 square metres of agricultural land, access track and drains (Manor Farm)	Iacqueline Alison Bowens 1 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018) James Andrew Brock 2 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018) The Royal Society For The Protection Of Birds The Lodge Potton Road SANDY Bedfordshire SG19 2DL (in respect of rights granted by Conveyance dated 26 September 1986)	
37-010	458 square metres of agricultural land and access track (Manor Farm)	Jacqueline Alison Bowens 1 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018)	

I	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
37-010 cont'd		James Andrew Brock 2 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018) The Royal Society For The Protection Of Birds The Lodge Potton Road SANDY Bedfordshire SG19 2DL (in respect of rights granted by Conveyance dated 26 September 1986)	
37-011	185 square metres of drain (east of Sandholme Lane) (excluding all interests of the Crown)	Unknown	
38-002	37 square metres of access track (north of Sandholme Lane) (excluding all interests of the Crown)	Unknown	
38-004	43 square metres of verge (north of Sandholme Lane) (excluding all interests of the Crown)	Unknown	
38-005	112 square metres of public road (Sandholme Lane) (excluding all interests of the Crown)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ Unknown	
38-006	533 square metres of public road (Sandholme Lane) (excluding all interests of the Crown)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 2			
	County of Lincolnshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
38-008	35612 square metres of agricultural land, copse and drain (north of Marsh Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)		
38-009	605 square metres of agricultural land and access track (south of Sandholme Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)		
39-001	201 square metres of access track and copse (north of Marsh Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown		
39-002	19 square metres of access track and copse (north of Marsh Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown		

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-003	208 square metres of public road (Marsh Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)	
39-004	80 square metres of access track (east of Marsh Lane) and public footpath (LL Kirt 877/1)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)	
39-005	5888 square metres of agricultural land and access track (south of Marsh Lane)	Unknown	
39-006	31216 square metres of agricultural land, access track and drain (south of Marsh Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)	
39-007	22813 square metres of agricultural land, access track and drain (north of Marsh Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2 County of LincoInshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-008	3313 square metres of agricultural land, access track and drain (north of Marsh Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown	
39-009	12 square metres of drain (north of Marsh Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON LincoInshire PE20 3PW (in respect of riparian rights) Unknown	
39-010	1752 square metres of access track (north of Marsh Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)	
39-011	30 square metres of access track (north of Marsh Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)	

1	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-011 cont'd		Unknown	
39-012	1014 square metres of agricultural land (north of Marsh Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)	
39-015	623 square metres of public road (Marsh Road)	Unknown	
39-016	69856 square metres of agricultural land and drains (south of Marsh Road)	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983)	

	Outer Dov	wsing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-016		Wendy Christine Tunnard	
cont'd		Sandholme House	
		Sandholme Lane	
		Frampton	
		BOSTON Lincolnshire	
		PE20 1AG	
		(in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	
20.047			
39-017	613 square metres of agricultural land (south of Marsh Road)	Richard Paul Tunnard Sandholme House	
		Sandholme Induse	
		Frampton	
		BOSTON	
		Lincolnshire	
		PE20 1AG	
		(in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	
		Unknown	
		(in respect of rights reserved by a conveyance dated 11 October 1963)	
		Unknown	
		(in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983)	
		Wendy Christine Tunnard	
		Sandholme House	
		Sandholme Lane	
		Frampton	
		BOSTON	
		Lincolnshire PE20 1AG	
		(in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	
20.010			
39-018	25 square metres of agricultural land and access track (south of Marsh Road)	Unknown	
39-019	6 square metres of access track (south of Marsh Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-021	16906 square metres of agricultural land (east of Bucklegate Lane)	Richard Paul TunnardSandholme HouseSandholme LaneFramptonBOSTONLincolnshirePE20 1AG(in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)Unknown(in respect of rights reserved by a conveyance dated 11 October 1963)Unknown(in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983)Wendy Christine TunnardSandholme HouseSandholme LaneFramptonBOSTONLincolnshirePE20 1AG(in respect of right of oway, light or other easements (if any) as stated in conveyance dated 29 April 1983)	
39-022	39 square metres of drain (north of Nidd's Lane)	Unknown	
39-023	112 square metres of agricultural land (south of Nidd's Lane and west of Marsh Road)	James Edward Cheer Burton House Nidds Lane Kirton BOSTON Lincolnshire PE20 1LZ (in respect of rights reserved as stated in Transfer dated 13 July 2007)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
39-023 cont'd		Joanna Louise Cheer Burton House Nidds Lane Kirton BOSTON Lincolnshire PE20 1LZ (in respect of rights reserved as stated in Transfer dated 13 July 2007)	
40-001	1985 square metres of agricultural land (south of Marsh Road)	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	
40-002	1240 square metres of agricultural land (south of Marsh Road and east of Bucklegate Lane)	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
40-002 cont'd		Unknown (in respect of rights reserved by a conveyance dated 11 October 1963)	
		Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983)	
		Wendy Christine Tunnard Sandholme House	
		Sandholme Lane Frampton BOSTON	
		Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	
40-003	23 square metres of access track and verges (east of Clough Lane) and public footpath (LL	Amelia Kate Gray	
40-003	Kirt 1/5)	Hundred Acre Farm	
	Kit 1/5)	Marsh Road	
		Kirton	
		BOSTON	
		Lincolnshire	
		PE20 1ND	
		(in respect of rights of access)	
		James David Gray	
		Hundred Acre Farm	
		Marsh Road	
		Kirton	
		BOSTON	
		Lincolnshire	
		PE20 1ND	
		(in respect of rights of access)	
		The Black Sluice Internal Drainage Board	
		North End Business Park	
		Station Road	
		Swineshead	
		BOSTON	
		Lincolnshire	
		PE20 3PW	
		(assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2			
	County of Lincolnshire			
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008		
40-003 cont'd		Unknown		
	3887 square metres of access track and verge (east of Clough Lane) and public footpath	Amelia Kate Gray Hundred Acre Farm		
	(LL Kirt 1/5)	Marsh Road Kirton		
		BOSTON Lincolnshire		
		PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018)		
		Environment Agency Horizon House		
		Deanery Road BRISTOL Avon		
		BS1 5AH (in respect of rights to maintain sea defences)		
		Howard Stanley Leader Seadyke Cottage		
		Seadyke Road Kirton		
		BOSTON Lincolnshire		
		PE20 1QE (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in Deed dated 27 May 2005)		
		James David Gray		
		Hundred Acre Farm Marsh Road Kirton		
		BOSTON Lincolnshire		
		PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018)		

r	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
40-006 cont'd		Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights listed in a Transfer dated 11 May 2018) Melanie Clare Leader Seadyke Road Kirton BoSTON Lincolnshire PE20 1QE (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in Deed dated 27 May 2005) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown (in respect of reservations, covenants and conditions as stated in Deed dated 23 June 1870)	
40-007	46713 square metres of agricultural land, access track, copse and drain (east of Clough Lane) and public footpath (LL Kirt 1/5)	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018)	

Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2		
		County of Lincolnshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
40-007 cont'd		Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights to maintain sea defences) Howard Stanley Leader Seadyke Road Kirton BOSTON LincoInshire PE20 1QE (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in Deed dated 27 May 2005) James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON LincoInshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018) LincoInshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018) LincoInshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018) LincoInshire Newland LINCOIN LincoInshire UNI 1Y1 (in respect of rights listed in a Transfer dated 11 May 2018)

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
40-007 cont'd		Melanie Clare Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in respect of restrictive covenants stated in Deed dated 27 May 2005) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown (in respect of reservations, covenants and conditions as stated in Deed dated 23 June 1870) Unknown	
40-008	263 square metres of agricultural land access track and drain (west of Clatterdyke Road)	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights to maintain sea defences)	

Outer Dowsing Offshore Wind Development Consent Order		
		BOOK OF REFERENCE - PART 2
		County of Lincolnshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
40-008 cont'd		Howard Stanley Leader Seadyke Road Kirton BOSTON Lincoinshire PE20 10E (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in Deed dated 27 May 2005) James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincoinshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018) Lincoinshire PE20 IND (in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018) Lincoinshire PE20 IND (in respect of rights in a transfer dated 11 May 2018) Newland LINCOLN Lincoinshire Int respect of rights listed in a Transfer dated 11 May 2018) Melanie Clare Leader Seadyke Road Kirton BOSTON Lincoinshire PE20 10E (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in Deed dated 27 May 2005)

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
Land Plans 40-008 cont'd	4982 square metres of access track and copse (east of Low Mill Lane)		
		PE20 1QE (in respect of a rights of access)	
		The Welland And Deepings Internal Drainage Board Deeping House	
		Welland Terrace SPALDING LincoInshire PE11 2TD	
		(in respect of a right to pass as stated in Conveyance dated 18 March 1953)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
40-010	58240 square metres of agricultural land, drains and access track (east of Low Mill Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 SAH (in respect of a right of access) Howard Stanley Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of a rights of access) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of a right to pass as stated in Conveyance dated 18 March 1953)	
41-001	904 square metres of agricultural land and access track (east of Low Mill Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of a right of access) Howard Stanley Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of a rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
41-001 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of a right to pass as stated in Conveyance dated 18 March 1953)	
41-002	499 square metres of drain (east of Low Mill Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (in respect of riparian rights) Unknown	
41-003	117929 square metres of agricultural land, access tracks and drains (east of Craven's Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown (in respect of rights granted in Conveyance dated 28 January 1985)	
42-001	72 square metres of access track (east of Low Mill Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
42-002	2399 square metres of access track (east of Low Mill Lane) and public footpath (LL Fosd 8/1)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown (in respect of rights granted in Conveyance dated 28 January 1985)	
42-003	14324 square metres of agricultural land, access track and hardstanding (south of Craven's Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights granted in Conveyance dated 28 January 1985)	
42-004	649 square metres of access track, copse, drains and agricultural land (east of Craven's Lane) and public footpath (LL Fosd 8/1)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
42-005	62 square metres of drain (south of Craven's Lane)	Unknown	
42-007	41 square metres of drain (north of Craven's Lane)	Unknown (in respect of rights granted in Conveyance dated 28 January 1985)	
42-008	80 square metres of verge (Craven's Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
42-009	59 square metres of drain (south of Craven's Lane)	Unknown	
42-011	13 square metres of access track (east of Wash Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Unknown	
42-014	50 square metres of agricultural land (east of Wash Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Unknown	
	3426 square metres of access track, drains and woodland (south of Craven's Lane) and public footpath (LL Fosd 8/1)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of a secondary flood defence) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (in respect of rights to access and maintain the drain known as Roman Bank Drain)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
42-015 cont'd		Unknown	
42-017	177 square metres of drain (east of Pullover Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Unknown	
42-018	34 square metres of drain (east of Pullover Lane)	Unknown	
42-021	1368 square metres of public road and verges (Wash Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Unknown	
42-022	478 square metres of drain (west of Wash Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Unknown	
42-024	261 square metres of drain (north of Wash Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
43-003	495 square metres of public roads and verges (Surfleet Bank)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully and in respect of street furniture) Unknown	
43-004	79 square metres of agricultural land (west of Main Road, A17)	Unknown	
43-005	4512 square metres of agricultural land (west of Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a gully)	
43-006	44 square metres of agricultural land (east of Five Towns Drain)	Unknown (in respect of covenants contained in Conveyance dated 04 November 1957)	
43-007	682 square metres of watercourse and banks (Five Towns Drain)	Unknown	
43-008	181 square metres of watercourse banks (Five Towns Drain)	Unknown (in respect of covenants contained in Conveyance dated 04 November 1957)	
43-011	952 square metres of public road, verges and footways (Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
43-013	132 square metres of access track, grassed area and shrubbery (Graves Farm)	Unknown	
43-014	2697 square metres of agricultural land, access track and grassed area (west of Main Road, A17)	Timothy James Gratton Graves Farm Main Road Fosdyke BOSTON PE20 2DB (in respect of a right to service media and right to enter as stated in Transfer dated 27 June 1997)	
43-015	2656 square metres of agricultural land (south of Graves Farm, west of Main Road, A17)	Unknown (in respect of rights in connection to the Dive Towns Drain contained in an undated document)	
43-016	930 square metres of watercourse and banks (Five Towns Drain, west of Main Road, A17)	Unknown	
43-017	288 square metres of watercourse banks (Five Towns Drain)	Unknown (in respect of covenants contained in Conveyance dated 04 November 1957)	
43-019	2301 square metres of copse and drain (west of Main Road, A17) and public bridleway (LL Fosd 2/2)	Unknown	
43-021	239 square metres of agricultural land and access track (south of Smeeton's Lane)	Unknown	
43-022	58 square metres of verge and drain (south of Wash Road and east of Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
		BOOK OF REFERENCE - PART 2	
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
43-023	144 square metres of grassed area (west of Main Road, A17 and east of Five Towns Drain)	Boston Borough Council Municipal Buildings West Street BOSTON Lincolnshire PE21 8QR (in respect of rights granted by Deed of Grant dated 16th June 1995) National Highways Limited Bridge House 1 Walnut Tree Close GUILDFORD Surrey GU1 4LZ (in respect of rights and restrictive covenants stated in Transfer dated 14 July 2017) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights as stated in conveyance dated 14 January 1954)	
43-024	1925 square metres of access track and verges (west of Main Road, A17) and public byway (LL Fosd 2/1)	Unknown	
43-025	568 square metres of access track and verges (west of Main Road, A17) and public byway (LL Fosd 2/1)	Unknown	
43-026	209 square metres of agricultural land (west of Main Road, A17)	Unknown	
43-027	4818 square metres of agricultural land and access track (west of Five Towns Drain)	Unknown (in respect of covenants contained in Conveyance dated 04 November 1957)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
43-029	87 square metres of access track and copse (west of Main Road, A17) (excluding all interest	ts Anglian Water Services Limited	
	of the Crown)	Lancaster House	
		Lancaster Way	
		Ermine Business Park	
		HUNTINGDON	
		Cambridgeshire	
		PE29 3NZ	
		(in respect of rights stated in Transfer Scheme document dated 1 September 1989)	
		Lincolnshire County Council	
		County Offices	
		Newland	
		LINCOLN	
		Lincolnshire	
		LN1 1YL	
		(in respect of unknown rights listed in Deed dated 14 November 1989)	
43-031	1214 square metres of access track (west of Moulton Washway, A17), public bridleway	Anglian Water Services Limited	
	(LL Fosd 2/2) and public byways (LL Fosd 2/1 and LL Fosd 3/1) (excluding all interests of	Lancaster House	
	the Crown)	Lancaster Way	
		Ermine Business Park	
		HUNTINGDON	
		Cambridgeshire	
		PE29 3NZ	
		(in respect of rights stated in Transfer Scheme document dated 1 September 1989)	
		Lincolnshire County Council	
		County Offices	
		Newland	
		LINCOLN	
		Lincolnshire	
		LN1 1YL	
		(in respect of unknown rights listed in Deed dated 14 November 1989)	
		The Welland And Deepings Internal Drainage Board	
		Deeping House	
		Welland Terrace	
		SPALDING	
		Lincolnshire	
		PE11 2TD	
		(assumed in respect of rights of access)	

	Outer Dowsi	ing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
43-034	263 square metres of agricultural land and access track (west of Main Road, A17) and public bridleway (LL Fosd 2/2)	Unknown	
43-036	8146 square metres of access track (west of Moulton Washway, A17), public byways (LL Fosd 3/1) (excluding all interests of the Crown)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights stated in Transfer Scheme document dated 1 September 1989) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire Lincolnshire LN1 1YL (in respect of unknown rights listed in Deed dated 14 November 1989)	
43-038	3842 square metres of hardstanding and copse (east of Moulton Washway, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right of way listed in Transfer dated 31 December 1929)	
43-039	2508 square metres of hardstanding (east of Moulton Washway, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL ((in respect of a right of way listed in Transfer dated 31 December 1929) Unknown (as Kings's Nominee in respect of bona vacantia land for Simpsons (Fosdyke) Limited, dissolved) (in respect of covenants relating to use of land listed in Deed dated 06 November 1996)	
43-040	42 square metres of access track (east of Main Road, A17)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
43-041	63 square metres of grassed area (east of Main Road, A17)	Unknown	
43-042	94 square metres of access track (east of Main Road, A17) (excluding all interests of the Crown)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of covenants and rights to discharge water from an adjoining highway listed in a Deed of Exchange dated 06 November 1996) Unknown (in respect of restrictive covenants stated in unknown document dated 1 January 1926)	
43-043	426 square metres of access track (east of Moulton Washway, A17) and public bridleway (LL Fosd 7/1)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right of way listed in Transfer dated 31 December 1929) Unknown (in respect of a right of way stated in an undated unknown document) Unknown (in respect of manorial rights as stated in Conveyance dated 30 June 1917) Unknown (in respect of manorial rights as stated in conveyance dated 31 December 1929)	
43-044	17 square metres of access splay (west of Main Road, A17 and south of River Welland)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)	
43-045	1 square metres of grassed area (west of Main Road, A17 and south of River Welland)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans		Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
43-046	26 square metres of access splay (west of Main Road, A17 and south of River Welland) and public bridleway (LL Fosd 6/1)	Alan Mowton Shady Nook Sold Main Road Fostyke BOSTON Lincolnshire PE20 2BU (In respect of assumed access) Alexander James Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of assumed access) Andrew George Hay Wragg Marsh Wragg Marsh Marsh Road Spalding Marsh Spalding Marsh Spalding Marsh Farm Wragg Marsh Farm Wragg Marsh Farm Wragg Marsh Road Spalding Marsh Spalding Marsh	

	Outer Dowsing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 2	
	County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
43-046		Environment Agency
cont'd		Horizon House
		Deanery Road
		BRISTOL
		Avon
		BS1 5AH
		(assumed in respect of rights of access)
		Hazel Rose Needham
		Tumbleweed
		Moulton Marsh
		SPALDING
		PE12 6LJ
		(in respect of assumed access)
		James Needham
		Sunny View
		Moulton Washway
		Fosdyke Bridge
		SPALDING
		PE12 6LH
		(in respect of assumed access)
		Mary Ellenor Hay
		Wragg Marsh Farm
		Wragg Marsh
		Marsh Road
		Spalding Marsh
		SPALDING
		Lincolnshire
		PE12 6HQ (in respect of assumed access)
		Robin Twigg
		16 Cook Road
		Gosberton
		SPALDING
		Lincolnshire
		PE11 4PD
		(in respect of assumed access)

	Οι	uter Dowsing Offshore Wind Development Consent Order
		BOOK OF REFERENCE - PART 2
		County of Lincolnshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
43-046 cont'd		Unknown
43-047	15874 square metres of access track (east of Moulton Washway, A17)	Alan Mowton Shady Nook S Old Main Road Fosdyke BOSTON LincoInshire PE20 28U (in respect of assumed access) Alexander James Hay Wragg Marsh Farm Wragg Marsh Farm Wragg Marsh SPALDING LincoInshire PE12 6HQ (in respect of assumed access) Andrew George Hay Wragg Marsh Marsh Road Spalding Marsh SpalDING LincoInshire PE12 6HQ (in respect of assumed access) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Mars Ermine Business Park HUNTINGDON Cambridgeshire PE22 3NZ (in respect of rights stated in Transfer Scheme document dated 1 September 1989)

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
43-047		Belinda Elizabeth Needham	
cont'd		Sunny View	
		Moulton Washway	
		Fosdyke Bridge	
		SPALDING	
		PE12 6LH	
		(in respect of assumed access)	
		Environment Agency	
		Horizon House	
		Deanery Road	
		BRISTOL	
		Avon	
		BS1 5AH	
		(in respect of rights of entry and covenants listed in Transfer dated 03 March 2021)	
		Hazel Rose Needham	
		Tumbleweed	
		Moulton Marsh	
		SPALDING	
		PE12 6LJ	
		(in respect of assumed access)	
		James Needham	
		Sunny View	
		Moulton Washway	
		Fosdyke Bridge	
		SPALDING	
		PE12 6LH	
		(in respect of assumed access)	
		Lincolnshire County Council	
		County Offices	
		Newland	
		LINCOLN	
		Lincolnshire	
		LN1 1YL	
		(in respect of rights stated in Deed dated 14 November 1989)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
43-047 cont'd		Mary Ellenor Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of assumed access) Robin Twigg 16 Cook Road Gosberton SPALDING Lincolnshire PE11 4PD (in respect of assumed access)	
44-002	42847 square metres of agricultural land and access track (south of Smeeton's Lane)	John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (in respect of rights reserved by Transfer dated 26 March 2010)	
44-005	2282 square metres of watercourse (Risegate Eau) and banks thereof	Unknown	
44-007	371 square metres of agricultural land (west of Smeeton's Lane)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (in respect of a right of way as contained in an unknown undated document)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
44-007 cont'd		John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (in respect of a right of way as contained in an unknown undated document)	
44-008	80 square metres of agricultural land (south of Sea Bank)	Unknown	
44-009	149 square metres of agricultural land and drain (south of Smeeton's Lane)	Unknown	
44-011	8 square metres of drain (south of Sea Bank)	Unknown	
44-013	196 square metres of drain (south of Sea Bank)	Unknown	
44-016	616 square metres of access track (west of Smeeton's Lane)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (in respect of a right of way as contained in an unknown undated document) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (in respect of a right of way as contained in an unknown undated document)	
44-019	102 square metres of banks (south of Risegate Eau)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
	1463 square metres of agricultural land (south of Sea Bank)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971)	
44-022	508 square metres of access track (south of Smeeton's Lane) and public byway (LL Alga 9/1)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)	
44-024	20 square metres of grassed area (south of Smeeton's Lane)	Unknown	
44-027	61 square metres of access track and verge (south of River Welland)	Alexander James Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access) Andrew George Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
44-027		George Hay & Sons Limited	
cont'd		Wykeham Abbey	
		The Chase	
		Wykeham	
		SPALDING	
		Lincolnshire	
		PE12 6HE	
		(in respect of rights of access)	
		Mary Ellenor Hay	
		Wragg Marsh Farm	
		Wragg Marsh	
		Marsh Road	
		Spalding Marsh	
		SPALDING	
		Lincolnshire	
		PE12 6HQ	
		(in respect of rights of access)	
44-028	17 square metres of access track and verge (south of River Welland)	Alexander James Hay	
		Wragg Marsh Farm	
		Wragg Marsh	
		Marsh Road	
		Spalding Marsh SPALDING	
		Lincolnshire	
		PE12 6HQ	
		(in respect of rights of access)	
		Andrew George Hay Wragg Marsh Farm	
		Wragg Marsh	
		Marsh Road	
		Spalding Marsh	
		SPALDING	
		Lincolnshire	
		PE12 6HQ	
		(in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
44-028 cont'd		George Hay & Sons Limited Wykeham Abbey The Chase Wykeham SPALDING Lincolnshire PE12 6HE (in respect of rights of access) Mary Ellenor Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access) Unknown	
44-029	5049 square metres of private accessway and scrubland (south of River Welland)	Alexander James Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access) Andrew George Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
44-029		George Hay & Sons Limited	
cont'd		Wykeham Abbey	
contra		The Chase	
		Wykeham	
		SPALDING	
		Lincolnshire	
		PE12 6HE	
		(in respect of rights of access)	
		Mary Ellenor Hay	
		Wragg Marsh Farm	
		Wragg Marsh	
		Marsh Road	
		Spalding Marsh	
		SPALDING	
		Lincolnshire	
		PE12 6HQ	
		(in respect of rights of access)	
45-001	5972 square metres of agricultural land and access track (south of Sea Bank and west of	Peter C. Thorold Limited	
	Marsh Lane) (excluding all interests of the Crown)	5 Bank House Farm	
		Cooks Road	
		Gosberton	
		SPALDING	
		Lincolnshire	
		PE11 4PE	
		(in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
	5597 square metres of agricultural land, access track and drain (east of A16 and south of	Centrica PLC	
	Sea Bank) (excluding all interests of the Crown)	Millstream	
		Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of rights and restrictive covenants granted by Deed dated 02 November 1972)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-002 cont'd		Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
	3692 square metres of agricultural land, access track and drain (east of A16 and south of Sea Bank) (excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972) Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
45-004	28 square metres of access splay (south of Marsh Lane) (excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972)	

	Outer Dows	ing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	County of Lincolnshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-004 cont'd		Unknown	
45-005	4235 square metres of agricultural land and access track (west of Marsh Lane) (excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972) Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
	2529 square metres of agricultural land, access track and drain (east of A16 and south of Sea Bank) (excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972) Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	

	Outer Dowsing Offshore Wind Development Consent Order	
		BOOK OF REFERENCE - PART 2 County of Lincolnshire
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
45-007	3 square metres of verge (south of Marsh Lane) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown
	1647 square metres of agricultural land (south of Marsh Lane) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)
	1464 square metres of agricultural land, access track and drain (east of A16 and south of Sea Bank) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-009 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-010	94 square metres of watercourse (Risegate Eau) and access track (south of Marsh Lane) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (in respect of riparian rights) Unknown	
45-011	28 square metres of watercourse (Risegate Eau) and access track (south of Marsh Lane) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
45-012	71 square metres of watercourse (Risegate Eau) and access track (south of Marsh Lane) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
-	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-013	657 square metres of agricultural land, access track and drain (south of Marsh Lane) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
45-014	2483 square metres of agricultural land, access track and drain (south of Marsh Lane) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
45-015	2454 square metres of agricultural land (south of Marsh Lane) (excluding all interests of the Crown)	 Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) 	

1	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-015 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (assumed in respect of rights of access)	
45-016	2589 square metres of agricultural land, access track and drain (south of Marsh Lane) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-017	15 square metres of drain (south of Marsh Lane) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-018	72 square metres of drain (south of Marsh Lane) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
45-019	1575 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-021	3667 square metres of agricultural land (south of Marsh Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-023	1 square metres of drain (south of Marsh Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-025	6163 square metres of agricultural land (south of Marsh Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-026	816 square metres of drain (south of Marsh Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
45-028	14356 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-029	718 square metres of drain (south of Marsh Lane) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-029 cont'd		Unknown	
45-030	27105 square metres of agricultural land (south of Marsh Lane) (excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972) Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
45-031	375 square metres of drain (east of A16) (excluding all interests of the Crown)	Unknown	
45-033	3472 square metres of agricultural land (south of Risegate Eau) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING LincoInshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
45-034	1036 square metres of drain (east of A16) (excluding all interests of the Crown)	Unknown	
45-041	259554 square metres of agricultural land, access track and drain (east of A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted in Transfer dated 29 April 1998)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-041 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-043	2023 square metres of watercourse (Risegate Eau) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
45-044	1312 square metres of watercourse (Risegate Eau) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
45-046	334 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	I	County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-046a	8 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	Centrica PLC Milistream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of rights and covenants granted by Deed dated 31 December 1971) Joanne Lesley Davis Welland Farm House Surfleet Marsh Surfleet SPALDING Lincolnshire PE11 40W (in respect of rights granted by Transfer dated 07 November 2013) Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Lincolnshire Source of rights granted in Transfer dated 29 April 1998) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BISTOL Avon BS2 OTB (in respect of rights granted by Deed dated 21 February 2007) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BISTOL Avon BS2 OTB (in respect of rights granted by Deed dated 21 February 2007)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-046a cont'd		Patricia Mary Marguerite Bealby 24 Catherine Place Pine Gardens MELTON MOWBRAY LE13 1FH (in respect of rights granted by Deed dated 22 March 2000) Russell Lee Davis Welland Farm House Surfleet Marsh Surfleet SPALDING Lincolnshire PE11 4DW (in respect of rights granted by Transfer dated 07 November 2013) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
45-047	417 square metres of drain (east of A16) (excluding all interests of the Crown)	Unknown	
45-049	6604 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
45-050	11140 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-051	5486 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
45-052	2371 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
45-054	22177 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
45-055	302 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
45-056	1695 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-057	2434 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
45-058	17760 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-065	4924 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-066	3071 square metres of access track and verge (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
Number on Land Plans	Description of Land	County of Lincolnshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-067	5686 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-068	6104 square metres of agricultural land (east of Marsh Drove)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-069	1496 square metres of drain (east of Marsh Drove)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
45-069 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (assumed in respect of rights of access)	
45-071	19 square metres of agricultural land (north of River Welland)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
45-072	2315 square metres of agricultural land (north of River Welland)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (assumed in respect of rights of access)	
45-073	946 square metres of drain (south of Marsh Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (assumed in respect of rights of access) Unknown	
46-001	22 square metres of access track (east of A16) (excluding all interests of the Crown)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
46-002	390 square metres of agricultural land (east of A16) (excluding all interests of the Crown)	Unknown	
46-003	480 square metres of agricultural land (east of A16) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING LincoInshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
46-004	1232 square metres of agricultural land (east of A16 and north of Bicker Creek Drain) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
46-005	1680 square metres of agricultural land (east of A16 and north of Bicker Creek Drain) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
46-006	4031 square metres of agricultural land, access track and drain (east of A16) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
46-007	85 square metres of drain (Bicker Creek) (excluding all interests of the Crown)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
46-009	1926 square metres of agricultural land and access track (east of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (assumed in respect of rights of access)	
46-010	2522 square metres of agricultural land and access track (east of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
46-011	15 square metres of access track (east of A16) and banks (Risegate Eau)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (in respect of assumed rights of access) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights and obligations listed in a Conveyance dated 30 November 1988)	
46-012	37 square metres of access track (east of A16) and banks (Risegate Eau)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (in respect of assumed rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
46-012 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights and obligations listed in a Conveyance dated 30 November 1988)	
46-013	200 square metres of banks (Risegate Eau)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted in Transfer dated 22 May 2001) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
46-017	823 square metres of agricultural land (east of The Reservoir, A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
46-018	2630 square metres of agricultural land, hedgerow and access track (east of The Reservoir, A16) (excluding all interests of the Crown)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
46-019	296 square metres of agricultural land, hedgerow and access track (east of The Reservoir, A16) (excluding all interests of the Crown)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000)	
46-020	2607 square metres of agricultural land, hedgerow and access track (east of The Reservoir, A16) (excluding all interests of the Crown)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
46-021	8453 square metres of agricultural land and access track (east of A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000)	
	190 square metres of agricultural land, hedgerow and access track (east of The Reservoir, A16) (excluding all interests of the Crown)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000)	
46-023	455 square metres of drain (north of Marsh Drove)	Unknown	
46-024	453 square metres of drain (north of Marsh Drove)	Unknown	
46-025	10257 square metres of agricultural land (north of Marsh Drove)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right to enter as stated in Transfer dated 25 March 2003) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
46-027	22 square metres of agricultural land and access track (west of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights and obligations listed in a Conveyance dated 30 November 1988)	
46-028	24 square metres of agricultural land and access track (west of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (in respect of assumed rights of access) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights and obligations listed in a Conveyance dated 30 November 1988)	
46-029	105 square metres of agricultural land, access track and banks (west of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (in respect of assumed rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
46-029 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights and obligations listed in a Conveyance dated 30 November 1988)	
46-030	4 square metres of access track (west of A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted in Transfer dated 22 May 2001) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
46-031	5236 square metres of agricultural land (west of A16) and banks (Risegate Eau)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
46-033	5992 square metres of watercourse (Risegate Eau)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
46-034	6284 square metres of drain and banks (west of A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted in Transfer dated 22 May 2001) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
46-035	83 square metres of drain and banks (west of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
46-036	1029 square metres of drain and copse (east of Gosberton Bank)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
46-038	26 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	
46-039	1018 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	
46-040	1696 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	
46-041	172 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	
46-042	31 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	
46-043	1029 square metres of drain and copse (east of Gosberton Bank)	Unknown	
46-045	2024 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
46-046	2762 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	
47-002	1 square metres of access splay (east of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
47-003	416 square metres of private road and verge (North of Surfleet Bank)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
47-004	1894 square metres of agricultural land and access track (east of A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right of way as stated in Deed dated 16 March 1999) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
47-005	4 square metres of private road (North of Surfleet Bank)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
47-006	4722 square metres of agricultural land, access track and hedgerow (Woad Farm)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right of way as stated in Deed dated 16 March 1999) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
47-008	742 square metres of drain (east of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
47-009	4252 square metres of access track and verge (east of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (in respect of a rights stated in Conveyance dated 05 October 1984) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
47-010	679 square metres of drain and copse (east of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
47-011	5302 square metres of agricultural land (east of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)	
47-014	9109 square metres of agricultural land, access track and drain (east of A16)	Alan Burdall Aston Leigh Surfleet Marsh SPALDING PE11 4DP (in respect of a right of way as stated in Transfer dated 08 June 2001)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
47-014 cont'd		Gloria June Burdall Aston Leigh Surfleet Marsh Surfleet SPALDING PE11 4DP (in respect of a right of way as stated in Transfer dated 08 June 2001) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
47-015	448 square metres of agricultural land, access track and drain (east of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
47-016	325 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
47-017	2938 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
47-019	417 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
47-020	576 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
47-021	32 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (assumed in respect of rights of access)	
47-022	4 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
47-023	31 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
47-024	61 square metres of drain (north of Marsh Drove)	Unknown	
47-026	3678 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)	
47-028	1880 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)	
47-029	13675 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
47-030	87 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
47-032	581 square metres of public road and verges (Marsh Road)	Unknown	
47-036	30 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
47-037	1793 square metres of agricultural land, hedgerow and drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
47-038	1081 square metres of agricultural land, hedgerow and drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
47-038 cont'd		Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
47-039	1991 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-001	17 square metres of verge (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
48-002	471 square metres of agricultural land (north of Marsh Drove)	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-003	1047 square metres of agricultural land, access track and hedgerow (north of Marsh Drove)	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-004	655 square metres of agricultural land, access track and hedgerow (north of Marsh Drove)	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-005	1312 square metres of agricultural land and grassed area (north of Marsh Drove)	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-006	22 square metres of agricultural land (north of Marsh Drove)	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-010	622 square metres of agricultural land (north of Marsh Drove)	Unknown	

1	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
48-011	484 square metres of agricultural land (north of Marsh Drove)	Unknown	
48-012	905 square metres of agricultural land (north of Marsh Drove)	Unknown	
48-013	297 square metres of agricultural land (north of Marsh Drove)	Unknown	
48-017	264 square metres of agricultural land (south of Marsh Drove)	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-019	6564 square metres of river (River Welland), foreshore, bed and banks thereof	Unknown	
48-020	3793 square metres of agricultural land, scrubland and hedgerow (north of Marsh Road) and public bridleway (LL Wstn 6/2)	The Dean And Chapter Of The Cathedral Church Of Christ In Oxford Of The Foundation Of King Henry The Eighth Christ Church St. Aldates Oxford Oxfordshire OX1 1DP (in respect of personal covenants contained in Conveyance dated 25 July 1949) Unknown (in respect of rights reserved in Transfer dated 01 September 1989)	
48-023	3044 square metres of agricultural land (north of Marsh Road)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
48-023 cont'd		Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of rights of access)	
48-024	17109 square metres of agricultural land (north of Marsh Road)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Unknown (in respect of restrictive covenants that may have been imposed thereon before 11 December 2013)	
48-025	14133 square metres of agricultural land (north of Marsh Road)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 SAH (assumed in respect of rights of access) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
49-002	1499 square metres of access track (north of Marsh Road) and public bridleway (LL Surf 9/1)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Unknown (in respect of restrictive covenants that may have been imposed thereon before 11 December 2013)	
49-003	509 square metres of private road and verges (West of Wragg Marsh Farm) and public bridleways (LL Wstn 4/1 and LL Wstn 5/1)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)	
49-004	61 square metres of private road and verge (Marsh Road)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)	

	Outer Dows	ing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 2		
Number on		County of Lincolnshire Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3)	
Land Plans	Description of Land	of the Planning Act 2008	
49-004		Unknown	
cont'd		(in respect of restrictive covenants that may have been imposed thereon before 21 March 2000)	
49-005	2286 square metres of agricultural land (north of Marsh Road and south of River Welland)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 SAH (assumed in respect of rights of access) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB	
49-006	5104 square metres of agricultural land (Wragg Marsh Farm)	(in respect of rights of access) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access)	

-	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
49-006 cont'd		Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of rights of access)	
49-007	94 square metres of verge (north of Marsh Road)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of rights of access)	
49-008	261 square metres of public road (Marsh Road)	Unknown	
49-009	151 square metres of verge (south of Marsh Road)	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
49-009 cont'd		Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of rights of access)	
49-010	38389 square metres of agricultural land and private road (Wragg Marsh Farm)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)	
49-011	1194 square metres of agricultural land (Wragg Marsh Farm)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
50-001	5639 square metres of agricultural land (east of Marsh Road)	Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 29 October 2004)	
	506159 square metres of building and hardstanding (Crowtree Farm), agricultural land and access track (east of Marsh Road), electricity cables and pylons and public footpath (LL Wstn 7/1)	Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 29 October 2004)	
50-003	203228 square metres of agricultural land and access track (east of Marsh Road), electricity cable, pylons and public footpath (LL Wstn 7/1)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)	
50-004	1958 square metres of agricultural land and drain (east of Marsh Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
50-004		Spalding Energy Company, Ltd.	
cont'd		Spalding Power Station	
		West Marsh Road	
		SPALDING	
		Lincolnshire	
		PE11 2BB	
		(in respect of legal easements granted by Lease dated 14 January 2005)	
		Spalding Energy Expansion Limited	
		30 Crown Place	
		LONDON	
		EC2A 4ES	
		(in respect of the rights granted by the Deed dated 26 February 2018)	
50-005	216275 square metres of agricultural land, access track, electricity cable and pylons (east of	National Grid Electricity Transmission PLC	
	Marsh Road)	1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of rights stated in Deed dated 29 November 2002)	
		Spalding Energy Company, Ltd.	
		Spalding Power Station	
		West Marsh Road	
		SPALDING	
		Lincolnshire	
		PE11 2BB	
		(in respect of legal easements granted by Lease dated 14 January 2005)	
50-007	40886 square metres of agricultural land, access track and drain (east of Marsh Road)	Unknown	
51-003	521 square metres of drain (east of Marsh Road)	Unknown	
51-005	17 square metres of drain (east of Marsh Road)	Unknown	
51-006	822 square metres of drain (east of Marsh Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
51-007	159921 square metres of agricultural land, drain and access track (east of Marsh Road), electricity cables and pylons	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights and restrictive covenants listed in a Deed dated 08 May 2003) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of the legal essements granted by the Lease dated 14 January 2005)	
		(in respect of the legal easements granted by the Lease dated 14 January 2005)	
51-008	2157 square metres of agricultural land and drain (east of Marsh Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights and restrictive covenants listed in a Deed dated 08 May 2003) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of the legal easements granted by the Lease dated 14 January 2005 and 26 February 2018) Spalding Energy Expansion Limited 30 Crown Place LONDON EC2A 4ES (in respect of the rights that are granted by the Deed dated 26 February 2018)	
51-009	189781 square metres of agricultural land, drain and access track (east of Marsh Road), electricity cables and pylons	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights and restrictive covenants listed in a Deed dated 08 May 2003)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
51-009 cont'd		Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of the legal easements granted by the Lease dated 14 January 2005) Unknown (in respect to rights relating to nine inch sanitary pipes detailed in a deed dated 29 April 1943)	
51-010	789 square metres of drain (east of Marsh Road)	Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of rights of access) Unknown	
51-011	963 square metres of agricultural land (east of Marsh Road)	Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by the Lease dated 29 October 2004) Spalding Energy Expansion Limited 30 Crown Place LONDON EC2A 4ES (in respect of rights and a restriction stated in Deed dated 26 February 2018)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
51-012	27076 square metres of agricultural land (east of Marsh Road)	Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 29 October 2004)	
51-013	5 square metres of drain (east of Marsh Road)	Unknown	
51-014	308 square metres of drain (east of Marsh Road)	Unknown	
51-015		Gordon Albert Mills St. Lamberts Hall Haligate Weston SPALDING Lincolnshire PE12 6RH (in respect of rights granted in Transfer dated 22 April 2003) Helen Juliette Mills St. Lamberts Hall Haligate Weston SPALDING Lincolnshire PE12 6RH (in respect of rights granted in Transfer dated 22 April 2003) Moulton Bulb Company Limited Long Lane Moulton SPALDING E12 6PP (in respect of restrictions listed in a Definitive Trust Deed dated 26 September 1986 and rights granted in Transfer dated 22 April 2003)	

Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008
51-015 cont'd		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights relating to electricity lines and pylons listed in a Deed of Grant dated 16 April 2003) Prudential Financial Planning Limited 10 Fenchurch Avenue LONDON EC3M 5AG (in respect of restrictions listed in a Definitive Trust Deed dated 26 September 1986)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Camven Jarley, Youngers Lane, Burgh Le Marsh, Skegness, PE24 5JQ	Pauline Janet Broughton Camven Jarley Youngers Lane Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JQ	
		Philip Charles Broughton Camven Jarley Youngers Lane Burgh Le Marsh SKEGNESS Lincolnshire PE24 5JQ	
N/A		Joyce Margaret Blanchard Orchard Farm West End Hogsthorpe SKEGNESS Lincolnshire PE24 5PA ()	
N/A	Ivy Lodge, Bracken Lane, Hogsthorpe, Skegness, PE24 5PB	Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5PB	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Mayfield House, Bracken Lane, Hogsthorpe, Skegness, PE24 5PB	Pauline Kirkby The Barn Langham Road Hogsthorpe SKEGNESS Lincolnshire PE24 5QF	
N/A	Big Tree Farm, Marsh Drove, Surfleet Marsh, PE11 4DW	Esmond Herbert Gadd Big Tree Farm Surfleet Marsh Surfleet SPALDING Lincolnshire PE11 4DW	
		Judith Gadd Big Tree Farm Surfleet Marsh Surfleet SPALDING Lincolnshire PE11 4DW	
N/A	Hawthorne Cottage, Sandholme Lane, Kirton, BOSTON,PE20 1NG	Donna Sillett Hawthorne Cottage Sandholme Lane Kirton BOSTON PE20 1NG	
		Lincolnshire Housing Partnership Limited L H P Cartergate House 26 Chantry Lane GRIMSBY DN31 2LJ	

Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2			
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Robert James Sillett Hawthorne Cottage Sandholme Lane Kirton BOSTON PE20 1NG	
N/A	West End Cottage, Hogsthorpe, Skegness, PE24 5PA and land lying to the south of West End	Jacqueline Collings West End Cottage West End Hogsthorpe SKEGNESS PE24 5PA Neil Collings West End Cottage West End Hogsthorpe SKEGNESS PE24 5PA	
N/A	Land on the north east side of Mumby Road, Hogsthorpe	Joyce Margaret Blanchard Orchard Farm West End Hogsthorpe SKEGNESS Lincolnshire PE24 5PA	
N/A	Swandyke Farm, Swandyke Lane, Freiston, Boston, PE22 0NZ	G-Veg Limited Mill Farm Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HY	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Lowfields Lodge, Double Bank, Freiston Ings, PE22 OPS	Saviour Deguara Clover Farm Main Road Langrick BOSTON PE22 7AW	
		Svetlana Fenech Clover Farm Main Road Langrick BOSTON PE22 7AW	
N/A	Faunt Bridge, Pode Lane, Old Leake, Boston, PE22 9NA	Susan Mary Jaques Faunt Bridge Farm Faunt Bridge Old Leake BOSTON Lincolnshire PE22 9QT ()	
N/A	The Bungalow, Faunt Bridge, Old Leake, PE22 9QT	Darren Graham Eric Blythe The Beeches Sibsey Road Old Leake BOSTON PE22 9QS	
N/A	Turkey Farm, Church Road, Old Leake, Boston, PE22 9PE	Amber Real Estate Investments (Agriculture) Limited Colmore Court 9 Colmore Row BIRMINGHAM West Midlands B3 2BJ	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	The Prefab, Church Road, Old Leake, Boston, PE22 9PD	Henry John Frank Lowis 25 Peck Avenue BOSTON Lincolnshire PE21 8DG	
N/A	Rose Villa, Church Road, Old Leake and Land lying to the east of Church Road, Old Leake	Dorothy May Holiday Rose Villa Church Road Old Leake BOSTON Lincolnshire PE22 9PD Robert Holiday Rose Villa Church Road Old Leake BOSTON Lincolnshire PE22 9PD	
N/A	Hawthorn Farm, Ivery Lane, Wrangle, Boston, PE22 9BB	Janet Codd Hawthorn Farm Ivery Lane Wrangle BOSTON Lincolnshire PE22 9BB	
N/A	The Laurels, Mill Hill Road, Friskney, PE22 8PD	Ronald Leslie Elvin The Laurels Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2 County of LincoInshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Lenick, Mill Hill, Friskney, PE22 8NG	Christopher Blevins Lenick Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	
		Hannah Elizabeth Blevins Lenick Mill Hill Road Friskney BOSTON Lincolnshire PE22 8PD	
N/A	Lynwood, Burgh Road, Friskney, PE22 8NS	Jonathan Harman Lynwood Burgh Road Friskney BOSTON Lincolnshire PE22 8NS Ruth Harman Lynwood	
		Burgh Road Friskney BOSTON LincoInshire PE22 8NS	

Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Decoy Farm, Burgh Road, Friskney, Boston, PE22 8NT	Maria Ann Gibson Decoy Farm Burgh Road Friskney BOSTON PE22 8NT	
		Peter Timothy Gibson Decoy Farm Burgh Road Friskney BOSTON PE22 8NT	
N/A	Border House, Burgh Road, Friskney, PE22 8NS	Gary Paul Miller Border House Burgh Road Friskney BOSTON LincoInshire PE22 8NS	
		Margaret Miller Border House Burgh Road Friskney BOSTON Lincolnshire PE22 8NS	
		Paul Miller Border House Burgh Road Friskney BOSTON Lincolnshire PE22 8NS	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2 County of LincoInshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	The Laurels, Burgh Road, Friskney	Brian Thoseby The Laurels Burgh Road Friskney BOSTON Lincolnshire PE22 8NS	
N/A	Waterside, east End, Thorpe St Peter, Skegness, PE24 4PQ	Kay Elizabeth Johnson Meadow Croft Croft Lane Croft SKEGNESS Lincolnshire PE24 4PF	
N/A	2 White Cottages, east End, Thorpe St Peter, Skegness, PE24 4PQ	Avril Weightman 2 White Cottage East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	
N/A	1 White Cottage, east End, Thorpe St Peter, Skegness, PE24 4PQ	Benjamin Frank Taylor 1 White Cottage East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A cont'd		Kelly Marie Scutt 1 White Cottage East End Thorpe St. Peter SKEGNESS Lincolnshire PE24 4PQ	
N/A	Nigella, Croft Lane, Croft, SKEGNESS, PE24 4PF	Denise Rewcastle Black Horse Bridge 2 Thorpe Bank Little Steeping SPILSBY Lincolnshire PE23 5BG Paul Raymond Rewcastle Black Horse Bridge 2 Thorpe Bank Little Steeping SPILSBY Lincolnshire PE23 5BG	
N/A	Kevean Paddocks, Collison Gate, Wainfleet	Jean Margaret Hart Kevean Cottage Collison Gate Wainfleet SKEGNESS Lincolnshire PE24 4LZ Kevin James Hart Kevean Cottage Collison Gate Wainfleet SKEGNESS Lincolnshire PE24 4LZ	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	Land to the north of The Gride	Gareth Matthew Adams Petersfield Farm The Gride Old Leake BOSTON Lincolnshire PE22 9SA	
N/A	Amblecote, West End, Hogsthorpe, PE24 5PA	Judy Adcock Amblecote West End Hogsthorpe SKEGNESS PE24 5PA	
		Paul Lesley Hartopp Amblecote West End Hogsthorpe SKEGNESS PE24 5PA	
N/A	Fieldview, Main Road, Fosdyke, Boston, PE20 2DB	Owen Thomas Elleray Fieldview Main Road Fosdyke BOSTON PE20 2DB	
N/A	Myrtle Cottage, Main Road, Fosdyke, PE20 2DB	Jerry Lee Lawes Myrtle Cottage Main Road Fosdyke Boston Lincolnshire PE20 2DB	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	The Moorings, Main Road, Fosdyke, Boston, PE20 2DB	Christopher Davison The Moorings Main Road Fosdyke BOSTON PE20 2DB	
		Gillian Davison The Moorings Main Road Fosdyke BOSTON PE20 2DB ()	
N/A	Graves Farm, Main Road, Fosdyke, Boston, PE20 2DB	Timothy James Gratton Graves Farm Main Road Fosdyke BOSTON PE20 2DB	
N/A	1, Middlecott Almshouses, Wash Road, Fosdyke, Boston, PE20 2DG	Anthony Paul Kindred 1 Middlecott Almshouses Wash Road Fosdyke BOSTON PE20 2DG	
		Lisa Anne Kindred 1 Middlecott Almshouses Wash Road Fosdyke BOSTON PE20 2DG	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 2		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	The Poplars, Sandholme Lane, Kirton, Boston, PE20 1NG	Robert James Sillett Hawthorne Cottage Sandholme Lane Kirton BOSTON PE20 1NG	
N/A	Laurel Cottage, Haven Bank, Wainfleet St Mary, Skegness, PE24 4JW	Daniel Swain 234 Roman Bank SKEGNESS PE25 1SJ	
		Holly Maria Sylvia Shillings 234 Roman Bank SKEGNESS PE25 1SJ	
N/A	Ivy Lodge, Bracken Lane, Hogsthorpe, Skegness, PE24 5PB	Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5PB	
		Benjamin John Maurice Lowe St. Catherines House All Saints Green Norwich NR1 3GA	
		Lindsay Jane Whitehead 2 Penrice Road Little Plumstead NORWICH NR13 5FP	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 2 County of Lincolnshire		
Number on Land Plans	Description of Land	Potential claims under S10 Compulsory Purchase Act 1965, Part 1 Land Compensation Act 1973 or 152 (3) of the Planning Act 2008	
N/A	2 Middlecott Almshouses, Wash Road, Fosdyke, PE20 2DG	Gweneth Brenda Miller 2 Middlecott Almshouses Wash Road Fosdyke BOSTON PE20 2DG Robert Miller 2 Middlecott Almshouses Wash Road Fosdyke BOSTON PE20 2DG	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-006	93177 square metres of agricultural land, drain, ponds and path (East of Roman Bank)	Alco Estates Limited Boathouse Sandhurst Road Sandilands MABLETHORPE Lincolnshire LN12 2RH (in respect of rights and restrictive covenants stated in Tranfer dated 30 September 2009) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted by Deed dated 14 January 1941)	
01-007	553 square metres of verge (Roman Bank)	Unknown (in respect of rights and restrictive covenants stated in Tranfer dated 30 September 2009)	
01-008	184 square metres of access track and verge (east of Roman Bank)	Unknown	
01-009	2521 square metres of public road and verges (Roman Bank)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus and abandoned underground sewerage apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-010	2154 square metres of verge (Roman Bank)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus and abandoned underground sewerage apparatus) Unknown	
01-011	177 square metres of public road, verge and drain (Roman Bank)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus and abandoned underground sewerage apparatus) Unknown	
01-012	569 square metres of agricultural land (west of Roman Bank)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-013	22897 square metres of agricultural land (west of Roman Bank)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown	
01-014	8249 square metres of agricultural land (west of Roman Bank)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown	
01-015	20085 square metres of agricultural land (west of Roman Bank)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-016	34738 square metres of agricultural land (west of Roman Bank)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
01-017	43296 square metres of agricultural land (west of Roman Bank)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
01-018	6314 square metres of agricultural land, drain and access track (west of Roman Bank)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of rights reserved by Conveyance dated 16 March 1965)	
01-019	812 square metres of agricultural land and drain (west of Roman Bank)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown	
02-001	26863 square metres of access track and drain (west of Roman Bank)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-001 cont'd		Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
02-002	13744 square metres of agricultural land and access track (west of Roman Bank)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground sewerage apparatus) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SNS 6PB (as beneficiary of registered land charges) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-003	719 square metres of agricultural land and drain (west of Roman Bank)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown	
	2278 square metres of agricultural land and hedgerow (west of Roman Bank) and public footpath (LL Chap 19/5)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-005	491 square metres of agricultural land (north of Ember Lane) and public footpath (LL Chap 19/5)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
02-006	36239 square metres of agricultural land (north of Ember Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of registered land charges) Unknown	
02-007	1386 square metres of public road and verges (Ember Lane) and public footpath (LL Chap 21/4)	Unknown	
02-008	751 square metres of agricultural land (south of Ember Lane)	Pridgeon Farms Limited Willows Farm Bradshaws Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5QL (in respect of a rights stated in Transfer dated 28 May 2009 and Conveyance dated 22 October 1979) Unknown (in respect of rights granted by Conveyance dated 11 October 1993)	
02-011	178 square metres of agricultural land (south of Ember Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-012	15650 square metres of agricultural land (south of Ember Lane) and public footpath (LL Chap 27/3)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground sewerage apparatus)	
02-015	17 square metres of hedgerow (south of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (assumed in respect of rights of access) Unknown	
02-016	1073 square metres of agricultural land (south of Ember Lane) and public footpath (LL Chap 27/3)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
02-017	1 square metres of agricultural land (south of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-017 cont'd		Unknown	
02-018	1 square metres of agricultural land (south of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
02-019	28 square metres of agricultural land (south of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
02-020	2153 square metres of agricultural land (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of reservation of mines and minerals and rights to work the same)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
02-021	16 square metres of agricultural land (south of Ember Lane)	Unknown	
03-001	47 square metres of agricultural land (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
03-002	485 square metres of agricultural land (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
03-003	30 square metres of agricultural land (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-004	3 square metres of agricultural land and drain (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of reservation of mines and minerals and rights to work the same)	
03-005	61 square metres of agricultural land and drain (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of reservation of mines and minerals and rights to work the same)	
03-006	248 square metres of agricultural land (east of Ember Lane)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of a rights as stated in Deed dated 28 February 1980 and and in Transfer dated 22 December 1997) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

		Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3
		County of Lincolnshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
03-007	106 square metres of agricultural land (east of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS Lincolnshire PE24 SRJ (In respect of a right to enter and service media rights stated in Transfer dated 28 May 2008) Catherine Ellen Hill Highfield Authorpe LOUTH Lincolnshire UN11 8PG (In respect of rights and restrictive covenants as stated in Transfer dated 28 May 2009) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire UN11 8UU (assumed in respect of rights of access) Richard John Hill Highfield Authorpe LOUTH Lincolnshire UN11 8PG (In tespect of rights and restrictive covenants as stated in Transfer dated 28 May 2009) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire UN11 8PG (In tespect of rights and restrictive covenants as stated in Transfer dated 28 May 2009)

		Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3	
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-008	1044 square metres of agricultural land (east of Ember Lane)	Allan Walter Clarke Hill View Farm Wigg Lane Chapel St. Leonards SKEGNESS LincoInshire PE24 5RJ (in respect of a right to enter and service media rights stated in Transfer dated 28 May 2008) Catherine Ellen Hill Highfield Authorpe LOUTH LincoInshire LN11 8PG (in respect of rights and restrictive covenants as stated in Transfer dated 28 May 2009) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN118 UU (assumed in respect of rights of access) Richard John Hill Highfield Authorpe LOUTH LincoInshire LN118 PG (in respect of rights and restrictive covenants as stated in Transfer dated 28 May 2009)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-009	172 square metres of watercourse (Wigg Drain)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
03-010	58924 square metres of agricultural land, hedgerows, access track (east of Ember Lane), watercourse (Wigg Drain) and public footpath (LL Hogs 34/4, LL Hogs 28/1)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	

	Outer E	Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3	
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
	1023 square metres of agricultural land and access track (east of Ember Lane)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of a rights as stated in Deed dated 28 February 1980 and and in Transfer dated 22 December 1997) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
03-012	8 square metres of access track (east of Ember Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
03-014	942 square metres of agricultural land (north of Langham Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
02.014		(Applications: Prescribed Forms and Procedures) Regulations 2009 Openreach Limited	
03-014 cont'd		6 Gracechurch Street	
cont u		LONDON	
		EC3V 0AT	
		(in respect of underground telecommunications apparatus)	
03-015	890 square metres of public road and verge (Langham Road)	Anglian Water Services Limited	
		Lancaster House	
		Lancaster Way	
		Ermine Business Park	
		HUNTINGDON	
		Cambridgeshire	
		PE29 3NZ	
		(in respect of underground water apparatus)	
		National Grid Electricity Distribution (East Midlands) PLC	
		Avonbank	
		Feeder Road	
		BRISTOL	
		Avon	
		BS2 OTB	
		(in respect of overhead electricity apparatus)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of underground telecommunications apparatus)	
		Unknown	
03-022	6180 square metres of agricultural land, hedgerow and drain (north of Lowgate Road)	Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of overhead telecommunications apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-024	194 square metres of agricultural land (north of Lowgate Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecoms apparatus) Unknown	
03-026	675 square metres of public road and verges (Lowgate Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown	
03-027	2424 square metres of agricultural land and drain (south of Lowgate Road)	Unknown (in respect of rights of drainage reserved by a Transfer dated 5 September 2000)	
03-028	23668 square metres of agricultural land and drain (west of Lowgate Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer apparatus) Unknown (in respect of rights of drainage reserved by a Transfer dated 5 September 2000)	
03-029	20779 square metres of agricultural land and access track (south of Lowgate Road) and public footpaths (LL Hogs 58/2, LL Hogs 57/1)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
03-029 cont'd		Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 19 April 2019)	
03-030	4889 square metres of agricultural land (north of Bracken Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Lease dated 12 December 2018)	
03-031	651 square metres of agricultural land and access track (north of Bracken Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 19 April 2019)	
04-001	242 square metres of agricultural land (west of Lowgate Road)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of the rights granted by the Deed of Grant dated 12 December 2018)	
04-002	621 square metres of agricultural land (north of Bracken Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Lease dated 12 December 2018)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-003	202 square metres of agricultural land and access track (west of Lowgate Road)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 19 April 2019)	
04-004	3291 square metres of agricultural land (north of Bracken Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in a Lease dated 26 October 2018)	
04-005	586 square metres of hedgerow, access track and drain (north of Bracken Lane)	Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of an assumed right of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-006	22399 square metres of agricultural land and electricity cable and pylon (north of Bracken Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 26 October 2018)	
04-007	513 square metres of agricultural land and access track (north of Bracken Lane)	Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of an assumed right of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SNS 6PB (in respect of rights stated in Caution dated 12 August 2021)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-007 cont'd		Unknown	
04-008	16 square metres of agricultural land and access track (north of Bracken lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
04-009	370 square metres of agricultural land (north of Bracken Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deed dated 26 October 2018)	
04-010	131 square metres of access track and drain (north of Bracken Lane)	Amanda Louise Green Ivy Lodge Bracken Lane Hogsthorpe SKEGNESS Lincolnshire PE24 5PB (in respect of an assumed right of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-010 cont'd		Jack Ward Jack Ward Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of an assumed right of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
04-011	2886 square metres of agricultural land (north of Bracken Lane)	L H Ward & Son Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of rights granted by Deed dated 09 April 2019) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deeds dated 9 April 2019 and 26 October 2018)	
04-012	22 square metres of drain (north of Mumby Road, A52)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-013	714 square metres of agricultural land, access track and copse (north of Mumby Road, A52)	L H Ward & Son Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of rights granted by Deed dated 09 April 2019) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deeds dated 9 April 2019 and 26 October 2018)	
	7467 square metres of agricultural land, access track and copse (north of Mumby Road, A52)	L H Ward & Son Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of rights granted by Deed dated 09 April 2019) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants stated in Deeds dated 9 April 2019 and 26 October 2018)	
04-015	483 square metres of agricultural land, access track and copse (north of Mumby Road, A52)	L H Ward & Son Hillcrest Cumberworth ALFORD Lincolnshire LN13 9LB (in respect of rights granted by Deed dated 09 April 2019)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-015 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SNS 6PB (in respect of rights and restrictive covenants stated in Deeds dated 9 April 2019 and 26 October 2018)	
04-016	14713 square metres of agricultural land (west of Bracken Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	
04-017	1499 square metres of public road, verges and hedgerow (Mumby Road, A52)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown	

	Outer Dow	sing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-018	29900 square metres of agricultural land and access track (south of Mumby Road, A52)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Agreement dated 31 December 1973) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	
04-019	219 square metres of drain (south of Bracken Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-020	3647 square metres of agricultural land and drain (north of Listoft Lane)	Andrew Michael Snell The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 5PG (in respect of rights granted in Conveyance dated 21 January 1993) Gillian Mary Sharpe The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 5PG (in respect of rights granted in Conveyance dated 21 January 1993) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB	
04-021	17206 square metres of agricultural land and drain (north of Listoft Lane)	H. Bradley & Sons (Produce) Limited Coots Farm Coots Lane Mumby ALFORD Lincolnshire LN13 9JZ (in respect of rights granted by Deed of Grant dated 12 December 2018) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunications apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Lond	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
04-022	1269 square metres of agricultural land and access splay (north of Listoft Lane)	Andrew Michael Snell	
		The Willows	
		Listoft Lane	
		Hogsthorpe	
		SKEGNESS	
		PE24 5PG	
		(in respect of rights granted in Conveyance dated 21 January 1993)	
		Gillian Mary Sharpe	
		The Willows	
		Listoft Lane	
		Hogsthorpe	
		SKEGNESS	
		PE24 5PG	
		(in respect of rights granted in Conveyance dated 21 January 1993)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of overhead telecommunications apparatus)	
		Triton Knoll Offshore Wind Farm Limited	
		Windmill Hill Business Park	
		Whitehill Way	
		SWINDON	
		Wiltshire	
		SN5 6PB	
		(in respect of rights granted by Deed of Grant dated 12 December 2018)	
04-023	3964 square metres of agricultural land and drain (north of Listoft Lane)	Andrew Michael Snell	
		The Willows	
		Listoft Lane	
		Hogsthorpe	
		SKEGNESS	
		PE24 5PG	
		(in respect of rights granted in Conveyance dated 21 January 1993)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
04-023 cont'd		Gillian Mary Sharpe The Willows Listoft Lane Hogsthorpe SKEGNESS PE24 5PG (in respect of rights granted in Conveyance dated 21 January 1993) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunications apparatus)	
05-001	407 square metres of agricultural land (south of Listoft Lane)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by Deed of Grant dated 12 December 2018)	
05-002	1379 square metres of public road (Listoft Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-002 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	
05-003	16258 square metres of agricultural land and drain (south of Listoft Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights granted by Deed of Grant dated 12 December 2018)	
05-004	222 square metres of drain (Willoughby High Drain, south of Listoft Lane)	Unknown	
05-005	1375 square metres of agricultural land and copse (west of Sloothby High Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-005 cont'd		Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Deed dated 26 October 2018)	
05-006	12089 square metres of agricultural land and copse (west of Sloothby High Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of rights) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SNS 6PB (in respect of rights stated in Deed dated 26 October 2018)	
05-007	1064 square metres of public road and verges (Sloothby High Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-007 cont'd		Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
	19597 square metres of agricultural land (east of Sloothby High Lane) and drain (Wyche Drain)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
05-009	106 square metres of drain (Wyche Drain, east of Sloothby Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-010	5883 square metres of agricultural land and copse (east of Sloothby High Lane)	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
05-011	4416 square metres of agricultural land and drain (east of Sloothby High Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
05-013	50 square metres of agricultural land (east of Sloothby High Lane)	Brian Edward Taylor Slackholme House Farm Slackholme End Hogsthorpe SKEGNESS Lincolnshire PE24 5NS (in respect of right to lay a copper pipe as stated in a Deed dated the 30 October 1961. Also in respect of right to pass as stated in a Transfer dated the 13 May 2004) British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989)	

	Outer D	owsing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
05-013		David Robert Norman Taylor	
cont'd		Slackholme House Farm	
		Slackholme End	
		Hogsthorpe	
		SKEGNESS	
		Lincolnshire	
		PE24 5NS	
		(in respect of right to lay a copper pipe as stated in a Deed dated the 30 October 1961. Also in respect of right to pass as stated in a Transfer	
		dated the 13 May 2004)	
		Maxine Hayley Taylor	
		Slackholme House Farm	
		Slackholme End	
		Hogsthorpe	
		SKEGNESS	
		Lincolnshire	
		PE24 5NS	
		(in respect of right to lay a copper pipe as stated in a Deed dated the 30 October 1961. Also in respect of a right to pass as stated in a	
		Transfer dated the 13 May 2004)	
05-014	784 square metres of public road (south Ings Lane)	Lindsey Marsh Drainage Board	
		Wellington House	
		Manby Park	
		Manby	
		LOUTH	
		Lincolnshire	
		LN11 8UU	
		(assumed in respect of rights of access)	
		Unknown	
05-016	3009 square metres of agricultural land and copse (east of Sloothby High Lane)	British Rail Pension Trustee Company Limited	
03-010	ישטט אינעמיב וויבנובא טו מצווגעונט מווע מווע נטאאב (כמאנ טו אוטטנוואא דוצוו במוופ)	100 Liverpool Street	
		LONDON	
		EC2M 2AT	
		(in respect of rights stated in Conveyance dated 29 September 1989)	
		(in respect of rights stated in conveyance dated 23 september 1303)	

	Outer Dow	rsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3	
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-016 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	
05-018	32131 square metres of agricultural land, copse and drains (east of South Ings Lane)	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
05-019	1827 square metres of agricultural land (east of South Ings Lane)	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
05-020	841 square metres of access track (east of South Ings Lane)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of rights pursuant to transfer dated 5 January 2021)	
	19541 square metres of agricultural land, access track, drain (east of South Ings Lane) and public footpath (LL Hogs 48/1)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of rights pursuant to transfer dated 5 January 2021)	
06-001	41 square metres of drain (east of South Ings Lane)	Unknown	
06-004	18 square metres of drain (east of South Ings Lane)	Unknown	
06-005	360 square metres of drain (east of South Ings Lane)	Unknown	
06-008	38463 square metres of agricultural land and drain (east of South Ings Lane)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of transfer dated 5 January 2021)	
06-014	4239 square metres of agricultural land and access track (east South Ings Lane)	Unknown (in respect of subjections as stated in conveyance dated 29 September 1989)	
06-016	2219 square metres of agricultural land and drain (east of South Ings Lane)	Unknown (in respect of subjections as stated in conveyance dated 29 September 1989)	
06-017	34903 square metres of agricultural land (east of South Ings Lane)	Unknown (in respect of subjections as stated in conveyance dated 29 September 1989)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
06-018	847 square metres of public road and verges (South Ings Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	
06-019	1583 square metres of agricultural land (west of South Ings Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989)	
06-020	641 square metres of agricultural land and access splay (east of South Ings Lane)	Unknown (in respect of subjections as stated in conveyance dated 29 September 1989)	
07-002	1283 square metres of agricultural land and drain (west of South Ings Lane)	British Rail Pension Trustee Company Limited 100 Liverpool Street LONDON EC2M 2AT (in respect of rights stated in Conveyance dated 29 September 1989)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-002		Lindsey Marsh Drainage Board	
cont'd		Wellington House	
		Manby Park	
		Manby	
		LOUTH	
		Lincolnshire	
		(assumed in respect of rights of access)	
07-003	61514 square metres of agricultural land and drain (west of South Ings Lane and north of	British Rail Pension Trustee Company Limited	
	Marsh Lane)	100 Liverpool Street	
		LONDON	
		EC2M 2AT	
		(in respect of rights stated in Conveyance dated 29 September 1989)	
		Lindsey Marsh Drainage Board	
		Wellington House	
		Manby Park	
		Manby	
		LOUTH	
		Lincolnshire	
		(assumed in respect of rights of access)	
		Triton Knoll Offshore Wind Farm Limited	
		Windmill Hill Business Park	
		Whitehill Way	
		SWINDON	
		Wiltshire	
		SN5 6PB	
		(in respect of rights stated in Deed dated 28 January 2019)	
07-004	2282 square metres of agricultural land (north of Marsh Lane)	British Rail Pension Trustee Company Limited	
		100 Liverpool Street	
		LONDON	
		EC2M 2AT	
		(in respect of rights stated in Conveyance dated 29 September 1989)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-004 cont'd		Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights stated in Deed dated 28 January 2019)	
07-005	199 square metres of drain (north of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
07-006	54064 square metres of agricultural land and copse (north of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of the rights granted by the Deed dated 07 December 1966)	
07-007	205 square metres of agricultural land and copse (north of Marsh Lane)	Unknown (in respect of the rights granted by the Deed dated 07 December 1966)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-008	8330 square metres of agricultural land (north of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of the rights granted by the Deed dated 07 December 1966)	
07-009	120 square metres of agricultural land and copse (north of Marsh Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of personal covenants stated in Transfer dated 5 January 1994) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
07-010	240 square metres of access splay (south of Marsh Lane)	David Mark Simpson Whitehouse Farm Addlethorpe SKEGNESS Lincolnshire PE24 4TU (in respect of an assumed right of access)	

Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-010 cont'd		Julia Clare Pound Ayshford Court Westleigh TIVERTON EX16 7HL (in respect of an assumed right of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire UN11 8UU (assumed in respect of rights of access) Mark Caudwell Limited Howlett House Farm Hogsthorpe Skegness LincoInshire PE24 SNS (in respect of an assumed right of access) Mark Leopold Caudwell Howlett House Slackholime End Hogsthorpe SKEGNESS LincoInshire PE24 SNS (in respect of an assumed right of access) Mark Leopold Caudwell Howlett House Slackholime End Hogsthorpe SKEGNESS LincoInshire PE24 SNS (in respect of an assumed right of access) UncoInshire PE24 SNS (in respect of an assumed right of access) Unknown (in respect of covenants contained in Conveyance dated 22 February 1991)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-011	1026 square metres of agricultural land (south of Marsh Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (In respect of water apparatus) David Mark Simpson Whitehouse Farm Addlethorpe SECENESS Lincolnshire PE24 ATU (In respect of an assumed right of way) Julia Clare Pound Aysthord Court Westeligh TVERTON EX16 7HL (In respect of an assumed right of access) Lindsey Marsh Drainage Board Weilington House Manby LOUTH Linoshtrie LV11 8UU (In respect of riparian rights) Mark Caudwell Limited Howlet House Farm Hogsthorpe Skegness Lincolnshire EX11 8US (In respect of riparian rights) Mark Caudwell Limited Howlet House Farm Hogsthorpe Skegness Lincolnshire <tr< td=""></tr<>	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3 County of LincoInshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
07-011		Mark Leopold Caudwell	
cont'd		Howlett House	
		Slackholme End	
		Hogsthorpe	
		SKEGNESS	
		LincoInshire PE24 5NS	
		(in respect of an assumed right of access)	
		(in respect of an assumed right of access)	
		Maureen Teresa Caudwell	
		Howlett House	
		Slackholme End	
		Hogsthorpe	
		SKEGNESS	
		LincoInshire	
		PE24 5NS	
		(in respect of an assumed right of access)	
		Unknown	
07-012	591 square metres of agricultural land (south of Marsh Lane)	Anglian Water Services Limited	
		Lancaster House	
		Lancaster Way	
		Ermine Business Park	
		HUNTINGDON	
		Cambridgeshire	
		PE29 3NZ	
		(in respect of water apparatus)	
		David Mark Simpson	
		Whitehouse Farm	
		Addlethorpe	
		SKEGNESS	
		Lincolnshire	
		PE24 4TU	
		(in respect of a right of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
07-012 cont'd		Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
08-001	17 square metres of access splay (north of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
08-002	4 square metres of access splay (north of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SNS 6PB (in respect of an option agreement dated 29 March 2019)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-002 cont'd		Unknown	
08-003	626 square metres of agricultural land and access track (north of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (in respect of rights and restrictive covenants granted by Deed dated 29 March 2019) Unknown (in respect of the rights granted by the Deed dated 07 December 1966)	
08-004	1 square metres of drain (north of Marsh Lane)	David Peter HandStone ArchesChapel LaneAddlethorpeSKEGNESSLincolnshirePE24 4TG(in respect of rights as stated in Conveyance dated 18 February 1991)Graham Kenneth HandGrange FarmChapel LaneAddlethorpeSKEGNESSLincolnshirePE24 4TG(in respect of rights as stated in Conveyance dated 18 February 1991)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-005	844 square metres of drain (north of Marsh Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
08-006	420 square metres of agricultural land (south of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	

		Outer Dowsing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-007	17857 square metres of agricultural land (south of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) RLS House Limited Alex House 260-268 Chapel Street SALFORD Greater Manchester M3 5JZ (in respect of informal rights agreement for car parking) Unknown (in respect of a right of way as contained in Conveyance dated 06 April 1962)	
08-008	39 square metres of agricultural land (south of Marsh Lane)	RLS House Limited Alex House 260-268 Chapel Street SALFORD Greater Manchester M3 5JZ (in respect of informal rights agreement for car parking) Unknown (in respect of a right of way as contained in Conveyance dated 06 April 1962)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-009	7624 square metres of agricultural land (south of Marsh Lane)	RLS House Limited Alex House 260-268 Chapel Street SALFORD Greater Manchester M3 5JZ (in respect of informal rights agreement for car parking) Unknown (in respect of a right of way as contained in Conveyance dated 06 April 1962)	
08-010	48 square metres of agricultural land (south of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) RLS House Limited Alex House 260-268 Chapel Street SALFORD Greater Manchester M3 5JZ (in respect of informal rights agreement for car parking) Unknown (in respect of a right of way as contained in Conveyance dated 06 April 1962)	
08-011	7 square metres of access track and drain (east of Skegness Stadium, Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-011 cont'd		Unknown	
08-012	77 square metres of access track (east of Skegness Stadium)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of a Wayleave Agreement relating to the construction and maintenance of an electric line dated 01 April 1992)	
08-013	99 square metres of drain (east of Skegness Stadium)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
08-014	29191 square metres of agricultural land (south of Marsh Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3 County of LincoInshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
08-014		National Grid Electricity Distribution (East Midlands) PLC	
cont'd		Avonbank	
		Feeder Road	
		BRISTOL	
		Avon	
1		BS2 OTB	
		(in respect of a Wayleave Agreement relating to the construction and maintenance of an electric line dated 01 April 1992)	
08-015	123 square metres of access track (south of Marsh Lane)	Lindsey Marsh Drainage Board	
		Wellington House	
		Manby Park	
		Manby	
		LOUTH	
		Lincolnshire	
		LN11 8UU	
		(in respect of riparian rights)	
		Unknown	
08.016		Andlen Weter Constant Limited	
08-016	48130 square metres of agricultural land, access track and drain (north of Ingoldmells Road)	Lancaster House	
		Lancaster House	
		Ermine Business Park	
		HUNTINGDON	
		Cambridgeshire	
		PE29 3NZ	
		(in respect of sewer apparatus)	
		Lindsey Marsh Drainage Board	
		Wellington House	
		Manby Park	
		Manby	
		LOUTH	
		Lincolnshire	
		LN11 8UU	
		(assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
08-016 cont'd		Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (In respect of rights as stated in deed dated 28 January 2019) Unknown (in respect of rights contained in Conveyance dated 08 September 1948)	
08-017	763 square metres of access track (north of Ingoldmells Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewer apparatus) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wittshire SNS 6PB (In respect of rights as stated in deed dated 28 January 2019) Unknown (in respect of rights contained in Conveyance dated 08 September 1948)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-001	26 square metres of access splay (south of Ingoldmells Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
09-002	976 square metres of public road and drain (Ingoldmells Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	
09-003	152 square metres of drain (south of Ingoldmells Road)	Unknown	

	Outer Dowsi	ing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-004	59860 square metres of agricultural land and drains (south of Ingoldmells Road and north of		
	Younger's Lane)	Wellington House	
		Manby Park	
		Manby	
		LOUTH	
		Lincolnshire	
		LN11 8UU	
		(in respect of riparian rights)	
		Unknown	
		(in respect of right of way for a drift road reserved by a conveyance dated 8 September 1948)	
09-005	9375 square metres of access track and drain (north of Younger's Lane and south of	Lindsey Marsh Drainage Board	
	Ingoldmells Road)	Wellington House	
		Manby Park	
		Manby	
		LOUTH	
		Lincolnshire	
		LN11 8UU	
		(in respect of riparian rights)	
		Unknown	
		(in respect of right of way for a drift road reserved by a conveyance dated 8 September 1948)	
09-006	167 square metres drain (north of Younger's Lane)	Unknown	
09-007	3644 square metres of agricultural land and copse (north of Younger's Lane)	National Grid Electricity Distribution (East Midlands) PLC	
		Avonbank	
		Feeder Road	
		BRISTOL	
		Avon	
		BS2 OTB	
		(in respect of overhead apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-008	2633 square metres of agricultural land (north of Younger's Lane)	J E Spence & Son Farms Ltd Mill Hill Farm Millfield Road Bratoft SKEGNESS Lincolnshire PE24 5AN (in respect of a right to install water pipes granted in Deed dated 06 June 1961)	
09-009	30 square metres of verge (north of Younger's Lane)	Unknown	
09-010	7 square metres of verge (north of Younger's Lane)	L.J. Fairburn & Son Limited Ivy House Farm Farlesthorpe Road Bilsby ALFORD Lincolnshire LN13 9PL (in respect of rights of way and covenants listed in a Conveyance dated 30 November 2000 & in relation to rights granted in a Conveyance dated 02 March 1981)	
09-012	269 square metres of hedgerow and copse (north of Younger's Lane)	Unknown	
09-015	29 square metres of access track (north of Younger's Lane)	Unknown	
09-016	223 square metres of verge and hedgerow (North of Younger's Lane)	Unknown	
09-017	718 square metres of public road and verge (Younger's Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-017 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
09-018	7608 square metres of agricultural land (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
09-019	729 square metres of agricultural land and verge (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
09-020	1011 square metres of agricultural land and copse (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
09-021	747 square metres of agricultural land (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
09-022	16419 square metres of agricultural land and drains (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-001	13692 square metres of agricultural land (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	
10-002	20400 square metres of agricultural land, copse and drains (south of Younger's Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
10-003	9204 square metres of agricultural land (west of Middlemarsh Road)	Annie Ruth Dalrymple The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (in respect of rights reserved by Transfer dated 22 December 2009)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
10.000		(Applications: Prescribed Forms and Procedures) Regulations 2009	
10-003		Claire Margaret Wilson	
cont'd		The Chestnuts Wainfleet Road	
		Burgh Le Marsh SKEGNESS	
		Lincolnshire	
		PE24 5AH	
		(in respect of rights reserved by Transfer dated 22 December 2009)	
		(in respect of rights reserved by transfer dated 22 betember 2005)	
		Luke Colin Martin Mackinder	
		The Chestnuts	
		Wainfleet Road	
		Burgh Le Marsh	
		SKEGNESS	
		Lincolnshire	
		PE24 5AH	
		(in respect of rights reserved by Transfer dated 22 December 2009)	
		Mary Ann Mackinder-Jonas	
		The Chestnuts	
		Wainfleet Road	
		Burgh Le Marsh	
		SKEGNESS	
		Lincolnshire	
		PE24 5AH	
		(in respect of rights reserved by Transfer dated 22 December 2009)	
		National Grid Electricity Distribution (East Midlands) PLC	
		Avonbank	
		Feeder Road	
		BRISTOL	
		Avon	
		BS2 OTB	
		(in respect of overhead electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-003 cont'd		SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (in respect of rights reserved by Transfer dated 22 December 2009)
10-004	9307 square metres of agricultural land, copse and drains (west of Middlemarsh Road)	Annie Ruth Dalrymple The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire P224 5AH (in respect of rights reserved by Transfer dated 22 December 2009) Claire Margaret Wilson The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (in respect of rights reserved by Transfer dated 22 December 2009) Luke Colin Martin Mackinder The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH (in respect of rights reserved by Transfer dated 22 December 2009) Luke Colin Martin Mackinder The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 5AH Lincolnshire PE24 5AH Lincolnshire PE24 5AH

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-004 cont'd		Mary Ann Mackinder-Jonas The Chestnuts Wainfleet Road Burgh Le Marsh SKEGNESS Lincolnshire PE24 SAH (in respect of rights reserved by Transfer dated 22 December 2009) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB SKEG Grain Limited Burgh Road West SKEGNESS Lincolnshire PE24 4UF (in respect of rights reserved by Transfer dated 22 December 2009)	
10-005	212 square metres of copse and drain (west of Middlemarsh Road)	Unknown	
10-006	26361 square metres of agricultural land and drains (north of Skegness Road, A158)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (in respect of rights reserved by Conveyance dated 05 February 1898)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-006 cont'd		Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINODON Cambridgeshire PE29 3NZ (In respect of water apparatus) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LN10 6YH (in respect of rights reserved by Conveyance dated 05 February 1898) Ian George Read Meadow House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (in respect of rights reserved by Conveyance dated 05 February 1898) Lindsey Marsh Drainage Board Wellington House Main Road Langton HORNCASTLE Lincolnshire LN9 5JT (in respect of rights reserved by Conveyance dated 05 February 1898) Lindsey Marsh Drainage Board Wellington House Manby Park Manby Park Manby LOUTH Lincolnshire LN1 8UU (in respect of riparian rights)	

	Outer Dows	sing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-007	3803 square metres of agricultural land, access tracks and drains (north of Skegness Road, A158)	Andrew Charles Read Roughton House Farm Moor Lane Roughton WOODHALL SPA Lincolnshire LIN10 6YH (in respect of rights reserved by Conveyance dated 05 February 1898) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Edward Martin Read Honeyhole Moor Lane Roughton WOODHALL SPA Lincolnshire LIN10 6YH (in respect of rights reserved by Conveyance dated 05 February 1898) Ian George Read Meadow House Main Road Langton HORKASTLE Lincolnshire Lincoln	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
Number	County of LincoInshire Names of all those entitled to enjoy easements or other private rights over land		
Number on Land Plans	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-007 cont'd	2 square metres of drain (north of Skegness Road, A158)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
10-009	29 square metres of access track and drain (north of Skegness Road, A158)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	

	Outer Dows	sing Offshore Wind Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of Lincolnshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
10-010	880 square metres of agricultural land (north of Skegness Road, A158) and public footpath (LL BurM 265/2)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)
10-011	244 square metres of drain (north of Skegness Road, A158)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown
10-015	31 square metres of copse (north of Skegness Road, A158)	Unknown
10-016	92 square metres of copse (north of Skegness Road, A158)	Unknown
10-017	2998 square metres of public road and verge (Skegness Road, A158)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
10-017 cont'd		Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of apparatus) Unknown	
10-018	23231 square metres of agricultural land (south of Skegness Road, A158)	Unknown	
10-019	24721 square metres of agricultural land (south of Skegness Road, A158)	Unknown	
10-020	7 square metres of agricultural land (south of Skegness Road, A158)	Unknown	

	Outer Dows	ing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3
		County of Lincolnshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
	2360 square metres of agricultural land (south of Skegness Road, A158) and public footpath (LL BurM 260/1)	
	97 square metres of agricultural land (south of Skegness Road, A158)	Unknown
11-002	163 square metres of agricultural land and drain (south of Skegness Road, A158)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 GPB (as beneficiary of an easement dated 21 June 2018) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-003	20065 square metres of agricultural land and drains (south of Skegness Road, A158)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 21 June 2018) Unknown	
11-004	958 square metres of agricultural land (south of Skegness Road, A158)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 21 June 2018) Unknown	
11-005	1202 square metres of agricultural land (south of Skegness Road, A158)	Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 2 July 2019) Unknown	
11-006	7646 square metres of agricultural land and hedgerow (west of Middlemarsh Road) and public footpath (LL BurM 261/3)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-006 cont'd		Triton Knoll Offshore Wind Farm Limited Windmill Hill Business Park Whitehill Way SWINDON Wiltshire SN5 6PB (as beneficiary of an easement dated 2 July 2019) Unknown	
	2825 square metres of agricultural land (west of Middlemarsh Road) and public footpath (LL BurM 261/3)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
11-008	3 square metres of drain and hedgerow (west of Middlemarsh Road)	Unknown	
11-010	18844 square metres of agricultural land and drain (north of Billgate Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	
11-011	167 square metres of agricultural land (north of Billgate Lane)	Unknown	
11-014	1359 square metres of agricultural land (north of Billgate Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
11-016	571 square metres of public road and verge (north of Billgate Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	
11-017	260 square metres of agricultural land (south of Billgate Lane)	Unknown	
11-019	7460 square metres of agricultural land (north of Middlemarsh Road) and public footpath (LL BurM 263/2)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	
11-020	881 square metres of public road and verges (Middlemarsh Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
11-020 cont'd		Unknown
11-021	698 square metres of agricultural land and access track (south of Middlemarsh Road)	Unknown
11-022	110 square metres of access track (south of Middlemarsh Road)	Unknown
11-023	20163 square metres of agricultural land (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown (in respect of rights granted by Deed dated 04 August 1981)
12-001	20 square metres of agricultural land (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)

	Outer E	Dowsing Offshore Wind Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of Lincolnshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-001 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus) Unknown
12-002	1500 square metres of agricultural land (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground and overhead electricity apparatus)
12-003	43 square metres of agricultural land and drain (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-003 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
12-004	240 square metres of agricultural land (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown (in respect of rights granted by Deed dated 04 August 1981)	
12-005	304 square metres of agricultural land (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-005 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
12-006	1271 square metres of agricultural land (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
12-007	481 square metres of drain (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-008	11156 square metres of agricultural land and drains (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
12-009	133 square metres of drain (east of Middlemarsh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
12-010	33143 square metres of agricultural land and drains (north of Low Road)	June Smith Park Farm Main Road Roughton WOODHALL SPA LincoInshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LIncoInshire LN11 8UU (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-010 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of deed of grant for underground and overhead electricity apparatus dated 6 May 1939) The Executor of the Estate of the Late Andrew Leslie Smith Park Farm Main Road Roughton WOODHALL SPA Lincolnshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)	
12-011	4704 square metres of private road, verges, agricultural land and access tracks (north of Low Road)	June Smith Park Farm Main Road Roughton WOODHALL SPA LincoInshire LN10 GYJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (in respect of riparian rights) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunications apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order	
		BOOK OF REFERENCE - PART 3
		County of Lincolnshire
		Names of all those entitled to enjoy easements or other private rights over land
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning
		(Applications: Prescribed Forms and Procedures) Regulations 2009
12-011		The Executor of the Estate of the Late Andrew Leslie Smith
cont'd		Park Farm
		Main Road
		Roughton
		WOODHALL SPA
		Lincolnshire
		LN10 GYJ
		(in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)
12-013	9271 square metres of private road, verges and agricultural land (north of Low Road)	June Smith
		Park Farm
		Main Road
		Roughton
		WOODHALL SPA
		Lincolnshire
		LN10 GYJ
		(in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)
		Lindsey Marsh Drainage Board
		Wellington House
		Manby Park
		Manby
		LOUTH
		Lincolnshire
		LN11 8UU
		(assumed in respect of rights of access)
		Openreach Limited
		6 Gracechurch Street
		LONDON
		EC3V 0AT
		(in respect of underground telecommunications apparatus)

	(Duter Dowsing Offshore Wind Development Consent Order	
		BOOK OF REFERENCE - PART 3	
	County of Lincolnshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
12-013		The Executor of the Estate of the Late Andrew Leslie Smith	
cont'd		Park Farm	
		Main Road	
		Roughton	
		WOODHALL SPA	
		Lincolnshire	
		LN10 6YJ	
		(in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)	
12-014	657 square metres of private road and verges (north of Low Road)	June Smith	
		Park Farm	
		Main Road	
		Roughton	
		WOODHALL SPA	
		Lincolnshire	
		LN10 GYJ	
		(in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)	
		Lindsey Marsh Drainage Board	
		Wellington House	
		Manby Park	
		Manby	
		LOUTH	
		Lincolnshire	
		LN11 8UU	
		(assumed in respect of rights of access)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of underground telecommunications apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-014 cont'd		The Executor of the Estate of the Late Andrew Leslie Smith Park Farm Main Road Roughton WOODHALL SPA LincoInshire LN10 GYJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)	
12-015	12 square metres of private road and verge (north of Low Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of overhead telecoms apparatus) Unknown	
12-016	1295 square metres of agricultural land, drain and verge (north of Low Road)	June Smith Park Farm Main Road Roughton WOODHALL SPA LincoInshire LN10 GYJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3	
		County of Lincolnshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
12-017	863 square metres of agricultural land, drain and verge (north of Low Road)	June Smith Park Farm Main Road Roughton WOODHALL SPA LincoInshire LN10 6YJ (in respect of water apparatus rights granted by Deed of Grant dated 23 August 1957)
12-018	705 square metres of public road and verges (Low Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
12-019	167 square metres of agricultural land (south of Low Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
12-020	25718 square metres of agricultural land and drains (west of Pinchbeck Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	
13-001	144 square metres of verge (north of Pinchbeck Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ	

	Outer Do	wsing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-001 cont'd		Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
13-002	581 square metres of public road, verges and drains (Pinchbeck Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 &UU (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-003	209 square metres of drain (south of Pinchbeck Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of drainage rights) Unknown	
13-004	52355 square metres of agricultural land and drains (south of Pinchbeck Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of rights granted in Wayleave Consent dated 23 June 1969)	
13-005	339 square metres of agricultural land and drain (south of Pinchbeck Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsi	ing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3	
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
	Drove)		
13-007	1601 square metres of access track (east of Gutheram Drove)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
13-008	1702 square metres of agricultural land and access track (north of Gutheram Drove)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
	637 square metres of agricultural land and drain (south of Gutheram Drove) 433 square metres of agricultural land (north of Croft Bank, A52)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) UNHNOWN	
13-011	37136 square metres of agricultural land, hedgerow and drains (north of Croft Bank, A52)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
13-014	143 square metres of agricultural land and verge (north of Croft Bank, A52)	Unknown	

	Outer Dov	wsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire
		County of Lincoinshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
13-015	670 square metres of agricultural land (south of Gutheram Drove)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)
13-016	33 square metres of agricultural land (south of Gutheram Drove)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus) Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-017	645 square metres of agricultural land and drains (south of Gutheram Drove) and public footpath (LL Crof 264/1)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
13-018	303 square metres of agricultural land and drain (south of Gutheram Drove)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
13-019	14335 square metres of agricultural land, access track and drain (south of Gutheram Drove) and public footpath (LL Crof 264/1)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
13-020	253 square metres of agricultural land and drain (south of Gutheram Drove)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
13-021	17907 square metres of agricultural land (east of Church Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
14-001	66 square metres of agricultural land (south of Gutheram Drove)	Unknown	
14-003	21 square metres of agricultural land (east of Church Lane)	Unknown	
14-004	76 square metres of agricultural land (east of Church Lane)	Unknown	
14-005	60 square metres of agricultural land (east of Church Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
14-006	755 square metres of public road and verges (Church Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-006 cont'd		Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC	
		Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
14-007	207 square metres of hedgerow (west of Church Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-007 cont'd		Unknown	
14-008	10649 square metres of agricultural land and drains (west of Church Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
14-009	49828 square metres of agricultural land, drains and verge (east of Croft Lane) and public footpath (LL Crof 276/2, LL Crof 276/3 and LL Crof 276/4)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of a restrictive covenant as stated in Deed dated 18 April 1977)	
14-010	2617 square metres of agricultural land and access track (west of Church Lane) and public footpath (LL Crof 276/4)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon SS2 OTB (in respect of overhead electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-010 cont'd		Unknown (in respect of a restrictive covenant as stated in Deed dated 18 April 1977)	
14-011	758 square metres of public roads and verges (Croft Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
14-012	318 square metres of watercourse (The Lymm)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-012 cont'd		Unknown	
14-013	1111 square metres of scrubland and hedgerow (west of Croft Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
14-014	777 square metres of public roads and verges (East End)	Anglian Water Services LimitedLancaster HouseLancaster WayErmine Business ParkHUNTINGDONCambridgeshirePE29 3NZ(in respect of underground water apparatus)Environment AgencyHorizon HouseDeanery RoadBRISTOLAvonBS1 5AH(assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
14-014 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus) Unknown	
14-015	62761 square metres of agricultural land and drains (east of Wainfleet Road, B1195)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
14-016	317 square metres of agricultural land (west of East End)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-001	1286 square metres of agricultural land (east of Wainfleet Road, B1195)	Unknown	
15-002	1 square metres of agricultural land (east of Wainfleet Road, B1195)	Unknown	
15-003	1173 square metres of agricultural land (east of Wainfleet Road, B1195)	Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	
15-004	10351 square metres of agricultural land (east of Wainfleet Road, B1195)	Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	
15-005	1901 square metres of access track, agricultural land and drain (north of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	
15-006	227 square metres of drain (east of Wainfleet Road, B1195)	Unknown	
15-008	42 square metres of verge (north of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-009	73 square metres of access track (north of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	
15-010	6 square metres of verge (Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
15-011	1551 square metres of public road, verges and drains (Wainfleet Road, B1195)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LINCOLN Lincolnshire LINCOLN Lincolnshire LINT YL (in respect of street furniture)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-011 cont'd		Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown	
15-012	204 square metres of agricultural land and drain (west of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
15-013	17709 square metres of agricultural land (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-013 cont'd		Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	
15-014	49 square metres of agricultural land (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	
15-015	293 square metres of drain (south of Wainfleet Road, B1195)	Unknown	
15-016	18594 square metres of agricultural land (south of Wainfleet Road, B1195)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
15-017	11326 square metres of agricultural land (south of Wainfleet Road, B1195)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-018	47 square metres of access track and drain (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus and proposed underground telecommunication apparatus) Unknown	
15-019	11 square metres of access track (south of Wainfleet Road, B1195)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecoms apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-019 cont'd		Unknown	
15-020	5 square metres of access track (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecoms apparatus) Unknown	
15-021	49 square metres of access track and drain (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-022	211 square metres of agricultural land (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown (in respect of rights of way listed in Conveyance dated 06 April 1921)	
15-023	148 square metres of agricultural land (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	

	Outer Dows	sing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3
		County of Lincolnshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
15-024	1132 square metres of agricultural land and access track (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)
15-025	903 square metres of agricultural land and access track (south of Wainfleet Road, B1195)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
		County of Lincoinsnire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-025 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
15-026	328 square metres of agricultural land and access track (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	
15-027	6 square metres of agricultural land and access track (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-027 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
15-028	9 square metres of agricultural land (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
15-029	18 square metres of agricultural land and access track (south of Wainfleet Road, B1195)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
		County of Enconstine	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-029 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
15-030	32 square metres of agricultural land (south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	
	476 square metres of agricultural land, access track (south of Wainfleet Road, B1195) and drain (Wainfleet Relief Channel)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-032	337 square metres of access track (south of Wainfleet Road, B1195) and drain (Wainfleet Relief Channel)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
15-033	354 square metres of agricultural land, access track (south of Wainfleet Road, B1195) and drain (Wainfleet Relief Channel)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
15-034	15 square metres of agricultural land (north of Brewster Lane and south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-035	4 square metres of agricultural land (north of Brewster Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) Unknown	
	29 square metres of agricultural land and drain (north of Brewster Lane and south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
15-037	865 square metres of agricultural land (north of Brewster Lane and south of Wainfleet Road, B1195)	Unknown	
15-038	2 square metres of agricultural land (north of Brewster Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH LincoInshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-038 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
	439 square metres of agricultural land and drain (north of Brewster Lane and south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
15-041	11110 square metres of agricultural land, access tracks (south of Wainfleet Road, B1195) and drains (Wainfleet Relief Channel)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-041 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
15-042	23156 square metres of agricultural land and drain (north of Brewster Lane and south of Wainfleet Road, B1195)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	
15-043	17872 square metres of agricultural land (north of Brewster Lane)	Unknown	
15-044	515 square metres of agricultural land and drain (north of Brewster Lane)	Unknown	
15-045	363 square metres of agricultural land (north of Brewster Lane)	Unknown	
15-046	929 square metres of agricultural land and drain (north of Brewster Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-047	1355 square metres of public road and verges (Brewster Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lindsey Marsh Drainage Board Wellington House Manby DUTH Lincolshire LN11 8UU (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
15-048	82 square metres of drain (south of Brewster Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-048 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
15-049	92 square metres of agricultural land (south of Brewster Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights)	
15-050	18382 square metres of agricultural land (south of Brewster Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
15-051	1035 square metres of agricultural land (south of Brewster Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
15-052	4600 square metres of agricultural land and drain (south of Brewster Lane and north of Collison Gate)	Unknown	
15-053	1975 square metres of railway (Wainfleet and Boston) and works	Unknown	
15-054	10247 square metres of agricultural land (north of Collison Gate)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus)	
15-055	172 square metres of railway (Wainfleet and Boston) and works	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus) Unknown	
15-056	6543 square metres of agricultural land (north of Collison Gate)	Network Rail Infrastructure Limited Waterloo General Office LONDON SE1 8SW (in respect of rail apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-001	6 square metres of access splay and drain (south of Collison Gate)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
16-002	45 square metres of access splay and drain (south of Collison Gate)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (assumed in respect of rights of access)	
16-003	1706 square metres of public road (Collison Gate)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of street furniture)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-003 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground and overhead electricity apparatus)	
16-004	172 square metres of drain (south of Collison Gate)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
16-005	19985 square metres of agricultural land (south of Collison Gate and east of Crow's Lane) 126 square metres of drain (south of Collison Gate)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of riparian rights) Unknown	
16-009	1197 square metres of public road (Mill Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-009 cont'd		Unknown	
16-010	1716 square metres of river (Steeping River), foreshore, bed and banks thereof	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)	
16-011	2231 square metres of river (Steeping River)	Unknown	
16-013	561 square metres of public road and access track (Mill Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-014	252 square metros of vorge (couth of Mill Lang)	Openreach Limited	
10-014	252 square metres of verge (south of Mill Lane)	6 Gracechurch Street LONDON EC3V 0AT	
		(in respect of underground telecommunication apparatus) Witham Fourth District Internal Drainage Board	
		47 Norfolk Street BOSTON	
		Lincolnshire PE21 6PP	
		(assumed in respect of rights of access)	
16-015	246 square metres of verge (south of Mill Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire	
		PE21 6PP (assumed in respect of rights of access)	
16-016	48443 square metres of agricultural land and drain (south of Mill Lane and west of Washdike Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	
		Unknown (in respect of rights contained in Conveyance dated 22 July 1919)	
		Unknown (in respect of a right of way contained in Conveyance dated 06 April 1907)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-016 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
16-017	3088 square metres of agricultural land and drain (north of Church Lane and west of Washdike Lane)	Unknown (in respect of rights contained in Conveyance dated 22 July 1919) Unknown (in respect of a right of way contained in Conveyance dated 06 April 1907) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
16-018	701 square metres of agricultural land (north of Church Lane and west of Washdike Lane)	Unknown (in respect of rights contained in Conveyance dated 22 July 1919) Unknown (in respect of a right of way contained in Conveyance dated 06 April 1907)	
16-019	518 square metres of agricultural land (north of Church Lane and west of Washdike Lane)	Unknown	
16-020	39 square metres of drain (north of Church Lane)	Unknown	
16-022	182 square metres of drain (north of Church Lane)	Unknown	
16-023	120 square metres of drain (north of Church Lane)	Unknown (in respect of rights contained in Conveyance dated 22 July 1919) Unknown (in respect of a right of way contained in Conveyance dated 06 April 1907)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
16-024	13592 square metres of agricultural land (north of Church Lane)	Unknown		
16-025	1638 square metres of public road, verge and copse (Church Lane)	Unknown		
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (assumed in respect of rights of access)		
16-026	33 square metres of agricultural land (south of Church Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)		
16-027	8061 square metres of agricultural land and drain (south of Church Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)		
16-028	212 square metres of agricultural land (south of Church Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)		

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
16-029	6389 square metres of agricultural land (south of Church Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
16-030	436 square metres of drain (south of Church Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
16-031	14513 square metres of agricultural land and path (south of Church Lane)	Unknown (in respect of rights relating to sewers, drains and water pipes listed in Deed of Gift dated 10 April 1992) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-001	1021 square metres of access track (north of Scald Gate) and public footpath (LL WStM 371/1)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-002	234 square metres of drain (south of Church Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-002 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 GPP (assumed in respect of rights of access)	
17-003	6428 square metres of agricultural land (south of Church Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (assumed in respect of rights of access)	
17-004	1 square metres of agricultural land (west of Hall Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-005	98 square metres of agricultural land (south of Church Lane)	Unknown (in respect of rights relating to sewers, drains and water pipes listed in Deed of Gift dated 10 April 1992) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (assumed in respect of rights of access)	
17-006	437 square metres of drain (south of Church Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-006 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-007	9897 square metres of agricultural land (south of Church Lane)	Unknown (in respect of rights relating to sewers, drains and water pipes listed in Deed of Gift dated 10 April 1992)	
17-008	28 square metres of drain (south of Church Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-009	250 square metres of drain (south of Church Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-010	1334 square metres of agricultural land (west of Hall Gate)	Unknown (in respect of rights of entry listed in Conveyance dated 31 December 1919) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-011	535 square metres of drain (south of Church Lane)	Unknown	
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-012	81 square metres of drain (west of Hall Gate)	Unknown	
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-013	11550 square metres of agricultural land (east of Burgh Road and south of Church Lane)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of drainage rights) Unknown (in respect of rights reserved by Conveyance dated 12 July 1919) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-014	10843 square metres of agricultural land (east of Burgh Road)	Sarah Elizabeth Scupholm 18 Rumbold Lane Wainfleet SKEGNESS PE24 4DS (in respect of restrictive covenants as imposed by restriction dated 18 August 2022) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-015	283 square metres of drain (north of Scald Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-016	12993 square metres of agricultural land and drain (north of Scald Gate)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-017	6306 square metres of drain and copse (north of Scald Gate)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-018	133 square metres of drain (north of Scald Gate)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-018 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-019	395 square metres of agricultural land (north of Scald Gate)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-020	27 square metres of agricultural land (north of Scald Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-021	165 square metres of drain (north of Scald Gate)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-021 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-022	731 square metres of public road and verge (Scald Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-023	241 square metres of drain (south of Scald Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-024	15081 square metres of agricultural land (south of Scald Gate)	Unknown (in respect of right of entry as stated by an Indenture dated 09 October 1919) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
17-025	10 square metres of access splay (north of Scald Gate)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-025 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
17-026	384 square metres of drain (south of Scald Gate)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of drainage rights) Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP	
17-027	35018 square metres of agricultural (east of Burgh Road)	(in respect of riparian rights) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	
17-028	89 square metres of agricultural land (east of Burgh Road)	Unknown	
17-030	198 square metres of drain and copse (east of Burgh Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
17-032	17 square metres of drain (east of Burgh Road)	Unknown	
17-033	248 square metres of agricultural land (east of Burgh Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of rights contained in Conveyance dated 16 February 1948)	
17-034	11589 square metres of agricultural land (east of Burgh Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of rights contained in Conveyance dated 16 February 1948)	
17-036	30 square metres of drain (east of Scald Gate)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
17-037	27 square metres of drain (east of Scald Gate)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus and proposed underground telecommunication apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-001	759 square metres of public road (Burgh Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	
18-002	163 square metres of drain (west of Burgh Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunication apparatus) Unknown	
18-004	2998 square metres of agricultural land (west of Burgh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of a right of way and drainage rights)	
18-005	345 square metres of drain (west of Burgh Road)	Unknown	

		Outer Dowsing Offshore Wind Development Consent Order
		BOOK OF REFERENCE - PART 3
		County of Lincolnshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
18-005 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)
18-006	5614 square metres of agricultural land (west of Burgh Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)
18-007	3783 square metres of agricultural land (west of Burgh Road)	John Charles Thomas Winslow Moorfield Burgh Road Friskney BOSTON PE22 8NS (in respect of rights to run services, use of land, entry, support and others listed in Conveyance dated 25 August 1995) Susan Helen Winslow Moorfield Burgh Road Friskney BOSTON PE22 8NS (in respect of rights to run services, use of land, entry, support and others listed in Conveyance dated 25 August 1995) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)

		Outer Dowsing Offshore Wind Development Consent Order		
		BOOK OF REFERENCE - PART 3		
		County of Lincolnshire		
Number on		Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished,		
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning		
		(Applications: Prescribed Forms and Procedures) Regulations 2009		
18-008	1960 square metres of agricultural land (west of Burgh Road)	John Charles Thomas Winslow		
		Moorfield		
		Burgh Road		
		Friskney		
		BOSTON		
		PE22 8NS		
		(in respect of rights to run services, use of land, entry, support and others listed in Conveyance dated 25 August 1995)		
		Susan Helen Winslow		
		Moorfield		
		Burgh Road		
		Friskney		
		BOSTON		
		PE22 8NS		
		(in respect of rights to run services, use of land, entry, support and others listed in Conveyance dated 25 August 1995)		
		Witham Fourth District Internal Drainage Board		
		47 Norfolk Street		
		BOSTON		
		Lincolnshire		
		PE21 6PP		
		(assumed in respect of rights of access)		
18-010	189 square metres of drain (west of Burgh Road)	Lindsey Marsh Drainage Board		
		Wellington House		
		Manby Park		
		Manby		
		LOUTH		
		Lincolnshire		
		LN11 8UU		
		(in respect of drainage rights)		
		Unknown		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-011	9164 square metres of agricultural land (west of Burgh Road)	Gary Brunt Ty Bach Burgh Road Friskney BOSTON PE22 8NT (in respect of covenants relating to maintenance of a boundary hedge listed in Conveyance dated 25 August 1983)	
18-012	3057 square metres of agricultural land and access track (west of Burgh Road)	Gary Brunt Ty Bach Burgh Road Friskney BOSTON PE22 8NT (in respect of covenants relating to maintenance of a boundary hedge listed in Conveyance dated 25 August 1983)	
18-013	174 square metres of drain (west of Burgh Road)	Lindsey Marsh Drainage Board Wellington House Manby Park Manby LOUTH Lincolnshire LN11 8UU (in respect of drainage rights) Unknown	
18-014	28313 square metres of agricultural land (north of Fen Bank)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of rights contained in Conveyance dated 16 February 1948)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-015	3368 square metres of access track (north of Fen Bank and west of Burgh Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of rights contained in Conveyance dated 16 February 1948)	
18-016	548 square metres of watercourse (Fodder Dike Bank)	Unknown (in respect of unknown restrictive covenants which may have been imposed prior to 6 July 2021) Boston & District Angling Association 13 Brand End Road Butterwick BOSTON PE22 0ET (in respect of fishing rights)	
18-017	371 square metres of drain (north of Fen Bank)	Unknown (in respect of unknown restrictive covenants which may have been imposed prior to 6 July 2021)	
18-018	1517 square metres of public road (Fen Bank)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	
18-021	215 square metres of drain (south of Fen Bank)	Unknown	
18-024	211 square metres of agricultural land (east of Burgh Road)	Unknown	
18-026	206 square metres of agricultural land (west of Burgh Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
L		-	
Number		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
18-027	51 square metres of access splay (west of Burgh Road)	(Applications: Prescribed Forms and Procedures) Regulations 2009 Unknown	
18-027	ST square metres of access splay (west of bulgh road)	UTIKIOWI	
18-030	199 square metres of drain (east of Cranberry Lane)	Unknown	
18-031	48 square metres of access splay (west of Burgh Road)	Unknown	
10 001			
18-032	22 square metres of access splay (west of Cranberry Lane)	Unknown	
		Witham Fourth District Internal Drainage Board	
		47 Norfolk Street	
		BOSTON	
		Lincolnshire	
		PE21 6PP	
		(assumed in respect of rights of access)	
18-033	48 square metres of agricultural land (east of Cranberry Lane)	Unknown	
10 055		(in respect of rights and restrictive covenants stated in an unknown document as may have been imposed before 5 December 2011)	
		(···· ································	
18-034	10876 square metres of agricultural land (east of Cranberry Lane)	Unknown	
		(in respect of rights and restrictive covenants stated in an unknown document as may have been imposed before 5 December 2011)	
18-035	124 square metres of agricultural land and access track (east of Cranberry Lane)	Unknown	
		(in respect of rights and restrictive covenants stated in an unknown document as may have been imposed before 5 December 2011)	
18-036	543 square metres of agricultural land (west of Cranberry Lane)	Colin Campbell	
	· · · · · · · · · · · · · · · · · · ·	1 Hudson Crescent	
		LEIGH-ON-SEA	
		SS9 5XT	
		(in respect of provisions and covenants listed in Transfer dated 03 September 2004)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
18-036 cont'd		Pamela Ann Campbell 1 Hudson Crescent LEIGH-ON-SEA SS9 5XT (in respect of provisions and covenants listed in Transfer dated 03 September 2004) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
18-037	1013 square metres of public road and verge (Cranberry Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
18-038	249 square metres of drain (west of Cranberry Lane)	Unknown	
19-001	9533 square metres of agricultural land (west of Cranberry Lane)	Colin Campbell 1 Hudson Crescent LEIGH-ON-SEA SS9 5XT (in respect of provisions and covenants listed in Transfer dated 03 September 2004) Pamela Ann Campbell 1 Hudson Crescent LEIGH-ON-SEA SS9 5XT (in respect of provisions and covenants listed in Transfer dated 03 September 2004)	

	Outer Dowsing Offshore Wind Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-001 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
19-002	168 square metres of drain (west of Cranberry Lane)	Unknown	
19-003	17041 square metres of agricultural land and access track (north of Mill Hill)	Unknown (in respect of unknown restrictive covenants which may have been imposed prior to 3 April 1996) Unknown (in respect of right to use conduits under the property together with a right of entry to clean, repair and replace as stated in transfer dated 29 March 1996) Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of a right of access and a right to lay, maintain and repair electricity apparatus as stated in Agreement dated 26 April 2024)	
19-004	194 square metres of copse (north of Mill Hill) (excluding all interests of the Crown)	Unknown	
19-005	2608 square metres of agricultural land (north of Mill Hill)	Unknown (n respect of unknown restrictive covenants which may have been imposed prior to 3 April 1996) Unknown (in respect of right to use conduits under the property together with a right of entry to clean, repair and replace as stated in transfer dated 29 March 1996)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-005 cont'd		Cornerstone Telecommunications Infrastructure Limited Hive 2 1530 Arlington Business Park Theale READING Berkshire RG7 4SA (in respect of a right of access and a right to lay, maintain and repair electricity apparatus as stated in Agreement dated 26 April 2024)	
19-006	25424 square metres of agricultural land (north of Mill Hill) (excluding all interests of the Crown)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) SG Kleinwort Hambros Bank Limited One Bank Street Canary Wharf LONDON E14 4SG (in respect of rights listed in Conveyance dated 29 September 1987)	
19-007	60 square metres of agricultural land (north of Mill Hill) (excluding all interests of the Crown)	SG Kleinwort Hambros Bank Limited One Bank Street Canary Wharf LONDON E14 4SG (in respect of rights listed in Conveyance dated 29 September 1987)	
19-008	181 square metres of drain (north of Mill Hill) (excluding all interests of the Crown)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-009	901 square metres of public road (Mill Hill) (excluding all interests of the Crown)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
19-011	246 square metres of drain (south of Mill Hill and north of Church Lane)	Unknown	
19-012	9 square metres of agricultural land (south of Mill Hill and east of Small End Road)	Unknown	
19-014	249 square metres of drain and copse (south of Mill Hill and east of Small End Road)	Unknown	
19-016	155 square metres of drain and copse (south of Mill Hill and east of Small End Road)	Unknown	
19-017	44 square metres of drain (south of Mill Hill and east of Small End Road)	Unknown	
	4524 square metres of agricultural land and copse (north of Church Lane and east of Small End Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-019	153 square metres of drain and copse (south of Mill Hill and east of Small End Road)	Unknown	
19-020	4818 square metres of agricultural land (east of Small End Road and north of Church Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
19-021	174 square metres of agricultural land (east of Small End Road and north of Church Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
19-022	1367 square metres of public road (Small End Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
19-023	877 square metres of agricultural land (west of Small End Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)	
19-024	37859 square metres of agricultural land and drain (west of Small End Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
19-025	16703 square metres of agricultural land (north of Skirmore Road)	Unknown	
20-001	5695 square metres of agricultural land (north of Skirmore Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
20-002	42 square metres of agricultural land and drain (north of Skirmore Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
20-003	14409 square metres of agricultural land (north of Skirmore Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
20-004	12067 square metres of agricultural land and drain (north of Skirmore Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
20-005	1140 square metres of public road, verges and drain (Skirmore Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
20-005 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunications apparatus) Unknown	
20-006	161 square metres of drain (south of Skirmore Road)	Unknown	
20-007	149 square metres of access splay (south of Skirmore Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
	51881 square metres of agricultural land (south of Skirmore Road and east of Patman's Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
20-009	99 square metres of agricultural land (south of Howgarth Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus and proposed underground telecommunication apparatus) Unknown	
21-001	32 square metres of access splay (west of Patman's Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
21-002	1123 square metres of agricultural land (west of Patman's Lane)	Roughton Farms Limited The Grange Eaudyke Road Friskney BOSTON Lincolnshire PE22 8RT (in respect of the rights reserved by conveyance dated 27 November 1975 and right to use drainage systems as stated in conveyance dated 28 October 2009)	
21-003	1418 square metres of public road and verges (Patman's Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 GPP (assumed in respect of rights of access)	
21-004	173 square metres of drain (west of Patman's Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
21-005	35215 square metres of agricultural land (west of Patman's Lane)	Roughton Farms Limited The Grange Eaudyke Road Friskney BOSTON Lincolnshire PE22 8RT (in respect of the rights reserved by conveyance dated 27 November 1975 and right to use drainage systems as stated in conveyance dated 28 October 2009)	
21-006	21128 square metres of agricultural land (east of Ivery Lane)	Unknown	
21-007	11 square metres of access splay (east of Ivery Lane)	Unknown	
21-008	1162 square metres of public road, verges and drain (Ivery Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
21-011	19862 square metres of agricultural road and drain (west of Ivery Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
22-001	6053 square metres of agricultural land and drain (west of Ivery Lane) (excluding all interests of the Crown)	Unknown (in respect of rights granted by Conveyance dated 29 August 1987) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
22-002	12 square metres of drain (west of Ivery Lane) (excluding all interests of the Crown)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
22-003	32 square metres of agricultural land and drain (west of Ivery Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
22-004	7797 square metres of agricultural land (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	Unknown (in respect of rights granted by Conveyance dated 29 August 1987)	
22-005	117 square metres of drain (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	Unknown	
22-006	29 square metres of agricultural land (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	Unknown (in respect of rights granted by Conveyance dated 29 August 1987)	
22-007	9 square metres of drain (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
22-008	41 square metres of agricultural land (west of Ivery Lane and south of Love Lane)	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 0DW (in respect of rights reserved in Transfer dated 22 February 1995) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
22-009	9803 square metres of agricultural land and copse (south of Love Lane)	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 ODW (in respect of rights reserved in Transfer dated 22 February 1995) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
22-010	1644 square metres of agricultural land (east of Broad Gate)	Andrew John Dickinson Old House Farm Crowhall Lane Benington BOSTON Lincolnshire PE22 ODW (in respect of rights reserved in Transfer dated 22 February 1995)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
22-010 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
22-011	150 square metres of drain (south of Love Lane)	Unknown	
22-013	381 square metres of drain (south of Love Lane)	Unknown	
22-015	15 square metres of drain (south of Love Lane)	Unknown	
22-019	8 square metres of drain (east of Broad Gate)	Unknown	
22-020	1306 square metres of agricultural land (east of Broad Gate)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
22-021	825 square metres of drain (east of Broad Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
22-022	948 square metres of agricultural land (east of Broad Gate)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
22-024	814 square metres of agricultural land (east of Broad Gate)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
22-025	15 square metres of access track (east of Broad Gate)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
22-026	12419 square metres of agricultural land (east of Broad Gate)	Unknown	
22-027	538 square metres of public road and verges (Broad Gate)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown	
22-028	13791 square metres of agricultural land (west of Broad Gate)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
22-029	2011 square metres of agricultural land (south of Cragmire Lane and west of Broad Gate)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
22-031	2601 square metres of agricultural land (south of Cragmire Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
22-032	2201 square metres of agricultural land and access track (south of Cragmire Lane)	Unknown (in respect of restrictive covenants as may have been imposed thereon before 20 July 2010)	
23-002	179 square metres of drain (east of Cragmire Lane)	Unknown	
23-004	13 square metres of drain (east of Cragmire Lane)	Unknown	
23-007	20 square metres of drain (south of Cragmire Lane)	Unknown	
23-009	1940 square metres of agricultural land and access track (east of Cragmire Lane)	Christopher William Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (in respect of rights reserved by Transfer dated 10 November 1995)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-009 cont'd		Jane Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (in respect of rights reserved by Transfer dated 10 November 1995) More Tank Edwards New Farm Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (in respect of rights reserved by Transfer dated 10 November 1995) Robert John Edwards The Mallards Soulby Lane Wrangle BOSTON Lincolnshire PE22 9BT (in respect of rights reserved by Transfer dated 10 November 1995)	
23-010	141 square metres of drain (east of Cragmire Lane)	Unknown	
23-011	847 square metres of public road and verges (Cragmire Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunications apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-012	12013 square metres of agricultural land (west of Cragmire Road)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
23-013	12266 square metres of agricultural land (north of Cragmire Lane and east of Double Bank) (excluding all interests of the Crown)	Unknown (in respect of rights granted by Conveyance dated 29 September 1987) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
23-014	1482 square metres of agricultural land (north of Cragmire Lane)(excluding all interests of the Crown)	Unknown (in respect of rights granted by Conveyance dated 29 September 1987) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
23-015	218 square metres of drain (east of Double Bank) (excluding all interests of the Crown)	Unknown	
23-016	14153 square metres of agricultural land (east of Common Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Do	wsing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-017	4615 square metres of drain (east of Double Bank)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
23-018	62 square metres of access splay (east of Double Bank)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 GPP (assumed in respect of rights of access)	
23-019	583 square metres of drain (east of Common Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunications apparatus) Unknown	
23-020	1138 square metres of public road, drain and verges (Common Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-022	10657 square metres of agricultural land (west of Common Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
23-023	12779 square metres of agricultural land (west of Double Bank)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	
23-024	13221 square metres of agricultural land (west of Common Road)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
23-025	604 square metres of agricultural land (west of Common Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-025 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
23-026	34 square metres of agricultural land (west of Common Road)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
23-027	9538 square metres of agricultural land (west of Common Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
23-028	31 square metres of agricultural land (west of Common Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
23-029	83 square metres of access splay (west of Common Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
23-030	28270 square metres of agricultural land and drain (east of Manor Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
23-032	396 square metres of public road and verges (Manor Lane)	Unknown	
23-033	328 square metres of public road and verges (Manor Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground apparatus)	
24-001	723 square metres of agricultural land (west of Manor Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
24-002	25766 square metres of agricultural land (east of Sea Dyke and west of Manor Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
24-003	969 square metres of public road, drain and verges (Sea Dyke)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	
24-004	179 square metres of access splay (west of Sea Dyke)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground electricity apparatus) Unknown	
24-005	52453 square metres of agricultural land, hedgerow, access track and drain (west of Sea Dyke)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
24-006	1253 square metres of access track and drain (west of Sea Dyke)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
24-006 cont'd		Unknown	
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
24-007	2588 square metres of agricultural land and access track (west of Seadyke)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
24-008	705 square metres of agricultural land (east of Church Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
24-009	705 square metres of access track (east of Church Road)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
24-010	20630 square metres of agricultural land, access track and drain (south of Fold Hill)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
24-011	335 square metres of access track (south of Fold Hill)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
24-012	1913 square metres of access track and drain (south of Fold Hill)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
24-013	24351 square metres of agricultural land and drain (east of Church Road)	National Grid Electricity Distribution (East Midlands) PLCAvonbankFeeder RoadBRISTOLAvonBS2 0TB(in respect of overhead electricity apparatus)Openreach Limited6 Gracechurch StreetLONDONEC3V 0AT(in respect of underground telecommunication apparatus)	
25-001	8 square metres of agricultural land (east of Church Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-001 cont'd		Unknown	
25-002	205 square metres of public road and verge (Church Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus)	
25-003	627 square metres of public road (Church Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
25-004	4516 square metres of agricultural land (west of Church Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-005	26572 square metres of agricultural land (west of Church Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)	
25-006	1218 square metres of public road (Sibsey Road, B1184)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-006 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
25-007	158 square metres of drain (west of Sibsey Road, B1184)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
25-009	784 square metres of public road and verge (Cowbroads Lane)	Unknown	
25-010	13742 square metres of agricultural land (south of The Gride)	Unknown	
25-018	234 square metres of agricultural land (north of Pode Lane)	Unknown	
25-019	5592 square metres of agricultural land (north of Pode Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
25-020	368 square metres of agricultural land (north of Pode Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-022	836 square metres of public road and verge (Pode Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus) Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (assumed in respect of rights of access)	
25-023	408 square metres of agricultural land (south of Pode Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
25-024	3015 square metres of agricultural land (south of Pode Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
25-025	1376 square metres of agricultural land and drain (south of Pode Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-026	81 square metres of drain (south of Pode Lane)	Unknown	
25-027	11 square metres of agricultural land (south of Pode Lane)	Unknown	
25-028	68 square metres of access track (south of Pode Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
25-029	18 square metres of agricultural land (south of Pode Lane)	Unknown (in respect of restrictive covenants as may have been imposed thereon before 20 July 2010) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
25-030	21731 square metres of agricultural land (south of Pode Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
25-031	12131 square metres of agricultural land and drain (south of Pode Lane and east of Skipmarsh Lane)	Unknown (in respect of restrictive covenants as may have been imposed thereon before 20 July 2010)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
25-031 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
25-032	7515 square metres of agricultural land and drain (east of Skipmarsh Lane)	Unknown	
25-033	15025 square metres of agricultural land and drain (east of Skipmarsh Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground sewerage apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
26-001	271 square metres of agricultural land (east of Skipmarsh Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground sewerage apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
26-003	758 square metres of public road and verges (Skipmarsh Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	
	153 square metres of agricultural land (west of Skipmarsh Lane) 24344 square metres of agricultural land (east of Southfields and west of Skipmarsh Lane)	Unknown National Grid Electricity Distribution (East Midlands) PLC	
		Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
26-006	1067 square metres of public road and verges (Southfields)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
26-007	88 square metres of access track and drain (west of Southfields)	47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access) Unknown	
26-008	14 square metres of access track and drain (west of Southfields)	Unknown	
26-009	3900 square metres of agricultural land (west of Southfields)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
26-010	318 square metres of drain (west of Southfields)	Unknown	
26-011	16555 square metres of agricultural land and drain (west of Southfields)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
26-013	827 square metres of public road and verges (Ings Drove)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
26-014	143 square metres of drain (south of Ings Drove)	B.S. Pension Fund Trustee Limited 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (in respect of covenants relating to use of land listed in Transfer dated 07 April 1998) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of rights stated in Conveyance dated 18 December 1981)	
26-015	321 square metres of agricultural land (south of Ings Drove)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
26-016	389 square metres of agricultural land (south of Ings Drove)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
26-017	56 square metres of drain (south of Ings Drove)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
26-018	37348 square metres of agricultural land (south of Ings Drove)	B.S. Pension Fund Trustee Limited 1 Ashley Road ALTRINCHAM Cheshire WA14 2DT (in respect of covenants relating to use of land listed in Transfer dated 07 April 1998) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of rights stated in Conveyance dated 18 December 1981) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
26-019	80 square metres of agricultural land (east of Ings Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
26-020	8 square metres of agricultural land (east of Ings Road)	Unknown	
26-021	291 square metres of agricultural land (east of Ings Road)	Unknown	
26-022	6766 square metres of agricultural land (east of Ings Road)	Unknown	
26-023	184 square metres of drain (east of Ings Road)	Unknown	
26-024	18921 square metres of agricultural land and drain (east of Ings Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	
26-025	9084 square metres of agricultural land (east of Ings Road)	Unknown	
27-001	898 square metres of public road and verge (Ings Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
27-001 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	
27-007	10 square metres of drain (west of Ings Road)	Unknown	
27-008	71 square metres of drain (west of Ings Road)	Unknown	
27-009	581 square metres of drain (west of Ings Road)	Unknown	
27-011	6902 square metres of agricultural land and drain (west of Ings Road and east of Ings Drove)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-012	1197 square metres of agricultural land (west of Ings Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
27-012 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-013	2480 square metres of agricultural land (east of Ings Drove)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (assumed in respect of rights of access)	
27-014	8 square metres of access splay (east of Ings Drove)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
27-015	8030 square metres of agricultural land and drain (east of Ings Drove)	Unknown (in respect of easements as affect the same by virtue of Section 15(1)(b) of the Endowments and Glebe Measure 1976) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-016	52 square metres of drain (east of Ings Drove)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
27-016 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-017	25 square metres of agricultural land (east of Ings Drove)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown (in respect of easements as affect the same by virtue of Section 15(1)(b) of the Endowments and Glebe Measure 1976) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-018	1406 square metres of agricultural land and drain (east of Ings Drove)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (in respect of riparian rights)	
27-019	6579 square metres of agricultural land and drain (east of Ings Drove)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
27-019 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
27-020	1789 square metres of agricultural land and access track (north of Ings Bank)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-021	13605 square metres of agricultural land (north of Ings Bank)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
27-022	12 square metres of access track (north of Ings Bank)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
27-023	169 square metres of drain (north of Ings Bank)	Unknown	
27-024	1211 square metres of public road (Ings Bank)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	
27-025	21 square metres of drain (south of Ings Bank and west of Double Bank)	Unknown	
28-001	257 square metres of drain (west of Double Bank)	Unknown	
28-003	11508 square metres of agricultural land (west of Double Bank)	Unknown (in respect of rights that granted by Conveyance dated 27 April 1979) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
28-004	6019 square metres of agricultural land (west of Double Bank)	Unknown (in respect of rights that granted by Conveyance dated 27 April 1979) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-005	211 square metres of drain (west of Double Bank)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON LincoInshire PE21 6PP (in respect of riparian rights)	
28-007	20289 square metres of agricultural land (east of Hobhole Bank)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
28-008	1766 square metres of agricultural land (north of Lowfields Lane)	Unknown	
28-009	497 square metres of agricultural land (north of Lowfields Lane)	Unknown	
28-012	185 square metres of drain (north of Lowfields Lane)	Unknown	
28-017	629 square metres of public road and verge (Lowfields Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
28-018	167 square metres of hedgerow and drain (south of Lowfields Lane)	Unknown	
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
28-019	74304 square metres of agricultural land (Oak House Farm)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
29-001	327 square metres of access track (Oak House Farm)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
29-002	41 square metres of access splay and drain (west of Swandyke Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
29-002 cont'd		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
29-003	676 square metres of agricultural land (Swandyke Farm)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	
29-004	515 square metres of hedgerow and drain (west of Swandyke Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
29-005	54368 square metres of agricultural land, hedgerow and access track (west of Swanhole Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a surface water drain) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of electricity apparatus)	

IDENTIFY INTERFERENCE - PART 3 County of Lincolnshire Number on Number on Description of Lind Number of all these entitled to any assessments or other private rights or subset (including private rights of marginal income related in the private rights or under frequentian 7(1)(1) of the infrastructure Phanning (Applications: Prescribed Forms and Procedures) Regulations 2009 29:007 6699 square metres of agricultural land and hedgerow (north of Wainfleet Road, AS2) County of first (County of first) (Including and counts) Regulations 2009 (Applications: Prescribed Forms and Procedures) Regulations 2009 29:008 91 square metres of footwarys and verge (Wainfleet Road, AS2) Operasch Limited 6 Garcechurdh Street LOMOON (In respect of a surface water drain) 29:009 864 square metres of public road, verges and footwarys (Wainfleet Road, AS2) Operasch Limited 6 Garcechurdh Street LOMOON (In respect of telecommunications apparatus) 29:009 864 square metres of public road, verges and footwarys (Wainfleet Road, AS2) Demension finite accaster Way Emine Roadings Park 29:009 864 square metres of public road, verges and footwarys (Wainfleet Road, AS2) Demension finite accaster Way Emine Roadings Park 29:009 864 square metres of public road, verges and footwarys (Wainfleet Road, AS2) Demension finite accaster Way Emine Roadings Park 29:009 864 square metres of public road, verges and footwarys (Wainfleet Road, AS2) Unecony Wainfleet Park <t< th=""><th></th><th colspan="3">Outer Dowsing Offshore Wind Development Consent Order</th></t<>		Outer Dowsing Offshore Wind Development Consent Order		
Number on Land Plans Description of Land Names of all those entitled to enjoy easements or other private rights of ravigation over waterly which it is proposed shall be extinguished, suspender of interfered with under Regulations 71(1) of the instructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 29-007 6699 square metres of agricultural land and hedgerow (north of Wainfleet Road, A52) Uncoinshire County Council Uncoinshire County Council UNCOUN Uncoinshire County Council UNCOUN 29-008 91 square metres of footways and verge (Wainfleet Road, A52) Openreach Limited 6 Graechurdt Street LONDON Openreach Limited 6 Graechurdt Street LONDON 29-008 864 square metres of public road, verges and footways (Wainfleet Road, A52) Openreach Limited ancechardt Street CONDON Cambridgestire Cambridgestire (In respect of a lumited Cambridgestire (In respect of audice House Cambridgestire PE3 342 Apglical Water Sparatus) 29-008 864 square metres of public road, verges and footways (Wainfleet Road, A52) Anglical Water Sparatus) 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Anglical Water Sparatus) Cambridgestire (In respect of lumited Ansay Park With Midding Cambridgestire PE3 342 (In respect of lumited Ansay Park With Midding CV 39 JU In respect of underground gas apparatus) Cambridgestire Verget Midlands CV 39 JU				
Number on Land Plans Description of Land (Including private rights of navigation over water) which it is proposed shall be extinguished. suspended or interfered with user Regulation X12(or land) 29-007 6699 square metres of agricultural land and hedgerow (north of Wainfleet Road, A52) Uncomshire County Outlices Newland Lincolnshire Unconshire Outly Offices Newland 29-008 91 square metres of footways and verge (Wainfleet Road, A52) Openeach Limited S Gradechurch Street LONDON Openeach Limited S Gradechurch Street LONDON 29-008 84 square metres of public road, verges and footways (Wainfleet Road, A52) Openeach Limited S Gradechurch Street LONDON Openeach Limited S Gradechurch Street LONDON 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Anglian Water Services Limited Lancaster House Lancaster House Anglian Water Services Limited Lancaster House Lancaster House			County of Lincolnshire	
Land Plans Description of Land suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009 29-007 6699 square metres of agricultural land and hedgerow (north of Wainfleet Road, A52) Lincolnshire County Council Courty Offices NewMand LINCOIN Lincolnshire UN1 31V (In respect of a surface water drain) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT 29-008 91 square metres of footways and verge (Wainfleet Road, A52) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunications apparatus) 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunications apparatus) 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Openreach Limited Lancaster House Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGON Cambridgeshire F223 302 (In respect of underground water apparatus) Cadent Gas Limited Ansty Fark HiGI Way Ansity COVENTRY West Midlands CY 9U (In respect of underground gas apparatus)				
Land mark Subpende or interfered with Luber Regulations (Lip(e) or the Intrastructure Valuing (Applications: Prescribed Forms and Procedures) Regulations 2009 29-007 6699 square metres of agricultural and and hedgerow (north of Wainfleet Road, A52) Uncolnshire County Council County Offices Newland Uncolnshire UN1 11: Uncolnshire County Offices Newland E Graceshurture Street County Offices Newland E Graceshurture Street County Offices Newland E Graceshurture Street County Council County Offices Newland E Graceshurture Street County Offices Newland E Graceshurture Street County Council County Offices Newland E Graceshurture Street Counceshurture Street Counceshurture E County Council E County Street Counceshurture Street Counderground water apparatus)	Number on	Description of Land		
29-007 6699 square metres of agricultural land and hedgerow (north of Wainfleet Road, A52) Lincolnshire county Officios Newland LINCOLN 29-008 91 square metres of footways and verge (Wainfleet Road, A52) Openreach Limited 6 Gracechurdt NoDON 29-008 91 square metres of footways and verge (Wainfleet Road, A52) Openreach Limited 6 Gracechurdt NoDON 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Arglian Water Services Limited Lancaster Way 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Arglian Water Services Limited Lancaster Way 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Arglian Water Services Limited Lancaster Way Emine Business Park HUNTINGDON Cambridgeshrie PE23 SNZ (In respect of underground water apparatus) Cadent Gas Limited Ansty Park Wet Mildands CVP SIJJ (In respect of underground gas apparatus)	Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
29-008 91 square metres of footways and verge (Wainfleet Road, A52) Openreach Limited Gracechurch Street LONDON 29-008 91 square metres of footways and verge (Wainfleet Road, A52) Openreach Limited Gracechurch Street LONDON 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Anglan Water Services Limited Loncaster Way Ermine Business Park 100 Loncaster Way Ermine Business Park Loncaster Way Ermine Business Park 100 Cadent Gas Limited Ansty Park Cadent Gas Limited Pictor of underground water apparatus) Cadent Gas Limited Ansty Park Cadent Gas Limited Pictor of Underground gas apparatus)	-			
Newland Newland UNCOLN Unconshire Linconshire Lin LA1 1YE, Graewater drain) 29-008 91 square metres of footways and verge (Wainfleet Road, A52) Operreach Limited 6 Graewater/units Street CONDON EG3V 0AT 100000 EG3V 0AT (in respect of telecommunications apparatus) 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Angle Street Stree	29-007	6699 square metres of agricultural land and hedgerow (north of Wainfleet Road, A52)		
Image: series of footways and verge (Wainfleet Road, A52)Uncounshire Un 1YL (in respect of a surface water drain)29-00891 square metres of footways and verge (Wainfleet Road, A52)Oefracehurn's treet LONDON (EX3) 0AT (in respect of telecommunications apparatus)29-009864 square metres of public road, verges and footways (Wainfleet Road, A52)Anglan Water Services Limited Lancaster Huse Lancaster Huse Lancaster Huse Lancaster Way Emime Lancaster Way Emime (in respect of underground water apparatus)29-009864 square metres of public road, verges and footways (Wainfleet Road, A52)Anglan Water Services Limited Lancaster Huse Lancaster Huse Lancaster Huse Lancaster Way Emime Lancaster Way Emime Lancaster Way Emime Cambridgeshire PE2 3NZ (in respect of underground water apparatus)29-009864 square metres of public road, verges and footways (Wainfleet Road, A52)Anglan Water Services Limited Lancaster Way Emime Lancaster Way Emime Cambridgeshire PE2 3NZ (in respect of underground water apparatus)29-009864 square metres of public road, verges and footways (Wainfleet Road, A52)Anglan Water Services Limited Lancaster Way Emime Cambridgeshire PE2 3NZ (in respect of underground water apparatus)20-009COVENTRY West Millands COVENTRY West Millands COVENTRYCOVENTRY West Millands COVENTRY				
Image: Second				
LN1 1YL (In respect of a surface water drain) 29-008 91 square metres of footways and verge (Wainfleet Road, A52) Openreach Limited 6 Gracechurch Street LONDON 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Anglian Water Services Limited Lancaster Way 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Anglian Water Services Limited Lancaster Way Cambridgeshire PE29 3N2 (in respect of underground water apparatus) Cambridgeshire PE29 3N2 (in respect of a suified Ansty Park Ansty Park Pilot Way Ansty Park Pilot Way CoveNTRY West Midlands CV7 9JU (in respect of underground gas apparatus)			LINCOLN	
Image: Contract of a surface water drain) Image: Contract of a surface water drain) 29-008 91 square metres of footways and verge (Wainfleet Road, A52) Operreach Limited 6 Gracechurch Street LONDON (C37 0AT (in respect of telecommunications apparatus) 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Anglian Water Services Limited Lancaster House Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Cadent Gas Limited Ansty COVENTRY Nosty COVENTRY Ansty Park Pilot Way Nosty COVENTRY				
29-008 91 square metres of footways and verge (Wainfleet Road, A52) Openreach Limited 6 Gracechurch Street LONDON EGV 0AT (in respect of telecommunications apparatus) 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Cader Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9IJ (in respect of underground gas apparatus)			LN1 1YL	
29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Cadent Gas Limited Ansty PCOVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)			(in respect of a surface water drain)	
29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Anglian Water Services Limited Lancaster House Lancaster Way Errnine Business Park HUNTINGDON Cambridgeshire (in respect of underground water apparatus) Cadent Gas Limited Ansty Park Pilot Way (in respect of underground gas apparatus)	29-008	91 square metres of footways and verge (Wainfleet Road, A52)	Openreach Limited	
EC3V 0AT (in respect of telecommunications apparatus) 29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Anglian Water Services Limited Lancaster House Lancaster House Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)			6 Gracechurch Street	
29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Anglian Water Services Limited Lancaster House Lancaster Way Larchine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)			LONDON	
29-009 864 square metres of public road, verges and footways (Wainfleet Road, A52) Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Cadent Gas Limited Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)			EC3V OAT	
Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)			(in respect of telecommunications apparatus)	
Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty Pilot Way Ansty West Midlands COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)	29-009	864 square metres of public road, verges and footways (Wainfleet Road, A52)	Anglian Water Services Limited	
Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)			Lancaster House	
HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)				
Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)				
PE29 3NZ (in respect of underground water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)				
(in respect of underground water apparatus) Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)				
Cadent Gas Limited Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)				
Ansty Park Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)			(in respect of underground water apparatus)	
Pilot Way Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)			Cadent Gas Limited	
Ansty COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)				
COVENTRY West Midlands CV7 9JU (in respect of underground gas apparatus)				
West Midlands CV7 9JU (in respect of underground gas apparatus)				
CV7 9JU (in respect of underground gas apparatus)				
(in respect of underground gas apparatus)				
Unknown			(in respect of underground gas apparatus)	
			Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
29-010	512 square metres of hedgerow (south of Wainfleet Road, A52)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	
29-011	250 square metres of hedgerow (south of Wainfleet Road, A52)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	
29-012	5604 square metres of agricultural land (south of Wainfleet Road, A52)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
30-004	28 square metres of access splay (west of Foxhole Lane)	Unknown	
30-006	207 square metres of access splay and drain (west of Foxhole Lane)	Unknown	
30-007	135 square metres of access splay and drain (west of Foxhole Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
30-008	915 square metres of public road and verge (Foxhole Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	
30-009	5727 square metres of agricultural land (east of Foxhole Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
30-016	168 square metres of drain (north of Butterwick Road)	Unknown	
30-019	73 square metres of access track and drain (north of Butterwick Road)	Unknown	
30-020	145 square metres of drain (north at Butterwick Road)	Unknown	
30-021	724 square metres of public road and verges (Butterwick Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
30-021 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus) Unknown	
30-022	315 square metres of access track and drain (south of Butterwick Road)	Unknown	
30-023	20273 square metres of agricultural land (south of Butterwick Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus)	
30-024	25374 square metres of agricultural land (north of Shore Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
31-001	217 square metres of access splay and drain (north of Shore Road)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
31-002	725 square metres of public road, verges and footways (Shore Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunication apparatus) Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire	
		PE21 6PP (assumed in respect of rights of access)	
	62198 square metres of agricultural land, hedgerow, access track and drain (east of Church End Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of sewerage apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
31-004 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
31-005	6 square metres of agricultural land (east of Church End Road)	Unknown	
31-006	1065 square metres of public road and verge (Church End Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water and sewerage apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
31-007	62 square metres of agricultural land (west of Church End Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
31-008	34988 square metres of agricultural land (west of Church End Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water and sewerage apparatus)	
31-009	906 square metres of public road, verges and drain (Clampgate Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	
32-001	89 square metres of agricultural land (west of Church End Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water and sewerage apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
32-001 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground and overhead telecommunication apparatus)	
32-003	10676 square metres of agricultural land (east of Hobhole Drain)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of drainage apparatus) Unknown	
		Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
32-004	7717 square metres of agricultural land (east of Grovefield Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of drainage apparatus and street furniture) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 GPP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
32-005	354 square metres of verge (Grovefield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
32-006	328 square metres of public road and verge (Grovefield Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
32-008	5401 square metres of agricultural land (west of Grovefield Lane)	Unknown (in respect of rights stated in Conveyance dated 26 March 1981) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
32-009	268 square metres of drain (west of Grovefield Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
32-010	13 square metres of drain (west of Grovefield Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
32-011	8076 square metres of agricultural land and drain (west of Grovefield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
32-012	646 square metres of agricultural land (west of Grovefield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
32-013	11 square metres of access splay (west of Grovefield Lane)	Unknown	
32-022	68 square metres of agricultural land (west of Grovefield Lane)	Unknown	
32-024	9 square metres of access splay (west of Grovefield Lane)	Unknown	
33-002	261 square metres of drain (north of Cut End Road)	Unknown	
33-003	24317 square metres of agricultural land and access track (east of Cut End Road)	Beverley Elizabeth Lister 2 Marsh Farm Cut End Road Fishtoft BOSTON PE22 0QZ (in respect of service media rights and a right to enter as stated in Transfers dated 20 August 1999 and 14 June 2001)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-003 cont'd		Robert John Lister c/o Beverley Elizabeth Lister 2 Marsh Farm Cut End Road Fishtoft BOSTON Lincolnshire PE22 OQZ (in respect of service media rights and a right to enter as stated in Transfers dated 20 August 1999 and 14 June 2001)	
33-007	48 square metres of access splay (east of Cut End Road)	Unknown	
33-008	8781 square metres of agricultural land (east of Cut End Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of proposed overhead telecommunication apparatus)	
33-009	134 square metres of verge (east of Cut End Road)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of proposed overhead telecommunication apparatus) Unknown	
33-010	677 square metres of public road, verge and drain (Cut End Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-010 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus and proposed overhead telecommunication apparatus) Unknown	
33-011	4507 square metres of agricultural land (east of Woad Lane and west of Cut End Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
33-012	10986 square metres of agricultural land (east of Woad Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-013	1394 square metres of agricultural land (east of Woad Lane)	Unknown (in respect of restrictive covenants dated 13 April 2007 stated in Register dated 20 October 2009)	
33-014	941 square metres of public road and verge (Woad Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	
33-016	15353 square metres of watercourse (Hobhole Drain) and copse (west of Woad Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown (in respect of unknown restrictive covenants which may have been imposed prior to 30 March 2021) Boston & District Angling Association 13 Brand End Road Butterwick BOSTON PE22 0ET (in respect of fishing rights)	
33-017	82 square metres of drain (west of Pinfold Lane)	Unknown	
33-018	259 square metres of drain (north of Pinfold Lane)	Unknown	
33-019	708 square metres of agricultural land, access track and verge (south of Pinfold Lane and east of Southfield Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-020	785 square metres of access track (south of Pinfold Lane and east of Southfield Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
33-021	1491 square metres of agricultural land (south of Pinfold Lane and east of Southfield Lane)	Unknown Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
33-022	45 square metres of agricultural land (south of Pinfold Lane)	Mark Bolland The White House Pinfold Lane Fishtoft BOSTON Lincolnshire PE21 OSL (in respect of drainage rights granted by a Transfer dated 24 November 2015) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
33-024	262 square metres of agricultural land (south of Pinfold Lane and east of Southfield Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-027	239 square metres of drain (east of Southfield Lane)	Unknown	
33-029	4652 square metres of agricultural land (east of Southfield Lane)	Unknown	
33-030	859 square metres of agricultural land (east of Southfield Lane)	Unknown	
33-032	2345 square metres of agricultural land, access track and drain (east of Southfield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
33-033	3 square metres of agricultural land (east of Southfield Lane)	Unknown	
33-034	12 square metres of agricultural land (east of Southfield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of a right of access to maintain and repair drains)	
33-035	485 square metres of drain (east of Southfield Lane)	Unknown	
33-036	7456 square metres of agricultural land (east of Southfield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-037	8666 square metres of agricultural land and hedgerow (east of Southfield Lane)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of a right of access to maintain and repair drains)	
33-039	9165 square metres of agricultural land and access track (south of Southfield Lane)	Betty Skipworth Nevadun Burton Croft Road Fishtoft BOSTON Lincolnshire PE21 0SA (in respect of covenants and rights reserved by Transfer dated 11 December 1995) Mark William Skipworth Old Orchard Seadyke Road Old Leake BOSTON Lincolnshire PE22 9HX (in respect of covenants and rights reserved by Transfer dated 11 December 1995) Richard Hardy (Fishtoft) Willoughby Huills Fishtoft BOSTON Willoughby Huills Fishtoft BOSTON Kichard Hardy (Fishtoft) Willoughby Huills Fishtoft BOSTON PE21 9SA (in respect of covenants and rights reserved by Transfer dated 11 December 1995)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
33-039 cont'd		The Executor of the Estate of the Late William Garfield Skipworth Nevadun Burton Croft Road Fishtoft BOSTON Lincolnshire PE21 0SA (in respect of covenants and rights reserved by Transfer dated 11 December 1995) Unknown (in respect of a right of way as stated in Conveyance dated 09 May 1968)	
	58974 square metres of agricultural land and access track (east of Scalp Road) and watercourse (The Graft)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of rights stated in Wayleave Agreement dated 17 September 1956) Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
34-001	2500 square metres of access track (east of Scalp Road)	Witham Fourth District Internal Drainage Board 47 Norfolk Street BOSTON Lincolnshire PE21 6PP (in respect of riparian rights)	
34-002	18 square metres of copse (east of Scalp Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-003	2217 square metres of private road, verge and hardstanding (Scalp Road) and public footpath (LL Fish 13/11)	(Applications: Prescribed Forms and Procedures) Regulations 2009 Anglian Water Services Limited Lancaster House Lancaster Way Ernline Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Environment Agency Horizon House Deanery Road BSI 5AH (assumed in respect of rights of access) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LIN11YL (in respect of street furniture) National Grid Electricity Distribution (East Midlands) PLC Avon BSI 50 L Avon Mational Grid Electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-004	7342 square metres of grassed area, shrubbery (west of Scalp Road) and public footpath (LL Fish 13/11)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Unknown	
34-005	1809 square metres of river (The Haven), foreshore, bed and banks thereof (excluding all interests of the Crown)	Unknown (in respect of rights reserved by Deed of Grant dated 25 July 1973)	
	19799 square metres of river (The Haven), foreshore, bed and banks thereof (excluding all interests of the Crown)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Boston Borough Council Municipal Buildings West Street BOSTON Lincolnshire PE21 8QR (in respect of rights granted by Deed of Grant dated 30 October 1936)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-006 cont'd		The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Deed of Grant dated 04 September 1964)	
34-007	104860 square metres of agricultural land and drain (Bleak House Farm)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1VL (in respect of drainage apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-007 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Conveyance dated 15 April 1966) The Royal Society For The Protection Of Birds The Lodge Potton Road SANDY Bedfordshire SG19 2DL (in respect of rights granted by Deed of Grant dated 14 March 2008) Unknown (in respect of rights granted by Deed of Grant dated 14 March 2008) Unknown (in respect of restrictive convenant not to build or erect anything that might damage the jetty work of the Haven of Boston or the ground or foreshore in favour of the Borough of Boston as stated in conveyance dated 1 March 1965)	
34-008	1296 square metres of agricultural land, access track, copse and outbuildings (Bleak House Farm)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-008 cont'd		Environment Agency Horizon House Deaneyr Road BRISTOL Avon BS1 SAH (assumed in respect of rights of access) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground and overhead telecommunication apparatus) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Conveyance dated 15 April 1966) Unknown (in respect of restrictive convenant not to build or erect anything that might damage the jetty work of the Haven of Boston or the ground or foreshore in favour of the Borough of Boston as stated in conveyance dated 1 March 1965)	
34-009	5554 square metres of agricultural land (Bleak House Farm)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Conveyance dated 15 April 1966)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-009 cont'd		Unknown (in respect of restrictive convenant not to build or erect anything that might damage the jetty work of the Haven of Boston or the ground or foreshore in favour of the Borough of Boston as stated in conveyance dated 6 August 1862)	
		Unknown (in respect of all rights as now exist over the outmarsh and adjoining bank as stated in conveyance dated 1 March 1965)	
34-010	6824 square metres of agricultural land and access track (Bleak House Farm)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 SAH (in respect of rights granted by Deed of Grant dated 14 March 2008) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights granted by Conveyance dated 15 April 1966) The Royal Society For The Protection Of Birds The Lodge Potton Road SANDY Bedfordshire SG19 2DL (in respect of rights granted by Deed of Grant dated 14 March 2008) Unknown (in respect of restrictive convenant not to build or erect anything that might damage the jetty work of the Haven of Boston or the ground or foreshore in favour of the Borough of Boston as stated in conveyance dated 6 August 1862) Unknown (in respect of all rights as now exist over the outmarsh and adjoining bank as stated in conveyance dated 1 March 1965)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-011	600 square metres of access track and copse (east of Wyberton Road)	A.E. Lenton Limited Church End Friskney BOSTON Lincolnshire PE22 8NE (in respect of rights to an accessway listed in Deed of Easement dated 18 June 2013) Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights stated in Transfer Scheme document dated 1 September 1989) Lincolnshire Wildlife Trust Banovallum House Manor House Street HORNCASTLE Lincolnshire LN9 5HF (in respect of a right to pass and repass stated in Conveyance dated 12 March 1986) Unknown (in respect of a right to pass and repass stated in Conveyance dated 30 July 1884)	
	133 square metres of access track, verge and copse (east of Wyberton Road) and public footpath (LL Wybe 2/4)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-013	120 square metres of copse (east of Wyberton Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	
34-014	1259 square metres of agricultural land and copse (east of Wyberton Road)	Unknown	
34-015	1105 square metres of public road and verges (Wyberton Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	
34-016	31 square metres of public road and verge (Wyberton Road)	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-016 cont'd		Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	
34-017	51 square metres of agricultural land (west of Wyberton Road)	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	

Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-018	2116 square metres of agricultural land (west of Wyberton Road)	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PEZ0 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PEZ0 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)
34-019	147 square metres of drain and copse (west of Wyberton Road)	Unknown
34-020	56 square metres of agricultural land and access track (west of Wyberton Road)	Unknown
34-021	140 square metres of agricultural land (west of Wyberton Road)	Unknown

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-022	6870 square metres of agricultural land (west of Wyberton Road)	Kim Lesley Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Lesley Jean Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Nicholas Alexander Sermon The Ship Wyberton Roads Wyberton Roads Wyberton Roads Wyberton Roads Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)	
	99628 square metres of agricultural land, hedgerows, copse and drains (north of Frampton Road)	Kim Lesley Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)	

Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
34-023 cont'd		Lesley Jean Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 IBA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) National Grid Electricity Distribution (East Midlands) PLC Avon BS2 OTB (in respect of underground and overhead electricity apparatus) National Grid Electricity Distribution (East Midlands) PLC Avon BS2 OTB (in respect of underground and overhead electricity apparatus) National Grid Electricity Distribution (East Midlands) PLC Avon BS2 OTB (in respect of underground and overhead electricity apparatus) National Grid Electricity Distribution (East Midlands) PLC Avon BS2 OTB (in respect of rights granted by Deed dated 19 August 2004) Nicholas Alexander Sermon The Ship Wyberton BOSTON Lincolnshire PE20 IBA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
34-023		The Black Sluice Internal Drainage Board	
cont'd		North End Business Park	
		Station Road	
		Swineshead	
		BOSTON	
		Lincolnshire	
		PE20 3PW	
		(in respect of riparian rights)	
		Unknown	
		(in respect of rights stated in Conveyance dated 11 January 1984)	
34-024	7288 square metres of agricultural land (west of Wyberton Road)	Kim Lesley Sermon	
		The Ship	
		Wyberton Roads	
		Wyberton	
		BOSTON	
		Lincolnshire	
		PE20 1BA	
		(in respect of rights stated in Conveyance dated 24 June 1987)	
		Lesley Jean Sermon	
		The Ship	
		Wyberton Roads	
		Wyberton	
		BOSTON	
		Lincolnshire	
		PE20 1BA	
		(in respect of rights stated in Conveyance dated 24 June 1987)	
		National Grid Electricity Distribution (East Midlands) PLC	
		Avonbank	
		Feeder Road	
		BRISTOL	
		Avon	
		BS2 OTB	
		(in respect of rights granted by Deed dated 19 August 2004)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
34-024 cont'd		Nicholas Alexander Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987) Unknown (in respect of rights stated in Conveyance dated 11 January 1984)	
34-025	26 square metres of copse (north of Wyberton Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	
35-002	49 square metres of access track (east of Wyberton Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
35-003	39 square metres of agricultural land and copse (north of Wyberton Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)	
35-004	897 square metres of agricultural land (east of Wiley's Lane)	Kim Lesley Sermon The Ship Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Lesley Jean Sermon The Ship Wyberton Roads Wyberton Roads Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Michelle Wright Marsh Farm Cottage Frampton Roads Frampton BOSTON Lincolnshire PE20 1AY (in respect of rights granted by Transfer dated 19 February 2003)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
35-004 cont'd		Neil WrightMarsh Farm CottageFrampton RoadsFramptonBOSTONLincolnshirePE20 1AY(in respect of rights granted by Transfer dated 19 February 2003)Nicholas Alexander SermonThe ShipWyberton RoadsWybertonBOSTONLincolnshirePE20 1BA(in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)The Black Sluice Internal Drainage BoardNorth End Business ParkStation RoadSvinesheadBOSTONLincolnshirePE20 3PW(assumed in respect of rights of access)	
35-005	7 square metres of access track and drain (south of Wiley's Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
35-006	547 square metres of access track (Wiley's Lane)	A.E. Lenton Limited	
		Church End	
		Friskney	
		BOSTON	
		Lincolnshire	
		PE22 8NE	
		(assumed in respect of rights of access)	
		The Black Sluice Internal Drainage Board	
		North End Business Park	
		Station Road	
		Swineshead	
		BOSTON	
		Lincolnshire	
		PE20 3PW	
		(assumed in respect of rights of access)	
35-007	15 square metres of access track (south of Wiley's Lane)	The Black Sluice Internal Drainage Board	
		North End Business Park	
		Station Road	
		Swineshead	
		BOSTON	
		Lincolnshire	
		PE20 3PW	
		(in respect of riparian rights)	
		Unknown	
35-008	1347 square metres of agricultural land and drain (west of Wiley's Lane)	Kim Lesley Sermon	
		The Ship	
		Wyberton Roads	
		Wyberton	
		BOSTON	
		Lincolnshire	
		PE20 1BA	
		(in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)	

Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
35-008 cont'd		Lesley Jean Sermon The Ship Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002) Michelle Wright Marsh Farm Cottage Frampton Roads Frampton BOSTON Lincolnshire PE20 1AY (in respect of rights granted by Transfer dated 19 February 2003) Neil Wright Marsh Farm Cottage Frampton BOSTON Lincolnshire PE20 1AY (in respect of rights granted by Transfer dated 19 February 2003) Neil Wright Marsh Farm Cottage Frampton BOSTON Lincolnshire PE20 1AY (in respect of rights granted by Transfer dated 19 February 2003) Nicholas Alexander Sermon The Ship Wyberton BOSTON Lincolnshire PE20 1BA (in respect of rights stated in Conveyance dated 24 June 1987 and Transfer dated 03 April 2002)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
35-008 cont'd		The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)	
36-002	433 square metres of agricultural land and access track (east of Millfield Lane East)	Unknown	
36-003	33201 square metres of agricultural land (north of Millfield Lane East)	W Dennis & Sons Ltd Kirton House Kirton BOSTON PE20 1JD (in respect of rights and restrictive covenants stated in Conveyance dated 13th April 1948)	
36-004	289 square metres of agricultural land (north of Millfield Lane East)	W Dennis & Sons Ltd Kirton House Kirton BOSTON PE20 1JD (in respect of rights and restrictive covenants stated in Conveyance dated 13th April 1948)	
36-005	34 square metres of agricultural land (west of Low Road)	Church Commissioners For England Church House Great Smith Street LONDON SW1P 3AZ (in respect of restrictions with the Endowments and Glebe Measure 1976 or some other Measure or authority)	
36-006	44 square metres of drain (west of Low Road)	Unknown	
36-007	20 square metres of access track (south of Millfield Lane East)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
36-008	92 square metres of access track (south of Millfield Lane East)	W Dennis & Sons Ltd Kirton House Kirton BOSTON PE20 1JD (in respect of rights and restrictive covenants stated in Conveyance dated 13th April 1948)	
37-001	625 square metres of public road (Frampton Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) The Black Sluice Internal Drainage Board North End Business Park Station Road SWINESS Station Road SWINESS STON Lincolnshire PE20 3PW (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
37-001 cont'd		Unknown	
37-002	264 square metres of copse (south of Frampton Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown	
37-003	10408 square metres of agricultural land and access track (south of Frampton Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown (in respect of rights granted by Conveyance dated 11 January 1984)	
37-004	196 square metres of drain (south of Frampton Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
37-005	41 square metres of access splay (south of Frampton Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown	
37-006	1486 square metres of access track (south of Frampton Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)	
37-007	11 square metres of agricultural land and drain (south of Frampton Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown	
37-008	200 square metres of agricultural land (south of Frampton Road)	Jacqueline Alison Bowens 1 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
37-008 cont'd		James Andrew Brock 2 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018)	
37-009	51823 square metres of agricultural land, access track and drains (Manor Farm)	Jacqueline Alison Bowens 1 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018) James Andrew Brock 2 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018) The Royal Society For The Protection Of Birds The Royal Society For The Protection Of Birds The Lodge Potton Road SANDY Bedfordshire SG19 2DL (in respect of rights granted by Conveyance dated 26 September 1986)	
37-010	458 square metres of agricultural land and access track (Manor Farm)	Jacqueline Alison Bowens 1 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
37-010 cont'd		James Andrew Brock 2 Manor Farm Cottages Sandholme Lane Frampton BOSTON PE20 1AF (in respect of rights granted by Transfers dated 29 April 2002 and 30 May 2018) The Royal Society For The Protection Of Birds The Lodge Potton Road SANDY Bedfordshire SG19 2DL (in respect of rights granted by Conveyance dated 26 September 1986)	
37-011	185 square metres of drain (east of Sandholme Lane) (excluding all interests of the Crown)	Unknown	
37-012	54857 square metres of agricultural land, access track and drain (east of Sandholme Lane) (excluding all interests of the Crown)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus)	
	2663 square metres of agricultural land and access track (east of Sandholme Lane) (excluding all interests of the Crown)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	

	Outer Dowcing Offichara Wind Dovelopment Consent Order		
	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
	37 square metres of access track (north of Sandholme Lane) (excluding all interests of the	Unknown	
	Crown)		
38-003	459 square metres of agricultural land and drain (north of Sandholme Lane) (excluding all	Openreach Limited	
	interests of the Crown)	6 Gracechurch Street	
		LONDON	
		EC3V OAT	
		(in respect of underground telecommunication apparatus)	
38-004	43 square metres of verge (north of Sandholme Lane) (excluding all interests of the Crown)	Unknown	
38-005	112 square metres of public road (Sandholme Lane) (excluding all interests of the Crown)	Anglian Water Services Limited	
		Lancaster House	
		Lancaster Way	
		Ermine Business Park	
		HUNTINGDON	
		Cambridgeshire	
		PE29 3NZ	
		Unknown	
38-006	533 square metres of public road (Sandholme Lane) (excluding all interests of the Crown)	Anglian Water Services Limited	
		Lancaster House	
		Lancaster Way	
		Ermine Business Park	
		HUNTINGDON	
		Cambridgeshire	
		PE29 3NZ	
		(in respect of underground water apparatus)	
		Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
38-008	35612 square metres of agricultural land, copse and drain (north of Marsh Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)	
38-009	605 square metres of agricultural land and access track (south of Sandholme Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)	
39-001	201 square metres of access track and copse (north of Marsh Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown	
39-002	19 square metres of access track and copse (north of Marsh Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-002 cont'd		Unknown	
39-003	208 square metres of public road (Marsh Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)	
39-004	80 square metres of access track (east of Marsh Lane) and public footpath (LL Kirt 877/1)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)	
39-005	5888 square metres of agricultural land and access track (south of Marsh Lane)	Unknown	
39-006	31216 square metres of agricultural land, access track and drain (south of Marsh Lane)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
	22813 square metres of agricultural land, access track and drain (north of Marsh Road) 3313 square metres of agricultural land, access track and drain (north of Marsh Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) The Black Sluice Internal Drainage Board North End Business Park	
		Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown	
39-009	12 square metres of drain (north of Marsh Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON LincoInshire PE20 3PW (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-010	1752 square metres of access track (north of Marsh Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)	
39-011	30 square metres of access track (north of Marsh Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of riparian rights) Unknown	
39-012	1014 square metres of agricultural land (north of Marsh Road)	The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access)	
39-015	623 square metres of public road (Marsh Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-015 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground and overhead telecommunication apparatus) Unknown	
39-016	69856 square metres of agricultural land and drains (south of Marsh Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (In respect of underground and overhead electricity apparatus) Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-016 cont'd		Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	
	613 square metres of agricultural land (south of Marsh Road)	Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	
39-018	25 square metres of agricultural land and access track (south of Marsh Road)	Unknown	
39-019	6 square metres of access track (south of Marsh Road)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-021	16906 square metres of agricultural land (east of Bucklegate Lane)	Openreach Limited 6 Gracechurch Street LONDON EG3V 0AT (in respect of underground and overhead telecommunication apparatus) Richard Paul Tunnard Sandholme House Sandholme House Sandholme House Sandholme House Sandholme House Sandholme Iane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963) Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983) Wendy Christine Tunnard Sandholme House Sandholme House Sandholme House Sandholme House Sandholme Lane Frampton BOSTON Uincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	
39-022	39 square metres of drain (north of Nidd's Lane)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
39-023	112 square metres of agricultural land (south of Nidd's Lane and west of Marsh Road)	James Edward Cheer Burton House Nidds Lane Kirton BOSTON Lincolnshire PE20 1LZ (in respect of rights reserved as stated in Transfer dated 13 July 2007) Joanna Louise Cheer Burton House Nidds Lane Kirton BOSTON Lincolnshire PE20 1LZ (in respect of rights reserved as stated in Transfer dated 13 July 2007) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunications apparatus)	
40-001	1985 square metres of agricultural land (south of Marsh Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus) Richard Paul Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
40-001 cont'd		Unknown (in respect of rights reserved by a conveyance dated 11 October 1963)	
		Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983)	
		Wendy Christine Tunnard Sandholme House Sandholme Lane	
		Frampton BOSTON Lincolnshire PE20 1AG	
		(in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	
40-002	1240 square metres of agricultural land (south of Marsh Road and east of Bucklegate Lane)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus) Richard Paul Tunnard	
		Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG	
		(in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983) Unknown (in respect of rights reserved by a conveyance dated 11 October 1963)	
		Unknown (in respect of rights reserved to light, air, drainage and passaged as stated by a conveyance dated 29 April 1983)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
40-002 cont'd		Wendy Christine Tunnard Sandholme House Sandholme Lane Frampton BOSTON Lincolnshire PE20 1AG (in respect of right of way, light or other easements (if any) as stated in conveyance dated 8 April 1983)	
40-003	23 square metres of access track and verges (east of Clough Lane) and public footpath (LL Kirt 1/5)	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND (in respect of rights of access) James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND (in respect of rights of access) The Black Sluice Internal Drainage Board North End Business Park Station Road Swienshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
40-006	3887 square metres of access track and verge (east of Clough Lane) and public footpath (LL Kirt 1/5)	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 SAH (in respect of rights to maintain sea defences) Howard Stanley Leader Seadyke Cottage Seadyke Cottage Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 10E (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in Deed dated 27 May 2005) James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON Lincolnshire PE20 1ND Lincolnshire PE20 1ND Lincolnshire PE20 1ND Lincolnshire	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
40-006		Lincolnshire County Council	
cont'd		County Offices	
		Newland	
		LINCOLN	
		Lincolnshire	
		LN1 1YL	
		(in respect of rights listed in a Transfer dated 11 May 2018)	
		Melanie Clare Leader	
		Seadyke Cottage	
		Seadyke Road	
		Kirton	
		BOSTON	
		Lincolnshire	
		PE20 1QE	
		(in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in Deed dated 27 May 2005)	
		National Grid Electricity Distribution (East Midlands) PLC	
		Avonbank	
		Feeder Road	
		BRISTOL	
		Avon	
		BS2 OTB	
		The Black Sluice Internal Drainage Board	
		North End Business Park	
		Station Road	
		Swineshead	
		BOSTON	
		Lincolnshire	
		PE20 3PW	
		(assumed in respect of rights of access)	
		Unknown	
		(in respect of reservations, covenants and conditions as stated in Deed dated 23 June 1870)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
40-007	46713 square metres of agricultural land, access track, copse and drain (east of Clough Lane) and public footpath (LL Kirt 1/5)	Amelia Kate Gray Hundred Acre Farm Marsh Road Kirton BOSTON LincoInshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 SAH (in respect of rights to maintain sea defences) Howard Stanley Leader Seadyke Road Kirton BOSTON LincoInshire PE20 1QE (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in Deed dated 27 May 2005) James David Gray Hundred Acre Farm Marsh Road Kirton BOSTON LincoInshire PE20 1ND (in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
40-007 cont'd		Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 YV (in respect of rights listed in a Transfer dated 11 May 2018) Melanie Clare Leader Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in respect of restrictive covenants stated in Deed dated 27 May 2005) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB The Black Sluice Internal Drainage Board North End Business Park Station Road Swineshead BOSTON Lincolnshire PE20 3PW (assumed in respect of rights of access) Unknown (in respect of reservations, covenants and conditions as stated in Deed dated 23 June 1870) Unknown	

	Outer Dow	sing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
40-008	263 square metres of agricultural land access track and drain (west of Clatterdyke Road)	Amelia Kate Gray	
		Hundred Acre Farm	
		Marsh Road	
		Kirton	
		BOSTON	
		Lincolnshire	
		PE20 1ND	
		(in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018)	
		Environment Agency	
		Horizon House	
		Deanery Road	
		BRISTOL	
		Avon	
		BS1 5AH	
		(in respect of rights to maintain sea defences)	
		Howard Stanley Leader	
		Seadyke Cottage	
		Seadyke Road	
		Kirton	
		BOSTON	
		Lincolnshire	
		PE20 1QE	
		(in respect of rights and covenants stated in Transfers dated 10 September 1986 & 11 May 2018 and in Deed dated 27 May 2005)	
		James David Gray	
		Hundred Acre Farm	
		Marsh Road	
		Kirton	
		BOSTON	
		Lincolnshire	
		PE20 1ND	
		(in respect of rights and restrictive covenants stated in Transfer and Pre-Emption agreement dated 17 December 2018)	
		BOSTON Lincolnshire PE20 1ND	

Outer Dowsing Offshore Wind Development Consent Order		
BOOK OF REFERENCE - PART 3 County of Lincolnshire		
private rights over land oposed shall be extinguished,		
he Infrastructure Planning		
Regulations 2009		
May 2018 and in Deed dated 27 May 2005)		
1870)		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
40-009	4982 square metres of access track and copse (east of Low Mill Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of a right of access)	
		Howard Stanley Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of a rights of access)	
		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus)	
		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of a right to pass as stated in Conveyance dated 18 March 1953)	
40-010	58240 square metres of agricultural land, drains and access track (east of Low Mill Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of a right of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
40-010 cont'd		Howard Stanley Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of a rights of access) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of a right to pass as stated in Conveyance dated 18 March 1953)	
41-001	904 square metres of agricultural land and access track (east of Low Mill Lane)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of a right of access) Howard Stanley Leader Seadyke Cottage Seadyke Road Kirton BOSTON Lincolnshire PE20 1QE (in respect of a rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
41-001 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (in respect of a right to pass as stated in Conveyance dated 18 March 1953)	
41-002	499 square metres of drain (east of Low Mill Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
41-003	117929 square metres of agricultural land, access tracks and drains (east of Craven's Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown (in respect of rights granted in Conveyance dated 28 January 1985)	
42-001	72 square metres of access track (east of Low Mill Lane)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
42-001 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
42-002	2399 square metres of access track (east of Low Mill Lane) and public footpath (LL Fosd 8/1)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown (in respect of rights granted in Conveyance dated 28 January 1985)	
42-003	14324 square metres of agricultural land, access track and hardstanding (south of Craven's Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
42-003 cont'd		Unknown (in respect of rights granted in Conveyance dated 28 January 1985)	
	649 square metres of access track, copse, drains and agricultural land (east of Craven's Lane) and public footpath (LL Fosd 8/1)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
42-005	62 square metres of drain (south of Craven's Lane)	Unknown	
42-007	41 square metres of drain (north of Craven's Lane)	Unknown (in respect of rights granted in Conveyance dated 28 January 1985)	
42-008	80 square metres of verge (Craven's Lane)	Unknown	
42-009	59 square metres of drain (south of Craven's Lane)	Unknown	

	Outer Dows	sing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3	
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
42-011	13 square metres of access track (east of Wash Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Unknown	
42-014	50 square metres of agricultural land (east of Wash Road)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of overhead telecommunications apparatus) Unknown	
42-015	3426 square metres of access track, drains and woodland (south of Craven's Lane) and public footpath (LL Fosd 8/1)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (in respect of a secondary flood defence)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
42-015 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights to access and maintain the drain known as Roman Bank Drain) Unknown	
42-017	177 square metres of drain (east of Pullover Lane)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of overhead telecommunications apparatus) Unknown	
42-018	34 square metres of drain (east of Pullover Lane)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
42.010	401 course matters of a minute wood and users (Dullawar Long)	(Applications: Prescribed Forms and Procedures) Regulations 2009 Openreach Limited	
42-019	401 square metres of private road and verges (Pullover Lane)	6 Gracechurch Street	
		LONDON	
		EC3V OAT	
		(in respect of underground telecommunication apparatus)	
		(in respect of underground teleconfinanciation apparatus)	
42-020	4290 square metres of agricultural land and drains (east of Wash Road)	Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of underground telecommunication apparatus)	
42-021	1368 square metres of public road and verges (Wash Road)	Anglian Water Services Limited	
		Lancaster House	
		Lancaster Way	
		Ermine Business Park	
		HUNTINGDON	
		Cambridgeshire	
		PE29 3NZ	
		(in respect of underground water apparatus)	
		Lincolnshire County Council	
		County Offices	
		Newland	
		LINCOLN	
		Lincolnshire	
		LN1 1YL	
		(in respect of rights to a gully)	
		Openreach Limited	
		6 Gracechurch Street	
		LONDON	
		EC3V 0AT	
		(in respect of overhead telecommunications apparatus)	
		Unknown	

	Outer Dov	vsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
42-022	478 square metres of drain (west of Wash Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Unknown
42-024	261 square metres of drain (north of Wash Road)	Unknown
43-002	1185 square metres of public roads, verges, copse and drains (Main Road, A17)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus)

		Outer Dowsing Offshore Wind Development Consent Order		
		BOOK OF REFERENCE - PART 3		
	County of Lincolnshire			
		Names of all those entitled to enjoy easements or other private rights over land		
umber on	Description of Lond	(including private rights of navigation over water) which it is proposed shall be extinguished,		
and Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning		
		(Applications: Prescribed Forms and Procedures) Regulations 2009		
13-003	495 square metres of public roads and verges (Surfleet Bank)	Anglian Water Services Limited		
		Lancaster House		
		Lancaster Way		
		Ermine Business Park		
		HUNTINGDON		
		Cambridgeshire		
		PE29 3NZ		
		(in respect of underground water apparatus)		
		Lincolnshire County Council		
		County Offices		
		Newland		
		LINCOLN		
		Lincolnshire		
		LN1 1YL		
		(in respect of rights to a gully and in respect of street furniture)		
		National Grid Electricity Distribution (East Midlands) PLC		
		Avonbank		
		Feeder Road		
		BRISTOL		
		Avon		
		BS2 0TB		
		(in respect of underground electricity apparatus)		
		Openreach Limited		
		6 Gracechurch Street		
		LONDON		
		EC3V 0AT		
		(in respect of underground telecommunications apparatus)		
		Unknown		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-004	79 square metres of agricultural land (west of Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of drainage apparatus and gully) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown	
43-005	4512 square metres of agricultural land (west of Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a gully)	
43-006	44 square metres of agricultural land (east of Five Towns Drain)	Unknown (in respect of covenants contained in Conveyance dated 04 November 1957)	
43-007	682 square metres of watercourse and banks (Five Towns Drain)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-008	181 square metres of watercourse banks (Five Towns Drain)	Unknown (in respect of covenants contained in Conveyance dated 04 November 1957)	
43-010	1652 square metres of public road, verges and footways (Main Road, A17)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus)	
43-011	952 square metres of public road, verges and footways (Main Road, A17)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of drainage apparatus and street furniture)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-011 cont'd		Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown	
43-012	1992 square metres of agricultural land (south of Graves Farm, west of Main Road, A17)	Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of overhead telecommunication apparatus)	
43-013	132 square metres of access track, grassed area and shrubbery (Graves Farm)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-014	2697 square metres of agricultural land, access track and grassed area (west of Main Road, A17)	Timothy James Gratton Graves Farm Main Road Fosdyke BOSTON PE20 2DB (in respect of a right to service media and right to enter as stated in Transfer dated 27 June 1997)	
43-015	2656 square metres of agricultural land (south of Graves Farm, west of Main Road, A17)	Unknown (in respect of rights in connection to the Dive Towns Drain contained in an undated document)	
43-016	930 square metres of watercourse and banks (Five Towns Drain, west of Main Road, A17)	Unknown	
43-017	288 square metres of watercourse banks (Five Towns Drain)	Unknown (in respect of covenants contained in Conveyance dated 04 November 1957)	
43-019	2301 square metres of copse and drain (west of Main Road, A17) and public bridleway (LL Fosd 2/2)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown	
43-021	239 square metres of agricultural land and access track (south of Smeeton's Lane)	Unknown	
43-022	58 square metres of verge and drain (south of Wash Road and east of Main Road, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights to a gully) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
43-023	144 square metres of grassed area (west of Main Road, A17 and east of Five Towns Drain)	Boston Borough Council	
		Municipal Buildings	
		West Street BOSTON	
		Lincolnshire	
		PE21 8QR	
		(in respect of rights granted by Deed of Grant dated 16th June 1995)	
		National Grid Electricity Distribution (East Midlands) PLC	
		Avonbank	
		Feeder Road	
		BRISTOL	
		Avon BS2 0TB	
		(in respect of underground electricity apparatus)	
		(in respect of underground electricity apparatus)	
		National Highways Limited	
		Bridge House	
		1 Walnut Tree Close	
		GUILDFORD	
		Surrey	
		GU1 4LZ	
		(in respect of rights and restrictive covenants stated in Transfer dated 14 July 2017)	
		Openreach Limited	
		6 Gracechurch Street LONDON	
		ECIVIDON EC3V 0AT	
		(in respect of underground telecommunications apparatus)	
		The Welland And Deepings Internal Drainage Board	
		Deeping House	
		Welland Terrace	
		SPALDING	
		Lincolnshire	
		PE11 2TD	
		(assumed in respect of rights of access)	
		Unknown	
		(in respect of rights as stated in conveyance dated 14 January 1954)	

	Outer Dows	ing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-024	1925 square metres of access track and verges (west of Main Road, A17) and public byway (LL Fosd 2/1)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown	
	568 square metres of access track and verges (west of Main Road, A17) and public byway (LL Fosd 2/1)	Unknown	
43-026	209 square metres of agricultural land (west of Main Road, A17)	Unknown	
43-027	4818 square metres of agricultural land and access track (west of Five Towns Drain)	Unknown (in respect of covenants contained in Conveyance dated 04 November 1957)	
	87 square metres of access track and copse (west of Main Road, A17) (excluding all interests of the Crown)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights stated in Transfer Scheme document dated 1 September 1989)	

	Outer Dows	ing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-029 cont'd		Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of unknown rights listed in Deed dated 14 November 1989)	
43-030	719 square metres of grassed area (west of Main Road, A17)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus)	
43-031	1214 square metres of access track (west of Moulton Washway, A17), public bridleway (LL Fosd 2/2) and public byways (LL Fosd 2/1 and LL Fosd 3/1) (excluding all interests of the Crown)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights stated in Transfer Scheme document dated 1 September 1989) Lincolnshire County Council County Offices Newland LINCOLN LINCOLN Lincolnshire LN1 1YL (in respect of unknown rights listed in Deed dated 14 November 1989) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-031 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
	263 square metres of agricultural land and access track (west of Main Road, A17) and public bridleway (LL Fosd 2/2)	Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown	
43-036	8146 square metres of access track (west of Moulton Washway, A17), public byways (LL Fosd 3/1) (excluding all interests of the Crown)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights stated in Transfer Scheme document dated 1 September 1989) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of unknown rights listed in Deed dated 14 November 1989)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-038	3842 square metres of hardstanding and copse (east of Moulton Washway, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right of way listed in Transfer dated 31 December 1929)	
43-039	2508 square metres of hardstanding (east of Moulton Washway, A17)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL ((in respect of a right of way listed in Transfer dated 31 December 1929) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground electricity apparatus) Unknown (as Kings's Nominee in respect of bona vacantia land for Simpsons (Fosdyke) Limited, dissolved) (in respect of covenants relating to use of land listed in Deed dated 06 November 1996)	
43-040	42 square metres of access track (east of Main Road, A17)	Unknown	
43-041	63 square metres of grassed area (east of Main Road, A17)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-042	94 square metres of access track (east of Main Road, A17) (excluding all interests of the Crown)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of covenants and rights to discharge water from an adjoining highway listed in a Deed of Exchange dated 06 November 1996) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Unknown (in respect of restrictive covenants stated in unknown document dated 1 January 1926)	
43-043	426 square metres of access track (east of Moulton Washway, A17) and public bridleway (LL Fosd 7/1)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right of way listed in Transfer dated 31 December 1929) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground and overhead telecommunication apparatus) Unknown (in respect of a right of way stated in an undated unknown document) Unknown (in respect of manorial rights as stated in Conveyance dated 30 June 1917) Unknown (in respect of manorial rights as stated in conveyance dated 31 December 1929)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-044	17 square metres of access splay (west of Main Road, A17 and south of River Welland)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of telecommunications apparatus)	
43-045	1 square metres of grassed area (west of Main Road, A17 and south of River Welland)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-046	26 square metres of access splay (west of Main Road, A17 and south of River Welland) and public bridleway (LL Fosd 6/1)		

Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
43-046 cont'd		Belinda Elizabeth Needham Sunny View Moulton Washway Fosdyke Bridge SPALDING PE12 6LH (in respect of assumed access) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Hazel Rose Needham Tumbleweed Moulton Marsh SPALDING PE12 6LI (in respect of assumed access) James Needham Sunny View Moulton Washway Fosdyke Bridge SPALDING PE12 6LI (in respect of assumed access) Mary Elenor Hay Wragg Marsh Farm Wragg Marsh SpaLDING SPALDING SPALDING SPALDING PE12 6L4 (in respect of assumed access)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3	
		County of Lincolnshire
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
43-046 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V OAT (in respect of underground telecommunication apparatus) Robin Twigg 16 Cook Road Gosberton SPALDING Lincolnshire PE11 4PD (in respect of assumed access)
43-047	15874 square metres of access track (east of Moulton Washway, A17)	Alan Mowton Shady Nook 5 Old Main Road Fosdyke BOSTON Lincolnshire PE20 2BU (in respect of assumed access)

Outer Dowsing Offshore Wind Development Consent Order			
		BOOK OF REFERENCE - PART 3	
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
42.047			
43-047 cont'd		Alexander James Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of assumed access) Andrew George Hay Wragg Marsh Farm Wragg Marsh Farm Spalding Marsh Spalding Marsh Spalding Marsh Spalding Marsh	
		PE12 6HQ (in respect of assumed access) Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of rights stated in Transfer Scheme document dated 1 September 1989) Belinda Elizabeth Needham Sunny View Moulton Washway Fosdyke Bridge SPALDING PE12 6LH (in respect of assumed access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-047 cont'd		Environment Agency Horizon House Deamery Road BRISTOL Avon BS1 5AH (in respect of rights of entry and covenants listed in Transfer dated 03 March 2021) Hazel Rose Needham Tumbleweed Moulton Marsh SPALDING PE12 6LI (in respect of assumed access) James Needham Sunny View Moulton Washway Foosdyke Bridge SPALDING PE12 6LH (in respect of assumed access) Lincolnshire County Offices Newland LINCOLN LIncolnshire LINCIN	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
43-047 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunications apparatus) Robin Twigg 16 Cook Road Gosberton SPALDING Lincolnshire PE11 4PD (in respect of assumed access)	
	42847 square metres of agricultural land and access track (south of Smeeton's Lane)	John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (in respect of rights reserved by Transfer dated 26 March 2010)	
44-005	2282 square metres of watercourse (Risegate Eau) and banks thereof	Unknown	
44-007	371 square metres of agricultural land (west of Smeeton's Lane)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING LincoInshire PE11 4QB (in respect of a right of way as contained in an unknown undated document)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
44-007 cont'd		John Morris Bratley Tanyard House Quadring Eaudyke SPALDING LincoInshire PE11 4QB (in respect of a right of way as contained in an unknown undated document)	
44-008	80 square metres of agricultural land (south of Sea Bank)	Unknown	
44-009	149 square metres of agricultural land and drain (south of Smeeton's Lane)	Unknown	
44-011	8 square metres of drain (south of Sea Bank)	Unknown	
44-013	196 square metres of drain (south of Sea Bank)	Unknown	
44-016	616 square metres of access track (west of Smeeton's Lane)	Carolyn Margaret Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (in respect of a right of way as contained in an unknown undated document) John Morris Bratley Tanyard House Quadring Eaudyke SPALDING Lincolnshire PE11 4QB (in respect of a right of way as contained in an unknown undated document)	
44-019	102 square metres of banks (south of Risegate Eau)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
44-020	1463 square metres of agricultural land (south of Sea Bank)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971)	
	508 square metres of access track (south of Smeeton's Lane) and public byway (LL Alga 9/1)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)	
44-024	20 square metres of grassed area (south of Smeeton's Lane)	Unknown	
44-027		Alexander James Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access) Andrew George Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access)	

		Outer Dowsing Offshore Wind Development Consent Order	
		BOOK OF REFERENCE - PART 3	
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
44-027 cont'd		Anglian Water Services Limited Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) George Hay & Sons Limited Wykeham Abbey The Chase Wykeham Abbey SPALDING Lincolnshire P112 6HE (in respect of rights of access) Mary Ellenor Hay Wragg Marsh Farm Wragg Marsh Spalling Marsh	

	Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 3 County of Lincolnshire			
		Names of all those entitled to enjoy easements or other private rights over land		
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,		
Land Plans	Description of Early	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning		
		(Applications: Prescribed Forms and Procedures) Regulations 2009		
44-028	17 square metres of access track and verge (south of River Welland)	Alexander James Hay		
		Wragg Marsh Farm		
		Wragg Marsh		
		Marsh Road		
		Spalding Marsh		
		SPALDING		
		Lincolnshire		
		PE12 6HQ		
		(in respect of rights of access)		
		Andrew George Hay		
		Wragg Marsh Farm		
		Wragg Marsh		
		Marsh Road		
		Spalding Marsh		
		SPALDING		
		Lincolnshire		
		PE12 6HQ		
		(in respect of rights of access)		
		Anglian Water Services Limited		
		Lancaster House		
		Lancaster Way		
		Ermine Business Park		
		HUNTINGDON		
		Cambridgeshire		
		PE29 3NZ		
		(in respect of underground water apparatus)		
		George Hay & Sons Limited		
		Wykeham Abbey		
		The Chase		
		Wykeham		
		SPALDING		
		Lincolnshire		
		PE12 6HE		
		(in respect of rights of access)		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
44-028 cont'd		Mary Ellenor Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING LincoInshire PE12 6HQ (in respect of rights of access) Unknown	
44-029	5049 square metres of private accessway and scrubland (south of River Welland)	Alexander James Hay Wragg Marsh Farm Wragg Marsh Marsh Road Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access) Andrew George Hay Wragg Marsh Farm Wragg Marsh Farm Wragg Marsh Spalding Marsh SPALDING Lincolnshire PE12 6HQ (in respect of rights of access)	

Outer Dowsing Offshore Wind Development Consent Order			
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
44-029 cont'd		Anglian Water Services Limited Lancaster May Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) George Hay & Sons Limited Wykeham Abbey The Chase Wykeham Abbey The Chase Wykeham Abbey PE12 GHE (in respect of rights of access) Mary Ellenor Hay Wragg Marsh Farm Wragg Marsh Farm Wragg Marsh Farm Bragg Marsh SPALDING Lincolnshire PE12 GHQ (in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground and overhead electricity apparatus)	

	Outer Dows	sing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-001	5972 square metres of agricultural land and access track (south of Sea Bank and west of Marsh Lane) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING LincoInshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
45-002	5597 square metres of agricultural land, access track and drain (east of A16 and south of Sea Bank) (excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of gas apparatus) Peter C. Thorold Limited S Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-002 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (assumed in respect of rights of access)	
45-003	3692 square metres of agricultural land, access track and drain (east of A16 and south of Sea Bank) (excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of gas apparatus) Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	

	Outer Dows	sing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-004	28 square metres of access splay (south of Marsh Lane) (excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of gas apparatus) Unknown	
	4235 square metres of agricultural land and access track (west of Marsh Lane) (excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of gas apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-005 cont'd		Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
45-006	2529 square metres of agricultural land, access track and drain (east of A16 and south of Sea Bank) (excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of gas apparatus) Peter C. Thorold Limited S Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-007	3 square metres of verge (south of Marsh Lane) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
45-008	1647 square metres of agricultural land (south of Marsh Lane) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-009	1464 square metres of agricultural land, access track and drain (east of A16 and south of Sea Bank) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-009 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-010	94 square metres of watercourse (Risegate Eau) and access track (south of Marsh Lane) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
45-011	28 square metres of watercourse (Risegate Eau) and access track (south of Marsh Lane) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
45-012	71 square metres of watercourse (Risegate Eau) and access track (south of Marsh Lane) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-013	657 square metres of agricultural land, access track and drain (south of Marsh Lane) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
45-014	2483 square metres of agricultural land, access track and drain (south of Marsh Lane) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	

	Outer Dows	ing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3	
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
	2454 square metres of agricultural land (south of Marsh Lane) (excluding all interests of the Crown)		
	2589 square metres of agricultural land, access track and drain (south of Marsh Lane) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	

		BOOK OF REFERENCE - PART 3		
		County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning		
		(Applications: Prescribed Forms and Procedures) Regulations 2009		
45-017 15	5 square metres of drain (south of Marsh Lane) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board Deeping House		
		Welland Terrace		
		SPALDING		
		Lincolnshire		
		PE11 2TD		
		(assumed in respect of rights of access)		
		Unknown		
45-018 72	2 square metres of drain (south of Marsh Lane) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board		
		Deeping House		
		Welland Terrace		
		SPALDING		
		Lincolnshire		
		PE11 2TD		
		(in respect of riparian rights)		
		Unknown		
45.040.45				
45-019 15	575 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House		
		Welland Terrace		
		SPALDING		
		Lincolnshire		
		PE11 2TD		
		(assumed in respect of rights of access)		
45-021 36	667 square metres of agricultural land (south of Marsh Lane)	The Welland And Deepings Internal Drainage Board		
		Deeping House		
		Welland Terrace		
		SPALDING		
		Lincolnshire		
		PE11 2TD		
		(assumed in respect of rights of access)		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-023	1 square metres of drain (south of Marsh Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
45-025	6163 square metres of agricultural land (south of Marsh Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (assumed in respect of rights of access)	
45-026	816 square metres of drain (south of Marsh Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
45-028	14356 square metres of agricultural land, drain and access track (east of The Reservoir, A16)) Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-028 cont'd		National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of underground gas apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-029	718 square metres of drain (south of Marsh Lane) (excluding all interests of the Crown)	National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of underground gas apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-030	27105 square metres of agricultural land (south of Marsh Lane) (excluding all interests of the Crown)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of rights and restrictive covenants granted by Deed dated 02 November 1972) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of gas pipeline apparatus) Peter C. Thorold Limited S Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
45-031	375 square metres of drain (east of A16) (excluding all interests of the Crown)	Unknown	
45-033	3472 square metres of agricultural land (south of Risegate Eau) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
45-034	1036 square metres of drain (east of A16) (excluding all interests of the Crown)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on		(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
45-041	259554 square metres of agricultural land, access track and drain (east of A16)	Lincolnshire County Council	
		County Offices	
		Newland	
		LINCOLN	
		Lincolnshire	
		LN1 1YL	
		(in respect of rights granted in Transfer dated 29 April 1998)	
		The Welland And Deepings Internal Drainage Board	
		Deeping House	
		Welland Terrace	
		SPALDING	
		Lincolnshire	
		PE11 2TD	
		(assumed in respect of rights of access)	
45-043	2023 square metres of watercourse (Risegate Eau) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board	
		Deeping House	
		Welland Terrace	
		SPALDING	
		Lincolnshire	
		PE11 2TD	
		(in respect of riparian rights)	
		Unknown	
45-044	1312 square metres of watercourse (Risegate Eau) (excluding all interests of the Crown)	The Welland And Deepings Internal Drainage Board	
		Deeping House	
		Welland Terrace	
		SPALDING	
		Lincolnshire PE11 2TD	
		(in respect of riparian rights)	
		Unknown	

	Outer Dows	sing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number or Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-046	334 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-046a	8 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	Centrica PLC Milistream Maidenhead Road WINDSOR Berkshire SL4 SGD (in respect of rights and covenants granted by Deed dated 31 December 1971) Joanne Lesley Davis Welland Farm House Surfleet Marsh Surfleet Marsh Surfleet SPALDING Lincolnshire PE11 4DW (in respect of rights granted by Transfer dated 07 November 2013) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LIN1 1YL (in respect of drainage apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LIN1 1YL (in respect of frainage apparatus) Lincolnshire Newland LINCOLN Lincolnshire Newland LINCOLN Lincolnshire Newland LINCOLN Lincolnshire Newland LINCOLN Lincolnshire Newland LINCOLN Lincolnshire Newland LINCOLN LIncolnshire Newland LINCOLN LIncolnshire Newland LINCOLN LIncolnshire Newland LINCOLN LIncolnshire Newland LINCOLN LINC	

Outer Dowsing Offshore Wind Development Consent Order			
BOOK OF REFERENCE - PART 3			
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-046a cont'd		National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 60A (in respect of underground gas apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of rights granted by Deed dated 21 February 2007) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of rights granted by Deed dated 21 February 2007) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of rights granted by Deed dated 21 February 2007) Patricia Mary Marguerite Bealby 24 Catherine Place Pine Gardens MELTON MOWBRAY LEI3 1 FH (in respect of rights granted by Deed dated 22 March 2000) Russell Lee Davis </td	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-046a cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
45-047	417 square metres of drain (east of A16) (excluding all interests of the Crown)	Unknown	
45-049	6604 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
45-050	11140 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	
45-051	5486 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)	

Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009
45-052	2371 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)
45-054	22177 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)
45-055	302 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)
45-056	1695 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)
45-057	2434 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights)

	Outer Dowsi	ing Offshore Wind Development Consent Order	
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-058	17760 square metres of agricultural land, drain and access track (east of The Reservoir, A16)		
		Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-065	4924 square metres of agricultural land, drain and access track (east of The Reservoir, A16)	Centrica PLC	
		Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of underground gas apparatus) The Welland And Deepings Internal Drainage Board	
45-066	3071 square metres of access track and verge (north of Marsh Drove)	Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace	
		SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
		BOOK OF REFERENCE - PART 3	
	County of Lincolnshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
45-067	5686 square metres of agricultural land, drain and access track (east of The Reservoir, A16)		
		Millstream	
		Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of rights granted by Deed dated 31 December 1971)	
		National Gas Transmission PLC	
		National Grid House	
		Warwick Technology Park	
		Gallows Hill	
		WARWICK	
		Warwickshire	
		CV34 6DA	
		(in respect of underground gas apparatus)	
		The Welland And Deepings Internal Drainage Board	
		Deeping House	
		Welland Terrace	
		SPALDING	
		Lincolnshire	
		PE11 2TD	
		(assumed in respect of rights of access)	
45-068	6104 square metres of agricultural land (east of Marsh Drove)	Centrica PLC	
		Millstream	
		Maidenhead Road	
		WINDSOR	
		Berkshire	
		SL4 5GD	
		(in respect of rights granted by Deed dated 31 December 1971)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-068 cont'd		National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of underground gas apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-069	1496 square metres of drain (east of Marsh Drove)	Centrica PLC Millstream Maidenhead Road WINDSOR Berkshire SL4 5GD (in respect of rights granted by Deed dated 31 December 1971) National Gas Transmission PLC National Grid House Warwick Technology Park Gallows Hill WARWICK Warwickshire CV34 6DA (in respect of underground gas apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
45-069 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-071	19 square metres of agricultural land (north of River Welland)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
45-072	2315 square metres of agricultural land (north of River Welland)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
45-073	946 square metres of drain (south of Marsh Lane)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
46-001	22 square metres of access track (east of A16) (excluding all interests of the Crown)	Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
46-002	390 square metres of agricultural land (east of A16) (excluding all interests of the Crown)	Unknown	
46-003	480 square metres of agricultural land (east of A16) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING LincoInshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
46-004	1232 square metres of agricultural land (east of A16 and north of Bicker Creek Drain) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
46-005	1680 square metres of agricultural land (east of A16 and north of Bicker Creek Drain) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	
46-006	4031 square metres of agricultural land, access track and drain (east of A16) (excluding all interests of the Crown)	Peter C. Thorold Limited 5 Bank House Farm Cooks Road Gosberton SPALDING Lincolnshire PE11 4PE (in respect of rights and covenants as stated in a Lease dated 27 January 2022)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3 County of Lincolnshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
46-007	85 square metres of drain (Bicker Creek) (excluding all interests of the Crown)	Unknown	
46-009	1926 square metres of agricultural land and access track (east of A16)	The Welland And Deepings Internal Drainage Board	
40 005	1520 square metres of agreated and and access track (cast of A10)	Deeping House	
		Welland Terrace	
		SPALDING	
		Lincolnshire	
		PE11 2TD	
		(assumed in respect of rights of access)	
46-010	2522 square metres of agricultural land and access track (east of A16)	The Welland And Deepings Internal Drainage Board	
		Deeping House	
		Welland Terrace	
		SPALDING	
		Lincolnshire	
		PE11 2TD	
		(assumed in respect of rights of access)	
46-011	15 square metres of access track (east of A16) and banks (Risegate Eau)	John Grant (Donington)	
		The Old Vicarage	
		Church Close	
		BOSTON	
		Lincolnshire	
		PE21 6NA	
		(in respect of assumed rights of access)	
		The Welland And Deepings Internal Drainage Board	
		Deeping House	
		Welland Terrace	
		SPALDING	
		Lincolnshire	
		PE11 2TD	
		(assumed in respect of rights of access)	
		Unknown	
		(in respect of rights and obligations listed in a Conveyance dated 30 November 1988)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
46-012	37 square metres of access track (east of A16) and banks (Risegate Eau)	John Grant (Donington)	
		The Old Vicarage	
		Church Close	
		BOSTON	
		Lincolnshire	
		PE21 6NA	
		(in respect of assumed rights of access)	
		The Welland And Deepings Internal Drainage Board	
		Deeping House	
		Welland Terrace	
		SPALDING	
		Lincolnshire	
		PE11 2TD	
		(assumed in respect of rights of access)	
		Unknown	
		(in respect of rights and obligations listed in a Conveyance dated 30 November 1988)	
46-013	200 square metres of banks (Risegate Eau)	Lincolnshire County Council	
		County Offices	
		Newland	
		LINCOLN	
		Lincolnshire	
		LN1 1YL	
		(in respect of rights granted in Transfer dated 22 May 2001)	
		The Welland And Deepings Internal Drainage Board	
		Deeping House	
		Welland Terrace	
		SPALDING	
		Lincolnshire	
		PE11 2TD (accumed in respect of rights of access)	
		(assumed in respect of rights of access)	
		Unknown	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
		Names of all those entitled to enjoy easements or other private rights over land	
Number on	Description of Land	(including private rights of navigation over water) which it is proposed shall be extinguished,	
Land Plans		suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
46-017	823 square metres of agricultural land (east of The Reservoir, A16)	Lincolnshire County Council	
		County Offices	
		Newland LINCOLN	
		Lincolnshire	
		LN1 1YL	
		(in respect of rights stated in Transfer dated 24 May 2000)	
46-018	2630 square metres of agricultural land, hedgerow and access track (east of The Reservoir,	Lincolnshire County Council	
	A16) (excluding all interests of the Crown)	County Offices	
		Newland	
		Lincolnshire LN1 1YL	
		(in respect of rights stated in Transfer dated 24 May 2000)	
		Lincolnshire County Council	
		County Offices	
		Newland	
		Lincolnshire	
		LN1 1YL (in respect of underground cable apparatus)	
		(in respect of underground cable apparatus)	
		National Grid Electricity Transmission PLC	
		1 - 3 Strand	
		London	
		WC2N 5EH	
		(in respect of overhead electricity apparatus)	
		The Welland And Deepings Internal Drainage Board	
		Deeping House	
		Welland Terrace	
		SPALDING	
		Lincolnshire	
		PE11 2TD	
		(assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
		BOOK OF REFERENCE - PART 3	
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
	296 square metres of agricultural land, hedgerow and access track (east of The Reservoir, A16) (excluding all interests of the Crown)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of underground cable apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect of overhead electricity apparatus)	
	2607 square metres of agricultural land, hedgerow and access track (east of The Reservoir, A16) (excluding all interests of the Crown)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of underground cable apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
46-020 cont'd		National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
46-021	8453 square metres of agricultural land and access track (east of A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of underground cable apparatus) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000)	
46-022	190 square metres of agricultural land, hedgerow and access track (east of The Reservoir, A16) (excluding all interests of the Crown)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights stated in Transfer dated 24 May 2000)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
46-022 cont'd		Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of underground cable apparatus)	
46-023	455 square metres of drain (north of Marsh Drove)	Unknown	
46-024	453 square metres of drain (north of Marsh Drove)	Unknown	
46-025	10257 square metres of agricultural land (north of Marsh Drove)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right to enter as stated in Transfer dated 25 March 2003) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)	
46-027	22 square metres of agricultural land and access track (west of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights and obligations listed in a Conveyance dated 30 November 1988)	

	Outer Dowsing Offshore Wind Development Consent Order		
	BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning	
		(Applications: Prescribed Forms and Procedures) Regulations 2009	
46-028	24 square metres of agricultural land and access track (west of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (in respect of assumed rights of access) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights and obligations listed in a Conveyance dated 30 November 1988)	
46-029	105 square metres of agricultural land, access track and banks (west of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (in respect of assumed rights of access) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of rights and obligations listed in a Conveyance dated 30 November 1988)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
46-030	4 square metres of access track (west of A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted in Transfer dated 22 May 2001) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
46-031	5236 square metres of agricultural land (west of A16) and banks (Risegate Eau)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
46-033	5992 square metres of watercourse (Risegate Eau)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
46-034	6284 square metres of drain and banks (west of A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of rights granted in Transfer dated 22 May 2001) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
46-035	83 square metres of drain and banks (west of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
46-036	1029 square metres of drain and copse (east of Gosberton Bank)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of electricity apparatus) Unknown	
46-038	26 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	
46-039	1018 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	
46-040	1696 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	
46-041	172 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	
46-042	31 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
46-043	1029 square metres of drain and copse (east of Gosberton Bank)	Unknown	
46-045	2024 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	
46-046	2762 square metres of agricultural land and access track (east of Gosberton Road)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights granted by the deed dated 13 August 1963)	
47-002	1 square metres of access splay (east of A16)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
47-003	416 square metres of private road and verge (North of Surfleet Bank)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-003 cont'd		Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
47-004	1894 square metres of agricultural land and access track (east of A16)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of underground cable apparatus and street furniture) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right of way as stated in Deed dated 16 March 1999)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-004 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)	
47-005	4 square metres of private road (North of Surfleet Bank)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
47-006	4722 square metres of agricultural land, access track and hedgerow (Woad Farm)	Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of a right of way as stated in Deed dated 16 March 1999) Lincolnshire County Council County Offices Newland LINCOLN Lincolnshire LN1 1YL (in respect of underground cable apparatus and street furniture)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-006 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
47-008	742 square metres of drain (east of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
47-009	4252 square metres of access track and verge (east of A16)	John Grant (Donington) The Old Vicarage Church Close BOSTON Lincolnshire PE21 6NA (in respect of a rights stated in Conveyance dated 05 October 1984) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-010	679 square metres of drain and copse (east of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
47-011	5302 square metres of agricultural land (east of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)	
47-014	9109 square metres of agricultural land, access track and drain (east of A16)	Alan Burdall Aston Leigh Surfleet Marsh Surfleet SPALDING PE11 4DP (in respect of a right of way as stated in Transfer dated 08 June 2001) Gloria June Burdall Aston Leigh Surfleet Marsh Surfleet SPALDING PE11 4DP (in respect of a right of way as stated in Transfer dated 08 June 2001) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N SEH (in respect of overhead electricity apparatus)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-014 cont'd		The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
47-015	448 square metres of agricultural land, access track and drain (east of A16)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
47-016	325 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
47-017	2938 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-019	417 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
47-020	576 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of riparian rights) Unknown	
47-021	32 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (assumed in respect of rights of access)	
47-022	4 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-023	31 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
47-024	61 square metres of drain (north of Marsh Drove)	Unknown	
47-026	3678 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)	
47-028	1880 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (in respect of rights of access and maintenance relating to a drain)	
47-029	13675 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING LincoInshire PE11 2TD (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-030	87 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access)	
47-032	581 square metres of public road and verges (Marsh Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of underground water apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of overhead electricity apparatus) Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of underground telecommunication apparatus) Unknown	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-036	30 square metres of drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
47-037	1793 square metres of agricultural land, hedgerow and drain (north of Marsh Drove)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
47-038	1081 square metres of agricultural land, hedgerow and drain (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
47-039	1991 square metres of agricultural land (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-001	17 square metres of verge (north of Marsh Drove)	The Welland And Deepings Internal Drainage Board Deeping House Welland Terrace SPALDING Lincolnshire PE11 2TD (assumed in respect of rights of access) Unknown	
48-002	471 square metres of agricultural land (north of Marsh Drove)	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-003	1047 square metres of agricultural land, access track and hedgerow (north of Marsh Drove)	National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-004	655 square metres of agricultural land, access track and hedgerow (north of Marsh Drove)	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-005	1312 square metres of agricultural land and grassed area (north of Marsh Drove)	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
48-006	22 square metres of agricultural land (north of Marsh Drove)	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-010	622 square metres of agricultural land (north of Marsh Drove)	Unknown	
48-011	484 square metres of agricultural land (north of Marsh Drove)	Unknown	
48-012	905 square metres of agricultural land (north of Marsh Drove)	Unknown	
48-013	297 square metres of agricultural land (north of Marsh Drove)	Unknown	
48-017	264 square metres of agricultural land (south of Marsh Drove)	Unknown (in respect of restrictive covenants that may have been imposed thereon before 10 October 1958)	
48-019	6564 square metres of river (River Welland), foreshore, bed and banks thereof	Unknown	
48-020	3793 square metres of agricultural land, scrubland and hedgerow (north of Marsh Road) and public bridleway (LL Wstn 6/2)	The Dean And Chapter Of The Cathedral Church Of Christ In Oxford Of The Foundation Of King Henry The Eighth Christ Church St. Aldates Oxford Oxfordshire OX1 1DP (in respect of personal covenants contained in Conveyance dated 25 July 1949) Unknown (in respect of rights reserved in Transfer dated 01 September 1989)	
48-023	3044 square metres of agricultural land (north of Marsh Road)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
48-023 cont'd		InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of rights of access)	
48-024	17109 square metres of agricultural land (north of Marsh Road)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Unknown (in respect of restrictive covenants that may have been imposed thereon before 11 December 2013)	
48-025	14133 square metres of agricultural land (north of Marsh Road)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
48-025 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of rights of access)	
49-002	1499 square metres of access track (north of Marsh Road) and public bridleway (LL Surf 9/1)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) Unknown (in respect of restrictive covenants that may have been imposed thereon before 11 December 2013)	
49-003	509 square metres of private road and verges (West of Wragg Marsh Farm) and public bridleways (LL Wstn 4/1 and LL Wstn 5/1)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3		
	County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
49-003 cont'd		Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)	
49-004	61 square metres of private road and verge (Marsh Road)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 SAH (assumed in respect of rights of access) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005) Unknown (in respect of restrictive covenants that may have been imposed thereon before 21 March 2000)	
49-005	2286 square metres of agricultural land (north of Marsh Road and south of River Welland)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order		
BOOK OF REFERENCE - PART		BOOK OF REFERENCE - PART 3	
		County of Lincolnshire	
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
49-005 cont'd		InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of rights of access)	
49-006	5104 square metres of agricultural land (Wragg Marsh Farm)	Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
49-006 cont'd		National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of underground and overhead electricity apparatus) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of rights of access)	
49-007	94 square metres of verge (north of Marsh Road)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) Environment Agency Horizon House Deanery Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS1 5AH (assumed in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
49-007 cont'd		Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of rights of access)	
49-008	261 square metres of public road (Marsh Road)	National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of overhead electricity apparatus) Unknown	
49-009	151 square metres of verge (south of Marsh Road)	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 0TB (in respect of electricity apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire		
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009	
49-009 cont'd		Openreach Limited 6 Gracechurch Street LONDON EC3V 0AT (in respect of apparatus) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of rights of access)	
49-010	38389 square metres of agricultural land and private road (Wragg Marsh Farm)	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire			
Land Plans suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning				
49-010 cont'd		Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)		
49-011	1194 square metres of agricultural land (Wragg Marsh Farm)	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)		
50-001	5639 square metres of agricultural land (east of Marsh Road)	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access)		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
50-001 cont'd		Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 29 October 2004)		
50-002	506159 square metres of building and hardstanding (Crowtree Farm), agricultural land and access track (east of Marsh Road), electricity cables and pylons and public footpath (LL Wstn 7/1)	Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of overhead electricity apparatus) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 28B (in respect of legal easements granted by Lease dated 29 October 2004)		

	Outer Dowsing Offshore Wind Development Consent Order				
	BOOK OF REFERENCE - PART 3 County of Lincolnshire				
		Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009			
50-003 203228 square metres of agricultural land and access track (east of Marsh Road), electricity InterGen (UK) Ltd 30 Crown Place IONDON EC2A 4ES (in respect of apparatus) National Grid Electricity Transmission PLC 1-3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002) Spaling Energy Company, Ltd. Spaling Power Station West Marsh Road SPALDING Lincolnshire PE11 28B (in respect of legal easements granted by Lease dated 14 January 2005)		InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB			
50-004	1958 square metres of agricultural land and drain (east of Marsh Road)	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire			
Number on Land Plans Description of Land		Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
50-004 cont'd Cont		Spalding Energy Expansion Limited 30 Crown Place LONDON EC2A 4ES		
50-005	216275 square metres of agricultural land, access track, electricity cable and pylons (east of Marsh Road)	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of apparatus) National Grid Electricity Distribution (East Midlands) PLC Avonbank Feeder Road BRISTOL Avon BS2 OTB (in respect of underground and overhead electricity apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights stated in Deed dated 29 November 2002) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of legal easements granted by Lease dated 14 January 2005)		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
50-006 43727 square metres of agricultural land, access and drain (east of Marsh Road)		Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park HUNTINGDON Cambridgeshire PE29 3NZ (in respect of water apparatus)		
50-007	40886 square metres of agricultural land, access track and drain (east of Marsh Road)	Unknown		
51-003	521 square metres of drain (east of Marsh Road)	Unknown		
51-005	17 square metres of drain (east of Marsh Road)	Unknown		
51-006	822 square metres of drain (east of Marsh Road)	Unknown		
51-007	159921 square metres of agricultural land, drain and access track (east of Marsh Road), electricity cables and pylons	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of apparatus) National Grid Electricity Transmission PLC 1 - 3 Strand London WC2N 5EH (in respect of rights and restrictive covenants listed in a Deed dated 08 May 2003) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of the legal easements granted by the Lease dated 14 January 2005)		

	Outer Dowsing Offshore Wind Development Consent Order				
	BOOK OF REFERENCE - PART 3				
	County of Lincolnshire				
	Names of all those entitled to enjoy easements or other private rights over land				
Number on	Description of land	(including private rights of navigation over water) which it is proposed shall be extinguished,			
Land Plans	Description of Land	suspended or interfered with under Regulation $7(1)(c)$ of the Infrastructure Planning			
		(Applications: Prescribed Forms and Procedures) Regulations 2009			
51-008	2157 square metres of agricultural land and drain (east of Marsh Road)	InterGen (UK) Ltd			
		30 Crown Place			
		LONDON			
		EC2A 4ES			
		(in respect of apparatus)			
		National Grid Electricity Transmission PLC			
		1 - 3 Strand			
		London			
		WC2N 5EH			
		(in respect of rights and restrictive covenants listed in a Deed dated 08 May 2003)			
		Spalding Energy Company, Ltd.			
		Spalding Power Station			
		West Marsh Road			
		SPALDING			
		Lincolnshire			
		PE11 2BB			
		(in respect of the legal easements granted by the Lease dated 14 January 2005 and 26 February 2018)			
		Spalding Energy Expansion Limited			
		30 Crown Place			
		LONDON EC2A 4ES			
		(in respect of the rights that are granted by the Deed dated 26 February 2018)			
		(in respect of the rights that are granted by the beed dated 201 ebidary 2018)			
	189781 square metres of agricultural land, drain and access track (east of Marsh Road),	InterGen (UK) Ltd			
	electricity cables and pylons	30 Crown Place			
		LONDON			
		EC2A 4ES			
		(in respect of apparatus)			
		National Grid Electricity Transmission PLC			
		1 - 3 Strand			
		London			
		WC2N 5EH			
		(in respect of rights and restrictive covenants listed in a Deed dated 08 May 2003)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of Lincolnshire			
Number on Land Plans(including private rights of navigation over water) which it is propriate suspended or interfered with under Regulation 7(1)(c) of the		Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
51-009 cont'd		Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of the legal easements granted by the Lease dated 14 January 2005) Unknown (in respect to rights relating to nine inch sanitary pipes detailed in a deed dated 29 April 1943)		
51-010	789 square metres of drain (east of Marsh Road)	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of apparatus) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB (in respect of rights of access) Unknown		
51-011	963 square metres of agricultural land (east of Marsh Road)	InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of rights of access)		

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 3 County of LincoInshire			
Number on Land Plans	Description of Land	Names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedures) Regulations 2009		
51-011 cont'd	Spalding Energy Company, Ltd.			
51-012 27076 square metres of agricultural land (east of Marsh Road) InterGen (UK) Ltd 30 Crown Place LONDON EC2A 4ES (in respect of apparatus) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire PE11 2BB		30 Crown Place LONDON EC2A 4ES (in respect of apparatus) Spalding Energy Company, Ltd. Spalding Power Station West Marsh Road SPALDING Lincolnshire		
51-013	5 square metres of drain (east of Marsh Road)	Unknown		
51-014	308 square metres of drain (east of Marsh Road)	Unknown		

	Outer Dowsing Offshore Wind Development Consent Order				
	BOOK OF REFERENCE - PART 3				
	County of Lincolnshire				
	Names of all those entitled to enjoy easements or other private rights over land				
Number on	Description of lead	(including private rights of navigation over water) which it is proposed shall be extinguished,			
Land Plans	Description of Land	suspended or interfered with under Regulation 7(1)(c) of the Infrastructure Planning			
		(Applications: Prescribed Forms and Procedures) Regulations 2009			
51-015	22693 square metres of agricultural land (east of Marsh Road), electricity cables and pylons	Gordon Albert Mills			
		St. Lamberts Hall			
		Hallgate			
		Weston			
		SPALDING			
		Lincolnshire			
		PE12 6RH			
		(in respect of rights granted in Transfer dated 22 April 2003)			
		Helen Juliette Mills			
		St. Lamberts Hall			
		Hallgate			
		Weston			
		SPALDING			
		Lincolnshire			
		PE12 6RH			
(ii		(in respect of rights granted in Transfer dated 22 April 2003)			
		Moulton Bulb Company Limited			
		Long Lane			
Moulton		Moulton			
		SPALDING			
		PE12 6PP			
		(in respect of restrictions listed in a Definitive Trust Deed dated 26 September 1986 and rights granted in Transfer dated 22 April 2003)			
		National Grid Electricity Transmission PLC			
		1 - 3 Strand			
		London			
		WC2N 5EH			
		(in respect of rights relating to electricity lines and pylons listed in a Deed of Grant dated 16 April 2003)			
		Prudential Financial Planning Limited			
		10 Fenchurch Avenue			
		LONDON			
		EC3M 5AG			
		(in respect of restrictions listed in a Definitive Trust Deed dated 26 September 1986)			

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of LincoInshire			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
01-001	А	Permanent Rights over 11097 square metres of foreshore seaward of the Mean High Water (East of Roman Bank) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
01-002	A	Permanent Rights over 104734 square metres of foreshore seaward of the Mean High Water, tidal pond and groynes (east of Roman Bank) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
19-004	D	Permanent Rights over 194 square metres of copse (north of Mill Hill) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
19-006	D	Permanent Rights over 25424 square metres of agricultural land (north of Mill Hill) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
19-007	к	Temporary Rights over 60 square metres of agricultural land (north of Mill Hill) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
19-008	D	Permanent Rights over 181 square metres of drain (north of Mill Hill) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
19-009	D	Permanent Rights over 901 square metres of public road (Mill Hill) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (assumed in respect of subsoil beneath public highway)	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Lincolnshire			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
22-001	к	Temporary Rights over 6053 square metres of agricultural land and drain (west of Ivery Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
22-002	к	Temporary Rights over 12 square metres of drain (west of Ivery Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
22-004	D	Permanent Rights over 7797 square metres of agricultural land (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
22-005	D	Permanent Rights over 117 square metres of drain (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
22-006	к	Temporary Rights over 29 square metres of agricultural land (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
22-007	к	Temporary Rights over 9 square metres of drain (west of Ivery Lane and south of Love Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
23-013	D	Permanent Rights over 12266 square metres of agricultural land (north of Cragmire Lane and east of Double Bank) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Lincolnshire			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
23-014	к	Temporary Rights over 1482 square metres of agricultural land (north of Cragmire Lane)(excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
23-015	D	Permanent Rights over 218 square metres of drain (east of Double Bank) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
34-005	D	Permanent Rights over 1809 square metres of river (The Haven), foreshore, bed and banks thereof (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
34-006	D	Permanent Rights over 19799 square metres of river (The Haven), foreshore, bed and banks thereof (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
37-011	D	Permanent Rights over 185 square metres of drain (east of Sandholme Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
37-012	D	Permanent Rights over 54857 square metres of agricultural land, access track and drain (east of Sandholme Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
38-001	К	Temporary Rights over 2663 square metres of agricultural land and access track (east of Sandholme Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Lincolnshire			
Number on Land PlansExtent of acquisition or useDescription of landThe owner of any Crown interest in the land which is proposed to be used for the purpose application is being made		The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made		
38-002	к	Temporary Rights over 37 square metres of access track (north of Sandholme Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
38-003	F	Temporary Rights over 459 square metres of agricultural land and drain (north of Sandholme Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
38-004	F	Temporary Rights over 43 square metres of verge (north of Sandholme Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
38-005	D	Permanent Rights over 112 square metres of public road (Sandholme Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
38-006	D	Permanent Rights over 533 square metres of public road (Sandholme Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (assumed in respect of subsoil beneath public highway)	
45-001	Freehold Acquisition	Freehold over 5972 square metres of agricultural land and access track (south of Sea Bank and west of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
45-002	J, L	Permanent Rights over 5597 square metres of agricultural land, access track and drain (east of A16 and south of Sea Bank) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Lincolnshire			
Number on Land PlansExtent of acquisition or useDescription of landThe owner of any Crown interest in the land which is proposed to be used for the p application is being made		The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made		
45-003	ſ	Permanent Rights over 3692 square metres of agricultural land, access track and drain (east of A16 and south of Sea Bank) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
45-004	J, L	Permanent Rights over 28 square metres of access splay (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
45-005	Freehold Acquisition	Freehold over 4235 square metres of agricultural land and access track (west of Marsh Lane) (excluding all interests of the Crown)		
45-006	ſ	Permanent Rights over 2529 square metres of agricultural land, access track and drain (east of A16 and south of Sea Bank) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
45-007	J, L	Permanent Rights over 3 square metres of verge (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
45-008	Freehold Acquisition	Freehold over 1647 square metres of agricultural land (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
45-009	J	Permanent Rights over 1464 square metres of agricultural land, access track and drain (east of A16 and south of Sea Bank) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	

Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Lincolnshire			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made
45-010	ſ	Permanent Rights over 94 square metres of watercourse (Risegate Eau) and access track (south of Marsh Lane) (excluding all interests of the Crown)	
45-011	J, L	Permanent Rights over 28 square metres of watercourse (Risegate Eau) and access track (south of Marsh Lane) (excluding all interests of the Crown)	
45-012	ſ	Permanent Rights over 71 square metres of watercourse (Risegate Eau) and access track (south of Marsh Lane) (excluding all interests of the Crown)	
45-013	J	Permanent Rights over 657 square metres of agricultural land, access track and drain (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
45-014	J, L	Permanent Rights over 2483 square metres of agricultural land, access track and drain (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
45-015	Freehold Acquisition	Freehold over 2454 square metres of agricultural land (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
45-016	J	Permanent Rights over 2589 square metres of agricultural land, access track and drain (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Lincolnshire			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
45-017	ſ	Permanent Rights over 15 square metres of drain (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
45-018	ſ	Permanent Rights over 72 square metres of drain (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
45-029	D	Permanent Rights over 718 square metres of drain (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
45-030	D	Permanent Rights over 27105 square metres of agricultural land (south of Marsh Lane) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	
45-031	D	Permanent Rights over 375 square metres of drain (east of A16) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
45-033	К	Temporary Rights over 3472 square metres of agricultural land (south of Risegate Eau) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH	

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of Lincolnshire			
Number on Land Plans	Extent of acquisition or use	Description of land	The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
45-034	ſ	Permanent Rights over 1036 square metres of drain (east of A16) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
45-043	ſ	Permanent Rights over 2023 square metres of watercourse (Risegate Eau) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
45-044	ſ	Permanent Rights over 1312 square metres of watercourse (Risegate Eau) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
45-047	ſ	Permanent Rights over 417 square metres of drain (east of A16) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
46-001	J, L	Permanent Rights over 22 square metres of access track (east of A16) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	
46-002	Freehold Acquisition	Freehold over 390 square metres of agricultural land (east of A16) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)	

Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 4 Land which is proposed to be used for the purposes of the order for which application is being made under Regulation 7(1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 County of LincoInshire			
Number on Land PlansExtent of acquisition or useDescription of landThe owner of any Crown interest in the land which is proposed to be used for the purposed application is being made		The owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made	
46-003	Freehold Acquisition	Freehold over 480 square metres of agricultural land (east of A16) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
46-004	J, L	Permanent Rights over 1232 square metres of agricultural land (east of A16 and north of Bicker Creek Drain) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
46-005	ſ	Permanent Rights over 1680 square metres of agricultural land (east of A16 and north of Bicker Creek Drain) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
46-006	Freehold Acquisition	Freehold over 4031 square metres of agricultural land, access track and drain (east of A16) (excluding all interests of the Crown)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH
46-007	ſ	Permanent Rights over 85 square metres of drain (Bicker Creek)	The King's Most Excellent Majesty In Right Of His Crown 1 St. James's Market LONDON SW1Y 4AH (as assumed owner)

	Outer Dowsing Offshore Wind Development Consent Order BOOK OF REFERENCE - PART 5 County of LincoInshire			
Number on Land Plans	Extent of acquisition or use	Description of land	Category of Land	
01-001	A	Permanent Rights over 11097 square metres of foreshore seaward of the Mean High Water (East of Roman Bank) (excluding all interests of the Crown)	Open Space	
01-002	A	Permanent Rights over 104734 square metres of foreshore seaward of the Mean High Water, tidal pond and groynes (east of Roman Bank) (excluding all interests of the Crown)	Open Space	
01-004	В	Permanent Rights over 1636 square metres of foreshore landward of the Mean High Water, scrubland and path (east of Roman Bank)	Open Space	
01-005	В	Permanent Rights over 83744 square metres of foreshore landward of the Mean High Water, scrubland and path (east of Roman Bank)	Open Space	
17-001	D	Permanent Rights over 1021 square metres of access track (north of Scald Gate) and public footpath (LL WStM 371/1)	Common Land	